

#### Designing Respectful Health Data Flows Centred on Data Contributors' Perspectives

Nunes Vilaza, Giovanna

Publication date: 2022

Document Version Publisher's PDF, also known as Version of record

Link back to DTU Orbit

*Citation (APA):* Nunes Vilaza, G. (2022). *Designing Respectful Health Data Flows Centred on Data Contributors' Perspectives*. DTU Health Technology.

#### **General rights**

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

• Users may download and print one copy of any publication from the public portal for the purpose of private study or research.

- You may not further distribute the material or use it for any profit-making activity or commercial gain
- You may freely distribute the URL identifying the publication in the public portal

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.



# Designing Respectful Health Data Flows Centred on Data Contributors' Perspectives

PhD Thesis

Giovanna Nunes Vilaza



PhD Thesis February, 2022

Author: Giovanna Nunes Vilaza

Published by: Technical University of Denmark Department of Health Technology Ørsteds Plads, Building 345B 2800 Kongens Lyngby, Denmark

## Preface

This PhD thesis represents the culmination of a 3-years doctoral project to fulfil the requirements for acquiring a PhD degree at the Technical University of Denmark (DTU). The research was supervised by Dr. Kevin Doherty, Prof. David Coyle and Prof. Jakob Bardram.

The PhD work was conducted at the Department of Health Technology (February 2019-February 2022) and the Department of Applied Mathematics and Computer Science (October 2018-January 2019). An external stay took place at the Insight Centre for Data Analytics, University College Dublin, in Ireland (September 2019-December 2019).

The project was a part of the Technology-Enabled Mental Health Innovative Training Network (TEAM-ITN), funded by the European Union's Horizon 2020 Research and Innovation Programme, under the Marie Sklodowska-Curie grant agreement 722561.

This thesis includes the following published research papers:

- Vilaza, Giovanna Nunes and Bardram, Jakob (2019). Sharing Access to Behavioural and Personal Health Data: Designers' Perspectives on Opportunities and Barriers. In Proceedings of the 13th EAI International Conference on Pervasive Computing Technologies for Healthcare.
- 2. Vilaza, Giovanna Nunes et al. (2020). *Futures for Health Research Data Platforms From the Participants' Perspectives*. In Proceedings of the 11th Nordic Conference on Human-Computer Interaction: Shaping Experiences, Shaping Society.
- 3. Vilaza, Giovanna Nunes (2020). *What is the Future of Data Sharing for Research?* ACM interactions 27, 4, 54–56.
- 4. Vilaza, Giovanna Nunes and McCashin, Darragh (2021). *Is the Automation of Digital Mental Health Ethical? Applying an Ethical Framework to Chatbots for Cognitive Behaviour Therapy*. Frontiers in Digital Health 3, 689736.
- 5. Vilaza, Giovanna Nunes et al. (2021). *Public Attitudes to Digital Health Research Repositories: Cross-sectional International Survey*. Journal of Medical Internet Research 23, 10, e31294.

Giovanna Nunes Vilaza

<u>(jievonmos Vilonos)</u> Signature

February 21st, 2022 Date

## Abstract

The increased use of digital health data has been praised as a revolutionary turn for the practice and study of medicine. Among the diverse range of health data sources, a significant percentage of research has focused upon data generated by citizens and recipients of healthcare services. This broad category comprises information as diverse as data collected within clinical infrastructures, during treatment, and throughout individuals' daily lives. A growing interest in digital health data has led to the creation of shared-access repositories that allow multiple stakeholders to utilise available data sources simultaneously. One of the primary aims of these repositories is to overcome challenges related to recruiting participants for research studies. A further ambition is to store mobile and wearable data to observe and learn from human behaviour at a population scale.

Although digital health data may contribute towards an in-depth understanding of human subjects, the choice of allowing data access to multiple users is not exempt from fundamental ethical considerations. If identifiable data used to monitor stigmatised health conditions is leaked, individuals may suffer the consequences of social discrimination. Health data can be abused if utilised to enable commercial profit from individual vulnerabilities or inform governments' authoritarian control over citizens. There is also a risk that unregulated data analysis is biased and invalid research findings ultimately affect how patients are treated in the future.

Given the undeniable gravity and significance of such ethical concerns, there is a need to consider more responsible data sharing practices within healthcare to protect individuals. This thesis aims to inform the design of future health data ecosystems by providing evidence in addition to a portfolio of approaches to support the prioritisation of ethical design choices aligned with the values of citizens and patients. The overarching goal of this work is to investigate how we can discover ethically-oriented and data contributor-centred requirements for designing health data interfaces. This research is grounded in Human-Computer Interaction (HCI) and user-centred design traditions while building upon the concept and practice of ethical design. Given this orientation, the methodolog-ical approach of this thesis work includes and connects literature reviews, user studies (both qualitative and quantitative), critical appraisals of technology and the prototyping of user interfaces.

This thesis makes the following contributions. First, this PhD delivers a review of state-ofthe-art health data repositories and discusses factors key to their acceptance. Second, it presents findings from a scoping review yielding a coherent up-to-date mapping of ethical thinking and doing in HCI. Third, it provides an in-depth understanding of data contributors' and technology creators' attitudes towards sharing data with and through health data repositories. Fourth, it illustrates in practice the application and teaching of ethicallycentred approaches to the design of health data interfaces. Finally, it presents a set of design considerations supporting the future development of increasingly ethical health data flows. Overall, this PhD demonstrates how requirements for respectful health data flows can integrate ethical and contributor centred.

## Dansk Resumé

Den øgede brug af digitale sundhedsdata er blevet set som en revolutionerende retning indenfor klinisk behandling og forskning. Blandt de mange forskellige kilder af sundhedsdata fokuserer en signifikant del af forskningen på data genereret af borgere og modtagere af sundhedsydelser. Denne brede kategori kan omfatte indsamlede informationer i kliniske infrastrukturer, under behandling eller gennem ens daglige liv. Den voksende interesse for sundhedsdata har ført til oprettelsen af arkiver med delt adgang, der tillader flere interessenter at tilgå den tilgængelige data. Et af de primære formål med disse arkiver er at overvinde udfordringer relateret til rekruttering af deltagere til kliniske studier. Et andet formål er at observere menneskelig adfærd i detaljer ved indsamlingen af data fra mobile og bærbare enheder.

Selvom personlige sundhedsdata kan bidrage til en dybdegående forståelse af menneskelig aktivitet, så er sundhedsdata ikke undtaget fra de grundlæggende etiske overvejelser. Data fra mobile og bærbare enheder kan fx. bruges som en indikator for mentale sundhedstilstande, der er stigmatiserede, hvilket kan føre til social diskrimination eller kompromittere den fremtidige beskæftigelsesegnethed. Tidligere studier har også fremhævet manglen på etisk begrundelse for at bruge menneskelige data til at gavne de virksomheder, der bygger robuste adfærdsmodeller ved at udnytte og tjene på individernes sårbarheder. Tidligere arbejde har også fremhævet manglen på etisk begrundelse for at bruge menneskelige data til at tjene virksomheder, der drager fordel af individuelle sårbarheder og regeringer, der udøver autoritær kontrol over borgerne. En lav indsats for at rekruttere en bredere mangfoldighed af bidragsydere til dataindsamlingen kan også medføre til en bias i datasæt og en uretfærdig fordeling af sundhedsydelserne.

I betragtning af den ubestridelige alvor og vigtigheden af disse etiske bekymringer, er der behov for at overveje mere ansvarlig datahåndtering indenfor sundhedsvæsenet. Denne ph.d.-afhandling har til formål at informere designet af digitale økosystemer i sundhedsvæsenet ved at levere evidens og en portefølje af tilgange til at prioritere etiske designvalg tilpasset borgernes og patienternes behov. Det overordnede mål er at undersøge, hvordan vi kan understøtte opdagelsen og handlingen af etiske krav såvel som krav fra bidragsydere ifm. dataindsamling ved design af interfaces til sundhedsdata. Arbejdet er funderet i menneske-computer-interaktion og brugercentrerede design traditioner, og er baseret på begrebet etisk design. Denne vinkel inkluderer en metodisk tilgang, som anvender en gennemgang af litteraturen, brugerundersøgelser (kvalitative og kvantitative), kritiske vurderinger af teknologien og prototyping af brugergrænseflader.

Dette projekt har følgende bidrag. For det første præsenterer denne ph.d.-afhandling en gennemgang af databaser og arkiver med sundhedsdata og diskuterer faktorer, der er nøglen til deres accept. For det andet bibringer ph.d.-afhandlingen en scoping-gennemgang, der giver en sammenhængende og opdateret kortlægning af etisk tænkning og handling i menneske-computer-interaktion. For det tredje giver afhandlingen en dybdegående forståelse af bidragsydernes til dataindsamling sammenholdt med teknologiskabernes holdninger til at dele data gennem arkiver med sundhedsdata. For det fjerde illustrerer afhandlingen i praksis, hvordan man anvender og underviser i værdi-centrerede tilgange til design af grænseflader for sundhedsdata. Endelig præsenterer afhandlingen et sæt designovervejelser, der understøtter den fremtidige udvikling af anvendelsen af sundhedsdata. Overordnet søger denne afhandling at demonstrere, hvordan etiske og bidragyder-centrerede perspektiver kan inddrages ved opstilling af krav til design af brugergrænse-flader for sundhedsdata.

## Acknowledgements

Much work, care and thought were put into turning this PhD project into a thesis. Nevertheless, none of the following pages would have become a reality without the support and love of many people. I list the following acknowledgements to thank those who have been fundamental in this journey.

First and foremost, I thank my family, who always, unconditionally and generously, has been by my side (virtually and physically). They are my inspiration, strength and home. I am incredibly privileged to have the family that I have.

I am thankful to my supervisors, who have trusted my work until the end. I am happy for being given this opportunity to immerse myself in a field of study that resonated deeply with who I am as a researcher, and I have learned a lot from each of you.

Thanks to the Marie Curie programme and the CACHET network, I could meet and learn from many academic colleagues who share the same goals, interests and frustrations as me, especially at TEAM and during my stay at UCD. I hope we keep growing together.

Thanks to DTU and the PhD School, I was provided with all I needed to produce this work. I am particularly grateful for having stable working conditions during a globally alarming pandemic. I wish more people could receive this type of support.

I would like to express my gratitude to the user studies' participants and the co-authors for making this thesis always a pleasant collaboration. I also thank the administrative workers at DTU Health Tech and my office peers. Numerous times, I needed help, and I have warmly received it.

I have also been incredibly fortunate to cross paths with many kind, cheerful, wholehearted people who have become loyal companions, therapists and friends in Denmark. Without you to share all the ups and downs, I would not have had the strength to keep going.

Finally, I am very grateful to the thesis examiners who promptly accepted the invitation to be part of the PhD committee. I hope this can be an enjoyable read.

## Contents

	Pref Abst Ackr	ace	iv v vii
Pa	art I	Introduction to the Paper Collection	1
1	Intro	oduction	2
	1.1	Context	2
	1.2	Motivation	3
	1.3	Research Questions	5
	1.4	Methodology	5
	1.5		6
	1.6	Contributions	7
2	Мар	ping the Literature   Health Data Repositories	9
	2.1		9
	2.2	Existing Platforms and Shared-Data Styles	9
		2.2.1 Top-down Initiatives	10
		2.2.2 Contributor-driven Initiatives	11
	2.3	Factors Influencing Willingness to Share Data	11
		2.3.1 Motivation	12
		2.3.2 Privacy	13
		2.3.3 Trust	13
		2.3.4 Consent	14
	2.4	In Summary	14
3	Мар	pping the Literature   Ethics and HCI	15
	3.1		15
	3.2	Method	16
	3.3	Results	16
		3.3.1 Characteristics of Sources of Evidence	17
		3.3.2 Objects of Ethical Consideration	17
		3.3.3 Approaches to Ethics in Design Research	18
		3.3.4 Recommendations for Putting Ethics into Action	20
	3.4	In Summary	22
4	Und	erstanding People   Interview Study with Data Contributors	23
	4.1		23
	4.2	Method	23
	4.3	Results	24
		4.3.1 Motivations to Contribute Data	24
		4.3.2 Trust in the Platform Governance	24
		4.3.3 Criteria for Sharing Each Data Type	25
	4.4	A Conceptual Framework	26
	4.5	In Summary	27

5	Und	lerstanding People   Survey Study with Data Contributors	29
	5.1		29
	5.2	Method	29
		5.2.1 Recruitment	29
		5.2.2 Survey Design	29
		5.2.3 Procedure	30
	5.3	Results	30
		5.3.1 Participants' Characteristics	30
		5.3.2 Experiences, Awareness and Attitudes	30
		5.3.3 Motivation to Participate	30
		5.3.4 Reasons for Concern	31
		5.3.5 Willingness to Share Different Data Types	31
		5.3.6 Preferred Data Access Options	32
		5.3.7 Cross-country Analysis	33
		5.3.8 Overview of Survey Findings	34
	5.4	In Summary	35
•			~ -
6		lerstanding People   Focus Group with Technology Creators	31
	6.2		37
	6.2		20
	0.5	Results	00 20
		6.3.2 Usors Are Deluctant to Share Data	20
		6.2.2 Usel's Ale Reluciant to Shale Data	29
	61		29
	0.4		39
_	_		
7	Pra	ctising Ethical Design   Health Data Interfaces	41
7	<b>Pra</b> 7.1	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41
7	<b>Pra</b> 7.1	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41 42
7	<b>Pra</b> 7.1	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41 42 42
7	<b>Pra</b> (7.1	ctising Ethical Design   Health Data InterfacesInterfaces for Research: Designing a Mobile Study App7.1.1Methods7.1.2Empirical Investigations7.1.3Conceptual Investigations	<b>41</b> 42 42 42
7	<b>Pra</b> ( 7.1	ctising Ethical Design   Health Data InterfacesInterfaces for Research: Designing a Mobile Study App7.1.1Methods7.1.2Empirical Investigations7.1.3Conceptual Investigations7.1.4Technical Investigations	<b>41</b> 42 42 42 43
7	<b>Pra</b> (7.1	ctising Ethical Design   Health Data InterfacesInterfaces for Research: Designing a Mobile Study App7.1.1Methods7.1.2Empirical Investigations7.1.3Conceptual Investigations7.1.4Technical Investigations7.1.5Section Summary	<b>41</b> 42 42 42 43 45
7	<b>Pra</b> (7.1	ctising Ethical Design   Health Data InterfacesInterfaces for Research: Designing a Mobile Study App7.1.1Methods7.1.2Empirical Investigations7.1.3Conceptual Investigations7.1.4Technical Investigations7.1.5Section SummaryInterfaces for Intervention: Analysing Chatbot Design	<b>41</b> 42 42 42 43 45 45
7	<b>Pra</b> (7.1 7.1	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 42 42 42 43 45 45 45
7	<b>Pra</b> (7.1 7.1	ctising Ethical Design   Health Data InterfacesInterfaces for Research: Designing a Mobile Study App7.1.1Methods7.1.2Empirical Investigations7.1.3Conceptual Investigations7.1.4Technical Investigations7.1.5Section SummaryInterfaces for Intervention: Analysing Chatbot Design7.2.1Methods7.2.2Results	<b>41</b> 42 42 43 45 45 46 46
7	<b>Pra</b> (7.1	ctising Ethical Design   Health Data InterfacesInterfaces for Research: Designing a Mobile Study App	<b>41</b> 42 42 43 45 45 46 46 46
7	<b>Pra</b> (7.1 7.2 7.3	ctising Ethical Design   Health Data InterfacesInterfaces for Research: Designing a Mobile Study App	<b>41</b> 42 42 43 45 45 46 46 49
7	<b>Pra</b> (7.1 7.2 7.3	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App         7.1.1       Methods         7.1.2       Empirical Investigations         7.1.3       Conceptual Investigations         7.1.4       Technical Investigations         7.1.5       Section Summary         7.1.1       Methods         7.1.2       Empirical Investigations         7.1.3       Conceptual Investigations         7.1.4       Technical Investigations         7.1.5       Section Summary         Interfaces for Intervention: Analysing Chatbot Design         7.2.1       Methods         7.2.2       Results         7.2.3       Section Summary         In Summary       In Summary	<b>41</b> 41 42 42 42 43 45 45 46 46 49 49 49
8	<ul> <li>Prac 7.1</li> <li>7.2</li> <li>7.3</li> <li>Prac 8.1</li> </ul>	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41 42 42 43 45 45 46 46 49 49 49 51
8	<b>Pra</b> 7.1 7.2 7.3 <b>Pra</b> 8.1 8.2	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41 42 42 42 43 45 45 45 46 49 49 49 <b>51</b> 51
8	<b>Pra</b> 7.1 7.2 7.3 <b>Pra</b> 8.1 8.2	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41 42 42 43 45 45 46 46 49 49 <b>51</b> 51 51
8	<b>Pra</b> 7.1 7.2 7.3 <b>Pra</b> 8.1 8.2 8.3	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41 42 42 43 45 45 46 49 49 49 <b>51</b> 51 51 52 53
8	<b>Pra</b> 7.1 7.2 7.3 <b>Pra</b> 8.1 8.2 8.3 8.4	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41 42 42 42 43 45 45 45 46 49 49 49 51 51 51 52 53 52
8	<b>Pra</b> (7.1) 7.2 7.3 <b>Pra</b> (8.1) 8.2 8.3 8.4	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41 42 42 43 45 45 46 46 49 49 49 <b>51</b> 51 51 52 53 53 53
8	<b>Pra</b> 7.1 7.2 7.3 <b>Pra</b> 8.1 8.2 8.3 8.4	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41 42 42 43 45 45 46 49 49 49 <b>51</b> 51 51 51 53 53 53
8	<b>Pra</b> 7.1 7.2 7.3 <b>Pra</b> 8.1 8.2 8.3 8.4	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41 42 42 43 45 46 46 49 49 49 51 51 51 51 52 53 53 54 6
8	<b>Pra</b> 7.1 7.2 7.3 <b>Pra</b> 8.1 8.2 8.3 8.4	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41 42 42 42 43 45 45 46 49 49 51 51 51 52 53 53 54 54 54 54
8	<b>Pra</b> 7.1 7.2 7.3 <b>Pra</b> 8.1 8.2 8.3 8.4	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App         7.1.1       Methods         7.1.2       Empirical Investigations         7.1.3       Conceptual Investigations         7.1.4       Technical Investigations         7.1.5       Section Summary         7.1.6       Section Summary         Interfaces for Intervention: Analysing Chatbot Design         7.2.1       Methods         7.2.2       Results         7.2.3       Section Summary         In Summary       Insummary         In Summary       Insummary         In Summary       Introduction         Teaching Context       Introduction         Results       Intervention: Raised Awareness         8.4.1       The Approach Raised Awareness         8.4.2       Principles Helped to Discover Concerns         8.4.3       Ethics Was a New Topic to Many         8.4.4       Some Perceived Ethics as Not Important	<b>41</b> 41 42 42 42 43 45 46 46 49 49 <b>51</b> 51 51 52 53 53 54 54 55 55
8	<b>Pra</b> 7.1 7.2 7.3 <b>Pra</b> 8.1 8.2 8.3 8.4	ctising Ethical Design   Health Data Interfaces         Interfaces for Research: Designing a Mobile Study App	<b>41</b> 41 42 42 43 45 46 49 49 49 <b>51</b> 51 51 52 53 53 54 55 55 55 55

	8.5	In Summary	56
9	<ul> <li>8.5 In Summary</li></ul>		56 57 58 58 58 59 60 61 61 61 61 62 62 62 62 62
		<ul> <li>9.3.2 Reflections on Theories and Design Philosophies</li> <li>9.3.3 Reflections on Teaching and Learning</li> <li>9.3.4 Reflections on Individual Responsibilities</li> </ul>	64 65 65
10	10 Conclusion 67		
10	0011		•••
Pa	rt II	Manuscripts in the Paper Collection	69
Pa 1	rt II Sum	Manuscripts in the Paper Collection	69 70
Pa 1 2	rt II Sum 2.1 2.2	Manuscripts in the Paper Collection mary of Contributions ers Included A Scoping Review of Ethics Across SIGCHI	69 70 73 73
Pa 1 2	rt II Sum 2.1 2.2 2.3	Manuscripts in the Paper Collection mary of Contributions ers Included A Scoping Review of Ethics Across SIGCHI	69 70 73 73
Pa 1 2	rt II Sum 2.1 2.2 2.3 2.4	Manuscripts in the Paper Collection mary of Contributions ers Included A Scoping Review of Ethics Across SIGCHI	69 70 73 73 100 115
Pa 1 2	rt II Sum 2.1 2.2 2.3 2.4 2.5 2.6 2.7	Manuscripts in the Paper Collection mary of Contributions ers Included A Scoping Review of Ethics Across SIGCHI	<b>69</b> <b>70</b> <b>73</b> 73 100 115 161 167 172 179
Pa 1 2	Pape         2.1         2.2         2.3         2.4         2.5         2.6         2.7         2.8	Manuscripts in the Paper Collection mary of Contributions ars Included A Scoping Review of Ethics Across SIGCHI Futures for Health Research Data Platforms From the Participants' Per- spectives Public Attitudes to Digital Health Research Repositories: Cross-sectional International Survey Sharing Access to Behavioural and Personal Health Data: Designers' Per- spectives on Opportunities and Barriers A Value-Sensitive Approach for Ethical Health Research Platforms Is the Automation of Digital Mental Health Ethical? Applying an Ethical Framework to Chatbots for Cognitive Behaviour Therapy Teaching User Experience Design Ethics to Engineering Students: Lessons Learned What is the Future of Data Sharing for Research?	<b>69</b> <b>70</b> <b>73</b> 73 100 115 161 167 172 179 192

## **List of Tables**

2.1	Factors affecting willingness to share data	12
4.1	Data types comprising each card for the sorting activity	24

# **List of Figures**

1.1 1.2	Infographic summarising the concepts of health data flows and interfaces . Diagram describing the thesis outline across chapters and research questions	3 6
2.1	Infographic explaining health data research repositories	10
3.1 3.2	PRISMA diagram displaying the process of selecting sources of evidence . Sample distribution according to publication venue (left) and year (right)	17 18
4.1	Conceptual framework for acceptance based on the interview findings	26
5.1 5.2 5.3 5.4	Distribution of answers for each motivation source	31 32 33 34
6.1	Participants' notes during the focus group	38
7.1 7.2	Health data interfaces supporting the value of agency	43 45
8.1 8.2	Example of a smartwatch prototype created by a group of students Infographic illustrating the five normative principles taught to the students .	52 53

## **List of Boxes**

8.1	Checklist template	54
8.2	Feedback form template	55

## List of Acronyms

- ACM Association for Computing Machinery
- AI Artificial Intelligence
- **CBT** Cognitive Behavioural Therapy
- CHI Conference on Human Factors in Computing Systems
- **DTU** Technical University of Denmark
- **EMA** Ecological Momentary Assessments
- **GDPR** General Data Protection Regulation
- HCI Human-Computer Interaction
- **IoT** Internet of Things
- IRB Institutional Review Board
- ML Machine Learning
- **RRI** Responsible Research and Innovation
- SIGCHI Special Interest Group on Computer-Human Interaction
- UX User Experience
- VSD Value-Sensitive Design

# Part I

# Introduction to the Paper Collection

# **1** Introduction

## 1.1 Context

Data is a precious resource for the healthcare sector. In recent years, the increased digitisation of medical records has made possible more informed decision-making by health professionals, care facilities and financing institutions (Dash et al. 2019). Through the collection of personal sensing data reflecting patients' experiences, habits, and lives, technology has opened up a plethora of possibilities for Artificial Intelligence (AI) and Machine Learning (ML) also to personalise treatments, optimise costs and automate processes (Odone et al. 2019). Practitioners, researchers and academic journals of medicine have praised the increased use of digital data as a revolutionary turn for the care and study of human health (Meskó et al. 2017; Mohr et al. 2018). The coronavirus (COVID-19) pandemic has further underscored the vital role of public health data to support rapid and effective responses (Foraker et al. 2021).

Amidst the vast diversity of health data sources, previous research has often focused upon data generated by citizens, patients and the recipients of healthcare services (Li et al. 2019b; Mentis et al. 2017; Ostherr et al. 2017). This broad data category comprises information collected inside clinical infrastructures (for example, electronic health records), during treatment (for example, therapeutic disclosures) or throughout individuals' daily life (for example, habits tracking). These data sources can be extracted both passively (for example, through mobile sensing) or actively (for example, through self-reporting of symptoms). Novel technology devices are continuously expanding data gathering possibilities in this regard. Computers and smartwatches now accompany wearables, bio-sensors and Internet of Things (IoT) devices for health surveillance (Kristoffersson et al. 2020; Rahaman et al. 2019).

A variety of diverse uses likewise exist for this category of health data, ranging from personal and social to clinical and public applications. For instance, many individuals selfmonitor sleeping hours, step counts and heart rate for their own personal purposes, seeking to acquire knowledge in support of their well-being (Epstein et al. 2015; Fleck et al. 2010; Khan et al. 2017). Health practitioners and clinicians, in turn, can utilise similar data to gain a fuller understanding of patients' routines in-between consultations (Malu et al. 2017; Chung et al. 2015). Patient groups and communities may also find social uses for health data by, for instance, sharing their symptoms via social networks as a means to obtain assistance, inspiration and encouragement (Wicks et al. 2010; Malu et al. 2017). Finally, one of the most prominent examples of health data usage lies in scientific research employing health data to improve prevention and treatment procedures for the general population (Collins et al. 2015). This last application of health data is the main focus of this thesis.

Therefore, while health data possesses the vast potential to support healthcare and scientific research, collecting and maintaining these data sets creates significant challenges in terms of high recruitment costs and the considerable effort associated with large scale data collection. This difficulty has led to the creation of shared-access repositories allowing multiple stakeholders to utilise data simultaneously, especially to support research initiatives (Broes et al. 2017; Hossain et al. 2016). The first among these repositories emulated the concept of bio-banks storing biological samples and genetic materials (Lo 2015; Broes et al. 2018), whilst the most recent also strive to enable the aggregation of digital sources of information (for example, from mobile and wearable sensing). Examples of such shared-access platforms range from the extensive All of Us Research Programme sponsored by the United States government (Collins et al. 2015) to the European COVID-19 Data Platform quickly created amidst the pandemic (Harrison et al. 2021). Smaller enterprises include the global open-data community Open Humans as maintained by its members (Greshake Tzovaras et al. 2019).

Health data in such a shared-access ecosystem is not static but rather 'flows' across entities interested in data access for different purposes and uses. Data collected from citizens and patients alike (data contributors) is valuable to multiple and many kinds of data consumers (researchers, clinicians, governments), meaning that it might be worthwhile maintaining data in repositories for future re-use (See Figure 1.1). The control of collected, stored, and shared data in such a fluid state usually occurs through user interface components employed by the stakeholders involved at each stage of these processes. These health data interfaces may consist, for instance, of mobile apps used by patients for personal data collection or web portals from where researchers obtain access to data sources. The institution of boundaries on these health data 'flows' as entailed in the design of the corresponding health data 'interfaces' is a goal at the very heart of the work of this PhD thesis.



Figure 1.1: Infographic summarising the concepts of health data flows and interfaces

## 1.2 Motivation

As discussed above, the opportunity to research and gain an in-depth understanding of human behaviour comprises one of the primary drivers for collecting and storing health data. However, many have argued that the predominant vision for shared-access repositories over-emphasises the value that data can bring to researchers, clinicians and gov-

ernments — consequently undermining the associated risks to the individuals disclosing their personal data (Kostkova et al. 2016). The intimate observation of affective states, behavioural changes and daily habits are in stark contrast to less invasive information more often employed within healthcare (for example, the number of hospital admissions, the working patterns of practitioners or even physiological data such as blood pressure examined in-clinic). Facilitating access to sensitive data sources can have alarming consequences for the safety, well-being and freedom of data contributors, as explained next (Petelka et al. 2020; Klingler et al. 2017).

First, a wide variety of well-informed perspectives have drawn attention to the commonplace neglect of individual privacy rights within health surveillance initiatives, especially when it comes to debilitating diseases and living conditions that individuals may prefer not to disclose (Rooksby et al. 2019). For instance, the leakage of identifiable information connected to stigmatised mental health conditions can lead to social discrimination and harassment (Murnane et al. 2018). Beyond possible damages to one's reputation, the consequences of undesirable exposures can include cyberattacks and stalking that put individuals' physical and financial safety at risk (Christin et al. 2011).

Another set of concerns often raised by academics and patient advocates alike pertains to issues of social justice and the exclusion of marginalised groups from digital health innovation (Sankar et al. 2017). A lack of significant and genuine efforts to recruit a broader diversity of data contributors is said to only contribute to the historically unfair distribution of benefits across populations, especially when it comes to conclusions derived from unfairly constructed datasets (Klingler et al. 2017). Perhaps worst of all, the abuse of data by authoritarian governments, insurance providers, recruitment agents and forensic investigators can prove unfairly more risky to vulnerable groups (Middleton et al. 2019).

Finally, past work has also highlighted a lack of ethical justification for human data being employed to serve businesses that build robust behavioural models to unlock profitmaking opportunities from personal vulnerabilities (Lupton 2014; Zuboff 2019). Drug advertisers, for example, may have a strong commercial interest in obtaining access to detailed information about patients and family members' diagnoses (Wilkes et al. 2000; Perera et al. 2011). Public concerns regarding such scenarios of data misuse can also directly affect individuals' willingness to contribute health data, which further hinders broad acceptance and aggravates this extant lack of data fairness (Skirpan et al. 2018c).

It is a significant challenge, therefore, to realise the potential benefits of health data sharing while also protecting data contributors from short and long term harm (lakovleva et al. 2019; Eyre et al. 2020; Clark et al. 2019). Contemporary research argues that any vision for more respectful and ethical health data flows must start by acknowledging and addressing sources of discomfort among those who agree to disclose their most personal information (Sankar et al. 2017). However, a lack of attention to these data contributors' perspectives currently hinders the creation and conception of health data technologies better able to attend to individuals' expectations, preferences and concerns (King 2019). This knowledge gap comprises an obstacle for appropriately understanding both the current perception of risks and positive attitudes individuals may hold towards the prospect of health data sharing.

Besides a more robust engagement with data contributors' perspectives, integrating ethical thinking into the conception of health data repositories could further help address the crucial issues that motivate this research. However, the consideration of ethical implications during technology design processes remains one of the most significant and fundamental challenges of the field of HCI, and not only concerning the context of digital health (Frauenberger et al. 2017). Many argue that because ethics itself has remained widely perceived as rigid, static and formalised in nature, its embodiment in digital systems development has proved slow. Professionals reportedly struggle to know how to go about putting 'ethical design' into practice (Stahl et al. 2016; Gray et al. 2021. Excepting a small number of argumentative essays (Nebeker et al. 2019; Roossien et al. 2021; Geneviève et al. 2019; Martinez-Martin et al. 2020), there exist few resources guiding how to translate ethical insights into concrete design alternatives for digital health applications — the second gap in knowledge motivating the work of this thesis.

## **1.3 Research Questions**

As seen above, even though current data sharing practices within digital health intend to benefit society, it is still unclear how the conception of health data interfaces might better respect the preferences of the individuals exposing themselves to risk (data contributors). Therefore, this thesis aims to inform the design of technology ecosystems entailing health data flows by offering evidence and a portfolio of approaches to support technology creators' prioritisation of citizens' needs. The following overarching research question drives this PhD thesis:

# "How might we discover and act upon ethical and data contributors' requirements to design respectful health data flows?"

In particular, we were interested in the specific case of designing user interfaces that can mediate data contributions for health research. As seen above, health data is increasingly being used for scientific investigations, making it a timely research topic. Therefore, in order to answer this overarching research question, we conducted a series of interconnected investigations that investigated the following sub-questions:

**RQ1.** What is the current landscape of data sharing for health research, and which are the contribution options for participants?

**RQ2.** How can technology creators surface, reflect and act upon ethical requirements for user interfaces mediating health data contributions?

**RQ3.** Which technology acceptance factors, core values and ethical principles can guide the design of respectful health data interfaces?

Each of these sub-questions adds a fundamental piece to the overarching goal of this work. The investigations carried out to answer these questions aim to provide a deeper understanding of existing health data repositories, how they currently shape data contribution, and how ethical thinking and data contributors' values could support technology creators in the conception of more respectful data exchanges in this context. The following section describes the methodology chosen to guide the research work.

### 1.4 Methodology

The work of this thesis stands on HCI and human-centred design traditions (Sharp 2003) while building upon the growing thread of 'in-action' ethics research (Frauenberger et al. 2017). In particular, this research focuses on the user group composed of those invited to contribute their data (citizens and patients) to health data repositories. Given this user-centred orientation, the methodological approach of this thesis includes and connects literature reviews, user studies (both qualitative and quantitative), critical appraisals of technology, conceptual design and user interface prototyping (See Figure 1.2). Adopting these different data gathering and analysis techniques aims to attain a deeper understanding of users to inform the initial stages of the interaction design process - establishing

requirements and proposing alternatives (sharp2003interaction). The following section describes the application of each specific method throughout the PhD course.

## 1.5 Thesis Outline

This thesis is composed of two parts. The first part (Chapters 1 to 10) provides the summary of the content from the research papers included in the thesis (See Figure 1.2). In this first part, chapters 1 to 8 summarise and connect context, motivation, research questions, methods, results and contributions of each of these research papers, setting the stage for the manuscripts found in their entirety in Part II. Then, chapters 9 and 10 close this first part and discuss implications for future works. The second part of the thesis presents the eight manuscripts (See Papers 2.1 to 2.8), either published or submitted, produced during the PhD project. The complete bibliography can be found at the end of this second part. An outline of the chapters is described next.



Figure 1.2: Diagram describing the thesis outline across chapters and research questions

Following this introductory chapter, Chapter 2 reviews a broad set of examples of health data repositories, including how these repositories configure different styles of data contribution for scientific investigations. Chapter 2 furthermore presents a comprehensive narrative review of factors for willingness to share data that have significant relevance to the acceptance of health data technologies. This chapter thus provides an in-depth look into the literature on digital health research and the importance of trust, privacy and reciprocity to this context. This chapter is based on the Papers 2.2 and 2.3.

Chapter 3 contributes a scoping review yielding a coherent up-to-date mapping of ethical thinking and doing in HCI. This review provides an overview of ethical concerns related to technology development and introduces methods, theories, and practical recommendations for putting ethics into action. Review findings constitute the foundation for the subsequent chapters, as they illustrate how to apply and explore matters of ethics in user-centred design. This chapter is a summary of the Paper 2.1.

Chapters 4 and 5 follow a mixed-methods approach to eliciting data contributors' attitudes towards data sharing with health data repositories and discovering which factors govern their decisions to participate. First, Chapter 4, based on the Paper 2.2, describes a qualitative investigation (interview study) illuminating what matters to 12 young adults when they are presented with a request for sharing different types of data with health research repositories. Then Chapter 5, based on the Paper 2.3, extends and complements these findings by engaging a much larger sample (1600 participants) through an online survey generating a quantitative understanding of principal sources of motivation, reasons for concern and data sharing preferences. Chapter 5 also contributes a comparison between two national contexts (Brazil and Denmark) and the impact of demographic characteristics on contributors' attitudes.

Chapter 6 then moves to the perspectives of designers and developers (technology creators) working on the conception of new digital health technologies. This chapter enabled an understanding of what this stakeholder group considers critical factors in data contributors' acceptance of data sharing through a focus group study. The enablers and barriers identified in this study complement the insights gathered from data contributors' perspectives, as the focus group points to particular areas where technology creators require support to conceive systems that respect data contributors' values. This chapter summarises the Paper 2.4.

Chapter 7 turns to more concrete design explorations, building again upon the findings of previous chapters. This chapter first engages with Value-Sensitive Design (VSD) to guide the application of ethical insights in the form of a prototype for a health research mobile app. This chapter then contributes several design reflections regarding yet another type of health data collection interface (psychotherapy chatbots), engaging with an Alethics framework to illustrate how the issues involved in sensitive data disclosures during digital therapy might be appraised. This chapter presents the insights from the Papers 2.5 and 2.6.

Chapter 8 returns to engaging technology creators through a pilot study conducted with university students to examine their learning about ethical User Experience (UX) design. This chapter, based on the Paper 2.7, provides insight into (future) technology creators' experiences when engaging with ethical principles for designing a smartwatch app to collect and share heart monitoring data with clinicians and researchers. This chapter thus contributes a report on the students' impressions after employing ethical frameworks, which expands the insights about ethical design practice acquired through our own first-person experiences described in Chapter 7.

Finally, Chapter 9 discusses the overall implications of the work of this thesis and presents a vision for ethical acceptance based on the Paper 2.8. In particular, this chapter presents a compilation of design recommendations for health data interfaces based on the findings of the user studies. It also features deliberations concerning how ethics can be applied and integrated into user-centred design methodologies, considering our learnings and reflections on the limitations of individualised ethical action we observed throughout this work. Chapter 10 then concludes this first part of the thesis with some final remarks.

## 1.6 Contributions

This PhD thesis provides the following contributions. First, it delivers an in-depth understanding of the health data repository context, including a review of crucial technology acceptance factors. Second, it provides a coherent mapping of pathways for ethical thinking and doing drawn from the HCI literature. Third, it presents various insights concerning the practice of health data sharing as directly expressed by data contributors and technology creators. Finally, this thesis illustrates how we might teach and apply ethically-centred approaches to user interface design. This combined evidence makes possible a more informed and critical discussion of alternative futures for health data technologies. Through these contributions, this PhD demonstrates how ethical and contributor centred perspectives can be integrated to support the design of more respectful health data flows.

## 2 Mapping the Literature | Health Data Repositories

As digital technologies enable ever-expanding access to information regarding human health, behaviour and well-being, many have looked towards developing data repositories to support the assimilation and re-use of datasets, especially for research purposes. Each type of repository has its style of enabling data transference, thus configuring different kinds of relationships between its users. These systems may offer significant advantages, yet their acceptance by data contributors depends on several factors.

In order to characterise this technological context - the main focus of the user studies described later in this thesis - this chapter presents an overview of types of data repositories primarily built to support health research. The chapter also summarises the challenges to public acceptance of these systems. This content draws from two cross-cutting papers: *"Futures for Health Research Data Platforms From the Participants' Perspectives"* (Vilaza et al. 2020) and *"Public Attitudes to Digital Health Research Repositories: Cross-sectional International Survey"* (Vilaza et al. 2021a), available in Part II as the Papers 2.2 and 2.3.

## 2.1 Introduction

Data about human health is fundamental for the progress of health sciences, as medicine strongly relies on the observation of connections between habits, symptoms and the effect of interventions (Bowling et al. 2005; Pagoto et al. 2013). Data sources for health research may come in different forms, from blood tests to medical images, but very often, researchers and clinicians gather data through in-person interviews and paper-based self-reports (Stone et al. 2007). In this regard, digital technologies are an appealing novelty, allowing the collection, storage and management of many and diverse kinds of information more efficiently than ever before (Hripcsak et al. 2015). Mobile apps employed to enable Ecological Momentary Assessments (EMA), for example, can sample patients' self-reported status in real-time (Van Berkel et al. 2017), with wearable sensing systems additionally extracting indicators of health from mobility to physiological signals (Trifan et al. 2019). Such digital data sources are now used in combination with traditional observations to support future improvements in care (Dinh-Le et al. 2019).

Thanks to these recent achievements of health digitisation, shared-access data repositories have increasingly emerged as means to facilitate the use of longitudinal data from patients and healthy individuals alike (Dey et al. 2017; Pencina et al. 2016). These platforms allow multiple clinicians, researchers and projects to make use of the same data for a wide variety of purposes (Veličković et al. 2017; McPadden et al. 2019). In particular, shared-access platforms offer advantages of reduced recruitment and data collection costs while still providing the information desired by clinicians, researchers and public workers (Song et al. 2010; Hamer et al. 2020). Configurations and designs of health data repositories are diverse, each posing challenges and opportunities. In this next section, we provide an overview of the types of data repositories currently implemented, in which data is used primarily (but not exclusively) for research purposes.

## 2.2 Existing Platforms and Shared-Data Styles

A significant percentage of research has been conducted about the diversity of data collection technologies (Van Berkel et al. 2017; Kumar et al. 2020) and possible personal and



Figure 2.1: Infographic explaining health data research repositories

social uses of health data (Wood et al. 2015; O'Doherty et al. 2016; Frost et al. 2008). However, a gap lies in research focused on the repositories that enable data to be reshared between multiple users, particularly health researchers. The following literature review aims to address this gap.

In order to develop a broad understanding of this technology ecosystem, we gathered and reviewed a set of existing health data platforms. The complete details of how we searched for examples of platforms can be found in the Paper 2.2. In particular, to shed light on the many possible styles of data contribution, we categorised the platforms we found according to the level of agency repository owners provide to data contributors. We referred to as 'top-down initiatives' the projects focused on attending to the demands of governments, institutions and researchers for access to large-scale, population-level data. In turn, we described as 'participant-driven initiatives' those attempts to bring participants, patients, and volunteers closer to data sharing practices by empowering these stakeholders to adopt a more active role in their participation. Next, we present specific examples of each of these broad categories.

### 2.2.1 Top-down Initiatives

Notable initiatives include national programmes that aim to gather representative samples of a country's population. An influential example is the UK Biobank, a repository that back in 2010 already had records of 500,000 patients (Sudlow et al. 2015). Denmark is also widely known for its well-established national registry, which digitally stores public health data from the entire Danish population (Schmidt et al. 2014). Other countries have similar national databases in development, including India (Sahay et al. 2020) and China (Li et al. 2019a). More recently, the All of Us Research Programme in the United States has been described as a pioneer for its aims to store not only clinical and hospital records but also patient-generated data from mobile devices and wearables (Sankar et al. 2017).

Other examples include repositories that re-purpose previously collected data. For instance, the mobile personal health tracking app 'Achievement' invites its users to voluntarily share daily behaviour logs with researchers associated with the privately-owned platform Evidation (Clay 2020). This approach comprising user 'opt-in' can also be found in the commercial DNA testing service 23andMe (King 2019) and the social network PatientsLikeMe (Wicks et al. 2010). Similarly, non-commercial repositories enable the re-use of data from previously conducted clinical trials, such as the non-profit Vivli (Li et al. 2019b) and the European Union's COVID-19 Data Portal (Alamo et al. 2020). These platforms function as data brokers, connecting researchers with the de-identified raw data samples from past studies (Bierer et al. 2016). In general, these 'top-down initiatives' are platforms for which data access is typically granted to institutions selected by repository owners without the participation of data contributors. In some instances, additional data consumers and stakeholders may later request access to data. In both situations, permissions are usually approved internally by those managing the repository, which means that those who contributed their data have little involvement in the process. Such practices prioritise the needs of the higher-level institutions behind the platforms, which raises ethical concerns regarding how much autonomy is given to data contributors to express their concerns and choices over data use (Kostkova et al. 2016). Another ethical issue is that commercial and national databases might have the potential to produce unfair applications, should, for instance, governments or companies hide an agenda that goes beyond employing data for healthcare purposes (Zuboff 2019).

### 2.2.2 Contributor-driven Initiatives

In contrast with these so-called top-down initiatives, other data repositories are configured and hosted by the individuals who collect data for their own personal use and choose to make their data available to others. Examples of these 'participant-driven' platforms include the community-based Open Humans project (Greshake Tzovaras et al. 2019), in the case of which users contribute to open-access projects analysing Fitbit, Twitter and Apple Health data. The idea of the Open-Humans project is to function as a community through which members propose ideas for data explorations and share tools and tips for data analysis.

Similarly, although less community-focused, in the Project Baseline (Maxmen 2017), the initiative to contribute to a data repository begins with the volunteers who independently access the platform select which studies to join. Project Baseline also attempts to get contributors more involved in data collection and sharing by offering data-driven insights and information about the research. A small number of other crowd-sourcing projects present similar features, such as participatory disease surveillance surveys in which individuals provide data to support epidemiological controls (Neto et al. 2020). The surge of COVID-19, for example, has triggered many more of such collaborative 'citizen science' projects in the last years as an attempt to boost access to population data (Garg et al. 2020).

One particularly prominent ethical challenge with such participant-driven platforms is that they often display low recruitment rates, thus compromising cohort diversity. The Open-Humans project, for instance, had 6,800 users in 2019, a stark contrast to the more than 500,000 patients of the UK Biobank (Estrada-Galiñanes et al. 2020). In addition, crowd-sourcing projects may have fewer mechanisms for privacy control, as compared to the tight filters applied by national programmes that pre-screen who gets to access data (Anhalt-Depies et al. 2019). Ideally, health data repositories should be able to successfully balance the broader reach of national programmes with the active participation of data contributors. The critical factors involved in a broader adoption of these systems is often studied through the lens of 'public acceptance' or 'technology acceptance', which is the topic of the second literature review we conducted.

## 2.3 Factors Influencing Willingness to Share Data

The above overview of platforms revealed the organisational and political characteristics of a set of health data platforms. In order to shed light on the human-centred and experiential factors, we reviewed previous related work focused on enablers and barriers for willingness to share data. It is crucial to understand prior research around data sharing to develop more ethical practices that respect the reasons why data contributors may feel uncomfortable disclosing information and prioritise their expectations for fair data exchanges (Nadal et al. 2020; Barkhuus 2012).

We gathered a comprehensive set of previous works presenting examination and discussion of what matters most to those collecting, sharing and re-sharing personal health and behavioural indicators (See Table 2.1<sup>1</sup>). These works include studies extending beyond the context of health data repositories to broadly illuminate data contributors' key concerns, expectations, and values. Details on the review and analysis process are available in the Paper 2.2.

Factors	Effects	References
Motivation	Motivation decreases if data col- lection is burdensome	Cohen et al. 2015; King et al. 2006
	Motivation decreases if the purpose is perceived as not useful	Kraft et al. 2018; Tan et al. 2018; Ziefle et al. 2016; Barkhuus et al. 2003
Privacy	Some types of data carry social stigma	Kelley et al. 2017; Rooksby et al. 2019
	Some types of data are per- ceived as sensitive	Rudnicka et al. 2019; Bhatia et al. 2018; Garcia-Ceja et al. 2018
Trust	Trust can be damaged by previous experiences	Anderson et al. 2012; Corbett et al. 2018; Adjekum et al. 2018
	Trust can be harmed by data misuse and lack of transparency	Middleton et al. 2019; Murmann et al. 2017
Consent	Consent forms are lengthy and complex	Luger et al. 2013; Ostherr et al. 2017
	Consent forms do not offer flexi- ble options	Hardy et al. 2018; Kaye et al. 2015; Nurgalieva et al. 2019

Table 2.1: Factors affecting willingness to share data

### 2.3.1 Motivation

The review revealed motivation to be essential for willingness to share data. One prominent reason why motivation is important is that it is challenging to keep individuals consistent with data collection requests in the long term if they do not have solid reasons for doing so (Rotman et al. 2012; Choe et al. 2014; Cohen et al. 2015). Understanding what motivates data contributors can lead to benefits more aligned to what is most valuable for them.

Previous research indicates that motivation can stem from *altruism* and a desire to contribute to a cause. Such charitable reasons include supporting science (Wiggins et al. 2019) and helping future patients (Gatny et al. 2012). The engagement of participants in decisions about research projects is also deemed as a motivating incentive (Shippee et al. 2015; Anderson et al. 2012). On the other hand, a different source of motivation discussed in prior studies refers to the practice of offering monetary incentives in the form of gift vouchers or cash (Lacetera et al. 2013; Benndorf et al. 2018). These types of incentives contrast in that some focus more on societal benefits whilst others target personal gains.

<sup>&</sup>lt;sup>1</sup>Based on Table 1 by Vilaza et al. 2020

Previous works have further discussed the role of other types of personal benefits in increasing motivation. For instance, getting access to DNA test results is often the primary motivation for individuals to share data with services such as 23AndMe King 2019, in a similar fashion that well-being maintenance is the driver of most individual self-tracking of behaviour (Fleck et al. 2010; Chung et al. 2017; Kersten-van Dijk et al. 2016). Despite benefits arising from such sources, others have warned about potential harms of such 'self-focused' approaches such as performance anxiety (West et al. 2016; Epstein et al. 2016; Chua et al. 2016).

#### 2.3.2 Privacy

Privacy is known as a significant barrier to personal data sharing, especially when it comes to behavioural and health-related data (Kostkova et al. 2016; Nurgalieva et al. 2019). Prior research has highlighted as prominent fears among data contributors that the leakage of sensitive diagnoses can lead to social discrimination, given that some types of personal data are challenging to be anonymised entirely, such as DNA samples (Middleton 2018). Government surveillance and undisclosed commercial interests are other significant concerns that may increase unwillingness to share data (Lemke et al. 2010; Middleton 2018). When data is used beyond the limits individuals have agreed upon, 'contextual integrity' is said to be broken, as information gathering and dissemination become untied to the governing norms of specific contexts (Nissenbaum 2011).

For these reasons, individual participants have often expressed a desire to impose limitations and boundaries to protect their autonomy (Raento et al. 2008; Rooksby et al. 2019) and social reputation (Kelley et al. 2017; Palen et al. 2003; Leary et al. 1990; Petelka et al. 2020). The process of data collection can also trigger feelings of paranoia and loss of freedom, which are direct harms to well-being (Vaizman et al. 2018; Saunders et al. 2017; Caine et al. 2012). The idea of a 'privacy paradox', however, states that there can be a mismatch between expressed willingness to share data and actual behaviours (Norberg et al. 2007). Regardless of the attitude of data contributors, privacy protection in respect of others' physical, emotional, and social integrity is known as a fundamental care commitment (Bagnasco et al. 2020).

Prior research shows that individuals may be willing to disclose certain information if they believe it can be useful to society and themselves (Ludford et al. 2007; Sannon et al. 2019; King 2019; Barkhuus et al. 2003). Agreeing to share personal data may depend, however, on the perceived sensitivity of the data request (Rudnicka et al. 2019). For instance, audio recordings (Di Matteo et al. 2018), browser history (Bhatia et al. 2018), social media activity (Garcia-Ceja et al. 2018), camera pictures (Rooksby et al. 2019), financial status (Weitzman et al. 2012), home address (Ludford et al. 2007), and sexually transmitted diseases (Weitzman et al. 2012) are among the information points perceived as least comfortable to disclose by data contributors.

### 2.3.3 Trust

The analysis shows that trust is fundamental for the acceptance of data sharing. When it comes to data repositories, trust in the institutions, organisations and companies managing that data is a significant factor for data contributors' willingness to share (Anderson et al. 2012; Pratap et al. 2019; Leon et al. 2013; Joinson et al. 2010). Previous research has found that academic institutions are trusted more often than insurance companies, governments and companies (Domaradzki et al. 2019). The freedom to choose what information is shared with whom can increase trust among individuals and is considered more ethical conduct than denying this right (Abdelhamid 2018; Petersen 2018; Kostkova et al. 2016).

The literature lists other ethically motivated enablers of trust which are particularly relevant to digital systems: customisable features, engaging interfaces and honest communication (Adjekum et al. 2018). In general, transparency has been frequently associated with trust, as previous works extensively promote the role that transparent information can play in promoting data sharing acceptance and ethics (Domaradzki et al. 2019; Murmann et al. 2017). Trust, however, is not a static construct. The development and maintenance of trust is a long term process, and it is not uncommon that public perceptions of trustworthiness change with time (Corbett et al. 2018). Evidence of data misuse can, for instance, damage trusting inclinations and decrease willingness to share data (Middleton et al. 2019).

#### 2.3.4 Consent

Before data collection and sharing, it is the usual and ethical practice to obtain voluntary consent from data contributors. The European General Data Protection Regulation (GDPR) has more strongly reinforced this requirement by demanding that any digital technology that collects personal data must obtain user consent (Voigt et al. 2017). At a minimum, consent forms should inform users about which information is being collected, how it will be shared and why the data is needed (Harari et al. 2016). However, previous research has emphatically argued that there is an ethical need to improve consent practices beyond these measures (Brehaut et al. 2012; Hutton et al. 2018). Consent forms are criticised for being overly lengthy and complex (Luger et al. 2013; Ostherr et al. 2017), whilst users are often not provided with flexible options (Hardy et al. 2018; Kaye et al. 2015; Nurgalieva et al. 2019).

## 2.4 In Summary

This chapter contributes an overview of multiple and diverse examples of health data repositories, accompanied by a review of factors identified as critical to the acceptance and willingness to share personal data. Although many have studied and proposed maximising technology adoption, if we solely focus on public acceptance and the possibility that people may share data despite disagreeing with terms and conditions, we might overlook critical ethical issues. This review already pinpoints a few ethical challenges caused by large-scale repositories' restrictive contribution style, including, for instance, the negligence of data contributors' autonomy. From a user-centred perspective, the experience of contributing to a health data repository should be comfortable and respectful. For this reason, to develop more acceptable and ethical health data research platforms, we need to understand how others have 'actioned' ethics in practice. In order to step ever closer to the aim of supporting the design of more ethically-informed health data flows, the next chapter presents a scoping review of ethics across HCI.

# 3 Mapping the Literature | Ethics and HCI

As seen across the previous chapters, to realise publicly acceptable systems that respect contributors' autonomy, well-being, and privacy, it is imperative to examine the conception of health data technologies through an ethical lens. For this reason, this chapter presents a scoping review conducted to develop a synthesis of just how the literature has examined and proposed the application of ethical reflection to research and design processes. As HCI is the academic discipline where human-centred design reflections are most often discussed, we focused this review on papers published within this key domain. Developing such a broad overview of ethical thinking and doing across this knowledge field consists of an ideal starting point for selecting and applying appropriate methodologies and theories to conceive more ethical systems. The content of this chapter is derived from the full paper titled *"A Scoping Review of Ethics Across SIGCHI"*, currently under review and available as the Paper 2.1 in Part II.

### 3.1 Introduction

Over the years, ethics has consistently proved a fundamental focus of human-centred design (Molich et al. 2001). Historically, user research ethics has been primarily studied in terms of respect for the persons involved in research processes (Benford et al. 2015). Today, ethics remains one of the grand technology design challenges, as illustrated by contemporary debates around algorithmic bias, data misuse, and deceptive user interfaces (Durrant et al. 2018; Davis 2020).

However, engaging in ethical decision-making in practice remains a challenge for technology creators (Barry et al. 2020). Ethics is in itself a complex philosophical subject with a varied history not always known in detail by computing professionals (Stahl et al. 2016). In broad terms, ethics seeks to address questions of human conduct by utilising conceptions of right and wrong, desirable and detrimental, which are already very difficult reflections in any context (Baggini 2018).

Formal approaches to ethics, such as the Association for Computing Machinery (ACM) Code of Ethics and Professional Conduct (Gotterbarn et al. 2018) and Institutional Review Board (IRB) procedures, aim at providing official guidance for ethical action. However, these are criticised for being overly bureaucratic, detached from the needs of research in the field, and insufficient for accounting for different oversight requirements across jurisdictions (Frauenberger et al. 2017; Benford et al. 2015). The reality is that the circumstances of each project can vary, leading perhaps to situations not predicted by top-down commandments (Nathan et al. 2016; McNamara et al. 2018).

Therefore, the continuous and proactive integration of critical perspectives during design processes is a necessary step to enabling ethical doing in practice (Shilton et al. 2017; Frauenberger et al. 2017). Technology creators, researchers and UX design professionals may find support for this task by consulting the academic literature. Scientific publications serve as formal sources of knowledge as to how to take an abstract concept such as ethics and integrate it into the cycle of the design, development and adoption of digital technologies (Tolmeijer et al. 2020). Previous systematic reviews, for instance, have discussed ethics in relation to a wide range of technologies, such as autonomous systems (Tolmeijer et al. 2020; Vandemeulebroucke et al. 2018; Zoshak et al. 2021), assistive devices (lenca et al. 2018; Burwell et al. 2017), and technologies for children (Van Mechelen et al. 2020).

Despite consisting of important sources of knowledge, findings from these previous systematic reviews are limited to their target application domain. A systematic review of computing ethics across Computer Science journals indexed by Scopus, as conducted by Stahl et al. 2016 comes closest to an aggregation of academic resources from multiple domains that could support researchers, designers and professionals in navigating ethical decision-making in their work. However, the authors of this previous review did not capture publications from important HCI conferences, where the majority of human-centred ethical reflections are likely published.

Given the relevance of ethical reflections spanning the field of HCI to human-centred design (Stephanidis et al. 2019), and the possibility that there might exist variable standards and expectations for ethical thinking across academic communities, there is therefore a need to conduct further reviews to complement current literature accounts. With this purpose in mind, a scoping review of ethics was conducted, focusing on how ethics is understood, employed and discussed within the HCI literature. The main goal of this review was to serve as a starting point for the process of designing new digital technologies and to provide insight into the ethical creation of health data ecosystems alike. The following research questions motivated and guided this literature review:

- 1. Which ethical considerations, technologies and user groups have been the subjects of previous papers?
- 2. Which approaches, methods and theories have been employed to explore ethical considerations?
- 3. Which practical recommendations for putting ethics into action have been reported and discussed?

## 3.2 Method

Seeking to answer the research questions above, we conducted a *scoping review*, which is a systematic literature review procedure used to produce a description of the scope of previous research about a topic (Munn et al. 2018). We followed the scoping review methodology established by Arksey and O'Malley (Arksey et al. 2005) and, in order to maintain rigour, we followed the Preferred Reporting Items for Systematic Reviews and Meta-Analyses extension for Scoping Reviews (PRISMA-ScR by Tricco et al. 2018) to structure our process. The complete description of the procedure conducted to select eligibility criteria, information sources, search queries and data items for analysis can be found in detail in the Paper 2.1.

This review targeted papers published at conferences sponsored by the ACM Special Interest Group on Computer-Human Interaction (SIGCHI). The motivation for focusing the search to SIGCHI venues came from the need to produce a diverse set of papers on where human and user-centred work is published within computer science while keeping the process manageable and transparently delimited instead of attempting to capture all HCI papers about ethics. Therefore, we searched the ACM Digital Library for full papers published from January 1st 2010 to December 31st 2020, containing the keyword 'ethic\*' within the title, abstract or authors' keywords list in order to identify papers with ethics as their core focus. Data was charted in a spread-sheet, and tables, visualisations and text-based narratives were used to describe the results.

## 3.3 Results

This review comprises a broad and diverse compilation of sources of knowledge and inspiration serving as a foundation for the ethical design of digital technologies broadly,

and in the context of the work of this thesis, health data ecosystems in particular. The initial search produced a corpus of 461 papers. Preliminary screening then excluded papers not peer-reviewed, full-length archival publications (314 manuscripts). The remaining 147 papers were subsequently assessed according to the inclusion criteria previously defined, resulting in the removal of 18 papers. This process then led to a final sample of 129 papers comprising the source of evidence for this review (See Figure 3.1<sup>1</sup>). The following section summarises the main findings of the review, and those more closely related to the thesis are more strongly emphasised. The complete description of results can be found on Paper 2.1.



Figure 3.1: PRISMA diagram displaying the process of selecting sources of evidence

### 3.3.1 Characteristics of Sources of Evidence

This sample consisted of 129 full-length papers published at SIGCHI-sponsored venues. The vast majority were published in the last three years: 2018 (17%, n=22), 2019 (20.9%, n=27) and 2020 (28.6%, n=37). Overall, most of the of papers (59.6%, n=77) were published at Conference on Human Factors in Computing Systems (CHI). Figure 3.2<sup>2</sup> shows the distribution of papers across publication venues and years in detail.

### 3.3.2 Objects of Ethical Consideration

Several types of technology proved the targets of ethical consideration (See Paper 2.1 for the complete list). Of particular interest to the work of this thesis are those papers (highlighted in bold for improving the reading experience) focused on **personal health** (Wärnestål et al. 2014; Shapiro et al. 2020; Rooksby et al. 2019; Barry et al. 2017; Purpura et al. 2011; Dahl et al. 2012; Sanches et al. 2019), web and mobile technologies used to collect **data for research** (McMillan et al. 2013; Morrison et al. 2012; Bowser et al. 2017; Snyder 2020; Pinder et al. 2017), **biosensors** (Howell et al. 2018; Fox et al. 2019; Curmi et al. 2014 and **DNA repositories** (Baig et al. 2020). The presence of papers directly related to the research topic of this thesis serves to highlight digital health data

<sup>&</sup>lt;sup>1</sup>Figure extracted from Paper 2.1

<sup>&</sup>lt;sup>2</sup>Figure extracted from Paper 2.1



Figure 3.2: Sample distribution according to publication venue (left) and year (right)

technologies as a primary source of broadly relevant ethical concern. Several papers also emphasised the importance of examining the relationship between ethics and technology creators - another key thesis subject. The sample included several papers about **HCI researchers** (Munteanu et al. 2015; Moncur 2013; Hirsch 2020), **industry professionals** (Gray et al. 2020; Chivukula et al. 2020), and **students** learning about technology design (DiPaola et al. 2020; Bilstrup et al. 2020; Foley et al. 2020; Shapiro et al. 2020; Sabie et al. 2019).

The review of these papers underlines a set of ethical concerns that technology creators need to consider in their design commitments to respect end-users genuinely. Those papers concerning digital health systems highlighted in particular harms to well-being and adverse psychological effects (Sanches et al. 2019; Hodge et al. 2020; Foley et al. 2020; Barry et al. 2017; Toussaint et al. 2020). Unsurprisingly, concerns regarding privacy invasion and security breaches were most often discussed in relation to passive sensing (McMillan et al. 2013; Morrison et al. 2012; Curmi et al. 2014; Bowser et al. 2017; Toussaint et al. 2020; Pierce et al. 2020; Seymour et al. 2020; Fritsch et al. 2018; Sailaja et al. 2018; Knowles et al. 2019), shared-access data repositories (Sun et al. 2019; Baig et al. 2020; Lin et al. 2020) and social media networks (Andalibi et al. 2020; Lyckvi et al. 2018). Concerns around social injustice most frequently referred to flawed AI decisions (Troiano et al. 2020; Madaio et al. 2020; Strengers et al. 2020; Veale et al. 2018; Wächter et al. 2018; Wu et al. 2019; Ekstrand et al. 2018), and were most often raised by studies examining the issue of the digital divide (Rodil et al. 2020; Sultana et al. 2018; Harvey et al. 2014; Irani et al. 2016). Stigma and social discrimination were very often discussed by those papers focused on individuals with mental illnesses (Snyder 2020; Rooksby et al. 2019; Sas et al. 2020 and sexual minorities Strengers et al. 2020).

### 3.3.3 Approaches to Ethics in Design Research

The findings of this review furthermore highlight how ethical considerations have been approached, explored and engaged with by the HCI literature. Most papers report empirical studies employed to consult stakeholders directly, although several design methods were also encountered which propose the possibility of being conducted without stakeholders' presence. In addition to empirical approaches, many other papers employ and engage with a wide variety of philosophical and theoretical frameworks. With these findings, we compiled the following vast portfolio of options for technology creators to apply in their

#### work.

#### **Empirical Approaches**

Many user studies across this sample employed well-known user research methods, such as interviews (Seymour et al. 2020), questionnaires (Skirpan et al. 2018c), focus groups (Lazar et al. 2016), and co-design workshops (Foley et al. 2020), conducted in the lab and the field (the complete list of methods can be found on Paper 2.1). Participants in these studies included previous or current end-users (Seymour et al. 2020), potential end-users (Scheutz et al. 2016, close contacts of end-users (Wärnestål et al. 2014), and technology creators and researchers (Moncur 2013). The majority of papers reported participatory workshops (21.7%, n=28), followed closely by field and interview studies which are also featured frequently (19.3%, n=25).

Speculative design was frequently employed to surface, extrapolate through, and raise awareness of how technologies can imbue values, ideologies and behavioural norms. Fictional scenarios and provocative prototypes are several of the most-common formats through which speculations are expressed (Toussaint et al. 2020; Pierce et al. 2020; Encinas et al. 2018; Snyder 2020; Knowles et al. 2019; Troiano et al. 2020). A small number of field studies go a step further, proposing 'activist systems' to realise societal changes in the real world (for example, systems targeting fairer conditions for Amazon's Mechanical Turk workers by Irani et al. 2013; Barbosa et al. 2019; Kasunic et al. 2019).

Distinct from approaches entailing the direct involvement of stakeholders, other papers promote the use of 'expert-only' methods: empathy-building tools, personas, stakeholder matrices and card-based materials to facilitate self-reflection and collaboration in teams (Sas et al. 2020; Ballard et al. 2019; Madaio et al. 2020). Critical appraisals of user interface features and scenarios have also been reported (Purpura et al. 2011; Fox et al. 2019; Lacey et al. 2019), as well as content analyses of online forum discussions (Ma et al. 2018) and manifestos (Fritsch et al. 2018). Several papers take stock of the literature and past projects to discuss ethical issues without involving stakeholders through argumentative essays (Brown et al. 2016; Mirnig et al. 2019), case studies (Irani et al. 2016) and literature reviews of specific subject areas (for example, a review of affective computing in HCI by Sanches et al. 2019).

#### **Theoretical Approaches**

Several papers discussed perspectives drawn from the domains of moral and social philosophy, in addition to political and design theory — composing approximately one-third of the sample. VSD was the framework most frequently found (10%, n=13). VSD theory proposes designers to accounts for 'human values' in a systematic manner (Dahl et al. 2012; DiPaola et al. 2020) by conceptualising as an agreed-upon vocabulary to guide and frame ethical reflection (for example, by considering how design decisions can affect the value of physical safety). Another perspective well-represented in the sample is Critical Theory (6.9%, n=9), an approach that challenges assumptions about power relations in society, often taking the form of Critical Design (Bardzell et al. 2013). Critical stances often question the role product design plays in social and cultural spheres and strive to foster debate in relation to an often-felt prioritised commercial focus (Irani et al. 2016.

Although less frequently, Care Ethics was also represented in the sample (for example, Sabie et al. 2019), which is a philosophical lens that shifts the moral focus towards embodied, situated and emergent relationships of mutual care (La Bellacasa 2017). When applied to design, Care Ethics foregrounds mutual commitments as fundamental for decision-making, in contrast with outlooks centred on norms and duties. Utilitarian ethics was encountered equally infrequently (for example, Niforatos et al. 2020). Utilitarianism holds that an optimal ethical choice can be achieved by 'calculating' the greatest good for the

greatest number of people (Boutilier et al. 2015). Given the difficulty of weighing positive and negative outcomes, utilitarianism risks turning decisions into 'moral arithmetic' (Mirnig et al. 2019).

Other theoretical orientations appear even more rarely within this sample. A small number of papers engage with Social Justice theory (Dombrowski et al. 2016), describing Social Justice-Oriented Interaction Design as a framework to help designers commit to plurality, advocacy, and political considerations. Similarly, Queer and Feminist theories appear only twice in regard to issues of inequality in design (Strengers et al. 2020; Nguyen et al. 2020). Virtue Ethics, which emphasises the cultivation of ethical wisdom through personal experience, appears in a single paper calling for greater scrutiny of the subjective assumptions underlying design choices (Barry et al. 2017). Biopolitics was likewise employed in a single instance as an analytical lens applied to question the authority of biosensing systems (Howell et al. 2018).

Similarly, Foucault's theory of 'care of the self' was used to emphasise technology's epistemic dangers as a source of power (Kou et al. 2019). Thanato-sensitive design was introduced as an approach to integrating the inevitability of mortality such that design can be more sensitive to this part of life (Massimi et al. 2011). Likewise, Somatic Ethics was explored in a single paper, positioning ethics as an experience to be lived through the body, not only the mind (Eriksson et al. 2020). Finally, amidst a predominant emphasis on Western ethical frameworks across the sample, Confucian Ethics was the sole non-Western perspective encountered, employed by a single paper exploring individual roles as sources of collective ethical obligations (Williams et al. 2020). Duty-based ethical frameworks were not explicitly applied by any single paper within the sample, although studies describing normative rules could be interpreted as examples of this deontological perspective, such as categorised guidelines (McMillan et al. 2013) and checklists (Knowles et al. 2019).

#### 3.3.4 Recommendations for Putting Ethics into Action

Finally, the review also presented several practical recommendations for tackling ethical concerns as they arise. Approaches spanned broad research participation, design choices, and responsibility roles. Therefore, this sample of the HCI literature provides examples of practical wisdom and knowledge that can be further applied to our goal of eliciting and acting upon ethical requirements in the context of health data ecosystems.

#### Ethics as Participation in User Research

The ethical involvement of people in design research is a major topic discussed across the sample. The following suggestions are particularly relevant for health data ecosystems design because they refer to protecting the best interests of individuals contributing with data for research. First, the literature shows a turn towards promoting an active engagement of participants concerning the configuration of projects (Vines et al. 2013; Bell et al. 2019). This recommendation includes that researchers make sure participants benefit from being part of a study, for example, by acknowledging participants as co-creators (McNally et al. 2016; Hodge et al. 2020), negotiating compensation options (Howard et al. 2019) and gifting working prototypes (Hodge et al. 2020. Some papers also propose ways for researchers to become more involved with the communities they study, for example, by volunteering with organisations (Foley et al. 2020).

Many other suggestions identified are noteworthy in that they would not always be considered by standard research procedures, such as writing consent forms in an easy to understand language (Sitbon et al. 2020), granting participants enough time to get acquainted with the technology before a study (Waycott et al. 2016), and allowing data access to be revoked anytime (Peacock et al. 2018; Benford et al. 2012. In the case of the re-use of publicly available data, several papers advises reducing the risk of re-identification (for example, Abbott et al. 2019) and informing participants about the new usage of data (for example, Bowser et al. 2017). If participating in and conducting research becomes an emotionally charged experience, several papers proposed that researchers should be trained to minimise and recognise sources of distress during their research (for example, Hirsch 2020). In addition, the presence of domain experts, counselling, group discussions, and a healthy work-life balance are other noteworthy suggestions (Moncur 2013; Massimi et al. 2011).

#### **Ethics as Design Choice**

According to several of the reviewed papers, design choices can be the 'vehicle' through which ethical concerns are addressed (See Paper 2.1 for a broad and detailed overview of the forms these design choices can take). We focus here, for the sake of brevity, on those most clearly related to the practice of health data ecosystem design. This analysis of the HCI literature pointed towards transparent communication of system intentions and technical limitations as supporting of user autonomy and the capacity of making informed choices (Shi et al. 2020; Baig et al. 2020). Privacy concerns, it is likewise suggested, may be managed should user interfaces provide mechanisms for data control (Rooksby et al. 2019; Sun et al. 2019). Interactions that trigger obsession, performance anxiety, and the over-optimisation of behaviour should be avoided in order to preserve user well-being, others argue (Howell et al. 2018; Andalibi et al. 2020).

Other papers furthermore comment that designers can attend to issues of social justice by prioritising the engagement of those misunderstood, misrepresented or expressing cultural difference, who are often subject to marginalisation (Sultana et al. 2018). Concerning this goal, the deployment of 'activist systems' is recommended as a fruitful way to start tackling social challenges (Barbosa et al. 2019). On the other hand, other papers argue that fairer practices of user interface design might not be sufficient to tackle social justice challenges, as the complex relationships between technologies and their surrounding socio-economic and political contexts can prove strong opposing forces (Veale et al. 2018; Madaio et al. 2020). These assorted challenges, despite initial solutions, serve to highlight just how complex the design of more respectful digital health systems can be should they strive to address such ethical challenges.

#### **Ethics as Taking Responsibility**

Another recurrent theme we found in the papers reviewed is the apparent expectation that professionals should adopt measures against unethical practices as part of their work (for example, Correll 2019. Papers also point to the need for researchers in academia to be responsible for ethical conduct beyond that which is anticipated by institutional ethics approval (for example, Munteanu et al. 2015). On the other hand, it is also discussed that individuals alone might have a limited capacity to act against business goals and roles within hierarchies (Chivukula et al. 2020; Gray et al. 2019).

Education is also encouraged as a path to supporting ethical awareness among individuals (Gray et al. 2018; Chivukula et al. 2019). One paper describing an open-ended project with local organisations shows how an 'ethics of care' was configured between students and users (Sabie et al. 2019). Similarly, another paper demonstrates that collaborative and self-experimentation classroom activities can initiate reflection about harmful data collection and visualisation practices (Shapiro et al. 2020). The use of cards to ignite group debate in regard to moral dilemmas and the creation of prototypes is yet another educational technique found within the sample (Bilstrup et al. 2020). The importance of enhancing end-users' critical gaze in relation to issues of ethics is also discussed (Di
Geronimo et al. 2020). Examples of how to achieve this include an immersive theatre experience (Skirpan et al. 2018b), a bio-metric mirror as provocative demo (Wouters et al. 2019) and an activity to examine the values embedded in a popular video-streaming platform (DiPaola et al. 2020).

In contrast, very few papers explicitly invoke responsibility from the public sector or urge the creation of more robust policies. Current regulations are, in fact, criticised by some, as overarching lists of rules said to be ultimately ineffective, given the challenge that it is to define guidelines in advance that multiple contexts can adopt (Figueiredo et al. 2020; Madaio et al. 2020). An exception to this argument can be found, however, in a paper praising the idea of Responsible Research and Innovation (RRI) as a valuable top-down framework to guide the implementation of ethics in HCI research (Grimpe et al. 2014). Questions of policy are equally relevant to health data repositories of all scales, particularly those about data misuse and accountability structures.

# 3.4 In Summary

The review presented in this chapter provides a valuable foundation for future works taking the abstract concept of 'ethics' and seeking to putting it into practice through technology design processes. The next step in the work of this thesis is to return our focus to the specific application context of health data ecosystems, using the knowledge sources identified in this review to support rich and informed engagement with participants, and develop a unique understanding of their expectations, preferences and values. The next chapter thus describes a user study conducted to examine the attitudes of health data contributors when presented with the scenario of contributing their personal information to a health data platform.

# 4 Understanding People | Interview Study with Data Contributors

As part of a user-centred research agenda, this chapter turns towards the people at the heart of the health data technology ecosystem: data contributors. We conducted a qualitative interview study to build upon the knowledge gained during the first stage of this research process (mapping the literature) and capture what matters most to this user group. In this study, we asked young adults about their motivations and concerns when presented with the opportunity to share different types of personal information with a health data platform. The study reported in this chapter has been published as *"Futures for Health Research Data Platforms From the Participants' Perspectives"* (Vilaza et al. 2020), and is attached to this thesis as Paper 2.2.

# 4.1 Introduction

As seen in the previous chapter, the acceptance of health data technologies can be compromised by ethical concerns, a lack of motivation to contribute, and trust issues. A lack of public acceptance has negative consequences; without broad inclusion and diversity, the potential benefits achieved through the use of health data may furthermore be unfairly distributed across populations (Sankar et al. 2017; lenca et al. 2020). Therefore, it is critical to understand data contributors' perspectives on collecting, sharing, and re-sharing personal health data. Previous research has investigated volunteers' willingness to contribute to bio-repositories (Domaradzki et al. 2019; Middleton et al. 2019; King 2019); however, there remains a gap in our understanding of users' willingness to contribute health data repositories that may store in particular more sensitive types of data, such as mobile sensing and behavioural monitoring. To address this critical gap in knowledge, we conducted the following study.

# 4.2 Method

This study employed a semi-structured interviewing method – a well-established design research methodology also suitable for investigating ethical concerns, as seen in Chapter 3. Interviews included a card sorting activity for inquiring participants about their comfort level with sharing specific data types (See Table 4.1<sup>1</sup>). The study was exempt from ethical approval, although conducted according to established ethical standards (informed consent, confidentiality, participant well-being) and followed Denmark's national code of scientific conduct.

In conducting this study, we chose to focus on young adults, given that this demographic group was found to be less open than other (older) groups to the idea of sharing health data (Goddard et al. 2009). There were 12 participants, being four females and eight males (age mean=23, std=1.8), from the United States, China, Hungary, Spain, France, Italy, Netherlands, and Scotland. In broad terms, participants were asked about their attitudes about eventually contributing to a health data repository that primarily stores and shares (de-identified) data for research purposes.

Data analysis employed an inductive approach, following Braun & Clarke's Thematic Analysis method (Blandford 2013; Braun et al. 2012) and identifying three main themes: rea-

<sup>&</sup>lt;sup>1</sup>Table extracted from Vilaza et al. 2020

sons to contribute health data to shared-access research repository, the ideal characteristics of a trustworthy repository and personal criteria for feeling comfortable with the prospect of sharing different types of information. The complete description of this study protocol is in the Paper 2.2.

Calls/text content	Financial status	Relationship
Mental illness	Address	Contagious illness
Genetic disorders	Emotional status	Family
Sleep patterns	Places visited	DNA
Alcohol/tobacco	Food	Local ambience
Current weather	Distances	Partner health
Physical status	Leisure	Physical illness
Physical exercises	Heart rate	Toilet habits
Hygiene	Blood/urine	On-screen time
Academic/work performance	Apps used	Calendar

Table 4.1: Data types comprising each card for the sorting activity

# 4.3 Results

# 4.3.1 Motivations to Contribute Data

Two main reasons to contribute data to health data repositories were identified: supporting public benefits and receiving personal benefits.

#### **Public benefits**

All participants praised how valuable the concept of a health data platform seems to be: "There could be a lot of good for a lot of people if healthcare providers could have more access to information about people's lives" (P3). Helping patients and researchers was a motivation to contribute expressed by most: "I would share my data with the idea that it can help research and it can help other people to be cured" (P9).

#### **Personal benefits**

The majority also felt motivated by the prospect of obtaining personal benefits, such as access to data to improve their health: *"If you have all the information about me, you can give me more specific tailored care, maybe even like getting life recommendations"* (P3). In contrast, monetary compensation was not perceived as essential by most: *"I don't think I would ask for financial compensation, I think it is more volunteering type of work"* (P7), unless there is a strenuous time commitment *"Probably if it takes a lot of time, it should be rewarded, it should be paid"* (P1).

# 4.3.2 Trust in the Platform Governance

Two main enablers of trust were mentioned recurrently in the interviews: more transparent communication and more choices for data control.

#### Transparency

Most participants mentioned that they would want to know who was managing the platform: "I need to know who are the people behind the programme. Is it the university, is it an external research centre, is it for hospitals?" (P2). It was also described as crucial by most to know in detail the purposes for data use in the future: "It is not good to share anything if we do not know how it can be used in the future" (P11). Others expressed that they would like to receive updates about research outcomes: "I would be interested to know if I helped and what conclusions were drawn as a result" (P8). Most commented that they would like to be informed about the security protocols in place: "Knowing some technical stuff would make me comfortable even though I don't understand that type of information" (P4). They would also like to know whether their data were to be stored anonymously: "If it is already anonymised in the database, then I would use the platform because I know that I am just going to be one in the millions" (P4).

#### Choice

Some participants expressed a desire to be asked for permission for their data to be reused on a case-by-case basis: "If they have new research and want to have access to my data, I want them to ask me if I want to release to the new research or not" (P7). Others added that they would also like to block access to specific data receivers: "I want to choose if I disagree with one of the studies and say: 'with this one I don't want to share'" (P10). Many also noted that they would prefer to be able to pause data collection or delete previously collected data: "Maybe in some situations, it is useful to stop recording some data. If you go to some places where you don't want to be followed, for example, you don't want people to know you are there" (P2). Other future possible data contributors expressed that they would feel more comfortable if their data were shared in the least detailed form necessary: "If it just to know what you are going to do in your day, how many hours you are walking, something like that, is fine" (P10).

# 4.3.3 Criteria for Sharing Each Data Type

Insight into data contributors' motivations and perceptions of trust enables the understanding of critical factors for public acceptance. However, it is also important to understand how these values relate to the prospect of being solicited to share different data types. Participants shared the criteria that would guide their decision to share or refrain from sharing an extensive list of information (See Table 4.1). These criteria included: evidence of the importance of the data type for research, perceived risk in case of a data leak (material loss and reputation damage), the burden of continuous data collection and the potential privacy invasion of their relatives, partners or colleagues.

#### **Relevance for research**

Most participants shared that they would be willing to share data if they knew why data of that type was essential for research: *"I would be happy to share that food consumption information. For me, that has a very tangible connection to health"* (P3). However, participants had divergent opinions regarding the relevance of different data items, with some willing to share items that others would not, as seen in the card sorting activity.

#### Potential for material damage

Many were concerned about the risk of physical attacks, for instance, if their location was leaked to malicious actors: *"If I am going back home at this time, then someone is just wait-ing for me at the door and just breaking into my room"* (P7). Participants also expressed that cyber-attacks are also concerning for them: *"Financial situation, no, because people maybe know how much money I have in my bank account and they would see if it is worth hacking it"* (P7). Both attacks could perhaps lead to financial loss, which is another reason they mention that contributes to fearing data sharing: *"If hackers are working for an insurance company, and they sell data to them, they could propose different plans"* (P6).

#### **Risks to reputation**

Each participant had their own set of data types that they believed posed risks to their reputation. Many were concerned about others knowing of their substance use or abuse: *"Alcohol, tobacco, I do not want people to know how much I drink because I drink too much"* (P8); particularly in the context of their work environment: *"I am afraid that in a potential data leak, my employer gains access and discriminates me in my employment"* (P11).

#### Invasion of others' privacy

The privacy of others was also an essential criterion for several participants: *"I think sharing calls and text content overlaps with the privacy of other people"* (P2). These participants believed that consent should be obtained from others in these cases: *"When it comes to the condition of my partner, it is tricky because I feel like it is their decision to share the information"* (P3).

#### The effort of data collection

Several participants also raised concerns regarding the burden of the commitment to collect data in the first place: *"I do not want to share data when I'm low on my battery or when I am short of data. I would not like this data to interfere with my daily life"* (P2); and perhaps particularly when in a vulnerable state: *"If someone is really sad or depressed, I think it is harder to share that"* (P10).

# 4.4 A Conceptual Framework

These findings show how data contributors reason about the prospect of eventually sharing their personal data with a shared-access repository for health research. The factors outlined above are critical to understanding what is essential for public acceptance. An ethical health data ecosystem design should elevate the experience of contributing by attending to these factors.

To highlight those factors shaping acceptance of health data platforms by data contributors, we developed this conceptual framework (See Figure 4.1<sup>2</sup>)). This framework serves as a tool to support technology creators' reflection on ethical health data ecosystem design. By understanding how different factors may affect acceptance at different stages, technology creators can be better equipped to make decisions that will more strongly support the values of those contributing their data. A core property of this framework is that it emphasises that technology acceptance is a continuous process that evolves with time, which means that it becomes necessary to consider the maintenance of users' trust after the initial phases (Nadal et al. 2020).



Figure 4.1: Conceptual framework for acceptance based on the interview findings

This conceptual framework describes how the experience of contributing to a health data repository can be understood as a sequence of acceptance phases: from the moment a

<sup>&</sup>lt;sup>2</sup>Figure from Vilaza et al. 2020

participant is first introduced to the possibility of contributing to the actual collection and sharing of data that comes with the commitment to participate. In the first phase, individuals seek reassurance that being part of a research platform will benefit themselves and others (motivation) without posing significant risks in their lives (trust). Once individuals agree to participate, they ponder which information types they feel comfortable sharing (willingness to share). Each participant applies different selection criteria during this deliberation process, based on their personal boundaries and risk perceptions. As participation in the repository progresses, individuals may change their perceptions of sharing data according to the initial conditions. For this reason, as highlighted in the framework, the maintenance of motivation and trust is a long-term process, requiring designers and researchers, from an ethical standpoint, to consider their impact on acceptance fluctuations.

# 4.5 In Summary

This chapter contributes a list of factors expressed by data contributors as critical for their acceptance of personal health data sharing with health data platforms. Results highlight the existence of clear motivations to contribute, potential enablers for trusting the platform and varied willingness to share each specific data type. To facilitate the process of translating these qualitative insights into design recommendations for future platforms better attend to contributors' needs and values, we proposed a conceptual framework (see Figure 4.1) connecting each of the identified factors to sequential stages of acceptance. The next chapter builds upon these findings by providing a complementary quantitative perspective on data contributors' attitudes expressed by a much larger and diverse sample.

# 5 Understanding People | Survey Study with Data Contributors

To better understand data contributors' attitudes, we next sought to expand the qualitative insights outlined in the previous chapter with a broader sample and a quantitative approach. This survey study was motivated by our need to ground our design suggestions not only in the perspectives of a target set of individuals but also in evidence collected from more diverse data contributors. We also wanted to compare the strength of specific sources of motivation, data access preferences, willingness to share digital and non-digital data items, and reasons for privacy concerns between contributors with different characteristics. The results reported in this chapter have been published in the form of the full-paper *"Public Attitudes to Digital Health Research Repositories: Cross-sectional International Survey"* (Vilaza et al. 2021a), available as Paper 2.3.

# 5.1 Introduction

This study builds upon the interview study described in Chapter 4 to gain a more nuanced understanding of the acceptance factors considered crucial by interview participants. As seen in our review of approaches to engaging with ethics (Chapter 3), surveys help make solid arguments for the existence and prevalence of ethical concerns across populations (Skirpan et al. 2018a). Surveys also allow the statistical analysis of relationships between individual characteristics and acceptance factors, which can be very relevant when considering differences between user groups. The survey study described next provided robust evidence on data contributors' attitudes.

# 5.2 Method

# 5.2.1 Recruitment

This collected responses from two national contexts: Brazil and Denmark. Recruitment occurred between March 2020 and December 2020, primarily targeting young adults but not prohibiting other age groups' responses. The link to the survey was distributed online via university forums and newsletters, email lists and social media (Facebook, Twitter, Whatsapp). A recruitment event was also organised at Technical University of Denmark (DTU), during which participants were compensated with a cup of coffee.

# 5.2.2 Survey Design

The design of the survey questions drew on several sources of knowledge, including the findings of the study described in the previous chapter (Vilaza et al. 2020; Middleton et al. 2019; Patil et al. 2016; Sanderson et al. 2017; Lemke et al. 2010; Ahram et al. 2014; Trinidad et al. 2012; Rathi et al. 2012; Papoutsi et al. 2015; Luchenski et al. 2012; Zanaboni et al. 2020; Porteri et al. 2014). Survey development was iterative and informed by feedback from experts (researchers, clinicians, statisticians) and members of the target audience (young adults). At the beginning of the survey, the following short description of a health data repository was provided: *"A research data repository is an online database containing data collected during research studies. In such repositories, de-identified data is to be re-used in the future by other research studies"*. The survey was deployed using the LimeSurvey platform, and data was hosted on a secure server at the DTU. More details on the rationale for each of the questions can be found in the Paper 2.3.

# 5.2.3 Procedure

The study was considered exempt from ethical approval in Denmark, although it received approval from the Institutional Review Board of the Universidade Federal do Rio Grande do Norte (UFRN). Participants were asked to provide consent at the beginning of the survey, in line with the European General Data Protection Regulation (GDPR) and following standard ethical conduct principles for online research. Participation was anonymous.

Data analysis and visualisation were conducted using the R Project for Statistical Computing (software for statistical computing and graphics). This analysis included frequency distributions, bivariate associations and binary regressions. Reference values followed that an odds ratio below 1.5 was considered weak and above 5.0 strong (Chen et al. 2010); and for a 95% confidence interval, results were significant if P<.05. Missing values from questions skipped were removed from the analysis.

# 5.3 Results

# 5.3.1 Participants' Characteristics

The survey was answered by 2299 participants, out of which 1963 completed all questions (1963/2299, 85.38%). For data analysis, only data from participants residing in Brazil (1017/1600, 63.56%) and Denmark (583/1600, 36.44%) were included (1600/1963, 81.50%). The majority of participants in the target sample were between 18 and 27 years old (933/1600, 58.31%), identified as being female (891/1600, 55.69%), had at least a university degree (992/1600, 62.00%), were in good, very good or excellent health (1407/1600, 87.94%) and were moderately, very, or extremely interested in health top-ics (1088/1600, 68.00%). The vast majority of the sample owned and used smartphones (1562/1600, 97.63%) and computers (1537/1600, 96.06%).

# 5.3.2 Experiences, Awareness and Attitudes

The vast majority of the sample positively perceived the type of health research data repository described in the survey that stores de-identified data (1339/1600, 83.69%). Approximately half of the participants had participated in a health research study already (763/1600, 47.69%), and those who participated were more likely to have a moderate to high interest in health topics (OR 2.35, 95% CI 1.88-2.93; P<.001). A small portion of the sample was aware of examples of research data repositories (459/1600, 28.69%), and those aware of examples were more likely to have a moderate to high interest in health topics (OR 3.02, 95% CI 2.30-3.96; P<.001) and to have been participants in previous health studies (OR 3.36, 95% CI 2.66-4.23; P<.001). Those who held a positive perception were also more likely to be aware of examples of research data repositories (OR 3.26, 95% CI 2.17-4.90; P<.001). Significant factors influencing participants' current perceptions of digital health research repositories were: an interest in health topics (OR 1.49, 95% CI 1.10-2.02; P=.01), previous participation in health research studies (OR 1.70, 95% CI 1.24-2.35; P=.001), and awareness of examples of existing repositories (OR 2.78, 95% CI 1.83-4.38; P<.001). The complete results of this logistic regression can be found on Table 2, Paper 2.3.

# 5.3.3 Motivation to Participate

The majority of participants felt very or extremely motivated regarding the prospect of helping future patients (1366/1600, 85.38%), helping researchers (1253/1600, 78.31%), receiving health results concerning themselves (1170/1600, 73.13%), and receiving general research results (1063/1600, 66.44%). Only a minority felt very or extremely motivated by financial compensation (505/1600, 31.56%). Those who positively perceived health data repositories were more likely to be motivated by all incentives suggested, except financial compensation. Those interested in health topics were more likely to be motivated



Figure 5.1: Distribution of answers for each motivation source

by receiving the results of research (OR 2.25, 95% CI 1.65-3.06; P<.001) and proposing questions to be investigated (OR 2.53, 95% CI 1.97-3.24; P<.001). The youngest segment (18-27 years old) was more likely to feel motivated by receiving financial compensation (OR 1.92, 95% CI 1.57-2.35; P<.001). Figure 5.1<sup>1</sup>) shows the distribution of responses.

# 5.3.4 Reasons for Concern

The majority of participants felt very or extremely concerned about their data being used for unethical projects (1219/1600, 76.19%) and profit-making without their consent (1096/1600, 68.50%). Being a victim of cyberattacks was very or extremely concerning for 65.94% (1055/1600). Only a few participants felt very or extremely concerned about the burden of being asked to share more data in the future (527/1600, 32.94%). No significant associations were found between specific participant groups and different reasons for concern. (See Figure  $5.2^2$ ).

# 5.3.5 Willingness to Share Different Data Types

Only a few participants felt uncomfortable sharing any data items (94/1600, 5.87%). Those who felt uncomfortable sharing any data items were more likely to have a negative or indifferent perception of health research repositories (OR 3.91, 95%CI, 2.49-6.14, P<.001). Most participants felt uncomfortable or very uncomfortable sharing the content of texts and calls (1206/1600, 75.37%). Fewer participants felt uncomfortable or very uncomfortable or very uncomfortable sharing the frequency of texts and calls (706/1600, 44.12%). Around half of the sample considered places visited (864/1600, 54%), apps used (775/1600, 48.43%) and DNA samples (750/1600, 46.87%) as uncomfortable or very uncomfortable data to share. On the other hand, most felt comfortable or very comfortable sharing the other data types: sleeping patterns, food consumption, alcohol consumption, diagnoses of physical illness, physical activity levels, stress levels, family health history, distances travelled, diagnosis of mental illness, blood samples and time spent on screen. Figure 5.3<sup>3</sup> shows the

<sup>&</sup>lt;sup>1</sup>Figure extracted from Vilaza et al. 2021a

<sup>&</sup>lt;sup>2</sup>Figure extracted from Vilaza et al. 2021a

<sup>&</sup>lt;sup>3</sup>Figure extracted from Vilaza et al. 2021a



Figure 5.2: Distribution of answers for each reason for concern

distribution of responses.

The following findings highlight that participants' willingness to share different data types can vary significantly. Those who had a positive perception of health research repositories were more likely to feel comfortable or very comfortable sharing all of the data items, except the content of calls and texts. Those concerned about being discriminated against were more likely to feel uncomfortable sharing data about diagnoses of mental illness (OR 2.26, 95%CI, 1.66-3.07, P<.001). Those uncomfortable sharing data about about app usage were more likely to be concerned about data being used for profit (OR 2.57, 95%CI, 1.91-3.46, P<.001) and not understanding terms and conditions (OR 2.22, 95%CI, 1.71-2.87, P<.001). Those not motivated or only slightly motivated by the possibility of receiving results about themselves were more likely to feel uncomfortable sharing their alcohol consumption (OR 5.76, 95%CI, 3.63-9.13, P<.001), distances travelled per day (OR 3.31, 95%CI, 2.29-4.80, P<.001), stress levels (OR 6.46, 95%CI, 4.43-9.44, P<.001) and physical activity levels (OR 6.78, 95%CI, 4.52-10.17, P<.001).

Age (OR 2.16, 95%CI, 1.28-3.70, P=.004), digital device ownership (OR 1.90, 95%CI, 1.14-3.26, P=.01), health status (OR 2.28, 95%CI, 1.24-3.98, P=.01) and current attitude regarding digital health research repositories (OR 3.77, 95%CI, 2.24-6.26, P<.001) were significant factors affecting participants' willingness to share any data with a health research repository. The complete results of this logistic regression can be found on Table 3, Paper 2.3. These findings highlight that participants' willingness to share different data types can vary significantly according to demographic factors, health status and current attitudes.

# 5.3.6 Preferred Data Access Options

The majority of participants stated they would like to receive information about which projects access their data in the future (1334/1600, 83.37%) and to be able to decide who obtains access to which aspects of their data (1181/1600, 73.81%). Never being contacted was desirable or very desirable to only 25.5% (408/1600), and allowing the owners of the repositories to decide who can access the data was desirable or very desirable on very desirable or access the data was desirable or very desirable on ly to 23.62% (378/1600). Giving public or academic institutions access to their



Figure 5.3: Distribution of willingness to share across different data types

data was perceived as desirable or very desirable by 48.93% (783/1600) of participants, although allowing private labs and companies access was considered desirable or very desirable by only 13.43% (215/1600). Figure 5.3<sup>4</sup> shows the distribution of responses.

Those who found it desirable to be informed about the identities of those using their data were more likely to have a positive perception of health data repositories (OR 2.45, 95%CI, 1.77-3.39, P<.001). Those concerned about data being used for unethical projects were more likely to find it desirable to have control over how their data would be used (OR 2.45, 95%CI, 1.80-3.42, P<.001) and to be informed (OR 3.09, 95%CI, 2.18-4.37, P<.001). Those concerned about data use for profit without their consent were more likely to find it undesirable to have private labs and companies access their data (OR 2.24, 95%CI, 1.69-2.96, P<.001).

# 5.3.7 Cross-country Analysis

No significant differences were found between the demographics distribution of participants residing in Brazil and Denmark, except that a higher percentage of females and a lower percentage of participants who had completed a higher education degree were present within the Brazilian sample. The vast majority of the Brazilian sample held a positive attitude towards health data repositories (927/1017, 91.15%), in contrast with a less significant majority in Denmark (412/583, 70.67%). The Brazilian sample was also more likely to feel motivated by receiving results about their health, proposing questions to be investigated, receiving research results, and helping researchers. The Brazilian sample was also more likely to be moderately, very, or extremely concerned by all options listed.

<sup>&</sup>lt;sup>4</sup>Figure extracted from Vilaza et al. 2021a



Figure 5.4: Distribution of answers for each access control option

Finally, the Brazilian sample was more likely to find it desirable to receive information about who is using their data (OR 5.51, 95% CI 4.12-7.37; P<.001), with most being more likely to find it desirable to never to be contacted (OR 3.63, 95% CI 2.87-4.60; P<.001). They were also more likely to find it desirable to enable repository managers to decide who gets access (OR 2.84, 95% CI 2.24-3.60; P<.001); and allow both private organisations (OR 3.73, 95% CI 2.77-5.04; P<.001) and public institutions to access data (OR 4.51, 95% CI 3.61-5.63; P<.001).

# 5.3.8 Overview of Survey Findings

Given the breadth of the variables contemplated by this study, we end this section by presenting a summary of results:

- A positive perception of health research data repositories is associated with higher levels of motivation and willingness to share. This positive perception can be further influenced by levels of personal interest in health, previous participation in health studies and awareness of examples of research repositories.
- The most potent sources of motivation for contributing data are: helping future patients, helping researchers and receiving results about personal health and research outcomes. On the other hand, receiving financial compensation appears as the weakest incentive.
- The most significant reasons for concern are the possibility of data being abused (for unethical research goals or profit) and the risk of cyber-attacks. The least concerning consequence is the possibility of being asked to provide more data in the future.
- The vast majority of participants state feeling comfortable with sharing at least one data item; however, most report feeling uncomfortable sharing the content and frequency of texts and calls, locations of places visited, and apps used.
- Those unwilling to share any data are more likely to have a negative perception of health research data repositories and low motivation levels; however, they are not

more likely to possess more substantial concerns than others.

- Those concerned about being discriminated against are more likely to feel uncomfortable with sharing data about mental illness diagnoses, with no significant associations being found between social discrimination and other data types.
- Those uncomfortable with sharing alcohol consumption, distances travelled per day, stress levels, and physical exercise are more likely not to feel motivated by receiving this information about themselves.
- The majority of participants find it desirable to known which projects access their data. Most also prefer to decide who gets access to their data. In contrast, not being contacted after providing consent is considered undesirable by most.
- Allowing public or academic institutions to access data is perceived as more desirable than allowing private labs and companies. Those concerned about unfair profiting practices are more likely to find it undesirable to grant data access to private organisations.
- The Brazilian sample is more likely to report a positive perception of health data repositories, higher motivation levels and higher levels of concern. The Danish sample is more likely to prefer not being contacted after providing consent and allowing repository managers to decide who can access data; however, the Danish sample is less likely to feel comfortable sharing alcohol consumption and stress levels.

The findings of this survey study confirm the importance of the factors derived from the qualitative study in Chapter 4, while they also expand our understanding of how motivating, concerning and desirable certain conditions for health data sharing can be. The implications of a more nuanced analysis of different factors include the possibility of fine-tuning design suggestions to focus on what matters most to potential contributors, thus supporting the work of technology creators. Overall, differences between national contexts and demographic groups demonstrate the need for data platforms to propose conditions for participation entailing diverse configurations of the contribution that may potentially accommodate the needs of a broad set of people. A more thorough discussion of how findings relate to previous literature can be found in the Paper 2.3 and a more detailed discussion of design recommendations derived from this study are available in Chapter 9.

# 5.4 In Summary

This survey study contributes novel insight into public attitudes regarding health data repositories in which data is used primarily for medical research; by enabling a quantitative understanding of that which matters to health data repository contributors. Together with the findings from Chapter 4, these results show that data contributors care about having their autonomy respected and are motivated about the benefits that might be offered to themselves and society. Findings from this research can be used to support design for health data technology acceptance, as well as more ethical use and sharing of health data. In order to further this work's contribution to the design of these systems and technology creators' capacity to action ethics, we next turn towards investigating the attitudes of those creating and managing digital health technologies - the topic of the next chapter.

# 6 Understanding People | Focus Group with Technology Creators

As this thesis has the overarching aim of supporting technology creators to build health data systems more attuned to data contributors' needs, we next turn to examine the technology creators' perspectives. To understand the enablers and barriers deemed most critical to supporting the acceptance of data contribution by this stakeholder group, we chose to conduct a focus group study. This additional viewpoint - as provided by interaction designers and software developers - expands and confirms findings derived from prior studies conducted with data contributors. The content of this chapter is based on the Paper 2.4: *"Sharing Access to Behavioural and Personal Health Data: Designers' Perspectives on Opportunities and Barriers"* (Vilaza et al. 2019).

# 6.1 Introduction

The health data ecosystem comprises many stakeholders whose needs and objectives can clash. Previous research has identified as the paramount necessity among those *consuming* health data for high data quality and robustness (Bourla et al. 2018; Kim et al. 2017). On the other hand, as seen in the previous chapters of this work, those *contributing* health data desire privacy protection and more flexible data sharing conditions. As argued in Chapter 2, the predominant vision of health data systems currently, however, caters primarily to the goals of researchers, clinicians and public workers, which at times means treating the needs of data contributors as of a lower priority.

Chapter 3 shows that engaging with those envisioning and building interactive systems is a meaningful way of putting ethics into practice. Professionals working on the conception of new health data technologies are often tasked with bridging the needs of data contributors and data consumers in order to create systems that capture what is essential for both (Marcu et al. 2011). In this context, technology creators can help raise technical limitations and design possibilities useful for devising future developments. However, previous health data repository research has rarely engaged with this stakeholder group. For this reason, we consulted a group of interaction designers and software developers to understand better their perspectives regarding the potential ethical benefits that health data repositories may yield and the factors that, based on their experiences, currently hinder such advances.

# 6.2 Method

A focus group study was conducted with six employees working at the Copenhagen Center for Health Technology. As they were part of the same research group, they were familiar with each other's work and had collaborated on projects ranging from a mobile app for depression treatment, a web-based tool for cognitive assessment and a heart rate sensor for monitoring patients with cardiovascular diseases. All participants had been working with digital health technologies for more than a year. Two worked as interaction designers (P2, P4) and four as software developers (P1, P3, P5, P6).

The focus group lasted for approximately two hours, was audio-recorded, and participants' notes during discussions were documented in the form of photographs (See Figure



Figure 6.1: Participants' notes during the focus group

6.1<sup>1</sup>). Audio data were transcribed and analysed using Braun & Clarke's Thematic Analysis Braun et al. 2012, in the process of which themes were inductively derived. More information on the study procedure can be found on the Paper 2.4.

# 6.3 Results

# 6.3.1 Data Should Be Used for Science

All participants believed that health data should primarily be used for increasing benefits for individuals and society: "Good purposes, research purposes, which are going to benefit society in general, going to benefit the health sector" (P5). Thanks to their technical knowledge, participants were able to articulate why data is an essential resource for health science: "Sharing data can help others in a bigger way or solve a bigger problem. For example, if I share my heart data, how my heart is functioning, that can help to build an artificial intelligence algorithm which can automatically detect diseases or that can help to build some model" (P3). They were also confident that even though data collected right now may not be relevant to healthy individuals, in the future, novel discoveries might be unlocked if people share data: "We might not know what to do with the data right now, but in four years, I might go down with stress. Then we can look at the four years of data leading up to me having this stress, and we could maybe figure out what is causing it, more accurately" (P4).

However, some participants admitted that entities with unethical purposes could also misuse health data: "A company or organisation that has my data can create a profile secretly and use it in the future for data analysis. They may already know about my behaviour before they actually meet me because they have a lot of data about me so that they can predict my actions in the future" (P1). One participant went further and stated that personal health data should not be used for profit: "It has to be good for all the people, and not for profit making like all the big corporations are doing in some way or another" (P2). Health developers' views on the merits of data sharing align significantly with those of health data contributors, as shared in the prior studies of this work.

<sup>&</sup>lt;sup>1</sup>Figure extracted from Vilaza et al. 2019

# 6.3.2 Users Are Reluctant to Share Data

All participants have experienced that some users are not willing to share health-related data despite the potential benefits: *"It is technically feasible, and we are very much looking forward to getting this kind of data because if you have more data, you can build more robust systems. But, that is the problem: you do not have the data because people are not willing to share it" (P3).* In particular, one participant mentioned that users from one of their previous projects were more reluctant to share data about their mental health: *"I think most people are happier sharing only physiological data, because of the stigma of sharing the mental data" (P6).* 

Technology creators, however, were aware that the reason for such unwillingness to share data is that users would prefer to keep their mental illness diagnosis private: *"We take the data from all of these psychiatry patients, but these patients are afraid of other people knowing what is going on with their mental health" (P2)*. One participant contrasted the collection of digital health data with traditional disclosures made during medical consultations: *"When you talk with your doctor, any data you share with your doctor, is confidential" (P4)*. Such findings indicate that technology creators are conscious of their users' concerns and aware of the reasons why users might be reluctant to share their data with digital repositories. This observation draws further evidence that data contributors often do not feel comfortable sharing data about their mental health, as indicated by the previous user studies in this thesis.

# 6.3.3 Health Data Systems Could Be Improved

Most participants were critical of standard practices in designing and developing data sharing systems. One participant drew attention to the lack of clarity of 'terms and conditions' documents: "The terms are sometimes not clear: when they want to collect it, like what sort of data they want to collect, it is very general the data agreement" (P1). The same participant also expressed frustration regarding the lack of real choice when it comes to consenting to access to data: "Even if you read them you cannot say to the service 'I will not give you consent', then you do not have any option besides not to use it. You have to give away your data. It is like a trade-off" (P1). This participant was also sceptical about the security measures taken by the developers of large-scale health data repositories, given that they had advanced knowledge of data protection techniques: "I am not sure that the software developers covered all issues, or if privacy and security were also considered when developing this platform" (P1).

Other participants described how health data systems could be improved in response to such critical views. One example was to increase transparency about data usage: *"The most important for me is the journey of the data, so who is going to use it, and at what point, who is going through the data"* (*P6*). However, even though most participants were vocal about their dissatisfaction with current standards for data sharing procedures, they felt they were not able to change them. Several expressed a belief that this should always be how things work: *"That is the point of sharing it, when you initiate it, you do not know what will happen next"* (*P3*). These comments stand in stark contrast to the enthusiasm about the new opportunities that health data could provide for building models and algorithms to support healthcare now and in the future.

# 6.4 In Summary

This chapter further supports the idea that ethical health data technologies must address contributors' concerns and provide protective measures against data misuse. Findings show that technology creators had personally experienced that data contributors can be reluctant to share their personal data, especially when it comes to data referring to mental

health. Developers and designers were also critical of how data sharing requests often provided contributors little choice or transparency. However, acting on such findings in the practice of technology design and development remains a significant challenge in itself — participants felt sceptical that some data sharing standards could ever change, and they did not explicitly suggest initiating this movement. We next explore and experiment with different approaches to realising the ethical design of health data user interfaces, rooted on the results of the user studies in the last three chapters.

# 7 Practising Ethical Design | Health Data Interfaces

User-centred design guidelines propose that initial research for understanding users' needs should lead to the recommendation of actionable requirements for the conception of design alternatives (Sharp 2003). Taking some of the methods for ethical design highlighted in Chapter 3 and the insights elicited through the user studies described Chapters 4 and 5, we here explore means of leveraging the combination of these prior sources of knowledge in the design of health data interfaces. Therefore, this chapter aims to understand how the process of translating ethical requirements into user interfaces unfolds.

This chapter explores in particular two approaches advocated by previous research as well-suitable to eliciting ethical guidance (See Chapter 3): Value-Sensitive Design (VSD) and the critical appraisal of technology design. These approaches share similarities in that they both employ conceptual constructs (values and principles) to guide ethical analysis, yet they also differ in how these constructs are selected. Whilst in VSD, values usually emerge through so-called 'empirical investigations' (bottom-up approaches), critical appraisals can rely on pre-defined lists of ethical principles from established frameworks (a top-down procedure). We here describe how we applied these methods to the context of two unique design cases, highlighting how it is possible to adopt ethical perspectives in the configuration of health data interfaces respectful of data contributors' needs.

First, the framework of VSD was applied to the design process of a mobile app to be used for health research (StudyApp), and in particular, to guide the proposal of user interfaces aligned with the values expressed by data contributors. This part of the chapter draws from and expands upon the Paper 2.5 *"A Value-Sensitive Approach for Ethical Health Research Platforms"*. Then, an AI-ethics framework (based on normative principles) was applied to analyse ethical improvements to the design of chatbots for psychotherapy, with particular attention being given to the question of data sharing. This second half of the chapter is a summary of the Paper 2.6 *"Is the Automation of Digital Mental Health Ethical? Applying an Ethical Framework to Chatbots for Cognitive Behaviour Therapy"* (Vilaza et al. 2021b).

# 7.1 Interfaces for Research: Designing a Mobile Study App

The design process described in this case study was part of the development of the StudyApp, a mobile application for health research studies, through which participants answer daily surveys and contribute passively collected data. This app design consists of an initial step in a much broader software development project that aims to build re-usable software components for a 'generic' data collection mobile app that will eventually cater for varied types of health research (Bardram 2020). This chapter describes the stage of developing conceptual user interface designs for this StudyApp to achieve the project's purpose while prioritising ethical perspectives.

To support this design task and enable the exploration of a specific approach to ethical design, we chose to adopt VSD. The rationale for selecting this framework started with observing that this is a popular framework employed often in HCI publications (See Chapter 3). We then thought that 'values' could suitably encapsulate the user studies' findings needed to inform the design process (See Chapters 4, 5 and 6). This way, VSD tied to-

gether with the investigation of data contributors' perspectives, was employed to support the proposal of user interface features for the StudyApp project, as described next.

#### 7.1.1 Methods

Value-sensitive design (VSD) is a design framework focused on human values and "what is important to people in their lives", with a focus on ethics and morality (Friedman et al. 2008). VSD argues that digital artefacts can shape human experience by enabling (or hindering) users' values. Values are posited as different from 'user needs' in that they represent conscious choices that are usually less volatile and circumstantial — they might be understood as core human aspirations rather than current human necessities (Hayes et al. 2009). Nevertheless, they are essential to human welfare.

Different stakeholder groups might have conflicting values. Navigating such value tensions is both a challenge for VSD and a way for providing constraints for the design process. In the case of digital health research platforms, there exists a tension between researchers' needs for high-quality evidence and contributors' concerns with privacy (See Chapter 2). Existing platforms have often attempted to overcome this tension by prioritising data collection while also employing data protection techniques to de-identify data and, thereby, protect participants' identities.

However, as seen in the findings of Chapters 4, 5 and 6, data contributors often have privacy concerns that can hinder their acceptance regardless of efforts to keep data secure. For this reason, there exists an opportunity to consider alternative approaches to data collection and sharing that emphasise participants' values such that they can feel more comfortable contributing. We aimed to craft a prototype that demonstrates an intention to respect what data contributors consider most important by engaging in the following VSD process.

# 7.1.2 Empirical Investigations

The VSD approach entails phases described as 'conceptual, empirical and technical investigations'. Empirical investigations consist of the study of the human context in which technologies are situated and the perspectives of users and stakeholders. In this case, findings from the user studies carried out in the initial phases of this thesis served as our 'empirical material' and evidence source.

The interview study described in Chapter 4 and the survey study of Chapter 5 investigated how data contributors perceive participation in a digital health research repository and how comfortable they feel sharing different data items. These studies revealed that the main reasons to contribute to health data repositories consist of helping patients and researchers, although there is also a strong interest in receiving other personal benefits in exchange for data contributions. Data access options perceived as desirable by data contributors include the control of data collection and use, driven by significant variation in individuals' willingness to share different types of information. We translated these key insights into a set of values of importance to data contributors through the process described next.

#### 7.1.3 Conceptual Investigations

According to VSD, conceptual investigations comprise analytic, theoretical and philosophical explorations of the constructs in focus, seeking to define, for instance, working definitions of values. During this phase, our conceptual investigations aimed to take our empirical findings and identify the values that would best characterise them through reflection. After carefully pondering over possible combinations of values, we decided to focus on two central aspects in our design process.

≡ MoodStu			Моос	• IStudy	E MoodStudy
About this project Get to know more	Data contro	ol settings	oice	ç	Data re-sharing settings Contribute to another study
GOALS The aim of the MoodStudy project is to improve care for individuals experiencing mood fluctuations by understanding thei habits and routines.	What informati and share with You can change	on do you w this research your preference	ant to c study?	ollect y time	Would you like to re-share the data you collected with another study? You can also decide about this later
READ MORE		COLLECT	SHARE		Depression and roga
PEOPLE	Alcohol			0	Yoga might improve depression
Per Bækgaard Associate Prof. at DTU Compute	Smoking			0	READ MORE
READ MORE	Mood			0	
FUNDING	Steps			0	University and Stress We want to investigate how
Inis project is funded by EIT Health as at Innovation Project. No profits are generate from this research and this is not a government initiative	d			0	stress sources vary across university students
READ MORE	Mood			0	READ MORE
HAVE A QUESTION ABOUT THE PROJEC CLICK HERE TO CONTACT US	FEELING UNSUR CLICK HE	E ABOUT THI RE TO CONT	ESE OP ACT US	TIONS?	HAVE YOU GOT QUESTIONS ABOUT THESE? CLICK HERE TO CONTACT US
	Back	Page 2/3	me)		
		raye 275			Fage 575
(a) Transparent project ir	lfo (b) Data	control s	settir	ngs	(c) Data re-sharing settings

Figure 7.1: Health data interfaces supporting the value of agency

First, we defined users' desire for increased control and choice concerning the use and sharing of their data as best encapsulated through the value of *agency*. We conceptualised agency in our case as "the ability to influence or control outcomes" (Ahearn 2001). In digital environments, a user's agency to take actions is often mediated through the affordances provided by computational systems and interface components (Harrell et al. 2009). In the context of digital health, user agency can thus be related to the mechanisms through which users can act upon choices over how data is collected, shared, and re-shared (Torous et al. 2019).

Second, we conceptualised users' desire to receive personal benefits, besides those societal in nature, in terms of the value of *reciprocity*. Reciprocity as a value here refers to the expectation of being rewarded for positive actions with other positive actions (Molm 2010). In the context of data contribution, reciprocity can take the form of material compensation; however, participants in our user studies often expressed an aspiration for more meaningful ways to be recognised for their contributions. They expressed interest in knowing how their data might be helpful to others, as well as a desire to build a relationship of mutual respect with data platform owners. Next, we describe how these two central values were used to inform a prototype created to illustrate our design proposal.

# 7.1.4 Technical Investigations

Technical investigations focus on how technologies can more readily support desirable values or render other values more challenging to realise. Value sensitive design proposes that technical inquiries may take the form of prototypes depicting how user interfaces support the values identified through conceptual and empirical investigations. We

took the values of agency and reciprocity as design rationales for the following prototyping process.

Following standard practices of user-centred design (Sharp 2003), we first sketched a low fidelity prototype to illustrate possible system features to enable agency and reciprocity. We used this low fidelity prototype to informally prompt discussions with another designer, a software developer and a researcher. Separately, we also showed the prototype to five young adults peers, unrelated to the project, asking for their input as potential beta users. Following feedback from these group discussions, we further designed a high fidelity interactive prototype to serve as input for the software developers' deliberations around the coding phase of the envisioned StudyApp.

#### **Designing for User Agency**

The value of agency led us to come up with health data interfaces to allow selection and customisation of the data to be collected, shared and re-shared (See Figure 7.1). First, an interactive form lets data contributors state their preferences about which data types they want to be collected and shared by ticking the respective checkboxes (See Figure 7.1b). This form also allows users to read more about why each data type is required for the research (clicking on the question-mark icon). Second, a similar form lets users express whether they would like to delete any data points from the platform permanently. Third, the value of 'agency' also drove the design of a data re-sharing request form to let users choose whether data already collected should be shared with other research projects (See Figure 7.1c).

In addition to these forms, a user interface informed the user about project goals, investigators, funding sources and the study duration (See Figure 7.1a). The 'read more' buttons display more detailed information in this interface. Users can contact the study owners at any time through the 'contact us' option in these interfaces. Users' agency was therefore enabled through choices regarding how data is collected, shared and re-used, while further supported by the possibility of contacting the research team at any moment.

#### **Designing for Reciprocity**

In terms of designing to foster a sense of reciprocity, we created several interfaces to allow users to more directly benefit from their contributions (See Figure 7.2). First, a personal health management interface provided visual summaries of the data collected, in line with the visions of personal health informatics, although not with the ambition to replace existing care structures (See Figure 7.2a). In addition to this feature and motivated by the need to support users regarding the interpretation of this data, we proposed another interface to allow users to schedule remote consultations with health experts best equipped to provide feedback on health indicators (See Figure 7.2c).

Reciprocity was also supported by channels for communication between researchers and study participants, thereby actively inviting data contributors to participate in the research process. The prototype included an interface displaying updates and news about the research projects and a button asking whether users would like to propose a theme for a research study (See Figure 7.2b). Other features to increase reciprocity and the feeling of being part of a community included social forums and learning materials. This design, therefore, pushed for a more participatory approach to health research studies by proposing user interfaces through which users could share their ideas and learn about the research. Similarly, the consent form was composed to show that participants are considered and respected as an integral part of the research process.



(a) Personal health tracking

(b) Project updates

(c) Clinical feedback



# 7.1.5 Section Summary

This application of VSD to a specific and typical example of a health data technology shows how an ethically-grounded approach can orient designers' focus towards values otherwise not customarily emphasised. The result is a prototype used as the starting point for the conception of a functional app, serving as an example of the application of an existing design methodology to ethical design leveraging results from studies conducted with users. The features proposed here probably apply not only to the specific mobile app but other health data interfaces designed to support data collection and sharing for health research. We next describe how we explored a contrasting yet complementary approach employing a list of pre-defined principles to inform future developments.

#### 7.2 Interfaces for Intervention: Analysing Chatbot Design

Another emerging and increasingly significant motivation for health data sharing pertain to disclosing emotions, symptoms, and personal events within a therapeutic context. As language modelling has become more sophisticated, a notable development in this space is the conception of conversational agents programmed to emulate the practices of psychotherapists (Lee et al. 2020). Chatbots for psychotherapy, especially for Cognitive Behavioural Therapy (CBT) are being increasingly promoted as credible tools that may enable broader access to psychological support and propitiate more comfortable means for self-disclosure (Inkster et al. 2018). Previous research has found, for instance, that endusers engage more with self-report and disclosure when interacting with automated virtual interviews (Lucas et al. 2017; Lucas et al. 2014). A well-known example of a chatbot for CBT is Woebot, which has had its effectiveness for depression and anxiety successfully

validated with a sample of university students (Fitzpatrick et al. 2017).

However, a recent review considered the responses of chatbots during suicide emergencies very limited and potentially harmful in nature (Vaidyam et al. 2019). Another particularly alarming issue related to digital interventions which collect sensitive information about mentally ill patients is the lack of trustworthy privacy policies (Mercurio et al. 2020). In the case of chatbots used for psychotherapy, disclosed symptoms and feelings represent confidential data sourced from individual lives that may lead to the realisation of those dangers discussed in previous chapters: data leaks potentially leading to cyber and physical attacks, social discrimination and data misuse without consent.

Ethical challenges, including inappropriate responses and data handling by these systems, make this an area in which users' well-being, and indeed lives, hinges upon the capacity to reflect on the critical requirements in this space. It is not surprising that the idea of automated psychotherapy is often met with scepticism by psychiatrists (Doraiswamy et al. 2020). There are increasing concerns regarding how AI developments can ever lead to safe, ethical and effective tools for psychological support (Thieme et al. 2020). Even though the primary goal of digital health interventions is to provide health support through digital interfaces, this data may be eventually re-shared for health research purposes, which points to all the ethical concerns extensively discussed throughout this thesis. Future developments, therefore, must consider these ethical issues if they are expected to be broadly and securely adopted by patients — as motivates the exploration described in this second part of this chapter.

# 7.2.1 Methods

The study of AI ethics has very often communicated and organised ethical considerations in terms of how they relate to core normative principles (Thieme et al. 2020). Previous research reviewing current AI ethics standards, guidelines and frameworks has argued that the debate over the impact of AI on society has primarily focused on five normative principles: beneficence, non-maleficence, autonomy, justice and explicability (Morley et al. 2021). In summary, it is commonly agreed that AI developments should strive to be beneficial to, and respectful of, people and the environment, robust and secure, respectful of human values, fair and explainable, accountable and understandable (Floridi et al. 2020; Royakkers et al. 2018). However, it is unclear how such normative principles might guide the appraisal of system features in practice.

For this reason, we apply an ethical framework to raise critical questions related to the specific context of automated psychotherapy using chatbots. As we also noted a similarity between using pre-defined 'principles' as a unit of ethical analysis and centring the design process around 'values', which we deemed as a valuable way to contrast these two approaches. The choice for which framework to apply was based on such a previous review of prevailing ethical principles present in AI-ethics frameworks in general (Morley et al. 2019). This framework served as a structure to guiding a group discussion between authors, who extensively brainstormed how each of the ethical principles might be unattended to by current systems and what future developments might take.

#### 7.2.2 Results

#### Beneficence

The principle of beneficence entails providing positive value to individuals and society. Beneficence in the general context of digital health interventions is often connected to the prospect of spreading access to psychological support to more individuals in need (Roberts et al. 2018). However, the lack of regulation in the digital health market gives grounds for data to be used for other purposes that are not aligned with this understanding

of beneficence. For instance, the use of chatbots for the collection of sensitive personal data might have the goal of feeding profiling algorithms that in turn can be used in predatory advertisement and exploitative businesses (Gentsch 2019).

In addition, in the particular case of automated CBT, benefits can only be achieved if there is evidence of the efficacy of the intervention. However, the vast majority of digital psychology interventions are still in the testing phase or have only been evaluated for a short time (Provoost et al. 2017; Bendig et al. 2019). Although scarce, previous controlled research into clinical outcomes indicates that it is possible to demonstrate efficacy rigorously, but it can take time (Ma et al. 2019).

Besides restricting its goals to the supply of health support and perhaps clinical research, creators developing CBT chatbots should strive to inform users about the intervention's intentions and any indications of its efficacy in leveraging the principle of beneficence. Conversational dialogues, which are characteristic of chatbots, could be the channel to explain to users about such information items. Other interfaces surrounding this interaction (for example, textual content within the mobile app) could also contain informative materials.

#### Non-maleficence

The principle of non-maleficence advocates that not harming is just as important as doing good. Chatbots for CBT collect and use data voluntarily disclosed by users during therapeutic dialogues, yet the negative consequences of data leakage and misuse infringe the principle of non-maleficence. Even if accidental, the disclosure of intimate therapeutic content can lead to social discrimination due to the stigma attributed to mental illness (Corrigan 2000; Penn et al. 2003). Reduced employment opportunities and forensic investigations may harm individuals if data collected during psychotherapy by chatbots are made available to the government or companies (Martinez-Martin et al. 2018b).

Interaction with a chatbot can also be harmful, especially if responses might be inappropriate for the context (for example, if a user is disclosing thoughts about suicide). Chatbot failures are aggravated by the fact that after a negative experience, users might be less willing to engage with in-person clinical support as well (Bauer et al. 2017). Nonmaleficence applied to the context of automated psychotherapy involves not only protecting data privacy but also considering if the feedback provided after a data disclosure might be detrimental to end users' health (Miner et al. 2017).

Above all, CBT-chatbots should strive to attend to non-maleficence by keeping data secure and private. In addition, unlike traditional in-person psychotherapy, many chatbots maintain a textual record of conversations, which means there should be ways to alert users of the risks this imposes. Chatbots' dialogues could be the vehicle for this type of communication, for instance.

#### Explicability

The principle of explicability refers to the capacity of outcomes to be made transparent and understandable. This principle has often been linked to privacy policies concerning data disclosures for digital health interventions. While some individuals may readily agree to use chatbots for digital psychotherapy, they might not fully understand the terms and conditions regarding using and re-using the data they disclose (Martinez-Martin et al. 2018a). Furthermore, it is not always clear to end-users the limitations of the responses of an artificial agent (Luger et al. 2016). In the case of psychotherapy chatbots, the lack of explicability to how dialogues are artificially produced might also have consequences for end-user's beliefs in therapeutic effectiveness.

As extensively discussed in previous chapters, past research has emphasised that word

choices and excessive length compromise the explicability of legal clauses and obfuscate aspects of high importance to users, such as how their identity is protected (Ostherr et al. 2017). For this reason, the consent provided is often deemed not valid if users do not grasp the data-sharing permissions entirely (Martinez-Martin et al. 2018a). Therefore, it is not sufficient to provide information to users - it is recommended that this information should come in language and a format that facilitates comprehension on behalf of all populations (Luger et al. 2013).

#### Autonomy

The principle of autonomy is related to the capacity of individuals to act and make choices freely. This principle is harmed when users cannot choose how data is used and re-used, as discussed in previous Chapters. To protect their autonomy, individuals may choose not to disclose certain aspects of their life out of fear of potential data breaches and disciplinary actions that might restrict their activities in the future (Rooksby et al. 2019). End-users should be offered ways to control how data is potentially shared beyond the therapeutic context. Data permissions could also possibly be requested through the chatbot dialogues.

Autonomy is also breached if users become liable to manipulation, as in the coercive selling of system upgrades so that users can get access to therapy (Gentsch 2019; Hiolle et al. 2012). Manipulation is a well-known form of unethical conduct in traditional psychotherapy, yet is currently less regulated within digital interventions (Koocher et al. 2008). A lack of autonomy can constrain the formation of a healthy therapeutic alliance grounded in trust, and for this reason, end-users should not be forced to accept conditions that may prove detrimental to their progress.

#### Justice

The principle of justice promotes equality, inclusiveness, diversity, and solidarity. The unequal inclusion of end-users from different backgrounds in research and development is a well-known source of algorithmic bias and injustice, leading to the creation of systems that fail to reflect the needs of minorities. In the particular case of chatbots, the data from which language models are built may be tainted with racist, sexist, and discriminatory speech that, if left unattended, can be harmful to end-users (Schlesinger et al. 2018). Such disrespectful health data interactions are aggravated by the absence of regulation and ethical standards for medical devices that attempt to present AI and health data use as innovative advantages (Martinez-Martin et al. 2018b).

The principle of justice can be understood in terms of responsibility attribution and accountability. Recent evaluations of AI ethics have highlighted that current regulations lack mechanisms to reinforce and punish negative consequences for ethics violations (Hagendorff 2020). Past research has pointed out that strict guidelines for the development of AI for medical devices are still missing as of today, thus further complicating matters of justice in this context (Bauer et al. 2017). Without definitive positions regarding accountability, it remains unclear which entities should be responsible for ethical issues related to therapeutic disclosures made through chatbot interaction (Berscheid et al. 2019).

It is challenging to consider the application of justice only in terms of user interface design. However, some choices that may help to attend to the principle of justice include the creation of datasets that respectfully address discriminatory speech and considering how pricing, hardware/software requirements, and language might hinder access to intervention. Another approach that goes beyond the design of user interfaces could be to adopt existing accountability and risk mitigation strategies employed by healthcare workers (Nathan et al. 2015).

# 7.2.3 Section Summary

Despite the strong appeal of automating psychotherapy, such endeavours must be considered with caution. In a similar fashion to the discussion of health data interfaces for *research purposes*, there are fundamental challenges to be tackled in the context of *health interventions*, which could be surfaced through the engagement with the five core principles. The use of the AI-ethics framework led to several suggestions pointing towards the use of chatbot dialogues as the interface through which general information and data permissions might be communicated and configured — an idea that avails of the specific features of this type of health data interface.

# 7.3 In Summary

In this chapter, human values and ethical principles allowed us to explore the unique circumstances through which data contributors may voluntarily disclose data through these interfaces. As a result of these methodological explorations, we contribute design implications for developing two specific health data collection interfaces: a mobile app designed to support health research studies and chatbots designed to enable psychotherapy. Regarding future research, the StudyApp will be developed further for a research project involving children with Obsessive-compulsive Disorder (OCD). A 9-week feasibility study will assess how ten families feel while using the app through the User Experience Questionnaire (Laugwitz et al. 2008) and Trust Scale (Gulati et al. 2019), and post-study semistructured interviews will guide further iterations. Similarly, future projects concerning conversational agents for mental health intend to follow up on some of the challenges brought forth by the analysis of chatbots for CBT this time with the engagement of actual end-users.

Even though empirical findings and explorations of methods provide insight into ethicallyoriented design, this may not be enough to shape practice. Technology creators might need to learn how and why to apply ethical thinking to user-centred design processes. Motivated by the vital role of educating future generations of technology creators, the next chapter turns to studying how others may learn and apply approaches similar to the one described in this chapter. 8 Practising Ethical Design | Educational Approach

As we have learned in our review of ethics across HCI presented in Chapter 3, computing ethics education is increasingly advocated for preparing university students for the complex design dilemmas they are likely to face in their future workplaces. Precisely how this can be best achieved, however, remains an open question. Our own experiences applying ethical thinking to technology design led us to explore how to teach others to do the same, thus contributing to the emergent ethics education efforts. We conducted a pilot study with engineering students from the Technical University of Denmark enrolled in an introductory UX Design class to observe how they engaged with ethical principles during a prototyping activity. This chapter describes the results of this pilot study and continues to move forward the work of this thesis by exploring technology creators' perspectives following their attempts to put ethical design into practice. The content of this chapter is drawn from the research paper *"Teaching User Experience Design Ethics to Engineering Students: Lessons Learned"*, currently under peer-review (See Paper 2.7).

# 8.1 Introduction

Ethics education comprises a much-needed foundation for the next generations of designers and developers (Skirpan et al. 2018a). The inclusion of 'computing ethics' in university programmes has been considered essential to preparing future professionals to face the increasingly urgent dilemmas on the role and impact of digital technologies in society (Gray et al. 2018; Chivukula et al. 2019). Pedagogical approaches for teaching ethics have included: first-hand experiential learning to facilitate empathy-building towards accessibility issues (El-Glaly et al. 2020), role-playing games for simulating the process of writing and adhering to privacy policies (Shilton et al. 2020), and science fiction writing to encourage moral imagination in relation to the drawbacks of AI (Burton et al. 2018).

Another possible teaching approach, however, less often explored by the literature pertains to the use of top-down ethical frameworks to guide students' design work. The development of 'ethics checklists' is an increasingly common practice among technology companies as means of attempting to alleviate the difficulty practitioners face when it comes to operationalising abstract principles (Madaio et al. 2020). Previous pilot studies indicated that ethical questions are a helpful starting point when it comes to exploring ethical considerations in the classroom, but similar approaches are yet to be evaluated in the context of health data ecosystems (Saltz et al. 2019). We conducted the following pilot study to understand how this approach might be applied to this application domain.

# 8.2 Teaching Context

The pilot study investigated how the students applied purposefully-designed teaching materials to support their design work during this one-week project. The study was conducted as a part of an introductory UX Design course, in which lean business models and design thinking were core components of the teaching program. Weekly classes consisted of one-hour lectures followed by 3 hours of supervised group work (with teaching assistants). After the class, students were asked to prepare and submit a set of deliverables (business model canvas, user story maps, interactive prototypes, a report on prototype evaluation) based on a design brief. The course was therefore structured using 'projectbased learning' pedagogy (Kokotsaki et al. 2016). The pilot study took place as part of a particular project in which students had to design in groups a prototype for a smartwatch app that would be used to collect, visualise and share heart rate data between patients, researchers and doctors (See Figure 8.1). The design brief was motivated by previous research arguing that wearable technologies can empower patients at risk by continuously monitoring their heart rate and alerting them and their doctors when episodes of cardiac arrhythmia arise (Kumar 2021). The assignment description also explained that data collected through the smartwatch was meant to keep clinicians informed about the status of their patients in between consultations and that data could serve as material for scientific investigations.



Figure 8.1: Example of a smartwatch prototype created by a group of students

# 8.2.1 Educational Materials

An essential part of this pilot study was the design of the teaching materials. Lecture slides were prepared with the goal of introducing students to the connection between ethics and UX design. The slides presented examples of user interface design choices that may infringe upon ethical principles as a way to illustrate why ethics is important. These examples of unethical design included interfaces that do not offer data sharing options, are not transparent about data use, attempt to exploit users for profit, actively exclude minorities and foster harmful behaviours, such as eating disorders. The principles being infringed upon in these instances were: choice, transparency, reciprocity, inclusion and well-being (See Figure 8.2<sup>1</sup>).

This work further builds upon the ethical principles chosen to structure our critical analysis of CBT-chatbots (Morley et al. 2019) and the framework by Nebeker et al. 2019 for actionable ethics in digital health. We adapted the names of some of the principles to a more friendly vocabulary so that the students could more easily develop a shared understanding of these concepts. It was explained to the students that ethical principles are one way of engaging with ethics during the design process and that the primary goal of learning about these principles is to facilitate the articulation of ethical requirements.

In addition to the lecture slides, two templates<sup>2</sup> were designed as tools to support group discussion and self-reflection. Students were asked to use the templates at any stage of their design process (before or after designing the prototype) and adapt these ethical inquiry templates to the particularities of their own projects.

The first template consisted of a checklist (See box 8.1) containing items to remind the

<sup>&</sup>lt;sup>1</sup>Figure extracted from Paper 2.7

<sup>&</sup>lt;sup>2</sup>Templates extracted from Paper 2.7



Figure 8.2: Infographic illustrating the five normative principles taught to the students

students of ways in which their prototypes could respect the above principles ("The system is transparent about how data is used and shared"). The second template was an online form (See box 8.2) that students could also adapt and send to their peers (or users) to collect external feedback, including questions such as "From not at all (0) to completely (7), how much do you think the system is transparent and clear about how it works? Why?".

# 8.3 Method

The pilot study evaluating the educational approach described above was carried out after the students delivered the prototypes and reports for the class assignment. Semistructured interviews were conducted with those who volunteered to participate in the evaluation study. The goal was to collect qualitative data from the students about their learning experiences.

Participants were 12 students at the Technical University of Denmark, taking a 13-week course on UX Design. A teaching assistant conducted interviews to ensure confidentiality from the professor responsible for the class. Audio recordings were transcribed, and an inductive Thematic Analysis was conducted following Braun and Clarke's framework (Braun et al. 2012). Further details on how the study was conducted can be found on Paper 2.7.

# 8.4 Results

# 8.4.1 The Approach Raised Awareness

Across the sample, UX design ethics was described as a topic not yet examined by many of the students until the UX design course: *"It is the first time I hear about ethics in design"* (P2). Nonetheless, at least half of the students expressed being interested in the subject: *"I have not thought about it, but as soon as I read it, I was like, okay, this is important, it is something that I really want to address because ethics is something that I care about"* (P11). The educational materials were said to be effective in bringing the topic to the students' attention: *"The lecture you gave raised some awareness. Since that lecture, ethics has been part of our work in the group"* (P1). The introduction to UX ethics also changed some of the students' perspectives about UX design: *"We thought about ethics, but maybe not in a good way. We did the opposite with the previous courses. We thought:* 

Choice
The system allows users to choose and select their preferences
The system does not force single options to users (for example, I accept)
The system allows users to change their mind and select a different option later
Transparency
The system provides enough information about how it works
The system is transparent about how data is used and shared
The system does not hide important information on purpose
Well-being
The system is designed to help users to improve their well-being
□ The system does not make users' well-being worst (for example, making them anx-
ious)
□ The system provides direct support to users in case they suffer physical or emotional
harm
Inclusion
$\hfill\square$ The system is directly targeted to attend the needs of at least one marginalised
group
The system does not exclusively target only over-represented groups
□ The system does not exclude people of different races, disabilities, gender and cul-
tures
Reciprocity
The system is useful by offering direct and significant benefits for the users
□ The system does not use manipulation techniques to get users to spend money
The system is not built to take advantage of users for business gains
Box 8.1: Checklist template

how can we be as evil as possible with this? How can we gather as much data? How can we blackmail the user the most? Now we think the opposite" (P6).

# 8.4.2 Principles Helped to Discover Concerns

Most students perceived the templates as helpful guides as to what to pay attention to when designing user interface features: *"What we did was to use the templates, and that is how we learned how to do it. Without the templates, we would not be able to know what to change"* (P5). Participants used the principles of the templates to discuss ethical issues as they attempted to prioritise ethical considerations in their prototypes: *"We tried to add more things to the smartwatch regarding transparency and freedom of choice"* (P4), and *"We were thinking about choice, being able to know what you are showing and having more settings. In the first app that we made, we did not have settings"* (P11). The students also discussed how they tried to address concerns related to potential harms to well-being: *"The whole point is trying to make the users not feel bad if they have not done something they should have done. The notification could like tell them to go out for a walk without trying to make them feel bad and just try to stay positive"* (P10).

# 8.4.3 Ethics Was a New Topic to Many

Given that the topic was a novelty to most, a lack of previous experience made the activity challenging: *"I feel hesitant, doubtful, concerned because I have never heard of the topic before"* (P2). The lack of specialised domain knowledge (healthcare) was also perceived as an obstacle: *"We felt a lack of knowledge because in this particular case, we need a doctor to say what is more important. Maybe it could be nice to have more health information because we know this is something we should take into consideration but we* 

#### From not at all (0) to completely (7), how much do you think the system...

- 1. ... allows users to choose and select their preferences? Why?
  - 2. ... is transparent and clear about how it works? Why?
  - 3. ... promotes well-being and protects users from harm? Why?
  - 4. ... can be used by diverse races, gender identities and physical mental abilities? Why?
  - 5. ... provides mutual benefits for owners and users without manipulation? Why?

#### Box 8.2: Feedback form template

do not know the potential damage" (P4). In particular, a participant expressed feeling incapable of knowing how to design for disabled users: "I don't know how to include the disabled. I think it is important, but I have no idea how. You must be the blind person to understand the blind person" (P2). Similarly, another student was also not aware of how blind users could use mobile phones: "One of the comments that we kept getting when we were reviewing each other's solutions was that blind people would not be able to use this but are blind users even able to use apps?" (P1).

#### 8.4.4 Some Perceived Ethics as Not Important

In contrast with the above findings, a small number of students commented that they believed that ethics should not be a priority in design: "I think it was a good add on to the course, but I do not consider it being a high priority"(P1). In their view, ethical ideals can hinder business opportunities: "I think it is rather unrealistic to incorporate ethics in such a corporate area. How would you ask big corporations or developing companies to be more aware of ethics if it is clear that their primary concern is money?" (P2). One participant, for instance, explicitly stated that ethics was an obstacle to profit: "Data is money, and all I ever wanted is to make money. So we need all the data even if you do not want to share it, that was our app's logic: money" (P7). As one of the participants alternatively remarked, getting a high grade was ultimately their own most important motivation: "In the end, we were caring about a good grade, so I am not going to lie this was the reason behind" (P3).

#### 8.4.5 Group Members Had Conflicts

Social dynamics played a role in how group discussions were held within teams. A few students mentioned that it was challenging to reach a consensus in the group: "We have been able to agree on many things, but we are a group of people who do not know each other very well, so we do not always turn out super compatible. It is hard to say: 'I think you should change all the work you just did' " (P12). A reason for disagreements could be found in the different levels of interest each member of the group had: "We were more interested in it than the others. It is not that they were against it, they just did not care" (P12). One of the students mentioned that in order to handle such conflicts, the group resorted to the feedback of other peers: "The moment one has a question and asks the group, but we cannot agree in a few minutes, we decide to validate the concept with others" (P5).

#### 8.4.6 Time Pressure Was Frustrating

Given the requirements for weekly deliverables, time was a scarce resource, which meant that students had to decide what to prioritise. However, ethics was perceived by many as a topic that demands a longer time frame: *"At that point, we were too busy and concerned with the next hand-in. We were just going to leave it because we did not have much time. We were not making great philosophical discussions about everything, but if we* 

had a longer time frame, we could do it" (P2). A suggestion made by one participant was to learn from examples in order to speed up the process: "Maybe we could see some examples of how to implement it more quickly" (P3). Besides a lack of time, a small number of participants expressed a desire to start the design process from a perspective of ethical deliberation rather than discussing ethics as an after-thought: "It felt stressful and frustrating because it was late in the process, and I feel like that is something that should have come earlier" (P11).

# 8.5 In Summary

This pilot study demonstrated the possibilities and limitations of teaching ethics by asking students to use ethical principles to guide their prototyping activities. Our pilot study effectively raised students' awareness, and the structured templates (a checklist and feedback form) provided helpful guidance. On the other hand, the challenges students experienced during the process serve as valuable insights for future education efforts. It became clear that there is a need for more extensive teaching methods, complemented by real-world user engagement if possible, and exercises for empathy-building. This exploration of the teaching of ethics complements the other approaches to ethics explored in this thesis by recognising the impact of grounding design decision-making in ethical principles and supporting the education of technology creators that may develop future ethical health data ecosystems. The next chapter serves to provide further reflection on this approach to supporting ethical design, as those others presented, explored and promoted through the work of this thesis.

# 9 Discussion

One of the primary and most persistent challenges in technology development to date pertains to integrating ethical concerns into design processes, especially when it comes to technologies that directly affect collective matters of civic life. Throughout this thesis, we have identified, discussed, and applied a portfolio of approaches to raising ethical considerations in practice. The evidence gathered from our engagement with young adults and young technology creators allows the integration of research findings into design recommendations for health data interfaces that respect data contributors' values.

This final chapter connects the results of this extensive research project set to "discover and act upon ethical requirements to design respectful health data flows" (See Chapter 1). Our road map started from a thorough understanding of the socio-technological context at hand (See Chapter 2), which then served as the foundation for the subsequent user studies (See Chapters 4, 5 and 6). With the knowledge gathered directly from data contributors and technology creators, we could conceptually envision and critically assess health data interface designs (See Chapter 7). We then moved to promote what we have learned to future generations of designers, researchers and developers (See Chapter 8). Ethics was, therefore, the core thread uniting these investigations and this engagement with ethical thinking was majorly supported by the scoping review we conducted (See Chapter 3).

The following sections highlight key implications derived throughout this PhD project and discuss our reflections about the overall process. These insights are also further linked to the relevant literature. As for the more detailed consideration of methodological limitations, this can be found in the respective discussion section in each of the papers available in Part II of this thesis.

# 9.1 Understanding What Matters to Data Contributors

In our review of existing health data repositories, we have discussed how digital health has traditionally focused on the needs of those consuming data: clinicians, researchers and public workers (See section 2.2 in Chapter 2). As a consequence, those who store and use the data (not those who produced it) have traditionally retained the power and knowledge that comes with data ownership (Shilton et al. 2021). The work of this thesis in its collective form makes a case for turning the focus towards understanding in greater depth the perspectives of those contributing data — in itself an essential step towards more ethical health data ecosystem design.

This shift equally requires and results in a revised conceptualisation of 'public acceptance', according to which individuals 'accept' to contribute because they feel comfortable disclosing their personal data and believe this act will be beneficial both to society and themselves. Data contributors are vital to health data repositories, yet they are much more than objects to be observed or a collection of entries in a database (Doherty et al. 2020). The narrative review of contributors' willingness to share data described in Chapter 2 and the combined work of the user studies presented across Chapters 4, 5 and 6 highlight and reinforce the critical nature of the three components for this ethically motivated acceptance: reciprocal exchange, privacy boundaries and trusting relationships.
## 9.1.1 Reciprocal Exchange

Ethical incentives are essential to communicating and realising the benefits of data sharing (See section 2.3.1 in Chapter 2). In contrast with usual research practices involving humans, the survey study conducted in this thesis work showed (across a sample of 1600 respondents) that material compensation does not represent the most appealing of motivations (See section 5.3.3 in Chapter 5). Options such as 'suggesting new research questions' were rated as very appealing for more participants than the possibility of receiving financial benefits, for example. Such evidence implies that ethical participation should probably entail meaningful incentives beyond money.

The prospect of helping future patients and future researchers appears as a powerful incentive. Both the survey and interview studies (See section 4.3.1 of Chapter 4) revealed that participants have a desire to help others through their data contributions, aligning with prior literature emphasising the importance of altruism and beneficence in motivating people to participate in research (Kraft et al. 2018; Tan et al. 2018; Ziefle et al. 2016). The focus group with technology creators conducted in the earlier stages of this thesis further complemented these findings with experts' commentary to a shared belief that generating scientific discovery should be the main reason leading people should agree with data sharing (See section 6.3.1 of Chapter 6). For this reason, it is crucial that data contributors are appropriately informed about how their contribution actually enables advanced clinical research.

Besides such demonstrations of selflessness, previous research about genetic data analysis has shown that learning about their own health can also be a significant motivation for volunteers (King 2019; Nobile et al. 2013). This observation was confirmed through the work of this thesis concerning a broader sampling of data types (for example, mobile sensing data). Participants in both the interview and survey studies demonstrated a strong interest in getting feedback about their health status. While this interest is likely connected to the innate human desire for growth and self-actualisation (Maslow 2013), the provision of personal benefits in return for data exchange points also to the value of 'reciprocity', as surfaced and discussed during the design of the StudyApp in the latter stages of this work (See section 7.1.4 Chapter 7). In the same way that individuals may therefore agree to disclose sensitive information with a chatbot to receive the benefits of therapy (See section 7.2.2 in Chapter 7), this thesis highlights reciprocal exchange as vital to the ethical acceptance of health data sharing; for research and intervention.

However, the way that personal health feedback is provided to patients and citizens must be mindful of individual preferences. A key finding of the survey study we conducted is that those uncomfortable with sharing alcohol consumption, levels of stress, and physical exercise are less likely to be interested in receiving feedback about themselves (See Chapter 5). Feelings of shame can probably explain this correlation, as participants in our interview study expressed feeling ashamed of their alcohol and tobacco consumption, for example (See Chapter 4). The technical students on our pilot teaching study have also observed that providing continuous heart rate measures for patients presenting cardiac arrhythmia can make them even more stressed and concerned (See Chapter 8). These findings together confirm that in order to return personal data in the form of health feedback ethically, technology creators must consider each individual's conditions, and a 'one-sizefits-all' generic design will probably not attend to such divergent needs (Bhat et al. 2020).

## 9.1.2 Privacy Boundaries

Much prior data sharing research has focused on the important question of privacy (See section 2.3.2 in Chapter 2). As part of this thesis, the interview and survey studies shed

light on data contributors' divergent views on the privacy risks associated with different data types, corroborating prior literature (Bhatia et al. 2018; Ahram et al. 2014). Our user studies extend these previous findings with a much more nuanced understanding of contributors' willingness to share both digital and non-digital sources of data. This analysis highlights that a desire for privacy protection is strongly related to the personal perception of risks - a finding elevating the need for technology creators to consider designs that target the specific reasons behind data contributors' reluctance to share each data type (Barkhuus 2012).

Data collected through mobile and wearable sensing technologies, for example, were perceived as uncomfortable to be disclosed by most, even more than the very revealing DNA samples (See section 5.3.5, Chapter 5). Our interview study indicated that participants were more reluctant to reveal information if they could not see a connection with research goals, and many were negatively surprised to learn that digital data sources, such as location tracking and texting patterns, are being used in scientific investigations (See section 4.3.3 in Chapter 4). This evidence is particularly relevant to digital health researchers intending to request access to mobile and wearable sensing data, as they might need to re-consider if the benefits of collecting such data sources surpass the discomfort people might have with this prospect.

Our survey results further indicated that the more granular the data types were, the more the contributors were unwilling to share them - even if the data was de-identified (See section 5.3.5, Chapter 5). The *frequency* of texts and calls was, for example, considered more comfortably shared than their *content* — participants in our interview study reasoning that such detailed data yielded additional risks to individual safety and freedom, including hacking and stalking concerns (See section 4.3.3, Chapter 4). Data contributors' discomfort sharing data sources that expose their location, social communication style, and phone usage can therefore compromise an ethical and broad public acceptance (Rooksby et al. 2019).

In particular, those unwilling to share data about diagnoses of mental illness were found to be more likely to fear social discrimination according to our survey study (See section 5.3.5, Chapter 5). Participants of our focus group study also raised the same concern, stating that end-users often perceive mental health data as a very sensitive type of information (See section 6.3.2 of Chapter 6). Similarly, we have also discussed the harms that the leakage of therapeutic disclosures made through automated psychotherapy technologies could represent to patients (See section 7.2.2, Chapter 2.6). When taking these insights together, it becomes evident that digital health innovations that target patients suffering with mental illnesses must consider what is at stake for these users who willingly open up about a part of their lives that can damage their reputation, employability and self-esteem (Murnane et al. 2018; Rooksby et al. 2019).

## 9.1.3 Trusting Relationships

This thesis shows that when it comes to designing health data flows, trust matters. Our survey study found that most data contributors fear that health data can be misused by unethical agents, which is a clear indication of lack of trust (See section 5.3.4 of Chapter 5). The Brazilian survey sample, in particular, perceived the idea of sharing data with public entities as less acceptable than the Danish one - a finding most likely related to the lower level of trust Brazilians display in government institutions (Zammar et al. 2010; Middleton et al. 2019). In contrast, people living in Denmark felt comfortable with the prospect of allowing repository owners to decide how their data is used (Duckert et al. 2022). These insights demonstrate that it is crucial for data contributors to believe that they can trust

those requesting access to their data for having ethical conduct, as negative experiences can erode trust as well as scandalous data abuse episodes (Murmann et al. 2017; Kizilcec 2016; Rieger et al. 2019).

Prior research frequently highlights transparency as a vital enabler of trust, especially when individuals have their initial expectations violated and need to clarify inconsistencies (Ziefle et al. 2016; Adjekum et al. 2018; Shklovski et al. 2014). Most of our survey respondents expressed a desire to receive continuous information about who is using their data (See section 5.3.6, Chapter 5). This finding can be explained on the basis that people often feel more comfortable if they know that their data will benefit the common good and not others' profit-making (Patil et al. 2016; Ostherr et al. 2017; Kraft et al. 2018; Bhatia et al. 2018). Our discussion about the principle of explicability concerning CBT-chatbot design (See section 7.2.2 of Chapter 7) also highlights that it is vital that technology can be seen to behave in ways that respect users' expectations. Therefore, the results of our studies with data contributors and technology creators in combination point to the urgency to consider transparent communication in future developments.

Another critical component for trust in this context is user autonomy. Comments made by data contributors across all phases of this work express the importance for them being allowed to have a choice over decisions made regarding who can use their data and for which goals (See section 4.3.2 of Chapter 4). Previous research has further emphasised that merely notifying individuals about possible data re-use is considered less acceptable by data contributors than requesting their permission for each new data usage - another evidence of the crucial role of respecting user autonomy (Ostherr et al. 2017; Ludman et al. 2010).

However, as shown by this thesis' review of health data repositories (See section 2.2, Chapter 2), most larger-scale initiatives still rely on a broad consent approach, which means that projects can use data at any time without contributors having any autonomy. For these reasons, we posit that shifting to more granular consent styles and increasing data control choices is vital in enabling freedom and choice. We explored such an idea in the design of the StudyApp (See section 7.1.4, Chapter 7) and the same way of thinking can probably be applied to other common forms of health data interfaces as well. In further exploring the translation of research findings to design implications, the next section expands recommendations for future developments for health data interfaces.

## 9.2 Designing Future Health Data Interfaces

The previous section emphasised and discussed the role of reciprocal exchanges, privacy boundaries and trustworthy relationships in fostering an ethical acceptance of health data technologies. Translating such abstract concepts into design choices is, however, a challenge in itself, as seen in our pilot study with technical degree students (See Chapter 8). This thesis offers paths forward by providing and discussing a set of broad but well-informed design alternatives based on the evidence gathered in the user studies and the literature. Whilst the subsequent implications for design might be particularly relevant to young adults, who were the main focus of the user studies conducted in this thesis, findings may still hold value to other populations, as findings appear much aligned with previous discussions in the literature.

The following design recommendations build upon an 'emancipatory' vision for the future of health data sharing. As outlined in the ACM Interactions Magazine article 2.8 (available in Part II), paths to public acceptance should be grounded upon the ethically justified need to collect personal data in the first place. Then, any initiative to collect, store and re-share

health data should act responsibly to correspond to the expectations of data contributors instead of attempting to coerce individuals to accept unfavourable terms of conditions. In this conceptual design vision, health data repositories position citizens and patients as 'permanent owners' of their personal health data, even after data is stored in a repository.

The evidence collected in this thesis serves as evidence for conceiving such respectful designs that may help achieve the desired societal benefit: better healthcare for all. Abusing this knowledge to coerce individuals to share data despite their concerns, merely to increase public acceptance, would not be an ethical path. The following recommendations aim to turn the experience of health data contribution into a rewarding interaction that prioritises individual well-being, safety, and freedom, as most of the existing platforms have so far focused on catering for the needs of data consumers - clinicians, researchers and public workers (See section 2.2 in Chapter 2).

## 9.2.1 Alternative Interface Modalities

The conceptualisation of health data as 'flows' and 'interfaces' allowed us to convey that when data 'flows' between entities, consumers and contributors interact with these flows through user interfaces. According to this abstraction, an 'interface' does not necessarily have to be materialised as a *digital* user interface such as the StudyApp we prototyped (See section 7.1.4 of Chapter 7). Non-digital options (for example, paper-based form and face-to-face encounters) can even potentially attend to a more diverse group of people who may not have access to digital devices, which is crucial to supporting inclusion and accessibility (De Sutter et al. 2021). Technology creators should stay open to blending digital and non-digital designs to realise optimal configurations for the interfaces composing health data ecosystems. Interfaces can consist of e-mails, letters, news, reports, landing web pages, chatbots (See Chapter 7) - each artefact thus mediating how people understand, take ownership of, and create value from data.

## 9.2.2 Transparent Communication

Regardless of the interface chosen, transparent communication is extremely important for health data interactions. Requirements surfaced through the user studies of this thesis include that the organisation(s) managing data repositories should be more transparent about: who is accessing data, the goals of individual projects and whether any entity might be profiting as a result. Moreover, most survey participants also expressed being interested in receiving updates about the outcomes of research conducted with their data (See section 5.3.3, Chapter 5). Depending on the setup of the health data repository and the preferences of its users, these information items could be transmitted through many different channels, such as web-pages, e-mail newsletters, data collection mobile apps and even paper-based postal correspondence. The most important is to provide this information in a way that data contributors can access it at any point, even after formally providing consent, in order to allow them to become aware of what is happening with their data and privacy (Duckert et al. 2022).

## 9.2.3 Interactive Learning Opportunities

There is also an opportunity for technology creators to use interactive learning materials creatively. Our interview study shows, for instance, that data contributors are intrigued about the connections between health outcomes and passively sensed behaviours (See Chapter 4), even though this seemed obvious to the technology creators in our focus group (See Chapter 6). The goal of adding easy-to-digest educational resources could be to invite data contributors to understand better how their contribution is relevant and become more aware of possible risks. In addition, even though it can be challenging to convey technical aspects of security and anonymisation protocols (Clark et al. 2019), our

user studies found that people desire to be able to understand how their data is protected and the potential risk of retracing personal identities. Insights and recommendations from science communication studies could further help on this endeavour, such as a stronger focus on not only conveying messages effectively but also engaging with the experimental, cultural and affective dimensions of the meaning-making process it triggers (Davies et al. 2019).

## 9.2.4 Personalised Health Feedback

The findings of this thesis furthermore underline the possibility of providing meaningful feedback in support of personal health goals, following the 'quantified-self' movement (Fleck et al. 2010). However, despite the appeal of data visualisation and quantitative measures of health, previous research argues that personal health tracking design should be careful not to trigger excessive self-scrutiny and low self-esteem (Chua et al. 2016; Murnane et al. 2018), especially among young adults who can become obsessed, anxious or deeply frustrated (Kelley et al. 2017). Some patients might also need the supervision of specialists to interpret health information appropriately, which means that simply giving people access to statistics and behavioural metrics can prove overly confusing and misleading (Mentis et al. 2017; West et al. 2016). Previous research has further advised against the promotion of an intense 'datafication' of well-being, as not every data point can be a reliable indicator of health (Chandwani et al. 2016; Bhat et al. 2020; Epstein et al. 2016). For these reasons, even though data contributors are interested in receiving personalised feedback and designing for that could be a way to promote 'reciprocal exchanges', it is fundamental to consider the arguments of this previous research before developing potentially harmful user interfaces.

## 9.2.5 Flexible and Continuous Data Control

Flexible data sharing is an option of significant public interest that surfaced in this work and the literature (Abdelhamid 2018; Kaye et al. 2015; Kuntsman et al. 2019). According to the user studies conducted in this thesis, data contributors desire to exert more control over which data is shared and with whom. Allowing such choices is a way to increase user autonomy (See Chapter 3) and an enabler for broader public acceptance (See Chapter 2). When contemplating specific ways to design for personalised data controls, however, it is important to avoid overwhelming contributors with requests, as not everyone shares the same desire to act upon their privacy choices and people can change their minds (Barkhuus 2012). For this reason, it might be wise to allow contributors themselves to choose how they prefer to express data access permissions, as discussed in the analysis of applications of technology in support of automated psychotherapy (See section 7.2.2 of Chapter 7). For instance, data contributors might choose to provide consent on a case-by-case basis or allow data to be always shared by default unless there is a specific receiver or purpose that the contributors might prefer to deny access permissions. Therefore, data contributors are allowed to flexibility decide which data types to share and how permissions for data re-sharing are to be solicited.

## 9.2.6 Active Community Engagement

Overall, the design implications discussed above suggest that data contributors should become more actively involved in the research process beyond simply sharing their data. The literature offers many other examples of meaningful opportunities for community engagement, including the possibility of proposing research questions, providing feedback on current projects and voting for research decisions - suggestions that the participants in our user studies have also endorsed (Bromley et al. 2015; Fleurence et al. 2013; Shippee et al. 2015). In the OpenHumans platform, for instance, members can engage in group discussions, browse through different projects and discuss ideas for data analysis (See

section 2.2 of Chapter 2). Inspired by health-related social networks, such as Patients-LikeMe, data contributors could also be interested in exchanging personal experiences with the platform with each other, clinicians or the repository owners (See section 2.2 of Chapter 2). Such additional features and communication channels would mean a shift from the traditional relationship between data contributors and data consumers towards a more balanced power dynamic, which is essential to enable ethical participation (Terry et al. 2011).

## 9.3 Supporting Technology Creators' Ethical Practices

The design implications discussed above serve as new and promising directions for an alternative and more ethical future for health data repository design. However, employing such insights in the real-world practice of design remains a significant additional challenge that requires engaging with and pondering methods, theories, and processes. This final discussion section contributes towards supporting the work of technology creators by offering reflections on the learnings acquired through first-person experiences and the extensive literature reviews conducted.

## 9.3.1 Reflections on Methodologies and Approaches

Ethical design is made complicated by the fact that technology creators need to make not only 'good' design decisions but also base these decisions on long-term consequences for users (Barry et al. 2017). Many have worried that it might be challenging to integrate even valid ethical reflections into the development of health data technologies due to a lack of professional awareness and willingness to incorporate ethical perspectives into the user-centred design process (Snyder 2020; Lupton 2014). The empirical user studies identified and conducted in this thesis surface noteworthy examples that corroborate the idea that, with proper motivation, it is possible to integrate an ethical perspective with commonplace user-centred design methods, such as interviews, surveys, and focus groups — the familiarity technology creators may already possess with these approaches proving an advantage (Sun et al. 2019; Figueiredo et al. 2020; Fiesler et al. 2018; Hirsch 2020; Skirpan et al. 2018c).

In this thesis, well-known user inquiry methods served as effective approaches to ethical design without requiring the complete remaking of research techniques and procedures. For instance, the design recommendations discussed in the previous section highlight not only the value that qualitative methods provide to understanding subjective, contextual, and intricate user experiences in terms of possible harms (See Chapter 4) but also how quantitative methods helped to reach a somewhat more objective basis for argumentation (See Chapter 5). Field studies evaluating ethically-informed designs 'in-the-wild' could further contribute towards the ecological validity of design suggestions, such as a future step planned for the StudyApp (See section 7.1 of Chapter 7).

Comparing and contrasting the two methods employed in Chapter 7, each allowed us to reflect on design choices differently. Using conceptual 'values' as anchors for the design rationale behind the StudyApp helped us to direct creativity towards enabling users to achieve their aspirations of higher agency and reciprocity through interactive elements within health data interfaces, such as data types selectors (See sections 7.1.4 and 7.1.4 of Chapter 7). Technology creators might find the VSD approach beneficial in their design processes, as it shifts the focus from user 'needs' to 'values', thus bringing a different, more ethically-engaged perspective on the definition of a 'system requirement'.

On the other hand, the employment of an ethical framework based on pre-established principles for the analysis of automated CBT led to the discovery of ethical considerations

beyond what might be addressed by user interface elements. Given that the AI-ethics framework we used was built from a previously acquired understanding of recurrent issues, this ethical analysis unfolded issues of justice and inclusion that the participants in our user studies did not state explicitly. However, this does not suggest a move away from attending to users but rather defend non-participatory approaches as also proving well-suited to facilitating earlier stage, conceptual, ethical discussions. Technology creators would do well to follow such procedures with empirical user studies, as developers and designers might not always fully grasp the lived experiences of end-users, as remarked by the students participating in our pilot study on the teaching of ethics (See Chapter 8).

## 9.3.2 Reflections on Theories and Design Philosophies

The range of theories, philosophies, and design orientations employed by HCI researchers to guide their engagement with ethics can prove immensely diverse, as seen in Chapter 2.1. Whilst some researchers argue that it is possible to make ethical decisions according to rational frameworks (McMillan et al. 2013), others construe ethics as an expression of care emerging from affective ties (Sabie et al. 2019). Some comment that perceptions of 'good' are born from individual virtue (Barry et al. 2017), while others emphasise the importance of socially established roles defining standards of conduct (Williams et al. 2020). Such disagreements, of course, predate HCI (Baggini 2018).

The lack of a definitive perspective on ethics, especially in a field as eclectic as HCI, can itself represent a risky position for technology creators, as without a common framework, it can difficult to reach agreements, especially when it comes to controversial subjects (Horst 2007). However, learning from different perspectives can lead to richer discussions if arguments are grounded in explicit deliberation of how diverse theories might conflict or complement a point-of-view (Mirnig et al. 2019). The view put forward by this work is that ethical thinking requires first attuning to the diverge range of possible moral philosophies and then wisely and coherently arguing why specific ideas, moral principles and actions are relevant to particular contexts (Baggini 2018). Although no theory can be considered the 'best' for ethical design, an informed and critical engagement with ethical theories is fundamental to putting ethics into practice.

A related ethos corroborating this argument for ethical pluralism can be found in the form of VSD. The VSD framework includes moral and ethical sensibilities to its working definition of 'human values' while allowing researchers to engage with consequentialist, deontological or Confucianist positions if appropriate. This approach does not deny the importance of fundamental principles for human flourishing (for example, human autonomy), yet it also encourages exploration of the reasons underlying divergent ways of thinking. Interestingly, VSD is the theoretical framework most often employed in the papers identified through our scoping review Chapter 2.1, which can be an indication that HCI demands theoretical frameworks that can accommodate plural perspectives while still attaining to the importance of key ethical values, such as autonomy, beneficence and justice.

When it comes to design philosophy, another perspective often found in HCI is Critical and Speculative Design (Sas et al. 2020; Ballard et al. 2019; Purpura et al. 2011), as our scoping review of ethics across HCI showed (See Chapter 3). The value in integrating Critical Theory with design lies in the way in which it brings unexpected insights to the process while also supporting the expression of design ideas and raising public awareness of problematic ideologies (Bardzell et al. 2013). Speculation serves as a means to fore-ground potentially hidden issues and communicates alternative (more ethical) versions for future systems, which are goals that technology creators might strive for if they are to commit to ethical practice (Dunne et al. 2013). Our applications of critical perspectives

applied to conceptual interface design serve as concrete examples of how this might be achieved in practice (See Chapter 7).

## 9.3.3 Reflections on Teaching and Learning

Creating a more ethical future for digital health requires professionals to learn how to move from discussion and debate to ethical design in practice. Building upon previous research identified through the scoping review of ethics reported in Chapter 3, we have attempted to cultivate future technology creators ethical awareness regarding the design of a smartwatch for heart rate monitoring (See the pilot study in Chapter 8). On the surface, our experience with teaching ethics to health data ecosystems was successful. Students gained more awareness of ethical design practices, and they felt that the list of ethical principles was a valuable resource for their creative process, which corroborates with prior work concerning the teaching of machine learning ethics through thematic questions (Saltz et al. 2019).

Despite these seemly supportive results, gaps in the students' diverse educational backgrounds and varied interest levels strongly affected their learning process. For instance, a student honestly stated that technology should be "as evil as possible", which only reinforces the importance of empathy development within technology creators. In particular, the ethical requirements that students had most difficulty approaching were related to accessibility and social justice, even though these are of undeniable importance for UX design (Inal et al. 2020; Costanza-Chock 2020). We have witnessed in practice how the navigation of ethical design has to be strongly tied with social awareness, goal setting and personal dispositions willing to embrace unavoidable complexities (See Chapter 8).

The work of this thesis serves equally to highlight a critical and yet pervasive challenge: respectful design cannot be achieved only through a series of deductive tasks or blind obedience to norms. It has been said that through consistent practice, individuals can eventually develop an 'ethical compass' that supports decision-making (Matchett 2009). This argument corroborates the idea that future and current technology creators probably need access to continuous and much more comprehensive ethics education to engage with ethical design genuinely. A concrete suggestion is for educational programs to combine theoretical exercises and hands-on classroom activities across several modules of university qualifications, thus going much beyond standalone lectures (Garrett et al. 2020). Another suggestion is to adopt such pedagogical approaches in the UX industry if suitable.

Furthermore, actively involving end-users in class could perhaps help technology creators and students understand the importance of learning about ethics. As demonstrated in a previous study, graduate students could develop an 'ethics of care' through a design project carried out in direct contact with communities and non-profit organisations (Sabie et al. 2019). As discussed in the first section of this chapter, the combination of participatory and non-participatory methods is probably the most advisable for a genuine engagement with ethics, and the evidence gathered through our studies points to the same conclusion. A single prototyping activity is far from sufficient to address personal misconceptions and cultivate an 'ethical designer character' (Lindberg et al. 2020). Finally, future researchers could examine further how to make ethical education an attractive topic to individuals who might not have previous experience with the open-ended and not straightforward nature of philosophical reasoning.

## 9.3.4 Reflections on Individual Responsibilities

This thesis and the literature in combination demonstrate that promoting engagement in ethical reflection requires that technology creators are provided not only with knowledge

but supportive environments favourable to the quest for more respectful designs. A challenge in this regard is that putting ethics in action is a demanding task for individuals, especially when it conflicts with dominant organisational goals (Gray et al. 2018). Previous studies have emphasised that the contexts in which technology creators are situated can hinder their capacity to act, as designing is usually an activity performed in teams, sponsored by business owners and funding agencies (Boyd et al. 2021). If ethical thinking is not supported by leaders, workplaces and the innovation culture, individuals might have to compromise their values (Friedman et al. 2019; Gray et al. 2020).

Therefore, although this thesis provides insight to support the enacting of ethical decisionmaking by research and design professionals, the responsibility for materialising ethical design, in reality, does not lie solely with these agents. As seen in the review of existing health data repositories, these platforms are often owned by national states, for-profit companies and research institutions (See Chapter 2). Without harmony across the roles of leaders, organisations, universities, governments and global entities to protect the interests of end-users, it becomes difficult to pursue and promote a broad ethical design agenda (Durrant et al. 2018; Shilton et al. 2017). In our focus group study, for instance, participants were not confident that data storage arrangements were secured entirely (See Chapter 6), which means that technical infrastructures probably need to be continuously evaluated and regulated by external entities (Horst 2007). Such regulatory frameworks do not necessarily need to become the overly rigid and bureaucratic structures criticised by previous research, but they are essential to strengthening ethical guidance in some jurisdictions that lack this foundation (Dow et al. 2019; Patil et al. 2016; Glerup et al. 2017).

Whilst the optimal role of regulations is still being consolidated and better understood by the professional community, it is paramount to find ways to protect data contributors from risk and empower technology creators to work towards more ethical designs. As health data has grown to become a resource with great value for many, this thesis calls for more substantial collective initiatives across academia, the public sector or as part of privately-owned endeavours. Suggestions include organising communities of practice sharing experiences and supporting leaders who promote working cultures that encourage critical thinking in design work. Further avenues for real-world change include developing communication tools to mediate ethical discussions across organisational hierarchies, including channels for citizens to express their concerns. In solidarity with the shared objective of improving healthcare for all, we hope this thesis serves as a source of inspiration, eye-opening perspective, and indeed hope for many others working towards the conception of more respectful health data flows.

## **10** Conclusion

In seeking to advance the development of health data interactions, this thesis has prioritised the position of data contributors to promote a shared understanding of how we might devise novel and respectful design possibilities rooted in user-centred research. Overcoming a prevailing stance that data sharing should primarily serve data consumers, the work carried out for this PhD has striven to offer an alternative vision in which those disclosing their personal data have their preferences and expectations prioritised. This thesis, therefore, serves as a step towards a future in which transparency, choice, mutual benefit, and trustworthiness are at the core of health data flow design.

The journey towards meeting this ambitious goal has included deep dives into the literature in parallel with inquiries entailing direct end-user engagement and the development of frameworks for critical appraisal and speculative prototyping. A structured search through academic knowledge sources concerning ethics across HCI has equipped the work of this thesis with a solid foundation regarding possible ways of thinking, doing and reporting on ethical design in practice. This thesis describes first-hand experiences with diverse yet complementary approaches (both qualitative and quantitative), enabling a rich depiction of that which matters most to data contributors.

This work reveals the extent to which citizens desire to become more actively engaged in learning about and acting upon how their data is collected, shared and used. Data contributors display varied needs for privacy boundaries, as disclosing certain data sources (for example, location, social communication and mental illness diagnosis) are far from being perceived as safe by many. Because of strong concerns regarding the risk of cyber-attacks and the possibility of data being abused by malicious agents, data contributors explicitly request more information and data control. These findings, in turn, serve to increase awareness among those responsible for creating and developing health data technologies about the reasons behind citizens' discomfort.

Furthermore, this thesis offers design recommendations that tap into the potential to transform the relationship between people who collect data and those who consume it. In particular, it is proposed that technology creators consider: communicating more openly about health research requirements, empowering data contributors to learn about clinical research, allowing flexible and customised data sharing options, and providing valuable personal benefits while reducing risks to the most vulnerable. These design suggestions should inspire new initiatives and encourage collective efforts towards more caring health data infrastructures.

Overall, this PhD project has sought to identify, express and outline pillars for a new vision of ethical digital health ecosystem design going forward. In striving to make this a reality, we have supported individual technology creators through research evidence, critical thinking frameworks and educational approaches. The most fundamental contribution aimed for this work is to emphasise the need to take consistent action so that the conception of any technology that directly affects citizens and patients' lives always stands in solidarity with those whom technical advances can harm. Hopefully, the contemporary movement to re-activate the role of ethics for user-centred design finds productive dispositions and welcoming minds, especially within innovation initiatives in the healthcare sector.

# Part II

# Manuscripts in the Paper Collection

## **1** Summary of Contributions

## A Scoping Review of Ethics Across SIGCHI

Submitted to: ACM Conference on Designing Interactive Systems

Motivated by the need for a structured overview of applied knowledge and practical approaches for navigating complex moral dilemmas, this paper contributes a scoping review of ethics as discussed within full-length SIGCHI papers containing the search term 'ethic\*' in their title, abstract or authors' keywords from 2010 to 2020. Findings show Value-Sensitive and Critical Design appearing as the most frequently applied orientations, and participatory approaches were found to be more prevalent than those without end-user input. This structured account of SIGCHI's engagement with ethics topics served as the foundation for the subsequent work of the thesis requiring an understanding of possible pathways to including critical ethical perspectives within user-centred design processes for health data interaction design.

## **Futures for Health Research Data Platforms From the Participants' Perspectives** Published in: *ACM Nordic Conference on Human Factors in Computing Systems*

This paper provides an in-depth understanding of health data repositories by reviewing a comprehensive set of existing platforms and then discussing factors impacting willingness to share data with health data platforms. Following this engagement with the literature, findings from a qualitative interview study with 12 young adults living in Denmark bring a contributor-centred perspective to supporting design suggestions for future developments. Based on the evidence gathered, this paper then proposes a conceptual framework structuring critical factors for acceptance (motivation, willingness to share and trust) across the different stages of the data contribution journey (recruitment, data collection and continuous participation).

## Public Attitudes to Digital Health Research Repositories: Cross-sectional International Survey

Published in: Journal of Internet Research

This paper expands and solidifies the qualitative insights of the prior interview study by gathering the perspective of 1600 individuals living in Denmark and Brazil through a quantitative approach (online survey). Data collected included participants' demographics, level of interest in health topics, past participation in health research studies, awareness of research data repositories, and current attitudes about the idea of shared-access repositories. Data analysis examined the relationships between these variables and participants' willingness to share different data types, as well as participants' preferred options for data control, principal reasons for concern, and strongest motivations to contribute. Findings contribute towards future developments with the aim of addressing and engaging with the concerns of those unwilling to participate, and engaging different population segments through means of contribution tailored to their preferences.

## Sharing Access to Behavioural and Personal Health Data: Designers' Perspectives on Opportunities and Barriers

## Published in: EAI International Conference on Pervasive Computing Technologies for Healthcare

This paper describes the results of a focus group study complementing the investigations of data contributors' perspectives by providing insights into the work practices and perspectives of those developing, designing and creating digital health technologies. By analysing the qualitative data collected during this focus group, findings reveal technology creators' inputs regarding enablers and barriers for a broader acceptance of systems that allow data to be shared and re-shared for clinical care and research. When combined with findings from the studies conducted with data contributors, this study confirms a need for health data systems to more explicitly support the values of transparency and choice.

## A Value-Sensitive Approach for Ethical Health Research Platforms

Presented at: ACM Conference on Computer-Supported Cooperative Work and Social Computing - Workshop: Beyond Checklist Approaches to Ethics in Design

This paper applies a Value-sensitive Design approach to designing a prototype for a mobile app to be used as a data collection interface for a shared-access platform for clinical support and health research. The resulting prototype demonstrates how the values of agency and reciprocity can be embedded in user interface features and system functionalities, including through continuous data control settings, personal feedback, clinical support and research study updates. The description of this experience with the valuesensitive design framework illustrates how it is possible to adopt an ethical lens to the process of creating user interfaces to be used by health data contributors.

# Is the Automation of Digital Mental Health Ethical? Applying an Ethical Framework to Chatbots for Cognitive Behaviour Therapy

Published in: Frontiers in Digital Health - Special Issue on Responsible Digital Health

This paper critically analyses how sensitive data from therapeutic disclosures can be ethically collected, stored and shared in the context of chatbots designed to automate psychotherapy. Possible harmful and positive aspects concerning this technological innovation were raised by applying an AI-ethics framework focused on five principles: (1) beneficence, (2) non-maleficence, (3) autonomy, (4) justice, and (5) explicability. The analysis complements the investigations carried out previously, demonstrating how applying ethical frameworks can offer pointers to possible improvements in future developments.

**Teaching User Experience Design Ethics to Engineering Students: Lessons Learned** Submitted to: *Frontiers in Computer Science - Special Issue on Teaching and Learning Human-Computer Interaction* 

This paper reports a pedagogical approach to support learning and teaching of User Experience Design ethics applied to the context of digital health. Through a qualitative investigation with engineering students enrolled in an introductory design course, this paper contributes insights concerning how future technology creators apply ethical principles in the task of designing a heart monitoring system. Results shed light on the benefits and limitations of teaching ethics this way and led to discussion and reflection on the challenges and possible paths forward for teaching and putting ethics into practice.

### What Is the Future of Data Sharing for Research?

Published in: ACM interactions

This argumentative essay draws a parallel between the acceptance of contact tracing apps for pandemic control and the acceptance of large-scale health data repositories. This discussion had the goal of underlining the severe implications of both blind acceptance and a lack of acceptance for society, and how for this reason, there is a need to be respectful of the needs of people contributing their personal data for the collective good. In particular, this essay questions the justification for building repositories of aggregated data that might mean better healthcare in the future but may also consist of a tool for mass control. Instead of seeking broad acceptance through coercive propaganda and imposed terms, the essay argues in favour of a power shift that would place citizens at the centre of decisions made about their data — as comprises the vision behind this thesis.

## 2 Papers Included

## 2.1 A Scoping Review of Ethics Across SIGCHI

## Authors:

Giovanna Nunes Vilaza, Kevin Doherty, Darragh McCashin, David Coyle, Jakob Bardram and Marguerite Barry

## Submitted to:

ACM SIGCHI Conference on Designing Interactive Systems (DIS)

### A Scoping Review of Ethics Across SIGCHI

GIOVANNA NUNES VILAZA, Technical University of Denmark, Denmark KEVIN DOHERTY, Technical University of Denmark, Denmark DARRAGH MCCASHIN, Dublin City University, Ireland DAVID COYLE, University College Dublin, Ireland JAKOB BARDRAM, Technical University of Denmark, Denmark MARGUERITE BARRY, University College Dublin, Ireland

Ethical deliberation has proved a consistent feature of Human-Computer Interaction (HCI) since its earliest years, spanning the ethical involvement of research participants to design choices impacting fairness, freedom and welfare. Despite growing discussions, applied knowledge and practical approaches for navigating complex moral dilemmas remain challenging to grasp. Motivated by the need for a structured overview, this paper contributes a scoping review of ethics as discussed within full-length SIGCHI papers containing the search term 'ethic\*' in their title, abstract or authors' keywords over the last ten years. Findings show increasing prioritisation of the topic, particularly within Artificial Intelligence. Value-Sensitive and Critical Design appear as the most frequently applied orientations, and participatory approaches are more prevalent than those without end-user input. Engaging with a spectrum from personal to societal concerns, the SIGCHI literature thus echos calls for critical perspectives on user-centred processes and the need to establish more sustainable responsibility structures.

CCS Concepts: • Social and professional topics  $\rightarrow$  Codes of ethics; • Human-centered computing  $\rightarrow$  Interaction design process and methods; • General and reference  $\rightarrow$  Surveys and overviews.

Additional Key Words and Phrases: Ethics; Design Research; Scoping Review.

#### **ACM Reference Format:**

Giovanna Nunes Vilaza, Kevin Doherty, Darragh McCashin, David Coyle, Jakob Bardram, and Marguerite Barry. 2022. A Scoping Review of Ethics Across SIGCHI. In *DIS '22: ACM Conference on Designing Interactive Systems, June 13–17, 2022, Virtual.* ACM, New York, NY, USA, 26 pages.

### 1 Introduction

As digital technologies have expanded their capacity to affect human lives at ever-increasing scales and in ever more intimate ways, the accompanying ethical dilemmas have grown only more prevalent: algorithmic bias, political interference, disinformation and hate online, the misuse of personal data and deceptive interfaces [40, 122]. Researchers, practitioners and professionals working with technology design are increasingly required to have an open and informed engagement with the concept of ethics, as they face complex questions of individual and collective values, power-knowledge asymmetries, and the legal and societal consequences of their choices [13, 52]. Many argue that integrating critical reflection into the daily work of design and research can serve as a productive approach to addressing such issues of ethics [16, 67, 147].

Manuscript submitted to ACM

Permission to make digital or hard copies of all or part of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. Copyrights for components of this work owned by others than ACM must be honored. Abstracting with credit is permitted. To copy otherwise, or republish, to post on servers or to redistribute to lists, requires prior specific permission and/or a fee. Request permissions from permissions@acm.org.

© 2022 Association for Computing Machinery.

Significant efforts have promoted the discussion of ethics within academia as means to support ethical work. Conference panels and workshops on the topic have maintained a regular presence since the very first HCI conferences [116], and ethics remains highlighted as one of the most important concerns of HCI today [155]. The ACM Code of Professional Conduct and the IEEE Code of ethics reflect this broad recognition of the prominent role of ethics through formal sources of guidance, advocating, for instance, that computing professionals should avoid causing harm and ensure that the public good is a central goal [71, 86].

Furthermore, a few authors conducted systematic reviews within specific application contexts, examining, for instance, how different ethical theories have been used to program autonomous machines [164]. Other authors have systematically characterised which ethical issues and arguments have been associated with brain-computer interfaces [26], robots for aged care [173] and assistive technologies for dementia [87]. Van et al. have revised mentions of the keyword 'ethics' across the ACM Interaction Design and Children Conference (IDC) [172]. In contrast to these reviews focused on particular technology domains, a much broader systematic review from 2016 has previously analysed Computer Science journals in order to extract insights into how computing ethics has been discussed by this particular section of the literature over the years [154].

While providing essential guidance to practitioners' and researchers', these past efforts in the form of systematic reviews have gaps. First, the search query used in the broader review of computing ethics from 2016 has not captured papers published in important HCI venues, such as CHI and DIS [154]. As HCI is at the centre of much ethical reflection concerning technology's design, use, and implication, this is a critical gap to address. Second, the other past reviews have specifically focused only on certain technology domains (for example, machine learning), which means they do not provide a comprehensive overview of ethical engagement across computing technologies as a whole. A systematic review of ethics across multiple HCI venues, including a diverse set of technology domains, could not only fulfil these gaps but, most importantly, provide general and informed advice on possible routes to action and ways to put ethics into practice.

Motivated by the identified research gaps and the pressing need to support individuals working towards more robust engagement with ethical technology development and design, we conducted a scoping review. For this review, we targeted full-length papers published at conferences sponsored by the ACM Special Interest Group on Computer-Human Interaction (SIGCHI) containing the search term 'ethics\*' in their title, abstract, or authors' keyword list. To be clear, instead of aiming to capture all papers within HCI, we sought to coherently, systematically and transparently identify a representative sample of papers discussing ethics in detail as a core focus of their HCI work.

Therefore, the main goal for this review is to provide a rigorous account of the landscape of ethics discussion across SIGCHI in order to support technology creators, designers and researchers looking to fortify orientations towards ethical thinking and doing. We carry out this work from the position of HCI, Software Engineering and Information Science Studies researchers, ranging from all levels of seniority, seeking to learn possible pathways for conceiving technologies that are not only user-friendly but ethically informed. Hence, the following research questions drive the scoping review:

- (1) Which ethical considerations, technologies and user groups have been the subjects of ethical engagements?
- (2) Which approaches, methods and theories have been employed to explore ethical considerations?
- (3) What practical recommendations for putting ethics into action have been reported and discussed?

This paper contributes a systematic and coherent overview of 'ethics' as provided by papers published at SIGCHIsponsored conferences. Next, we describe and motivate our choice for a scoping literature review methodology. Then, we present the findings gathered through the identified sample of papers, followed by a discussion of the implications and limitation of results, as well as directions for future work.

### 2 Methodology

Seeking to provide a comprehensive account of HCI publications that could support professionals in their ethical practice, we identified and analysed relevant literature sources about ethics through a scoping review methodology. A *scoping review* is a rigorous literature review procedure that produces a systematic description of previous research about a topic, refraining from judging the quality or weight of evidence provided by individual papers. It is an effective method for identifying knowledge gaps and clarifying concepts across a body of literature and an ideal methodology to initially appraise the field (in contrast to traditional reviews where evidence is critically evaluated) [120].This review follows the established scoping review procedure of Arksey, and O'Malley [4], and reports the review procedure and its results using best-practice items recommended by the latest version of the Preferred Reporting Items for Systematic Reviews and Meta-Analyses extension for Scoping Reviews<sup>1</sup> (PRISMA-ScR, 2018) [167].

#### 2.1 Eligibility Criteria

We searched the ACM Digital Library for papers published from January 1st 2010 to December 31st 2020. Using the ACM Digital Library search filter 'sponsored by' SIGCHI (Special Interest Group on Computer-Human Interaction), we then searched for publications containing the keyword 'ethic\*' within the title, abstract or authors' keywords. Employing the wild card \* means that any word starting with the letters 'ethic' was included in our search results (e.g. ethical, ethics or ethically). Only peer-reviewed, full-length archival publications were eligible for inclusion: a standard literature reviewing practice to ensure the analysis of mature work. These eligibility criteria were driven by the need to produce a rigorous overview of ethics within a transparently delimited, reproducible and manageable scope. See Section 4.4 for more detailed reflections on our rationale.

#### 2.2 Information Sources

The ACM Digital Library was our exclusive information source. We did not contact other authors to identify additional sources. Our most recent search of the ACM Digital Library was conducted on January 27th 2021.

#### 2.3 Search Query

The full query syntax used for the ACM Digital Library search was: "query": Title:(ethic\*) OR Abstract:(ethic\*) OR Keyword:(ethic\*) "filter": Sponsor: sigchi,Publication Date: (01/01/2010 TO 12/31/2020),ACM Content: DL.

#### 2.4 Selection of Sources of Evidence

The initial sample produced by the query search was screened to ensure the sole inclusion of peer-reviewed, full-length archival publications. Then, based on the main research questions presented in Section 1, authors agreed upon a concise set of exclusion criteria. Papers that, despite mentioning ethics in title, abstract or authors' keywords, did not engage with the term in the main text would be removed from the sample. Papers using the term ethics to refer to a body of values governing a particular culture or group but not directly addressing technology ethics (e.g., hunting ethics) would also be removed, as would papers only very briefly mentioning the word ethics in order to justify their chosen research

<sup>&</sup>lt;sup>1</sup>The PRISMA-ScR [167] is a standardised guideline for the description of scoping review procedures based on the framework outlined by Arksey and O'Malley [4].

approach without reflecting on ethical considerations to any significant extent within the main text. These criteria were applied to all sources of evidence initially identified, leading to the removal of papers fulfilling any of the above conditions.

#### 2.5 Data Charting Process

The first and second authors discussed and agreed upon a set of predefined data items to be extracted from each paper, based on the overarching research questions listed in Section 1. The first author then charted the data items from the complete set of papers using a spreadsheet, and the third author independently charted 10% of papers in a separate spreadsheet to validate the charting process. There were no substantial disagreements at this stage. Then, the second author independently reviewed the data-charting spreadsheet and iterated upon the results with the first author in order to resolve any disagreements.

#### 2.6 Data Items

Driven by the research questions listed in Section 1, the following predefined data items (variables) were extracted from each paper and organised using a spreadsheet:

- characteristics of sources of evidence:
  - publication year (e.g., 2020)
  - publication venue (e.g., DIS)
- subject matters of ethical considerations:
  - ethical concerns (e.g., privacy invasions)
  - technology types (e.g., public displays)
  - groups of end-users (e.g., children)
- approaches to ethics in design and research:
  - empirical approaches (e.g., focus group)
  - theoretical approaches (e.g., Value-Sensitive Design)
- sample recommendations for putting ethics into action

#### 2.7 Synthesis of Results

For handling and summarising the charted data, we employed a table and two data visualisations to illustrate the distribution of sources of evidence (papers) by publication year and venue. We similarly produced more tables to show the distribution technology types, end-user groups and empirical methods for approaching ethics present within the sample. Next to these tabulated results, a descriptive summary in text-form explained how the results related to the review questions. Findings from other data charting items were reported only in descriptive narrative form.

Results from charting the data items of ethical concerns, technology types, end-user groups and sample recommendations were also classified according to key conceptual categories, as recommended when collating, summarising and reporting scoping review results [4]. These categories were produced by conceptually grouping related findings through an iterative and inductive approach, without employing any particular taxonomies as guides. For example, three broad groups were inductively formed based on conceptual affinity (ethical participation in research, ethics as design choice, and ethics as taking responsibility), and within each group, the recommendations of each paper (or sometimes more than one paper) were depicted in narrative form. After classifying results according to these distinct conceptual groups, relationships between ethical concerns and different technology types and end-user groups were identified and summarised in descriptive text-form.

### 3 Findings

Following PRISMA-ScR [167] for the reporting of scoping review results, we first present the process of selecting sources of evidence. We then describe the characteristics of these sources of evidence in terms of publication venue and year. And finally, we present the synthesis of the results organised by research question.

#### 3.1 Selection of Sources of Evidence

The search query employed resulted in an initial sample of 461 papers. The meta-data of this set were then screened to ensure the sole inclusion of peer-reviewed, full-length archival publications. This process led to the removal of 314 manuscripts comprising late-breaking works, workshop calls, works in progress, tutorials, Special Interest Group meetings, keynotes, case studies, courses, doctoral consortia, interactivity publications, research landscapes and awards. Pictorials were not removed as they are considered archived, full-length publications at the annual Designing Interactive Systems (DIS) conference.

The remaining 147 full papers were analysed in relation to the exclusion criteria defined in the previous section (See section 2.4). This resulted in the removal of the following papers: papers that do not engage with the term in the main text (n=5, [76, 78, 111, 148, 187]); papers that use the term ethics only to refer to the values of a group (e.g., the ethics of hunting) rather than the ethics of digital technology research and design (n=4, [62, 101, 157, 165]); and papers that mention the word ethics only very briefly to justify research conduct without further engaging with the topic (n=9, [6, 31, 41, 91, 103, 123, 135, 143, 186]). This process resulted in a final sample of 129 papers relevant to our research questions and constituting the complete source of evidence (See Figure 1).

#### 3.2 Characteristics of Sources of Evidence

As described above, the final review sample includes 129 full-length papers published at SIGCHI-sponsored venues (See Table 1). The majority of papers were published in the last three years: 2018 (17%, n=22), 2019 (20.9%, n=27) and 2020 (28.6%, n=37). The years between 2010 and 2017, in contrast, account for only 33.3% of the sample (n=43). The majority of papers within the sample were published at CHI (59.6%, n=77), with the second-largest group of papers published at DIS (22.1%, n=17)  $^2$ . Figure 2 illustrates the distribution of papers across publication venues and years.

#### 3.3 Subject Matters of Ethical Considerations

Our first research questions was devised to identify the subject matters of ethical considerations concerning specific ethical concerns, technology types and groups of people across this sample in order to discern predominant and under-explored areas of focus. Several ethical concerns were identified, and we classified these concerns in broad groups that even though do not represent a formal taxonomy, provide a valuable depiction of recurrent themes. Examples of autonomy violations include privacy invasions and lack of data sharing choices [1–3, 7, 9, 20, 23, 24, 27, 32, 38, 39, 60, 64, 70, 84, 95, 98, 102, 104, 113, 114, 118, 121, 124, 128, 130, 137, 138, 144, 145, 150, 151, 160, 163, 166, 170, 172, 176, 177, 179, 181, 183, 184]. Hidden persuasion, often found in 'dark patterns' (malicious user interface

<sup>&</sup>lt;sup>2</sup>The complete list of SIGCHI-sponsored conferences is available at https://sigchi.org/conferences/

Vilaza et al.



Fig. 1. Selection of sources of evidence (flow diagram)

Year	Instances	Venue and References
2010	0	-
2011	3	CHI [8, 109, 131]
2012	4	CHI [17, 66, 118], DIS [39]
2013	8	CHI [21, 77, 88, 113, 117, 176], ECCE [42], PerDis [98]
2014	9	CHI [53, 77, 79, 180], DIS [38, 158], CSCW [59, 108], CHI PLAY [174]
2015	2	CHI [121], HRI [107]
2016	12	CHI [24, 89, 114, 171, 179, 181], DIS [48, 100], CSCW [177], HRI [33, 85, 142]
2017	5	CHI [12, 75], CSCW [23], HRI [5], MobileHCI [129]
2018	22	CHI [2, 56, 60, 70, 74, 84, 99, 105, 110, 126, 150, 151, 159, 163, 175], DIS [182], HAI [139, 178], RecSys [55], AutomativeUI [112], MobileHCI [104], TVX [137]
2019	27	CHI [1, 10, 14, 29, 32, 34, 36, 72, 83, 96, 115, 130, 134, 136, 138, 160, 185], DIS [9, 64, 184], HRI [90, 92, 95, 97, 140], CHI PLAY [153], CHIIR [18]
2020	37	CHI [3, 7, 30, 35, 44, 57, 63, 81, 94, 106, 119, 124, 125, 128, 142, 145, 146, 156, 169], DIS [20, 73, 82, 141, 144, 152, 166], HRI [47, 183], IDC [27, 46, 93, 170, 172], CHIIR [102, 149], VRST [132], HT [61]

Table 1. Publication year, venue and reference number for each source of evidence included within the sample.

A Scoping Review of Ethics Across SIGCHI



Fig. 2. Sample distribution according to the publication venue (left) and year (right).

patterns), can also be considered disrespectful to autonomy, as these are intentionally employed to engage users in performing actions against their best interests and without their awareness [3, 5, 8, 23, 29, 30, 34–36, 44, 55, 72– 74, 85, 90, 95, 97, 100, 129, 131, 137, 139, 140, 146, 170–172, 183]. Harm to integrity and well-being can be identified through evidences of physical and emotional distress [3, 10, 12, 14, 17, 21, 30, 39, 47, 53, 63, 64, 66, 81–84, 88, 92, 93, 95, 96, 99, 100, 104, 105, 108, 109, 117, 121, 131, 134, 138, 141, 142, 149, 151–153, 166, 169, 170, 172, 174, 176, 177, 180–182, 184]. Potential fatal consequences are yet another way individuals can be physically harmed, and their safety is at risk (e.g., accidents with autonomous vehicles) [2, 5, 33, 90, 94, 107, 112, 115, 119, 125, 178, 183]). Issues related to social injustice encompass the negative consequences of exclusion, misrepresentation, stigma, bias and oppression [3, 9, 10, 18, 20, 32, 48, 55, 56, 64, 75, 79, 88, 89, 99, 105, 106, 108, 110, 125, 126, 132, 134, 136, 138, 142, 144, 151, 152, 156, 159, 160, 168, 169, 175– 178, 182, 184]. Similarly, instances of unethical monetisation practices, such as the selling of data outside its original context are also related to social (un)fairness [7, 27, 34, 35, 44, 46, 59–61, 72–74, 136, 137, 145, 150, 153, 158, 172, 174, 185]. Most papers explicitly mention and name at least one ethical concern, even though a few papers provide more general discussions that could apply to any type of ethical issue [11, 42, 57, 77].

The first research question also sought to identify which technologies and groups of people have been the targets of ethical considerations. Regarding technology types, this sample includes papers discussing ethics in relation to artificial intelligence (AI) applications , mobile and web applications, social networks and forums, the internet of things (e.g., smart spaces), public installations and displays, wearables and biosensors (e.g., fashion technologies), games, data repositories and broadcast media. AI systems are the most discussed technology type across the sample, with a considerable number of papers focusing on embodied agents (e.g., social robots), although other AI applications (e.g., recommender systems), are also present (See Table 2). Several papers in the sample do not focus on any technology in particular [1, 11, 14, 18, 24, 48, 63, 66, 77, 79, 81, 83, 93, 99, 114, 117, 121, 149, 159, 163, 172, 176, 177]. Very few papers discuss more than one technology type [64, 96, 151].

Papers tend to focus less frequently on specific end-user groups than on technology types. However, this sample did include papers about children/teenagers, independent workers, patients or individuals with health conditions, minorities, women, older adults, rural populations, researchers and non-human end-users (e.g., birds) (See Table 3). The most frequently discussed group of users across this evidence base are children and teenagers, followed by workers. Several papers also targeted intersecting groups such as women with impaired hearing [182] and older adults with dementia [63, 82]. Aside from these papers which discuss ethical considerations in relation to end-users of digital

Technology Types	Instances	References	
Artificial Intelligence (AI)	32	social robots [5, 27, 30, 33, 47, 85, 90, 97, 100, 107, 139, 140, 142, 168, 170, 178, 181, 183], autonomous vehicles [112, 115, 151, 185], conversational agents [146, 156], recommender systems [55, 185], facial recognition [184], in general [9, 20, 42, 106, 125]	
Mobile and web applications	23	personal health [12, 39, 131, 134, 138, 145, 180], research [23, 113, 118, 129, 152], services [29, 34, 35, 44, 72–74, 105, 136, 151, 171]	
Social networks and forums	12	crowd-sourcing [10, 88, 89, 108, 110], social media [3, 32, 56, 60, 104, 150, 151]	
Public installations and displays	8	ambient displays [8, 130], installations [17, 29, 53, 94, 98, 179]	
Internet of Things (IoT)	7	smart spaces [64, 128, 144, 158], smart TVs [137], educational tools [95], in general [70]	
Wearables and biosensors	6	biosensing [38, 64, 84], fashion technologies [166, 182], body integration [119]	
Games	6	playful narratives [92], virtual reality [132], intentionally idle [153], with cultural references [174], with sexual content [124], multiplayer [96]	
Broadcast media	6	video streaming platforms [46, 59, 61, 149], documentaries [75], news media [96]	
Data repositories	5	historical archives [102], DNA repositories [7], learning analyt- ics [160], visualisation [36], in general [151]	
Others	5	shape-changing interfaces [2], end-of-life technologies [21, 109], drones [57, 151]	

Table 2. Technology types as the target of ethics discussions.

technology products, a further group of papers discuss ethics in relation to research participants and researchers [1, 24, 81, 81, 83, 117, 121, 176, 177], UX professionals [34, 35, 72–74], and students learning about technology design [20, 46, 63, 136, 145].

By examining our results, we were able to identify some relationships between ethical concerns and technology types or groups of end-users. Adverse psychological effects and potential harms to individual well-being are in particular surfaced in relation to studies involving women [12, 64, 182], children [27, 47, 104, 170, 174] and the elderly [30, 39, 100, 181]. Papers discussing these concerns are likewise most often focused on technologies for personal health [12, 63, 82, 138, 166] and games [153]. Safety risks, accidents and the possibility of severe physical hazards of technology use are most often associated with social robots and autonomous vehicle interactions [5, 33, 112, 142, 183].

Concerns regarding privacy invasion and security breaches are most often encountered in contexts involving data collection and sharing, such as mobile and wearable sensing [23, 38, 113, 118, 166], Internet of Things (IoT) devices [70, 95, 128, 137, 144], shared-access data repositories [7, 102, 160] and social media networks [3, 104]. Questions of privacy have also been raised however in relation to interactions taking place in public space, as in the case of interactive installations [130, 179]. Many papers also discuss concerns around persuasion and deception in connection to services that aim at increasing profit margins by manipulating end-users' actions [34, 44, 73, 74]. In addition, emotional manipulation is frequently discussed in papers concerning autonomous agents such as social robots [85, 90, 97, 100, 139,

A Scoping Review of Ethics Across SIGCHI

End-Users	Instances	References	
Children and teenagers	14	in general [27, 46, 47, 61, 66, 93, 95, 104, 114, 126, 170, 172, 174, 180]	
Workers	10	Amazon Mechanical Turk workers [10, 88, 89, 92, 108], musi- cians [59, 158], Uber drivers [105] and home health aides [169], shipping industry [185].	
Patients/with health conditions	9	neurological illnesses [14, 39, 63, 82], mental health diagnosis [134, 141, 152], substance addiction [32], cancer [180]	
Minorities	5	gender and sexual [156], people with specific communication [99, 182] and learning needs [18, 149]	
Women	4	pregnant [12], with impaired hearing [182], in general [64, 159]	
Older adults	4	[1, 30, 63, 82, 100, 181]	
Rural populations	3	in Namibia [132], South Africa [21] and Bangladesh [159]	
Non-human	1	birds [94]	

Table 3. End-users that have raised design questions and considerations of ethics.

140] and chatbots [146]. More rarely, ill-intended persuasion is connected with data visualisations [36] and potentially unethical practices of nudging behaviour [29].

When it comes to broader societal issues, concerns around injustice, unfairness, exclusion, and bias feature most often in relation to AI [55, 106, 156, 168, 175, 178, 185]. Questions of justice are also often raised in studies involving gig workers (e.g., Amazon Mechanical Turk workers) [10, 88, 92, 105, 108, 169]. Ethnographic studies involving rural and indigenous populations also tend to focus on related topics including the digital divide, power asymmetries and digital imperialism [79, 89, 132, 159]. Issues of social stigma and discrimination are similarly often raised in studies involving individuals with mental illnesses [134, 141, 152] and sexual minorities [156].

#### 3.4 Approaches to Ethics in Design Research

The second research question relates to how ethical considerations have been approached, explored and engaged with across the sample. Most papers report empirical studies devised to raise and address ethical issues, often employing user research methods to consult stakeholders directly, although at times also employing design methods without stakeholders' presence such as tools for team discussion. In addition to (or instead of) such empirical explorations however, other papers employ philosophical and theoretical frameworks to inform and support their understanding of ethics. And these diverse approaches to ethics employed across the sample we describe next.

*3.4.1 Empirical Approaches* Many papers in the sample report on empirical studies conducted to surface ethical considerations (See Table 4). These studies frequently employ user research methods well-known to the HCI community (interviews, questionnaires, focus groups, observations, and co-design workshops), conducted both in the lab and in the field. Participants in these empirical studies include previous or current end-users [3, 7, 10, 59, 88, 92, 110, 114, 118, 144], potential or future end-users [14, 21, 27, 30, 39, 63, 66, 75, 84, 90, 95, 99, 100, 104, 125, 126, 129, 132, 134, 136, 139, 140, 142, 149, 151, 152, 158, 159, 163, 169–171, 174, 178, 180–182, 184], close contacts to end-users [170, 180], technology creators and researchers [2, 9, 33–35, 72, 106, 117, 119, 137, 141, 153, 175, 177, 185], and multiple stakeholders at

once [23, 38, 53, 160]. A small number of papers report on experiments and surveys conducted not with end-users but with paid, crowd-sourced participant samples (e.g., Amazon Mechanical Turk) [90, 107, 139, 146, 168].

The majority of papers report findings from participatory and co-design workshops, followed closely by field and interview studies which are also featured frequently (See Table 4). A small number of field studies additionally propose 'activist systems' designed to elevate real-world ethical concerns while striving to realise change. In particular, this sample features examples of a platform for citizen-led governance experiments [110], an immersive theatrical installation designed to raise ethical awareness among the audience [150] and systems targeting fairer conditions for workers crowd-sourced via marketplaces such as Amazon's Mechanical Turk [10, 88, 92]. In contrast, labs and online experiments are less frequently reported and this sample features a small number of surveys (online questionnaires).

Distinct from empirical approaches entailing the direct involvement of stakeholder at various degrees, some papers promote the use of 'expert-only methods'. Empathy-building tools, personas, stakeholder matrices and card-based materials have all been used to facilitate self-reflection and collaboration between teams and students [9, 20, 46, 61, 106, 141, 145, 185]. Critical appraisals of design features and speculative scenarios have also been reported [5, 35, 39, 44, 47, 55, 56, 64, 74, 85, 97, 113, 124, 128, 129, 131, 152, 156, 166]. This sample also features content analyses of online forum discussions [60, 73, 105, 108] and presentations of manifestos [70], but these are rare.

Across the sample and diverse methodologies reported, speculation has often been used as a material or stimulus to surface, extrapolate from, and raise awareness of how technologies can become imbued, even invisibly, with values, ideologies and behavioural norms. In particular, speculation has been employed in critical conceptual designs [5, 56, 64, 128, 131, 156, 166, 183], participatory workshops [27, 95, 152, 169, 185] and user inquiry methods [168, 171], often in the form of fictional scenarios and provocative prototypes.

Lastly, several papers take stock of the literature and past projects to review and discuss ethical issues. This includes examples of argumentative essays [11, 18, 24, 48, 77, 81, 109, 115, 176], analyses of case studies, historical and contemporary examples [12, 17, 36, 83, 89, 96, 98, 102, 121] and literature reviews of specific subjects and technology domains [1, 29, 79, 93, 130, 138, 172]. Several of these papers define, propose and argue for theoretical orientations, as described in greater detail in the next section.

*3.4.2 Theoretical Approaches* Although less frequently encountered than papers presenting empirical methods and findings, various papers contained within this sample also present theories, philosophies and design orientations as means to structure discussions or lend more conceptual framings to empirical explorations. These 'theoretical approaches' include engagements with moral philosophy (e.g., Utilitarian and Virtue Ethics), social and political philosophies (e.g., Critical Theory and Social Justice), theories of design (e.g., Value Sensitive Design) and social theories (e.g., Biopolitics). Papers engaging with theory in one form or another make up approximately one third of the sample.

The framework most frequently employed across the sample was Value-Sensitive Design (VSD) [9, 33, 34, 39, 42, 46, 53, 56, 72, 74, 83, 93, 105], which can be considered a theory of design and a formative design framework [80]. The VSD approach seeks to account for 'human values' in a principled and systematic manner throughout the design process [69]. Following this approach, human values are conceptualised as an agreed-upon vocabulary defining that which is important to each stakeholder group (e.g., security, transparency, accountability). In our sample, studies employ VSD to identify and engage with human needs, elicit system requirements and prioritise features, in relation to conflicting stakeholder perspectives in particular.

Another theoretical perspective well-represented in this sample is Critical Theory, often taking the form of Critical Design, Design Fictions and Speculative Design [11, 38, 56, 64, 128, 131, 150, 168, 185]. Critical theory is an approach to

A Scoping Review of Ethics Across SIGCHI

Methods	Instances	References
Workshops & Focus Groups	28	[2, 9, 14, 20, 27, 46, 61, 63, 66, 75, 95, 99, 100, 104, 106, 119, 126, 136, 141, 145, 149, 152, 169, 170, 174, 180, 182, 185]
Field Studies	25	[10, 21, 30, 34, 35, 38, 39, 63, 72, 84, 88, 92, 94, 104, 110, 118, 129, 132, 134, 140, 150, 158, 163, 181, 184]
Critical Prototypes & Expert Evaluations	20	[5, 35, 39, 44, 47, 55, 56, 64, 74, 85, 97, 113, 124, 128, 129, 131, 152, 156, 166, 183]
Interview Studies	19	[3, 7, 23, 33, 39, 53, 59, 82, 114, 117, 134, 137, 144, 153, 159, 160, 171, 175, 180]
Experiments	11	[8, 42, 90, 107, 112, 125, 139, 146, 149, 171, 178]
Argumentative Essays	11	[11, 18, 24, 48, 57, 77, 81, 109, 115, 176, 179]
Case Studies	9	[12, 17, 36, 83, 89, 96, 98, 102, 121]
Surveys	8	[7, 93, 114, 142, 144, 151, 168, 177]
Literature Reviews	7	[1, 29, 79, 93, 130, 138, 172]
Content Analysis	6	[32, 60, 70, 73, 105, 108]

Table 4. Papers which have used specific methods to identify, investigate and engage with ethical considerations.

social philosophy that challenges assumptions and conceptions about power relations in society [11]. Critical design brings this perspective to bear on the role product design plays in social and cultural spheres, often to foster debate and increase ethical awareness in lieu of an often-prioritised commercial focus [89]. Moral and technical imaginations are central to such a critical stance, which explains the frequent association of critical theory with design materialisations through speculative concepts and design fiction.

A perspective encountered less frequently across our sample is Care Ethics [83, 136, 145, 153], a philosophical outlook that shifts the moral focus towards embodied, situated and emergent relationships of mutual care [43], and as such a stark contrast to ethical philosophies centred on principles, norms and duties, such as deontology. When applied to design processes, care ethics with its relational and responsive approach foregrounds empathy and reciprocal commitments to each other as fundamental for decision-making [136, 145].

Meanwhile, utilitarian ethical frameworks are also found less frequently [107, 115, 125, 178]. Utilitarianism is a form of consequentialism which holds that an optimal ethical choice produces the greatest good for the greatest number of people by following a sort of 'moral arithmetic' in order to calculate the benefits of each outcome [22]. This philosophical perspective is most often encountered in this sample in the context of studies involving AI as a means to inform the programming of autonomous agents [107, 125, 178]. Another paper however notes the common criticism of consequentialism as unhelpful for decision making, given the difficulty (and perhaps impossibility) of weighing all positive and negative outcomes [115].

Other possible theoretical orientations, however, appear even more rarely within this sample. A few papers explicitly engage with Social Justice theory [48, 79, 169], describing Social Justice-Oriented Interaction Design as a framework that can help designers commit to plurality, advocacy, and political considerations [48]. Similarly, Queer and Feminist theories twice appear in regard to issues of inequality in the design of technology for minority groups [124, 156]. Virtue Ethics, which focuses on the cultivation of ethical wisdom through experience, appears in a single paper concerning mobile

technology design to support mental health, resulting in a call for greater scrutiny of the subjective assumptions and hierarchies of expertise underlying design choices [12]. And, Somatic Ethics is explored in a single paper depicting the experience of dancing with drones and positioning ethics as choreographing movements to be experienced somatically (through the body) [57].

Biopolitics is employed in a single instance as an analytical lens to drive reflection on issues of authority in biosensing designs [84]. Similarly, Foucault's theory of 'care of the self' is used to emphasise technology's epistemic dangers as a source of power [96]. And, Thanato-sensitive design is introduced, describing an approach to integrating the inevitability of mortality so that user-centred design can be more sensitive to those in bereavement and decrease harm in that sense [109]. Finally, amidst a predominant emphasis on Western-centric ethical frameworks across the sample, Confucian Ethics is the sole non-Western perspective featured, in a single paper exploring individual roles as the source of ethical obligations and character growth in relation to the design of morally competent artificial agents [183]. Duty-based ethical frameworks (Kantian deontology) are not explicitly applied in any papers within this sample, although studies describing normative rules devised to guide actions could be interpreted as examples of this perspective, such as calls for 'categorised ethical guidelines' [113] and 'checklists' [95].

#### 3.5 Recommendations for Putting Ethics into Action

Our final research question pertains to the compilation of practical recommendations for putting ethics into action. As described in section 2.7, the iterative and inductive analysis of the recommendations identified across the sample resulted in three conceptual groups. The main topics in these groups relate to research participation, design choices and responsibility claims - as described next.

*3.5.1 Ethics as Participation in Research* This sample offers several practical recommendations for the ethical involvement of people in design research. Some of these suggestions already comprise minimum research ethics standards in some, although not all, jurisdictions [54]. Several papers provide recommendations in relation to ethical procedures for obtaining informed consent from participants prior to data collection, advising that consent forms are written in accessible language [113, 149], participants granted enough time to become familiar with the technology before a study [181] and that consent should be revisited throughout a project [1, 24, 63, 81, 114, 126, 163, 177]. When it comes to making use of publicly available data for research purposes, studies advise reducing the risk of re-identification by reporting results at a low level of detail [1, 32, 113, 177], and if data is shared outside of initial agreements, to inform participants [23, 113]. In the case of public installations, others recommend allowing participants to withdraw at any time [17, 98, 179].

Many papers also advise researchers to prioritise the benefits of participation by, for example, acknowledging participants as co-creators [114] or authors [82], negotiating compensation in line with participants' preferences [83], gifting participants working prototypes [30, 63, 82, 136] or celebrating the conclusion of a project with an event [82]. Authors furthermore advise inviting participants to engage more directly in the configuration of research projects from the very start [14, 75, 83, 99, 110, 176]. Several papers also emphasise the need to identify means to foster, ethically, participant diversity and to become genuinely involved with the communities we study, by, for example, volunteering with organisations [18, 63, 82], yet without falling for the 'design saviour complex' which can ultimately worsen uneven power relations with end-users [89].

On a related note, several papers highlight that participating in and conducting research can prove an emotionally charged experience, and propose appropriately preparing researchers to minimise possible sources of emotional discomfort to participants and promptly recognise signs of participant distress during the research [63, 81, 99, 109, 149]. Similarly, some papers promote support for researchers' own well-being through counselling, group discussions, and a healthy work-life balance [63, 82, 117, 163]. Authors likewise advise incorporating domain experts and health professionals into teams researching sensitive contexts [66, 81, 109, 121]. A final and inclusive recommendation provided by authors in support of ethical HCI research is to articulate decisions made in support of well-being during published works as means of promoting awareness [63, 117].

*3.5.2 Ethics as Design Choice* Many papers across the sample frame design choices as means to address ethical concerns, even though they also acknowledge that it is challenging to make detailed recommendations, as each specific context requires an approach tailored to its characteristics. A large number argue that user autonomy can be increased through more transparent communication of both system intentions and technical limitations for example [7, 17, 29, 30, 36, 60, 70, 79, 113, 131, 146]. Several also suggest that privacy concerns can be addressed by providing mechanisms for data control and the collection of less intrusive forms of data [7, 30, 36, 60, 118, 124, 128, 130, 134, 138, 160].

A considerable group of papers draw attention to the potential adverse consequences of technology for users' well-being. Authors in turn advise that designers avoid reinforcing trends towards the over-optimisation of behaviour [3, 29, 36, 84, 131, 134, 138] and consult experts if unfamiliar with how the particularities of a health domain can negatively impact users [14, 21, 53, 66, 94, 109, 138]. When it comes to safety risks, authors advise the careful consideration of the how to mitigate the possibility of physical hazards [112, 115].

Another, slightly smaller, group of papers argue that design can and should serve as a means of accounting for and addressing social justice concerns by involving minorities in the design process and becoming more aware of the role designed products can play in perpetuating systems of oppression [48, 138, 156, 159, 168]. Other papers recommend and deploy 'activist systems' as means to tackle complex social challenges [10, 88]. As many on the other hand highlight however, it is essential to recognise that concepts such as fairness are not easily translated into design features, and authors argue that the complex relationships between technologies and their surrounding socio-economic and political contexts can limit what could be possible to address by means of user interface design [106, 115, 175].

3.5.3 Ethics as Taking Responsibility Many papers additionally encourage professionals to adopt measures against unethical practices by remaining mindful of intentions and reflecting upon the consequences of their choices [36, 56, 77, 81, 96, 98, 118]. Researchers within academic settings have likewise been advised to accept responsibility for ethical conduct beyond what is required and anticipated by institutional ethics procedures [121]. In contrast, however, a small group of papers warn that an individual's capacity to engage in ethical decision-making can be severely constrained by tensions with business goals or roles within enterprise hierarchies [35, 72]. For this reason, authors recommend that more emphasis is placed on the creation of strategies to enhance individuals' capacity to act [72].

Education is often endorsed across the sample as a means of fostering ethical awareness among future generations of HCI and UX professionals [34–36, 72, 74]. Approaches recommended in this regard include the cultivation of an 'ethics of care' among students through the organisation of open-ended projects with small civic organisations [136], collaborative classroom critique of data collection and visualisation practices following first-hand experience [145], and the use of card-based tools to help students reflect on moral dilemmas and build new prototypes in response [20]. Across this subset of papers, authors frequently suggest that classroom activities should be designed to enhance students' awareness of the societal consequences of technology development, and that their effectiveness can be observed by evidence of an increased sensitivity to ethical issues on students' behalf.

A small number of papers also suggest raising end-users' awareness of the risks and social consequences entailed in technology adoption [44]. Attempts to educate the public have been realised in creative ways including an immersive theatre experience designed to engage the public in discussions concerning personal data misuse by third parties [150], a provocative demo employing a bio-metric mirror to entice reactions from passersby [184], and the invitation of school students to discuss the values embedded in a well-known video streaming platform [46]. These papers suggest that strategies devised to invite end-users to reflect on ethical concerns should strive to prove appealing and, if deemed suitable, also provocative, in order to capture end-users' attention and leave behind an impactful, take-home message.

Relatively few papers in contrast invoke responsibility from the higher ranks of organisations or the public sector, by calling, for example, for more robust regulation as means to shape future technological developments or suggesting concrete policy implications [2, 3, 119, 146, 151]. Several papers do criticise current regulations for failing to attend to ethical issues, arguing however that overarching rules are hard to apply if not customised to particular contexts [61, 106]. A single exception praises policies derived from the Responsible Research and Innovation (RRI) agenda [77]. Even though very few papers actively advocate for and discuss ways to increase institutional and national responsibilities beyond the individual, this sample also does not explicitly suggest that higher-level entities are exempt from their share of responsibility.

### 4 Discussion

This review was motivated by the increasingly urgent need to support researchers, practitioners and professionals working with technology design and user research. By assimilating knowledge of just how to put ethics 'into practice', results shows a plurality of ways of thinking about, approaching and relating to the topic. This review thus demonstrates the potential for ethical thinking to drive, shape and crystallise responsible design and research. This final section revisits the research questions, reflects on the review's findings and discusses how they relate to literature beyond the sample. These reflections further illuminate future directions to commit to the topic more meaningfully, strengthen explorations of ethical requirements, and create more harmonic structures for exercising responsibility.

#### 4.1 Establishing Priorities and Commitments

Our first research question sought to identify the ethical concerns discussed by published research within SIGCHI. Examining these results, it became evident that SIGCHI is concerned with issues ranging from the very personal to the societal; from the emotional work involved in the creation of digital technologies [81] to the consequences of algorithmic bias in aggravating social oppression [48]. This finding is well-aligned with a previous review of ethics in Computer Science journals [154] that also identified similar and diverse ethical discussions across its sample. Increased positioning of social justice as pertinent to technology ethics is likewise much in agreement with the contemporary literature on the politics of artefacts [40], design justice [37] and data feminism [45]. This finding should serve as an encouragement to designers and researchers engaging more actively in contemplation of how technology artefacts can shape the most personal aspects of our experience in connection to historical, cultural and geopolitical contexts - a commitment also brought forth by others within the broader HCI community [19].

The first research question also inquired into which technology types have proved the focus of publications about ethics at SIGCHI – findings revealing the distinctive prevalence of discussions pertaining to artificial intelligence (AI). The capacity of AI to 'think', persuade and influence decisions, sometimes in unfair and unpredictable ways, makes these systems particularly prone to moral dilemmas. The definition of ethical requirements for the AI domain is rendered additionally complex by the general lack of transparency in regard to algorithmic results and performance metrics [25]. Prior research makes the case that strategies to improve AI ethics demand much more than technical work [15, 175]. The debate about AI ethics extends beyond this sample and HCI, as researchers question just how much power digital systems should have over civic life and if damaging outcomes can ever be avoided [28, 58]. Such reflections only reinforce that critical thinking within this domain is essential to informing the definition and placement of ethical boundaries for autonomous systems [161].

Furthermore, we have examined the groups of people involved in the participatory studies, revealing a surprising gap. Despite demonstrating concern for issues of justice and fairness, the sample seldom features the direct participation of social and ethnic minorities [132]; much research instead engages children, patients, and older adults. Even though this gap might not reflect the HCI literature more broadly [162], it represents an opportunity to emphasise that by not prioritising an ethical engagement with those potentially marginalised, design and research fail to attend to the particularities of groups who are misunderstood, misrepresented and express cultural difference [176]. Discussions on diversity and inclusion are among the most important conversations happening in HCI as of today, and despite challenges in devising and conducting ethical engagements with potentially marginalised populations, it is crucial to elevate this as a high priority in research agendas.

Such gathering of ethical matters at SIGCHI makes evident that technology design should account for users' vulnerabilities and intersections, just as much as it already accounts for the craft of 'appealing' interfaces [149, 181]. Ethical concerns surfaced by the sample illustrate the capacity of technologies of all sorts to disrespect human needs for autonomy, happiness and dignity – in clear contrast with visions of HCI as motivated by the idea of 'sustaining human flourishing' [127]. Ethics is a growing trend at SIGCHI, yet for ethics to become more accepted as a measure of success, we, designers and researchers, might need to reconsider standard assumptions of what 'good' technology design means. The broad range of ethical concerns, technology types and groups identified as subject matters for ethical consideration only confirms that the goal of creating 'useful tools for productive work' and 'delightful experiences' should not to trap user-centred design in a narrow definition of scope that potentially devalues the importance of ethics [49].

#### 4.2 Adding Structure to Explorations of Ethics

The second research question led to the investigation of exactly how ethical considerations have been approached, both theoretically and empirically, at SIGCHI. Our analysis indicates that it is possible to put ethics into practice by adopting methods familiar to the HCI community (e.g., interviews, field studies, lab experiments) and whose procedures, in principle, do not differ substantially from their traditional employment [133]. For instance, across this sample, user studies are often used to critically assess the potential negative impact of technologies, often through qualitative methods to understand subjective, contextual and intricate experiences. Quantitative methods, including surveys and lab experiments, also have their place, providing a somewhat more objective argument for the prevalence of ethical issues. Other approaches have skirted the direct involvement of end-users by yielding tools to facilitate team discussion and proving their value as approaches to conduct a critical appraisal of empirical and secondary data sources to support more informed decisions.

Although there exist examples of papers identifying ethical issues as by-products of their work [2], studies contained within this sample typically engage in research and design processes with the mindset that ethical concerns are in and of themselves primary study objects. This observation constitutes an essential point for individuals seeking to make ethics a more integral aspect of their work: ethics is best enacted if considered a core component of the research design process, and a guiding viewpoint, from the very beginning. The vast spectrum of empirical approaches provides, collectively, a portfolio of various enablers of reflexivity - diverse methods for adoption at stages throughout the design

life-cycle - in contrast with a more transitory, checkbox approach [106], positioning ethical analysis as the core of the design.

In addition to highlighting the possible adoption of well-established user research methods, findings also demonstrate the significance of the so-called 'speculative approaches' to envisioning preferable futures and safely surfacing unethical outcomes [51]. Papers contained within the sample often connect speculation to critical design to foreground the intentions of technology creators, illuminate potentially hidden ideologies, and propose under-explored design visions [11, 64, 131, 168]. Beyond speculating about the future, another possibility is to consider evaluating ethical interventions 'in-the-wild'. This sample offered a few examples of field studies deploying activities and systems designed to drive change in the real world [88], which are a valuable source of inspiration for a type of activism that HCI could strive for [65].

This review surfaces examples of theories, philosophies, and design orientations argued as proper structures for directing engagement with ethics. Different views, of course, conceptualise ethics in considerably different ways. Rather than arguing which yields the definitive perspective on complex moral challenges, learning from this pluralism of perspectives is most likely the most fruitful approach, permitting the deliberation of diverse positions without necessarily falling into a complete relativism. A utilitarian stance, for instance, takes consequences as the dominant aspect to be discussed concerning autonomous agents [107, 125, 178]. Yet, it has been argued that there is value in considering alternative theories to confront automation dilemmas and search for their causes, not only the possible outcomes [115].

Regarding theoretical pluralism, this review also reveals a critical gap previously highlighted as a significant bias in HCI's engagement with ethics [188]. The vast majority of the sample references 'Western-centric' literature, theories, and philosophical orientations, indicating the lack of theoretical diversity across the sample, a finding also remarked by previous AI-ethics reviews [164]. Future work should strive to learn more from other worldviews (e.g., collectivism [183]) as a way to expand theoretical possibilities.

Review findings provide sufficient grounds to argue that it is crucial to continue expanding theoretical knowledge to support the specific ethical challenges of technology design and research. The theoretical and methodological framework most frequently employed across this SIGCHI sample is Value-Sensitive Design, possibly a consequence of its origins in HCI [68]. However, a few papers also introduce theories from other fields, such as Biopolitics [84], which demonstrate how direct engagement with theoretical foundations are advantageous for ethics in practice. Therefore, these findings show that both developing novel theories with HCI roots and building stronger interdisciplinary connections with other fields (e.g., social sciences, philosophy, gender studies) can be highly beneficial to support ethical reasoning applied to the context of digital technologies.

Despite the presence of such references, most of the sample does little to engage with theory actively (See Section 3.4.2), a gap also reported in a previous review of ethics in Computer Science more broadly [154]. While our scope is too narrow to draw conclusions about the field of HCI as a whole, when discussions of ethics are carried out without formal definitions and rely only on 'intuitive' understanding of the everyday use of the term, an opportunity is missed to gain value from and add to established sources of knowledge that could function as reference points for more robust analyses. Rather than making ethics a rigid, static and formalised concept, a theoretical foundation can be game-changing for more robust argumentation and decision-making [67].

#### 4.3 Harmonising Roles and Responsibilities

This review has gathered a considerable collection of recommendations for putting ethics into practice. For instance, this SIGCHI sample offers many valuable insights into how to involve stakeholders in research processes and how product design might tackle ethical concerns. Nevertheless, these findings also shed light on the challenges of allocating roles and responsibilities for putting ethics into action.

The recommendations featured across this sample most often imply that the responsibility for ethical decision-making lies with research and design professionals [121]. More rarely featured, although also present, are reflections concerning means of raising users' capacity to question the ethics of the technologies they use and ultimately determine whether to adopt them or not [60]. However, this sample appears to place less emphasis on the need for stronger regulations, external evaluation, and legal accountability as means to protect and guide individuals [113]. For example, institutional ethics committees - which outline standard procedures for the planning and conduct of research - have faced criticism for providing little support when ethical issues do arise in practice [16].

Although there are valid arguments for advocating for individual responsibility - at least as a practice of consciousnessraising [96] - it is not yet fully clear to what extent this is possible given the constraints, demands and formalities of the contexts in which individuals are situated [72]. Despite the original intention of this paper to better equip individuals with this structured literature overview, the limited capacity of individuals to act combined with the complexity entailed in making ethical decisions points to the need to turn to the power of collectives. The formation of communities of practice and a professional culture that encourages critical thinking will probably be vital to supporting the creation of technologies attuned to fairer futures [89, 110, 136]. The influence of sources of funding for research and development should also be more emphatically discussed, as they serve an essential role in enabling individuals to work towards more meaningful and ethical goals [50, 176].

Therefore, there seems to be a significant role for tools and design processes that facilitate 'ethics mediation' [72] between professionals, teams, organisations, and ultimately end-users. Researchers and designers need to be empowered to express the contingencies of ethical decision-making more effectively, especially within the UX industry and in publication venues. Pursuing and promoting an ethics agenda also requires leadership. Further discussions beyond the SIGCHI-community regarding individual, collective and institutional roles should be encouraged in order to gradually establish a framework for organising responsibilities and expectations across entities [52].

#### 4.4 Limitations and Reflections

Despite being motivated by the desire to produce a manageable body of knowledge that would also be sufficient to start answering questions that demand a systematic investigation, this review has limitations. We have deliberately avoided over-generalising findings throughout the text as our contribution is limited to the pre-defined goals and scope. We here reflect and comment on the methodological choices made to acknowledge and discuss the boundaries of our research scope, especially when compared to the vast field of HCI.

Our review focuses only on SIGCHI-sponsored conferences, which means that findings reflect only this specific segment of the broader HCI literature. We justify our choice first by emphasising that our goal was not to provide a definitive account of ethical discourse in HCI but gather a sample of approaches and possible pathways for understanding and applying ethics to technology design research. Future systematic reviews could encompass other HCI venues and different literature sources to add to, and perhaps critically contrast with, the insights extracted through this review of SIGCHI.

Another limitation is that this review includes only archival peer-reviewed full papers. This approach has allowed for a more cohesive analysis by ensuring that each paper included in the sample had a similar structure and underwent a similar submission and acceptance process. However, this choice meant that insights from maturing sources of knowledge such as late-breaking works and workshops were not contemplated. For this reason, we emphasise here that any gaps and trends identified in our sample cannot be interpreted as gaps of SIGCHI papers. Future works could consider reviewing these other papers to understand better how ethics discussions are distributed across specific publication types. The review is also limited by the search strategy focused on title, abstract or keywords. This decision was made for pragmatic reasons to avoid including a large number of papers irrelevant to the research question, such as those that mention only "approval by an ethics committee". However, this choice meant that those papers that happened to not mention the word 'ethics' in these query fields were not identified in our search. Future research conducting a quantitative analysis of how many papers without ethics in their title, abstract and keywords extensively engage with the topic could lead to an insightful discovery.

Finally, a difficult choice in the design of this review strategy concerned whether to include other keywords related to ethics, such as 'justice', 'values' or 'rights' in the search query. Upon careful consideration and discussion, we concluded that any particular set of terms about ethics is at risk of failing to prove exhaustive. We also could not devise unbiased objective criteria to justify the inclusion of specific terms over others, and we deemed it inappropriate to claim a paper is 'about ethics' if the authors chose not to use the keyword themselves. We encourage future works to build upon this work and expand the search criteria to gather even more sources of knowledge to guide ethical practice and better characterise the ethics-related landscape in the diverse discipline of HCI.

#### 5 Conclusion

Intending to gain a deeper understanding of how a fundamental philosophical concept applies to the diverse field of HCI, this paper contributes a rigorous scoping review of 'ethics' as approached by the SIGCHI literature. Findings highlight a growing interest in the topic through recent years, a trend that only reinforces the importance of ethics for technology design and research. As SIGCHI-sponsored conferences are prominent venues guiding HCI research worldwide, the sources of evidence examined in this paper pave the way for future ethical engagements.

This review reveals a plurality of perspectives and approaches comprising diverse possible paths for recognising and preventing ethical concerns. Learning from this pluralism can allow more robust debates and well-informed argumentation. Similarly, the diverse catalogue of identified methods brings about a vision for ethically-centred design processes in which critical thinking connects user inquiries to design explorations. The opportunity now arises to adopt and develop theoretical and methodological perspectives that support such an applied and reflexive type of ethics. As means to transform insight into positive real-life change, beyond method and theory, it is also essential to consider novel approaches to increase the harmony between individual, collective and institutional responsibilities.

As this sample clearly shows, every unique configuration of context, technology and users has its own requirements and challenges, but ethical thinking often converges in topics of human autonomy, beneficence, welfare and justice. Attending to such core ethical principles will prove beneficial in the conception of technologies more attuned to recurrently prevalent human values. As seen in the review, understanding broader socio-political contexts is just as important as engaging with individual particularities so that ethics can protect those who might be at risk and establish fairer relationships of power.

We commend the pervasive and consistent consideration of ethics evident across this sample. At the same time, we aspire that this work intensifies critical engagement with ethics across the broader HCI community. This sample

offers a vast collection of recommendations regarding opportunities to engage with the often 'elusive' concept of ethics, which in aggregate may serve as a valuable resource for a diverse audience of readers, from students and early-stage researchers to seasoned academics and organisational leaders. We hope to see many more future works consulting and building upon these references aggregated through this review.

As a final reflection, some may wonder why we, as researchers, part of the SIGCHI community, should dare to care about ethics in the first place. It can be challenging to cultivate ethical awareness within and across individuals in a knowledge field historically driven by fast and bold technical innovation. Committing to ethical thinking means leaving behind a position of indifference for a path undeniably challenging and rarely permissive of definitive answers. We may find motivation and inspiration, however, in recognising that this is perhaps the only means of pursuing and achieving the human flourishing for which HCI has striven all along. Inhabiting privileged positions as producers and propagators of knowledge, we have a shared responsibility to shape our actions in the world and society. Our choices should serve not only our own goals and desires but the common good, in solidarity with the aspirations and needs of those who comprise our surroundings and, indeed, our planet.

#### Acknowledgments

#### References

- Jacob Abbott, Haley MacLeod, Novia Nurain, Gustave Ekobe, and Sameer Patil. 2019. Local standards for anonymization practices in health, wellness, accessibility, and aging research at CHI. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–14.
- [2] Jason Alexander, Anne Roudaut, Jürgen Steimle, Kasper Hornbæk, Miguel Bruns Alonso, Sean Follmer, and Timothy Merritt. 2018. Grand challenges in shape-changing interface research. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–14.
- [3] Nazanin Andalibi and Justin Buss. 2020. The Human in Emotion Recognition on Social Media: Attitudes, Outcomes, Risks. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–16.
- [4] Hilary Arksey and Lisa O'Malley. 2005. Scoping studies: towards a methodological framework. International journal of social research methodology 8, 1 (2005), 19–32.
- [5] Thomas Arnold and Matthias Scheutz. 2017. Beyond moral dilemmas: exploring the ethical landscape in HRI. In 2017 12th ACM/IEEE International Conference on Human-Robot Interaction (HRI. IEEE, 445–452.
- [6] Thomas Arnold and Matthias Scheutz. 2018. Observing robot touch in context: How does touch and attitude affect perceptions of a robot's social qualities?. In Proceedings of the 2018 ACM/IEEE International Conference on Human-Robot Interaction. 352–360.
- [7] Khadija Baig, Reham Mohamed, Anna-Lena Theus, and Sonia Chiasson. 2020. "I'm hoping they're an ethical company that won't do anything that I'll regret" Users Perceptions of At-home DNA Testing Companies. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–13.
- [8] Madeline Balaam, Geraldine Fitzpatrick, Judith Good, and Eric Harris. 2011. Enhancing interactional synchrony with an ambient display. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. 867–876.
- [9] Stephanie Ballard, Karen M Chappell, and Kristen Kennedy. 2019. Judgment call the game: Using value sensitive design and design fiction to surface ethical concerns related to technology. In Proceedings of the 2019 on Designing Interactive Systems Conference. 421–433.
- [10] Natã M Barbosa and Monchu Chen. 2019. Rehumanized crowdsourcing: a labeling framework addressing bias and ethics in machine learning. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–12.
- [11] Jeffrey Bardzell and Shaowen Bardzell. 2013. What is" critical" about critical design?. In Proceedings of the SIGCHI conference on human factors in computing systems. 3297–3306.
- [12] Marguerite Barry, Kevin Doherty, Jose Marcano Belisario, Josip Car, Cecily Morrison, and Gavin Doherty. 2017. mHealth for maternal mental health: everyday wisdom in ethical design. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems. 2708–2756.
- [13] Marguerite Barry, Aphra Kerr, and Oliver Smith. 2020. Ethics on the ground: from principles to practice. In Proceedings of the 2020 Conference on Fairness, Accountability, and Transparency. 688–688.
- [14] Jeanette Bell and Tuck Wah Leong. 2019. Collaborative futures: Co-designing research methods for younger people living with dementia. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–13.
- [15] Emily M Bender, Timnit Gebru, Angelina McMillan-Major, and Shmargaret Shmitchell. 2021. On the Dangers of Stochastic Parrots: Can Language Models Be Too Big?. In Proceedings of the 2021 ACM Conference on Fairness, Accountability, and Transparency. 610–623.
- [16] Steve Benford, Chris Greenhalgh, Bob Anderson, Rachel Jacobs, Mike Golembewski, Marina Jirotka, Bernd Carsten Stahl, Job Timmermans, Gabriella Giannachi, Matt Adams, et al. 2015. The ethical implications of HCI's turn to the cultural. ACM Transactions on Computer-Human

Interaction (TOCHI) 22, 5 (2015), 1-37.

- [17] Steve Benford, Chris Greenhalgh, Gabriella Giannachi, Brendan Walker, Joe Marshall, and Tom Rodden. 2012. Uncomfortable interactions. In Proceedings of the sigchi conference on human factors in computing systems. 2005–2014.
- [18] Gerd Berget and Andrew MacFarlane. 2019. Experimental methods in IIR: The tension between rigour and ethics in studies involving users with dyslexia. In Proceedings of the 2019 Conference on Human Information Interaction and Retrieval. 93–101.
- [19] Karthik S Bhat and Neha Kumar. 2020. Sociocultural Dimensions of Tracking Health and Taking Care. Proceedings of the ACM on Human-Computer Interaction 4, CSCW2 (2020), 1–24.
- [20] Karl-Emil Kjær Bilstrup, Magnus H Kaspersen, and Marianne Graves Petersen. 2020. Staging Reflections on Ethical Dilemmas in Machine Learning: A Card-Based Design Workshop for High School Students. In Proceedings of the 2020 ACM Designing Interactive Systems Conference. 1211–1222.
- [21] Jon Bird, Peter Byass, Kathleen Kahn, Paul Mee, and Edward Fottrell. 2013. A matter of life and death: practical and ethical constraints in the development of a mobile verbal autopsy tool. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. 1489–1498.
- [22] Craig Boutilier, Ioannis Caragiannis, Simi Haber, Tyler Lu, Ariel D Procaccia, and Or Sheffet. 2015. Optimal social choice functions: A utilitarian view. Artificial Intelligence 227 (2015), 190–213.
- [23] Anne Bowser, Katie Shilton, Jenny Preece, and Elizabeth Warrick. 2017. Accounting for privacy in citizen science: Ethical research in a context of openness. In Proceedings of the 2017 ACM Conference on Computer Supported Cooperative Work and Social Computing. 2124–2136.
- [24] Barry Brown, Alexandra Weilenmann, Donald McMillan, and Airi Lampinen. 2016. Five provocations for ethical HCI research. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. 852–863.
- [25] Joy Buolamwini and Timnit Gebru. 2018. Gender shades: Intersectional accuracy disparities in commercial gender classification. In Conference on fairness, accountability and transparency. PMLR, 77–91.
- [26] Sasha Burwell, Matthew Sample, and Eric Racine. 2017. Ethical aspects of brain computer interfaces: a scoping review. BMC medical ethics 18, 1 (2017), 1–11.
- [27] Bengisu Cagiltay, Hui-Ru Ho, Joseph E Michaelis, and Bilge Mutlu. 2020. Investigating family perceptions and design preferences for an in-home robot. In Proceedings of the Interaction Design and Children Conference. 229–242.
- [28] Rafael A Calvo, Dorian Peters, Karina Vold, and Richard M Ryan. 2020. Supporting human autonomy in AI systems: A framework for ethical enquiry. In *Ethics of Digital Well-Being*. Springer, 31–54.
- [29] Ana Caraban, Evangelos Karapanos, Daniel Gonçalves, and Pedro Campos. 2019. 23 ways to nudge: A review of technology-mediated nudging in human-computer interaction. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–15.
- [30] Felix Carros, Johanna Meurer, Diana Löffler, David Unbehaun, Sarah Matthies, Inga Koch, Rainer Wieching, Dave Randall, Marc Hassenzahl, and Volker Wulf. 2020. Exploring Human-Robot Interaction with the Elderly: Results from a Ten-Week Case Study in a Care Home. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–12.
- [31] Stevie Chancellor, Zhiyuan Lin, Erica L Goodman, Stephanie Zerwas, and Munmun De Choudhury. 2016. Quantifying and predicting mental illness severity in online pro-eating disorder communities. In Proceedings of the 19th ACM conference on computer-supported cooperative work & social computing. 1171–1184.
- [32] Stevie Chancellor, George Nitzburg, Andrea Hu, Francisco Zampieri, and Munmun De Choudhury. 2019. Discovering alternative treatments for opioid use recovery using social media. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–15.
- [33] EunJeong Cheon and Norman Makoto Su. 2016. Integrating roboticist values into a Value Sensitive Design framework for humanoid robots. In 2016 11th ACM/IEEE International Conference on Human-Robot Interaction (HRI). IEEE, 375–382.
- [34] Shruthi Sai Chivukula, Colin M Gray, and Jason A Brier. 2019. Analyzing value discovery in design decisions through ethicography. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–12.
- [35] Shruthi Sai Chivukula, Chris Rhys Watkins, Rhea Manocha, Jingle Chen, and Colin M Gray. 2020. Dimensions of UX Practice that Shape Ethical Awareness. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–13.
- [36] Michael Correll. 2019. Ethical dimensions of visualization research. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–13.
- [37] Sasha Costanza-Chock. 2020. Design justice: Community-led practices to build the worlds we need. MIT Press.
- [38] Franco Curmi, Maria Angela Ferrario, and Jon Whittle. 2014. Sharing real-time biometric data across social networks: Requirements for research experiments. In Proceedings of the 2014 conference on Designing interactive systems. 657–666.
- [39] Yngve Dahl and Kristine Holbø. 2012. Value biases of sensor-based assistive technology: case study of a GPS tracking system used in dementia care. In Proceedings of the designing interactive systems conference. 572–581.
- [40] Jenny L Davis. 2020. How Artifacts Afford: The Power and Politics of Everyday Things. MIT Press.
- [41] Munmun De Choudhury, Emre Kiciman, Mark Dredze, Glen Coppersmith, and Mrinal Kumar. 2016. Discovering shifts to suicidal ideation from mental health content in social media. In Proceedings of the 2016 CHI conference on human factors in computing systems. 2098–2110.
- [42] Tjerk de Greef, Arvind Mohabir, Ibo van der Poel, and Mark Neerincx. 2013. sCEthics: embedding ethical values in cognitive engineering. In proceedings of the 31st European conference on cognitive ergonomics. 1–8.
- [43] Maria Puig de La Bellacasa. 2017. Matters of care: Speculative ethics in more than human worlds. Vol. 41. U of Minnesota Press.
- [44] Linda Di Geronimo, Larissa Braz, Enrico Fregnan, Fabio Palomba, and Alberto Bacchelli. 2020. UI dark patterns and where to find them: a study on mobile applications and user perception. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–14.
#### A Scoping Review of Ethics Across SIGCHI

[45] Catherine D'Ignazio and Lauren F Klein. 2020. Data feminism. MIT Press.

- [46] Daniella DiPaola, Blakeley H Payne, and Cynthia Breazeal. 2020. Decoding design agendas: an ethical design activity for middle school students. In Proceedings of the Interaction Design and Children Conference. 1–10.
- [47] Anna Dobrosovestnova and Glenda Hannibal. 2020. Teachers' Disappointment: Theoretical Perspective on the Inclusion of Ambivalent Emotions in Human-Robot Interactions in Education. In Proceedings of the 2020 ACM/IEEE International Conference on Human-Robot Interaction. 471–480.
- [48] Lynn Dombrowski, Ellie Harmon, and Sarah Fox. 2016. Social justice-oriented interaction design: Outlining key design strategies and commitments. In Proceedings of the 2016 ACM Conference on Designing Interactive Systems. 656–671.
- [49] Paul Dourish. 2019. User experience as legitimacy trap. Interactions 26, 6 (2019), 46-49.
- [50] Andy Dow, Rob Comber, and John Vines. 2019. Communities to the left of me, bureaucrats to the right... here I am, stuck in the middle. *interactions* 26, 5 (2019), 26–33.
- [51] Anthony Dunne and Fiona Raby. 2013. Speculative everything: design, fiction, and social dreaming. MIT press.
- [52] Abigail Durrant and David Kirk. 2018. On Ethical Responsiveness: Being Answerable to Others as an HCI Researcher. Interacting with Computers 30, 2 (2018), 99–115.
- [53] Abigail C Durrant, David S Kirk, and Stuart Reeves. 2014. Human values in curating a human rights media archive. In Proceedings of the SIGCHI Conference on Human factors in Computing Systems. 2685–2694.
- [54] Economic and Social Research Council. 2019. Research ethics. Retrieved Sep 15, 2020 from https://esrc.ukri.org/funding/guidance-for-applicants/ research-ethics
- [55] Michael D Ekstrand, Mucun Tian, Mohammed R Imran Kazi, Hoda Mehrpouyan, and Daniel Kluver. 2018. Exploring author gender in book rating and recommendation. In Proceedings of the 12th ACM conference on recommender systems. 242–250.
- [56] Enrique Encinas, Mark Blythe, Shaun Lawson, John Vines, Jayne Wallace, and Pam Briggs. 2018. Making Problems in Design Research: The Case of Teen Shoplifters on Tumblr. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–12.
- [57] Sara Eriksson, Kristina Höök, Richard Shusterman, Dag Svanes, Carl Unander-Scharin, and Åsa Unander-Scharin. 2020. Ethics in Movement: Shaping and Being Shaped in Human-Drone Interaction. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–14.
- [58] Amitai Etzioni and Oren Etzioni. 2017. Incorporating ethics into artificial intelligence. *The Journal of Ethics* 21, 4 (2017), 403–418.
  [59] Casey Fiesler and Amy S Bruckman. 2014. Remixers' understandings of fair use online. In *Proceedings of the 17th ACM conference on Computer*
- supported cooperative work & social computing. 1023–1032.
- [60] Casey Fiesler and Blake Hallinan. 2018. "We Are the Product" Public Reactions to Online Data Sharing and Privacy Controversies in the Media. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–13.
- [61] Flavio Figueiredo, Felipe Giori, Guilherme Soares, Mariana Arantes, Jussara M Almeida, and Fabricio Benevenuto. 2020. Understanding Targeted Video-Ads in Children's Content. In Proceedings of the 31st ACM Conference on Hypertext and Social Media. 151–160.
- [62] Martin Flintham, Richard Hyde, Paul Tennent, Jan-Hinrik Meyer-Sahling, and Stuart Moran. 2020. Now Wash Your Hands: Understanding Food Legislation Compliance in a Virtual Reality Restaurant Kitchen. In Proceedings of the Annual Symposium on Computer-Human Interaction in Play. 169–180.
- [63] Sarah Foley, Nadia Pantidi, and John McCarthy. 2020. Student Engagement in Sensitive Design Contexts: A Case Study in Dementia Care. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–13.
- [64] Sarah Fox, Noura Howell, Richmond Wong, and Franchesca Spektor. 2019. Vivewell: Speculating near-future menstrual tracking through current data practices. In Proceedings of the 2019 on Designing Interactive Systems Conference. 541–552.
- [65] Sarah Fox, Catherine Lim, Tad Hirsch, and Daniela K Rosner. 2020. Accounting for Design Activism: On the Positionality and Politics of Designerly Intervention. Design Issues 36, 1 (2020), 5–18.
- [66] Christopher Frauenberger, Judith Good, Wendy Keay-Bright, and Helen Pain. 2012. Interpreting input from children: a designerly approach. In Proceedings of the SIGCHI conference on human factors in computing systems. 2377–2386.
- [67] Christopher Frauenberger, Marjo Rauhala, and Geraldine Fitzpatrick. 2017. In-action ethics. Interacting with Computers 29, 2 (2017), 220-236.
- [68] Batya Friedman. 1996. Value-sensitive design. interactions 3, 6 (1996), 16–23.
- [69] Batya Friedman and David G Hendry. 2019. Value sensitive design: Shaping technology with moral imagination. Mit Press.
- [70] Ester Fritsch, Irina Shklovski, and Rachel Douglas-Jones. 2018. Calling for a revolution: An analysis of IoT manifestos. In Proceedings of the 2018 CHI conference on human factors in computing systems. 1–13.
- [71] DW Gotterbarn, Bo Brinkman, Catherine Flick, Michael S Kirkpatrick, Keith Miller, Kate Vazansky, and Marty J Wolf. 2018. ACM code of ethics and professional conduct. (2018).
- [72] Colin M Gray and Shruthi Sai Chivukula. 2019. Ethical mediation in UX practice. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–11.
- [73] Colin M Gray, Shruthi Sai Chivukula, and Ahreum Lee. 2020. What Kind of Work Do" Asshole Designers" Create? Describing Properties of Ethical Concern on Reddit. In Proceedings of the 2020 ACM Designing Interactive Systems Conference. 61–73.
- [74] Colin M Gray, Yubo Kou, Bryan Battles, Joseph Hoggatt, and Austin L Toombs. 2018. The dark (patterns) side of UX design. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–14.
- [75] David Philip Green, Simon Bowen, Jonathan Hook, and Peter Wright. 2017. Enabling Polyvocality in Interactive Documentaries through" Structural Participation". In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems. 6317–6329.

#### DIS '22, June 13-17, 2022, Virtual

- [76] Catherine Grevet and Eric Gilbert. 2015. Piggyback prototyping: Using existing, large-scale social computing systems to prototype new ones. In Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems. 4047–4056.
- [77] Barbara Grimpe, Mark Hartswood, and Marina Jirotka. 2014. Towards a closer dialogue between policy and practice: responsible design in HCI. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. 2965–2974.
- [78] Jennefer Hart, Ioanna Iacovides, Anne Adams, Manuel Oliveira, and Maria Margoudi. 2017. Understanding engagement within the context of a safety critical game. In Proceedings of the Annual Symposium on Computer-Human Interaction in Play. 253–264.
- [79] John Harvey, David Golightly, and Andrew Smith. 2014. HCI as a means to prosociality in the economy. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. 2955–2964.
- [80] David G Hendry, Batya Friedman, and Stephanie Ballard. 2021. Value sensitive design as a formative framework. Ethics and Information Technology 23, 1 (2021), 39–44.
- [81] Tad Hirsch. 2020. Practicing Without a License: Design Research as Psychotherapy. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–11.
- [82] James Hodge, Sarah Foley, Rens Brankaert, Gail Kenning, Amanda Lazar, Jennifer Boger, and Kellie Morrissey. 2020. Relational, Flexible, Everyday: Learning from Ethics in Dementia Research. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–16.
- [83] Dorothy Howard and Lilly Irani. 2019. Ways of Knowing When Research Subjects Care. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–16.
- [84] Noura Howell, Laura Devendorf, Tomás Alfonso Vega Gálvez, Rundong Tian, and Kimiko Ryokai. 2018. Tensions of data-driven reflection: A case study of real-time emotional biosensing. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–13.
- [85] Andreas Huber, Astrid Weiss, and Marjo Rauhala. 2016. The ethical risk of attachment how to identify, investigate and predict potential ethical risks in the development of social companion robots. In 2016 11th ACM/IEEE International Conference on Human-Robot Interaction (HRI). IEEE, 367–374.
- [86] IEEE. 2020. IEEE Code of Ethics. Retrieved Sep 15, 2020 from https://www.ieee.org/about/corporate/governance/p7-8.html
- [87] Marcello Ienca, Tenzin Wangmo, Fabrice Jotterand, Reto W Kressig, and Bernice Elger. 2018. Ethical design of intelligent assistive technologies for dementia: a descriptive review. Science and engineering ethics 24, 4 (2018), 1035–1055.
- [88] Lilly C Irani and M Six Silberman. 2013. Turkopticon: Interrupting worker invisibility in amazon mechanical turk. In Proceedings of the SIGCHI conference on human factors in computing systems. 611–620.
- [89] Lilly C Irani and M Six Silberman. 2016. Stories We Tell About Labor: Turkopticon and the Trouble with" Design". In Proceedings of the 2016 CHI conference on human factors in computing systems. 4573–4586.
- [90] Ryan Blake Jackson and Tom Williams. 2019. Language-capable robots may inadvertently weaken human moral norms. In 2019 14th ACM/IEEE International Conference on Human-Robot Interaction (HRI). IEEE, 401–410.
- [91] Rikke Hagensby Jensen, Yolande Strengers, Jesper Kjeldskov, Larissa Nicholls, and Mikael B Skov. 2018. Designing the desirable smart home: A study of household experiences and energy consumption impacts. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–14.
- [92] Anna Kasunic, Chun-Wei Chiang, Geoff Kaufman, and Saiph Savage. 2019. Turker Tales: Integrating Tangential Play into Crowd Work. In Proceedings of the 2019 on Designing Interactive Systems Conference. 21–34.
- [93] Saba Kawas, Ye Yuan, Akeiylah DeWitt, Qiao Jin, Susanne Kirchner, Abigail Bilger, Ethan Grantham, Julie A Kientz, Andrea Tartaro, and Svetlana Yarosh. 2020. Another decade of IDC research: Examining and reflecting on values and ethics. In Proceedings of the Interaction Design and Children Conference. 205–215.
- [94] Rébecca Kleinberger, Anne HK Harrington, Lydia Yu, Akito van Troyer, David Su, Janet M Baker, and Gabriel Miller. 2020. Interspecies Interactions Mediated by Technology: An Avian Case Study at the Zoo. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–12.
- [95] Bran Knowles, Sophie Beck, Joe Finney, James Devine, and Joseph Lindley. 2019. A scenario-based methodology for exploring risks: children and programmable IoT. In Proceedings of the 2019 on Designing Interactive Systems Conference. 751–761.
- [96] Yubo Kou, Xinning Gui, Yunan Chen, and Bonnie Nardi. 2019. Turn to the Self in Human-Computer Interaction: Care of the Self in Negotiating the Human-Technology Relationship. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–15.
- [97] Cherie Lacey and Catherine Caudwell. 2019. Cuteness as a 'Dark Pattern'in home robots. In 2019 14th ACM/IEEE International Conference on Human-Robot Interaction (HRI). IEEE, 374–381.
- [98] Marc Langheinrich, Albrecht Schmidt, Nigel Davies, and Rui José. 2013. A practical framework for ethics: the PD-net approach to supporting ethics compliance in public display studies. In Proceedings of the 2nd ACM International Symposium on Pervasive Displays. 139–143.
- [99] Amanda Lazar, Jessica L Feuston, Caroline Edasis, and Anne Marie Piper. 2018. Making as expression: Informing design with people with complex communication needs through art therapy. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–16.
- [100] Amanda Lazar, Hilaire J Thompson, Anne Marie Piper, and George Demiris. 2016. Rethinking the design of robotic pets for older adults. In Proceedings of the 2016 ACM Conference on Designing Interactive Systems. 1034–1046.
- [101] Gilly Leshed and Phoebe Sengers. 2011. " I lie to myself that i have freedom in my own schedule" productivity tools and experiences of busyness. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. 905–914.
- [102] Jimmy Lin, Ian Milligan, Douglas W Oard, Nick Ruest, and Katie Shilton. 2020. We Could, but Should We? Ethical Considerations for Providing Access to GeoCities and Other Historical Digital Collections. In Proceedings of the 2020 Conference on Human Information Interaction and Retrieval. 135–144.

#### A Scoping Review of Ethics Across SIGCHI

- [103] J Derek Lomas, Jodi Forlizzi, Nikhil Poonwala, Nirmal Patel, Sharan Shodhan, Kishan Patel, Ken Koedinger, and Emma Brunskill. 2016. Interface
- design optimization as a multi-armed bandit problem. In Proceedings of the 2016 CHI conference on human factors in computing systems. 4142–4153.
  [104] Sus Lyckvi and Olof Torgersson. 2018. Privacy and design ethics vs designing for curiosity, communication and children: Lessons learned. In Proceedings of the 20th International Conference on Human-Computer Interaction with Mobile Devices and Services. 1–8.
- [105] Ning F Ma, Chien Wen Yuan, Moojan Ghafurian, and Benjamin V Hanrahan. 2018. Using Stakeholder Theory to Examine Drivers' Stake in Uber. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–12.
- [106] Michael A Madaio, Luke Stark, Jennifer Wortman Vaughan, and Hanna Wallach. 2020. Co-Designing Checklists to Understand Organizational Challenges and Opportunities around Fairness in AI. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–14.
- [107] Bertram F Malle, Matthias Scheutz, Thomas Arnold, John Voiklis, and Corey Cusimano. 2015. Sacrifice one for the good of many? People apply different moral norms to human and robot agents. In 2015 10th ACM/IEEE International Conference on Human-Robot Interaction (HRI). IEEE, 117–124.
- [108] David Martin, Benjamin V Hanrahan, Jacki O'Neill, and Neha Gupta. 2014. Being a turker. In Proceedings of the 17th ACM conference on Computer supported cooperative work & social computing. 224–235.
- [109] Michael Massimi, William Odom, Richard Banks, and David Kirk. 2011. Matters of life and death: locating the end of life in lifespan-oriented hci research. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. 987–996.
- [110] J Nathan Matias and Merry Mou. 2018. CivilServant: Community-led experiments in platform governance. In Proceedings of the 2018 CHI conference on human factors in computing systems. 1–13.
- [111] J Nathan Matias, Sarah Szalavitz, and Ethan Zuckerman. 2017. FollowBias: supporting behavior change toward gender equality by networked gatekeepers on social media. In Proceedings of the 2017 ACM Conference on Computer Supported Cooperative Work and Social Computing. 1082–1095.
- [112] Steffen Maurer, Rainer Erbach, Issam Kraiem, Susanne Kuhnert, Petra Grimm, and Enrico Rukzio. 2018. Designing a guardian angel: Giving an automated vehicle the possibility to override its driver. In Proceedings of the 10th international conference on automotive user interfaces and interactive vehicular applications. 341–350.
- [113] Donald McMillan, Alistair Morrison, and Matthew Chalmers. 2013. Categorised ethical guidelines for large scale mobile HCI. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. 1853–1862.
- [114] Brenna McNally, Mona Leigh Guha, Matthew Louis Mauriello, and Allison Druin. 2016. Children's perspectives on ethical issues surrounding their past involvement on a participatory design team. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. 3595–3606.
- [115] Alexander G Mirnig and Alexander Meschtscherjakov. 2019. Trolled by the trolley problem: On what matters for ethical decision making in automated vehicles. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–10.
- [116] Rolf Molich, Brenda Laurel, Carolyn Snyder, Whitney Quesenbery, and Chauncey E Wilson. 2001. Ethics in HCL. In CHI'01 Extended Abstracts on Human Factors in Computing Systems. 217–218.
- [117] Wendy Moncur. 2013. The emotional wellbeing of researchers: considerations for practice. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. 1883–1890.
- [118] Alistair Morrison, Donald McMillan, Stuart Reeves, Scott Sherwood, and Matthew Chalmers. 2012. A hybrid mass participation approach to mobile software trials. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. 1311–1320.
- [119] Florian Floyd Mueller, Pedro Lopes, Paul Strohmeier, Wendy Ju, Caitlyn Seim, Martin Weigel, Suranga Nanayakkara, Marianna Obrist, Zhuying Li, Joseph Delfa, et al. 2020. Next Steps for Human-Computer Integration. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–15.
- [120] Zachary Munn, Micah DJ Peters, Cindy Stern, Catalin Tufanaru, Alexa McArthur, and Edoardo Aromataris. 2018. Systematic review or scoping review? Guidance for authors when choosing between a systematic or scoping review approach. BMC medical research methodology 18, 1 (2018), 1–7.
- [121] Cosmin Munteanu, Heather Molyneaux, Wendy Moncur, Mario Romero, Susan O'Donnell, and John Vines. 2015. Situational ethics: Re-thinking approaches to formal ethics requirements for human-computer interaction. In Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems. 105–114.
- [122] Lisa P Nathan, Anja Thieme, Deborah Tatar, and Stacy Branham. 2016. Disruptions, dilemmas and paradoxes: Ethical matter (s) in design research. Interacting with Computers (2016), 1–9.
- [123] Carman Neustaedter and Jennifer Golbeck. 2013. Exploring pet video chat: the remote awareness and interaction needs of families with dogs and cats. In Proceedings of the 2013 conference on Computer supported cooperative work. 1549–1554.
- [124] Josef Nguyen and Bonnie Ruberg. 2020. Challenges of Designing Consent: Consent Mechanics in Video Games as Models for Interactive User Agency. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–13.
- [125] Evangelos Niforatos, Adam Palma, Roman Gluszny, Athanasios Vourvopoulos, and Fotis Liarokapis. 2020. Would you do it?: Enacting Moral Dilemmas in Virtual Reality for Understanding Ethical Decision-Making. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–12.
- [126] Sean Peacock, Robert Anderson, and Clara Crivellaro. 2018. Streets for people: engaging children in placemaking through a socio-technical process. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–14.
- [127] Dorian Peters, Rafael A Calvo, and Richard M Ryan. 2018. Designing for motivation, engagement and wellbeing in digital experience. Frontiers in psychology 9 (2018), 797.

#### DIS '22, June 13-17, 2022, Virtual

- [128] James Pierce, Richmond Y Wong, and Nick Merrill. 2020. Sensor Illumination: Exploring Design Qualities and Ethical Implications of Smart Cameras and Image/Video Analytics. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–19.
- [129] Charlie Pinder, Jo Vermeulen, Benjamin R Cowan, Russell Beale, and Robert J Hendley. 2017. Exploring the feasibility of subliminal priming on smartphones. In Proceedings of the 19th International Conference on Human-Computer Interaction with Mobile Devices and Services. 1–15.
- [130] Henning Pohl, Andreea Muresan, and Kasper Hornbæk. 2019. Charting subtle interaction in the hci literature. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–15.
- [131] Stephen Purpura, Victoria Schwanda, Kaiton Williams, William Stubler, and Phoebe Sengers. 2011. Fit4life: the design of a persuasive technology promoting healthy behavior and ideal weight. In Proceedings of the SIGCHI conference on human factors in computing systems. 423–432.
- [132] Kasper Rodil, Donovan Maasz, and Heike Winschiers-Theophilus. 2020. Moving Virtual Reality out of its Comfort Zone and Into the African Kalahari Desert Field: Experiences From Technological Co-Exploration With an Indigenous San Community in Namibia. In 26th ACM Symposium on Virtual Reality Software and Technology. 1–10.
- [133] Yvonne Rogers, Helen Sharp, and Jenny Preece. 2011. Interaction design: beyond human-computer interaction. John Wiley & Sons.
- [134] John Rooksby, Alistair Morrison, and Dave Murray-Rust. 2019. Student Perspectives on Digital Phenotyping. (2019).
- [135] Daniel Roth, Gary Bente, Peter Kullmann, David Mal, Chris Felix Purps, Kai Vogeley, and Marc Erich Latoschik. 2019. Technologies for social augmentations in user-embodied virtual reality. In 25th ACM Symposium on Virtual Reality Software and Technology. 1–12.
- [136] Samar Sabie and Tapan Parikh. 2019. Cultivating Care through Ambiguity: Lessons from a Service Learning Course. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–14.
- [137] Neelima Sailaja, Andy Crabtree, Derek McAuley, and Phil Stenton. 2018. Explicating the challenges of providing novel media experiences driven by user personal data. In Proceedings of the 2018 ACM International Conference on Interactive Experiences for TV and Online Video. 101–113.
- [138] Pedro Sanches, Axel Janson, Pavel Karpashevich, Camille Nadal, Chengcheng Qu, Claudia Daudén Roquet, Muhammad Umair, Charles Windlin, Gavin Doherty, Kristina Höök, et al. 2019. HCI and Affective Health: Taking stock of a decade of studies and charting future research directions. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–17.
- [139] Elaheh Sanoubari, Denise Y Geiskkovitch, Diljot S Garcha, Shahed A Sabab, Kenny Hong, James E Young, Andrea Bunt, and Pourang Irani. 2018. Subliminal Priming in Human-Agent Interaction: Can Agents Use Single-Frame Visuals in Video Feeds to Shape User Perceptions?. In Proceedings of the 6th International Conference on Human-Agent Interaction. 205–213.
- [140] Elaheh Sanoubari, Stela H Seo, Diljot Garcha, James E Young, and Verónica Loureiro-Rodríguez. 2019. Good Robot Design or Machiavellian? An In-the-Wild Robot Leveraging Minimal Knowledge of Passersby's Culture. In 2019 14th ACM/IEEE International Conference on Human-Robot Interaction (HRI). IEEE, 382–391.
- [141] Corina Sas, Kobi Hartley, and Muhammad Umair. 2020. ManneqKit Cards: A Kinesthetic Empathic Design Tool Communicating Depression Experiences. In Proceedings of the 2020 ACM Designing Interactive Systems Conference. 1479–1493.
- [142] Matthias Scheutz and Thomas Arnold. 2016. Are we ready for sex robots?. In 2016 11th ACM/IEEE International Conference on Human-Robot Interaction (HRI). IEEE, 351–358.
- [143] Joseph Seering, Tianmi Fang, Luca Damasco, Mianhong'Cherie' Chen, Likang Sun, and Geoff Kaufman. 2019. Designing user interface elements to improve the quality and civility of discourse in online commenting behaviors. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–14.
- [144] William Seymour, Reuben Binns, Petr Slovak, Max Van Kleek, and Nigel Shadbolt. 2020. Strangers in the Room: Unpacking Perceptions of 'Smartness' and Related Ethical Concerns in the Home. In Proceedings of the 2020 ACM Designing Interactive Systems Conference. 841–854.
- [145] Ben Rydal Shapiro, Amanda Meng, Cody O'Donnell, Charlotte Lou, Edwin Zhao, Bianca Dankwa, and Andrew Hostetler. 2020. Re-Shape: A method to teach data ethics for data science education. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–13.
- [146] Weiyan Shi, Xuewei Wang, Yoo Jung Oh, Jingwen Zhang, Saurav Sahay, and Zhou Yu. 2020. Effects of Persuasive Dialogues: Testing Bot Identities and Inquiry Strategies. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–13.
- [147] Katie Shilton and Sara Anderson. 2017. Blended, not bossy: Ethics roles, responsibilities and expertise in design. Interacting with computers 29, 1 (2017), 71–79.
- [148] David Sirkin, Nikolas Martelaro, Mishel Johns, and Wendy Ju. 2017. Toward measurement of situation awareness in autonomous vehicles. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems. 405–415.
- [149] Laurianne Sitbon, Benoit Favre, Margot Brereton, Stewart Koplick, and Lauren Fell. 2020. Engaging the Abilities of Participants with Intellectual Disabilityin IIR Research. In Proceedings of the 2020 Conference on Human Information Interaction and Retrieval. 103–112.
- [150] Michael Warren Skirpan, Jacqueline Cameron, and Tom Yeh. 2018. More than a show: Using personalized immersive theater to educate and engage the public in technology ethics. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–13.
- [151] Michael Warren Skirpan, Tom Yeh, and Casey Fiesler. 2018. What's at Stake: Characterizing Risk Perceptions of Emerging Technologies. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–12.
- [152] Jaime Snyder. 2020. Visualizing Personal Rhythms: A Critical Visual Analysis of Mental Health in Flux. In Proceedings of the 2020 ACM Designing Interactive Systems Conference. 269–281.
- [153] Katta Spiel, Sultan A Alharthi, Andrew Jian-lan Cen, Jessica Hammer, Lennart E Nacke, Z O Toups, and Tess Tanenbaum. 2019. "It Started as a Joke" On the Design of Idle Games. In Proceedings of the Annual Symposium on Computer-Human Interaction in Play. 495–508.

#### A Scoping Review of Ethics Across SIGCHI

- [154] Bernd Carsten Stahl, Job Timmermans, and Brent Daniel Mittelstadt. 2016. The ethics of computing: A survey of the computing-oriented literature. ACM Computing Surveys (CSUR) 48, 4 (2016), 1–38.
- [155] Constantine Stephanidis, Gavriel Salvendy, Margherita Antona, Jessie YC Chen, Jianming Dong, Vincent G Duffy, Xiaowen Fang, Cali Fidopiastis, Gino Fragomeni, Limin Paul Fu, et al. 2019. Seven HCI grand challenges. International Journal of Human–Computer Interaction 35, 14 (2019), 1229–1269.
- [156] Yolande Strengers, Lizhen Qu, Qiongkai Xu, and Jarrod Knibbe. 2020. Adhering, Steering, and Queering: Treatment of Gender in Natural Language Generation. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–14.
- [157] Norman Makoto Su and EunJeong Cheon. 2017. Reconsidering nature: the dialectics of fair chase in the practices of American midwest hunters. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems. 6089–6100.
- [158] Norman Makoto Su and Bryan Duggan. 2014. TuneTracker: tensions in the surveillance of traditional music. In Proceedings of the 2014 conference on Designing interactive systems. 845–854.
- [159] Sharifa Sultana, François Guimbretière, Phoebe Sengers, and Nicola Dell. 2018. Design within a patriarchal society: Opportunities and challenges in designing for rural women in bangladesh. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–13.
- [160] Kaiwen Sun, Abraham H Mhaidli, Sonakshi Watel, Christopher A Brooks, and Florian Schaub. 2019. It's My Data! Tensions Among Stakeholders of a Learning Analytics Dashboard. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–14.
- [161] Susumu Tachi. 2015. Telexistence. In Virtual Realities. Springer, 229–259.
- [162] Franziska Tachtler, Konstantin Aal, Tanja Ertl, Daniel Diethei, Jasmin Niess, Mohammed Khwaja, Reem Talhouk, Giovanna Nunes Vilaza, Shaimaa Lazem, Aneesha Singh, Marguerite Barry, Volker Wulf, and Geraldine Fitzpatrick. 2021. Artificially Intelligent Technology for the Margins: A Multidisciplinary Design Agenda. Association for Computing Machinery, New York, NY, USA. https://doi.org/10.1145/3411763.3441333
- [163] Jennyfer Lawrence Taylor, Alessandro Soro, Paul Roe, Anita Lee Hong, and Margot Brereton. 2018. "Debrief O'Clock" Planning, Recording, and Making Sense of a Day in the Field in Design Research. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. 1–14.
- [164] Suzanne Tolmeijer, Markus Kneer, Cristina Sarasua, Markus Christen, and Abraham Bernstein. 2020. Implementations in machine ethics: A survey. ACM Computing Surveys (CSUR) 53, 6 (2020), 1–38.
- [165] Austin L Toombs, Shaowen Bardzell, and Jeffrey Bardzell. 2015. The proper care and feeding of hackerspaces: Care ethics and cultures of making. In Proceedings of the 33rd annual ACM conference on human factors in computing systems. 629–638.
- [166] Lianne Toussaint and Marina Toeters. 2020. Keeping the Data within the Garment: Balancing Sensing and Actuating in Fashion Technology. In Proceedings of the 2020 ACM Designing Interactive Systems Conference. 2229–2238.
- [167] Andrea C Tricco, Erin Lillie, Wasifa Zarin, Kelly K O'Brien, Heather Colquhoun, Danielle Levac, David Moher, Micah DJ Peters, Tanya Horsley, Laura Weeks, et al. 2018. PRISMA extension for scoping reviews (PRISMA-ScR): checklist and explanation. Annals of internal medicine 169, 7 (2018), 467–473.
- [168] Giovanni Maria Troiano, Matthew Wood, and Casper Harteveld. 2020. " And This, Kids, Is How I Met Your Mother": Consumerist, Mundane, and Uncanny Futures with Sex Robots. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–17.
- [169] Emily Tseng, Fabian Okeke, Madeline Sterling, and Nicola Dell. 2020. "We can learn. Why not?" Designing Technologies to Engender Equity for Home Health Aides. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–14.
- [170] Gijs van Ewijk, Matthijs Smakman, and Elly A Konijn. 2020. Teachers' perspectives on social robots in education: an exploratory case study. In Proceedings of the Interaction Design and Children Conference. 273–280.
- [171] Max Van Kleek, Dave Murray-Rust, Amy Guy, Kieron O'hara, and Nigel Shadbolt. 2016. Computationally mediated pro-social deception. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. 552–563.
- [172] Maarten Van Mechelen, Gökçe Elif Baykal, Christian Dindler, Eva Eriksson, and Ole Sejer Iversen. 2020. 18 Years of ethics in child-computer interaction research: a systematic literature review. In Proceedings of the Interaction Design and Children Conference. 161–183.
- [173] Tijs Vandemeulebroucke, Bernadette Dierckx de Casterlé, and Chris Gastmans. 2018. The use of care robots in aged care: A systematic review of argument-based ethics literature. Archives of gerontology and geriatrics 74 (2018), 15–25.
- [174] Asimina Vasalou, Rilla Khaled, Daniel Gooch, and Laura Benton. 2014. Problematizing cultural appropriation. In Proceedings of the first ACM SIGCHI annual symposium on Computer-human interaction in play. 267–276.
- [175] Michael Veale, Max Van Kleek, and Reuben Binns. 2018. Fairness and accountability design needs for algorithmic support in high-stakes public sector decision-making. In Proceedings of the 2018 chi conference on human factors in computing systems. 1–14.
- [176] John Vines, Rachel Clarke, Peter Wright, John McCarthy, and Patrick Olivier. 2013. Configuring participation: on how we involve people in design. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. 429–438.
- [177] Jessica Vitak, Katie Shilton, and Zahra Ashktorab. 2016. Beyond the Belmont principles: Ethical challenges, practices, and beliefs in the online data research community. In Proceedings of the 19th ACM conference on computer-supported cooperative work & social computing. 941–953.
- [178] Laura Wächter and Felix Lindner. 2018. An explorative comparison of blame attributions to companion robots across various moral dilemmas. In Proceedings of The 6th International Conference on Human-Agent Interaction. 269–276.
- [179] Annika Waern. 2016. The ethics of unaware participation in public interventions. In proceedings of the 2016 CHI conference on human factors in computing systems. 803–814.
- [180] Pontus Wärnestål, Petra Svedberg, and Jens Nygren. 2014. Co-constructing child personas for health-promoting services with vulnerable children. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. 3767–3776.

#### DIS '22, June 13-17, 2022, Virtual

- [181] Jenny Waycott, Frank Vetere, Sonja Pedell, Amee Morgans, Elizabeth Ozanne, and Lars Kulik. 2016. Not for me: Older adults choosing not to participate in a social isolation intervention. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. 745–757.
- [182] Danielle Wilde and Patrizia Marti. 2018. Exploring aesthetic enhancement of wearable technologies for deaf women. In Proceedings of the 2018 designing interactive systems conference. 201–213.
- [183] Tom Williams, Qin Zhu, Ruchen Wen, and Ewart J de Visser. 2020. The confucian matador: three defenses against the mechanical bull. In Companion of the 2020 ACM/IEEE International Conference on Human-Robot Interaction. 25–33.
- [184] Niels Wouters, Ryan Kelly, Eduardo Velloso, Katrin Wolf, Hasan Shahid Ferdous, Joshua Newn, Zaher Joukhadar, and Frank Vetere. 2019. Biometric Mirror: Exploring Ethical Opinions towards Facial Analysis and Automated Decision-Making. In Proceedings of the 2019 on Designing Interactive Systems Conference. 447–461.
- [185] Yiying Wu, Sus Lyckvi, and Virpi Roto. 2019. "What is Fair Shipping, Anyway?" Using Design Fiction to Raise Ethical Awareness in an Industrial Context. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–13.
- [186] Susan P Wyche, Sarita Yardi Schoenebeck, and Andrea Forte. 2013. "Facebook is a luxury" an exploratory study of social media use in rural Kenya. In Proceedings of the 2013 conference on Computer supported cooperative work. 33–44.
- [187] Dohyeon Yeo, Gwangbin Kim, and Seungjun Kim. 2020. Toward immersive self-driving simulations: Reports from a user study across six platforms. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–12.
- [188] John Zoshak and Kristin Dew. 2021. Beyond Kant and Bentham: How Ethical Theories are being used in Artificial Moral Agents. In Proceedings of the 2021 CHI Conference on Human Factors in Computing Systems. 1–15.

## 2.2 Futures for Health Research Data Platforms From the Participants' Perspectives

## Full Reference:

Giovanna Nunes Vilaza, Raju Maharjan, David Coyle, and Jakob Bardram. 2020. Futures for Health Research Data Platforms From the Participants' Perspectives. In *Proceedings of the 11th Nordic Conference on Human-Computer Interaction: Shaping Experiences, Shaping Society (NordiCHI '20), October 25–29, 2020, Tallinn, Estonia*. ACM, New York, NY, USA, 14 pages. https://doi.org/10.1145/3419249.3420110

## **Copyrights:**

Open-access article. Republished under the Creative Commons license. Copyright held by the authors.

## Futures for Health Research Data Platforms From the Participants' Perspectives

Giovanna Nunes Vilaza Technical University of Denmark Kgs. Lyngby, Denmark gnvi@dtu.dk

> David Coyle University College Dublin Dublin, Ireland d.coyle@ucd.ie

### ABSTRACT

In clinical cohort studies, researchers analyse the life history of population groups to understand the evolution of diseases. Health research data platforms came to facilitate such studies as they allow multiple projects to share access to cohorts' non-identifiable health information. Some latest initiatives are also starting to include mobile-generated data in their research programmes. Although seemly beneficial, it is not yet clear how potential participants feel about contributing to the new platforms: there is a need to investigate potential factors related to the acceptance in this specific context. In this paper, previous works from related contexts were brought together and, along with a qualitative study, composed a participant-centred perspective of enablers and barriers for contribution. We found that there is an apparent misalignment between current implementations and participants' preferences, leading us to propose design guidelines for future developments which can make participation more ethical and engaging.

## **CCS CONCEPTS**

• Human-centered computing  $\rightarrow$  Empirical studies in HCI; • Applied computing  $\rightarrow$  Health informatics.

## **KEYWORDS**

health research; data platforms; personal sensing; acceptance; privacy; ethics.

#### **ACM Reference Format:**

Giovanna Nunes Vilaza, Raju Maharjan, David Coyle, and Jakob Bardram. 2020. Futures for Health Research Data Platforms From the Participants' Perspectives. In Proceedings of the 11th Nordic Conference on Human-Computer Interaction: Shaping Experiences, Shaping Society (NordiCHI '20), October 25–29, 2020, Tallinn, Estonia. ACM, New York, NY, USA, 14 pages. https: //doi.org/10.1145/3419249.3420110

© 2020 Copyright held by the owner/author(s).

ACM ISBN 978-1-4503-7579-5/20/10.

https://doi.org/10.1145/3419249.3420110

Raju Maharjan Technical University of Denmark Kgs. Lyngby, Denmark rajm@dtu.dk

Jakob Bardram Technical University of Denmark Kgs. Lyngby, Denmark jakba@dtu.dk

## **1 INTRODUCTION**

Clinical cohort studies are used to investigate the natural history of diseases. By analysing demographic, behavioural and health data in chronological order, clinical researchers look for correlations, causes and effects [88]. Some of the challenges of these studies are the recruitment of large samples and the costly collection of longitudinal data [135].

Recently, health research data repositories have emerged to facilitate this process. They consist of large-scale shared-access databases containing different types of information on patients and healthy individuals, allowing multiple researchers to conduct analytical studies without having to recruit new participants [36, 112, 141]. Examples of data stored in such repositories may range from Electronic Health Records (e.g. Danish Civil Registration System [128]) to daily habits and behaviours (e.g. Evidation [44]).

Even though such platforms aim at benefiting society and healthcare, previous research in other contexts found that when it comes to sharing behavioural and sensitive data, people have several concerns. In the context of information sharing for health treatments, a benevolent end goal is not enough for patients to agree with disclosing every data type to any receiver [106]. Similarly, in the context of genetic and bio-repositories, individuals feel concerned about their DNA data being misused in the future even though they understand the need for research [96].

Given such evidence from other contexts, individual concerns may represent a risk for the future of the new platforms: lack of acceptance can deter potential participation and compromise cohort diversity [125]. While a large body of research has focused on willingness to share data with bio-repositories [39], genetic testers [70] and clinicians [69], there is a gap in research that focuses on the specific context of the emerging large-scale platforms and research programmes. The latest initiatives differ from bio-repositories because they contemplate the use of continuous behavioural monitoring, which means it is not only DNA being stored but a complete picture of one's daily life.

Therefore, there is a need to investigate how potential participants feel about contributing to shared-access repositories, especially those which include behavioural monitoring. Ethical, legal, and social issues related to the topic have been previously discussed [125], but not much has been considered in terms of the

Permission to make digital or hard copies of part or all of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. Copyrights for third-party components of this work must be honored. For all other uses, contact the owner/author(s). NordiCHI '20, October 25–29, 2020, Tallinn, Estonia

participants' preferences, concerns and priorities. Data repositories for health research have seen a rise in the last years [33], and since the COVID-19 pandemic, the expectation is that they will become even more predominant worldwide [48]. However, the context in question is underexplored in Human-Computer Interaction (HCI) [70]. It is time to consider how participants want to be engaged in such initiatives [59], as a better understanding of their viewpoint will enable the design of acceptable features aligned with people's concerns and expectations.

The goal of this paper is to investigate the perspectives of those who are not yet participants in these large-scale platforms and inspect their willingness to share different types of data. In order to achieve this goal, an empirical study with young adults was conducted. The motivation to focus on young adults came from the observation that those under 40 years old are often less favourable towards contributing to research repositories [50] and expect more control over how data is used [133], which makes them a more challenging group in terms of acceptance. Through the empirical study, our goal was to tease out their reasons for a potential resistance so that future developments can take these factors into account.

Moreover, this paper also provided a comprehensive compilation of the literature related to acceptance, privacy and motivation, as well as a summary of representative examples of existing platforms. The goal of this extensive narrative review was to put together findings and theories from scattered contexts, and show how they provide the foundation for the understanding of the participation in the new initiatives. Such aggregation is a useful resource to ground future research as well. Finally, we also contribute to design guidelines which are put in contrast with the current platforms to demonstrate what can be done to better align with participants' preferences.

### 2 LITERATURE REVIEW

The review was conducted in two parts. First, to characterise the spectrum of existing platforms, a set of examples was identified. Some of them had their origin in traditional research, but others came from a commercial or non-profit context, which means they were not listed in academic search engines. Therefore, searches were conducted online, on Google Scholar, via Twitter. Even though not systematic, this was considered as the most appropriate and feasible approach, given the heterogeneous field of search. Platforms were selected and categorised according to their distinctive characteristics to compose a diverse and representative set. Different from other reviews [9], particular emphasis was given to repositories which offered more agency to its participants. Only platforms with details available in English were included, which could mean that some examples, operating locally in their own language, were not detected.

Second, previous works about behavioural monitoring and research participation were identified and reviewed. Google Scholar, ScienceDirect, PubMed and the ACM library were searched for a combination of the keywords: participation in research, open-access repositories, bio-banks, genetic data-banks, motivation, willingness to share, acceptance, trust, privacy, informed consent, behavioural monitoring, self-tracking, research engagement. Papers' reference lists were also consulted (snowballing). We aimed at including a broad set of reference, in order to illustrate that many how different factors can influence participation, but we were also mindful of the quality of the publication venues. The choice of topics to highlight in the narrative emerged from several rounds of analysis and discussions among authors until the core elements relevant to the context were traced. Particular emphasis was placed on identifying key challenges for acceptance, willingness to share and consent. A summary of these can be seen in Table 1.

#### 2.1 Health research data platforms

A health research data platform is a repository which stores healthrelated data from population groups, to be used for analytical studies, often by multiple projects [92]. The development of such repositories represents a significant advance on how research can be conducted. Studies using data from existing platforms can provide evidence about the relationship between behaviours and health outcomes in an efficient and less costly way [54]. These repositories have been implemented differently, and the involvement of participants also differs among them.

2.1.1 National initiatives. A well-established example of a national research repository is the Danish Civil Registration System, which holds all medical registers for the entire Danish population [128]. Large-scale cohort studies have been conducted using this data, without much involvement of the patients. Researchers who want to get access to the data need to submit their projects for internal approval. Similar centralised national-level approaches are under development in India [124] and China [80].

Countries that do not have centralised health records have invested in the recruitment of volunteers for bio-banks. The *UK Biobank*, by the Wellcome Trust, collected clinical data from 500,000 patients until 2010 [136]. Their web page shows studies approved and published [89], as well as summaries of the available data [16]. Researchers can request access to this data, but the involvement of patients ended after data collection.

A more recent initiative is the *All of Us Research Programme*, which aims at recruiting one million volunteers and contemplates the collection of mobile-generated data [125]. According to their protocol [107], they want to allow volunteers to use the data to keep track of their health. Access is meant to be provided only to a group of researchers selected by the Programme.

2.1.2 Data portals for researchers. Another type of repository gathers data from previously conducted clinical trials. For instance, the non-profit organisation *Vivli* [25] works as a 'neutral broker' by connecting institutions and companies who are willing to share results from past projects [15]. Each organisation has its own data-sharing policies and access conditions, but Vivli run de-identification algorithms to the personal data beforehand. Participants receive the benefits offered by the individual clinical trials but are not involved in the processes at Vivli. Other data portals have aggregated information about genes, proteins and published papers, like the European Union's COVID-19 Data Portal [117]. In these platforms, it is up to the researchers to share their data, not the participants.

2.1.3 Opt-in sharing. Individuals may start to collect data for themselves but agree to share it for research afterwards. For instance, *Achievement* [2] is a commercial mobile app for personal health

Acceptance and Trust	People trust some institutions more than others	[34, 39, 118, 132]
	Trust can be hindered by previous experiences	[3, 5, 34]
	Acceptance decreases if data collection is hard	[32, 71]
	Acceptance decreases if the purpose is not useful	[75, 78, 137, 152]
Willingness to share	People worry about data misuse	[78, 95, 96]
	People want to preserve their reputation	[65, 77, 111, 119, 121]
	Some types of data are more sensitive	[14, 47, 61, 86, 123, 146]
	Some people are more concerned than others	[8, 63, 86]
Consent and Ethics	Consent forms are lengthy and complex	[84, 110]
	People cannot understand the risks	[31, 94, 115, 131]
	Lack of flexible sharing options and control	[5, 27, 57, 64, 73, 91, 106]
	Lack of transparency about data use	[39, 99, 142]

Table 1: Challenges identified in the literature, which can represent barriers to participation in health research platforms.

tracking in which users earn points by logging daily behaviours, answering surveys and being passively tracked. They can opt-in to share this data with *Evidation* [44], a platform which uses the information for clinical studies. Before the surge of COVID-19, not many details could be found about the on-going projects of Evidation, besides a publication list, but currently, they post weekly updates for the public. A similar opt-in approach is also used by the DNA genetic testing company *23andMe* [70] and the social network *PatientsLikeMe* [149]. In none of these examples are participants in control of how the data is used once they give their consent to share it for research.

2.1.4 Participant-driven. In some platforms, the initiative to join an initiative and collect data for that comes from the participants themselves. An example is the community-based Open Humans [53], in which projects are posted, and members are invited to contribute with their Fitbit, Twitter or symptoms data. Members can also propose projects of their own, and share tools for self-experimentation. With the surge of COVID-19, organisers have been hosting online meetings as a way to open the discussion about pandemic-related projects. A similar example is the recent Project Baseline, from Alphabet (parent company of Google) [90]. Their projects range from virtual mood tracking to in-person bio-sample collection. Since the rise of COVID-19, they partnered with local governments to send tests and share results with Departments of Public Health but so far, this data does not seem to be openly available. Besides such platforms which aggregate calls for several projects, there are also crowd-sourcing projects focused on health, also called 'participatory disease surveillance', in which individuals are invited to self-report symptoms and answer surveys for epidemiological control [103]. The COVID-19 emergency increased the number of such initiatives all over the world [48].

## 2.2 Behavioural monitoring and research participation

As it can be observed from the above overview, and from previous studies [29, 38, 47, 52, 85], smartphones and wearable data are starting to be considered for health research. Also called 'digital phenotyping', such data sources can provide information about users' environments [38], mental states [52], physical status [29] and daily habits [85]. This data can be collected passively (personal sensing) or manually through self-reports.

When it comes to using this data for health research, individuals are asked to contribute with their personal (and often sensitive) data for a purpose which at first may not directly benefit them it is meant to help society. In this complex context, there might be a desire to contribute, but privacy concerns might hinder the willingness to share data. Besides, the goal of many of the platforms is to allow data to be reused across multiple projects. Within genetic data repositories, there are several discussions about how consent should be asked in this situation [39]. This section, therefore, reviews factors which are key in the context of data platforms storing health and behavioural information for research beyond the scope of a single project.

2.2.1 Motivation. In the context of participating in research, motivation is the force which drives individuals to contribute and may come from different sources. It is common to reward participants with coupons, lottery tickets and gift cards [76], which are external sources of motivation. Money payments is another potential source of motivation to share personal data [13]. However, individuals might feel more motivated if their behaviours are self-determined: they act because they believe in the inherent value of their action [35]. A common finding in related works about willingness to participate in research is that people are motivated by contributing to science [150], helping patients [49] and seeing the outcomes [39].

In order to enhance motivation, researchers have looked for ways to get participants to be more engaged in the process [21, NordiCHI '20, October 25-29, 2020, Tallinn, Estonia

40, 46]. It has been discussed that participants should take part in agenda-setting, study design, recruitment, project evaluation [130], as well as decisions about the usage of the data [5]. By being more integrated with the project, they may feel more responsible for the results [7].

Getting test results is a primary motivation of those who send samples for genetic testing, such as 23AndMe [70] and previous works have pointed out to the use of personal data for reflection and behavioural change [45, 68, 82]. In addition to learning from the data collected, people can also feel motivated by comparing themselves to others [87], sharing their goals [30], and receiving feedback on their progress [51, 66]. However, there can also be negative consequences for using these as motivation drivers. When looking at their stats, some may experience anxiety [81, 147], feel guilty for not keeping up [42], lower their self-esteem [28, 100], and make incorrect interpretations [93, 109, 147]. Another issue is that when it comes to self-tracking, motivation may be necessary for keeping adherence to the data collection. It can be challenging for participants to stay consistent after joining a project [122], especially if they have to provide inputs frequently [26, 32].

2.2.2 *Privacy.* In the same way that motivation is an important enabler to participation in the research, privacy is a major barrier [74]. People worry about the potential for discrimination, undisclosed marketing interests, government surveillance and data used for generating profit [78, 95, 96]. Being constantly monitored can lead to feelings of surveillance [140], paranoia [127], and loss of freedom [24].

Privacy boundaries are often defined to protect internal values of autonomy, social acceptance and impression management [119]. In a study about students' perceptions on disclosing self-tracked data, they mentioned experiencing social pressure, stigma and embarrassment about their data [65]. The fear of a data breach is related to the potential damage to one's reputation and how others may react [111, 121]. By selecting which aspects of their lives to disclose, people can also control the image others have of themselves [77, 113].

However, willingness to share can depend considerably on the context [104], and there is a known mismatch between stated intentions and actual behaviours (privacy paradox) [105]. Therefore, even though privacy concerns may be a barrier to participation, in the context of research, individuals apply a 'usefulness heuristic': they prefer to share information that they know will be useful [83], and that they believe will not increase the risk for personal harm [121]. The Amazon Mechanical Turks were found to base their decision to share personal data for research on pay rate, requester, purpose, and perceived sensitivity of the request. However, they also engage in privacy-protective behaviours, such as abandoning tasks or providing inaccurate data [126].

Some factors can affect the intensity of privacy concerns. Previous research has found that more sensitive data types are disclosed less often [86, 123], such as audio recordings [37], browser history [14], message/phone logs and social media activity [47], camera pictures [121], financial information [146], home address [83], feelings of loneliness [61], sexually transmitted diseases [146], toilet use [12], and any health-related data in general [141]. Privacy is also perceived differently across individuals [22]. Personality traits

of agreeableness, conscientiousness, and openness to experience, can affect privacy perceptions [63]. Privacy fundamentalists are less likely to disclose information [86] as well as anxious individuals [8]. The benefits obtained from information disclosure are also a factor that can modulate the importance people attribute to privacy risks [70].

2.2.3 Acceptance. The decision to share data for research consists of weighing the motivation to contribute against the potential risks [83, 134]. It is clear that in the context of data sharing for research, motivation and privacy are critical components for acceptance. It is only when individuals 'accept' to contribute to a health research platform that their participation begins, but these acceptance should also be considered in the longer term, as it would be undesirable to have people dropping out [101].

The literature has extensively discussed factors for technology acceptance in general - usability and usefulness are known to be key factors for it [71]. In the context of participation in research, acceptance can increase if the burden of data input is low, which can be achieved by high usability [32, 57]. Regarding usefulness, people are more accepting of risks if they see their benefits, which can be for future patients [75], global causes [78, 152], researchers [137] or themselves [70]. However, usefulness can be highly contextual. Individuals may be less willing to participate if they think information is not relevant for the purpose [121, 123, 147]. On the other hand, individuals who are health-motivated [60], familiar with research [96] and who have disabilities [12] can be more accepting, because they understand how data can be useful.

The acceptance of information sharing with research repositories is also closely related to trust, especially on those who will keep and use the data [5, 118]. People may be willing to leave concerns aside when they trust those requesting their information [62, 79]. Scientific institutions are often trusted more than commercial, governmental and insurance institutions holding health information [39]. However, the development of trust is a continuous process, and trust can change over time [34]. Some enablers of trust in digital systems have been listed as fair data access, recommendation by others, customised design features, initial face-to-face contact, stakeholder engagement and improved communication [3].

## 2.3 Ethics and informed consent

It is well-known that before using any data from research subjects, the ethical approach is to obtain their consent, especially when data is collected through smartphones [55]. Subjects should also be allowed to withdraw from any study effortlessly [31]. The recent European General Data Protection Regulation (GDPR) established that in any data sharing procedure, people need to be at least informed about the information being shared, with whom and for what reasons [144]. However, consent forms are often overly complex [84], lengthy [110] and not efficient in informing participants [19, 58]. Many people do not understand what they are signing up for [145], how much information can be extracted from data [31], what data from sensors can reveal about them [94], how risky it is if someone identifies them [131], and how easily their identities can be revealed [115].

As a consequence, recent works have pointed to a desire for even more transparent and understandable information about data Futures for Health Research Data Platforms From the Participants' Perspectives

usage [39, 99, 142]. Participants of genetic repositories have mentioned being interested in receiving more details about the goals of the projects [139], reports on privacy risks [145] and notifications about when data is used [114]. However, it is not trivial to provide an appropriate level of information without confusing users or making it a burden [43, 72]. The literature on how to improve the process of asking for consent is extensive, especially within research [41]. Suggestions have been made to go beyond textual forms and consider face-to-face encounters [4, 114], communication channels (e.g. chats and forum) [129], visual info-graphics [97] and interactive tools [64, 102].

Furthermore, it has been argued that people should be allowed more control over how their data is used [5, 27, 57, 64], such as the level of detail disclosed [91, 106], and how long data is retained [73]. It has been suggested that willingness to share may increase if users can choose what information is shared and to whom [1, 148], and customised settings can enhance trust [114]. User interfaces that allow a selective disclosure of data have been proposed as a way to support automatic sharing rules in order to reduce the burden of continually editing preferences [100].

However, from the review of the existing platforms, it could be observed that most of the large-scale repositories often rely on models of *broad consent* [120], which means that participants provide their consent to any future study using their data for an indeterminate period of time (e.g. All of Us, Evidation). This contrasts with the idea of *narrow consent*, which has been found to be a preferred option over broad consent in some cases [67]. This type of consent consists of consulting participants for each project using the data (e.g. Open Humans), but restricts the re-use across multiple researchers and the combination of data sources. A consent type in between narrow and broad is called the *tiered consent* [20], in which consent is only asked in certain situations, such as if studies vary significantly in nature. None of the platforms seems to follow this type of consent yet - such mid-term appears to be missing in this space.

## **3 EMPIRICAL STUDY**

Our literature review identified enablers and barriers for participation in research and data sharing overall, but it could not identify qualitative works specifically on health research repositories that consider continuous behaviour monitoring. Moreover, the overview of the existing repositories identified different strategies to health data platforms, but it remains unclear if the expectations of participants match the available options. This empirical study is to explore individuals' attitudes and preferences regarding this scenario. The study consisted of semi-structured interviews and a card sorting activity with potential future participants who had not contributed to any research data repository yet.

#### 3.1 Sample

A sample of 12 participants, aged between 19 and 26 (mean=23, std=1.8) was recruited through word-of-mouth. Recruitment stopped when data saturation was reached - no new themes emerged from the last two interviews. Participants were four females and eight males, from France, China, Spain, United States, Hungary, Netherlands, Italy and Scotland. One was a full-time developer, three were

part-time developers, three were doing internships (in wind energy, marine biology and photonics), two were exchange students, and two were finishing their Masters'. They reported being healthy in general, did not mention experiencing data breaches in the past, never been part of a clinical study, and four of them tracked physical activity occasionally. The study was exempt from ethical approval but was conducted according to accepted ethical standards (informed consent, benefit- not harm, confidentiality) and followed the national code of scientific conduct.

#### 3.2 Study design

Semi-structured interviews and card sorting activities were conducted with each of the participants individually. All of the interviews had the same set of questions and cards (within-subjects design).

#### 3.3 Materials

Before the interview, a set of cards was prepared. Each card contained a different 'information item' (see Table 2). The choice of items was based on the analysis of previous works about data sensitivity and willingness to share [12, 14, 37, 47, 61, 83, 86, 121, 123, 141, 146], data collected in the different platforms, and mobile sensing frameworks for behavioural monitoring [23]. Three medical doctors were consulted to confirm that the card set contained only data relevant to health research.

#### 3.4 Procedure

In order to introduce participants about the concept of a health platform, on the day of the interviews, participants were first shown a 1 minute and 55 seconds video about precision medicine, made by the All of Us organisers <sup>1</sup>. The short video did not provide any in-depth details about the platform; it was an introduction to the idea of having a data repository for health research. Participants were explained that the video was meant to be just an illustrative example.

Participants were asked to describe were their initial opinions about health research platforms. Then, they were asked about potential reasons to contribute to and general concerns. Interview questions deliberately elicited both positive and negative perspectives to try to mitigate the bias of social acquiescence.

After these initial questions, participants were asked to do the card sorting. For each card, they were asked if they would feel comfortable with having the information depicted on the card stored in a repository. They were asked to place the cards in "yes", "no" and "maybe" piles and explain their decision. After the card sorting, they were asked what they would like to happen once their data had been shared data and what they expected to hear from the platform. The interview ended with the collection of demographic information. Interviews lasted 45-70 minutes. They were conducted in-person (before the COVID-19 crisis), audio-recorded, and transcribed. The card sorting activity was documented in pictures. Data analysis was conducted using Thematic Analysis [17, 18], with themes being iteratively generated from the data collected (inductive approach). Transcripts were read several times, during which similar quotes

<sup>&</sup>lt;sup>1</sup>https://www.youtube.com/watch?v=ti50nS7B5vI

NordiCHI '20, October 25-29, 2020, Tallinn, Estonia

Calls/text content	Calendar	Financial status	Relationship	Address	Blood/urine			
Mental illness	Grades	Genetic disorders	Emotional status	Family	Partner health			
Toilet habits	Hygiene	Sleep patterns	Places visited	DNA	Contagious illness			
Alcohol/ tobacco	Food	Local ambience	On-screen time	Distances	Physical exercises			
Physical status	Leisure	Current weather	Physical illness	App	Heart rate			
Table 2: List of information items (data types) shown as cards to participants								

Table 2: List of information items (data types) shown as cards to participants.

were given codes, and groups of related codes were associated with themes.

## 4 FINDINGS

The interview study had the goal of investigating how young individuals perceive the idea of contributing to a health research repository and how they feel about sharing different data items. The result of the analysis showed that three main themes were recurrently discussed. These were: reasons to contribute to a research repository, characteristics of an ideal platform and the criteria for sharing the different types of information.

#### 4.1 Motivations to contribute

When participants discussed the possibility of contributing to a health research platform in the future, they discussed the potential benefits of this choice. These fell under two categories: public and personal benefits.

4.1.1 Public benefits. Without exception, all of the participants praised the importance of health data repositories for society. Their initial comments often highlighted the usefulness of such idea: "There could be a lot of good for a lot of people if healthcare providers could have more access to information about people's lives" (P3). Such altruistic motivation, of helping researchers and patients, was the first mentioned by the majority of the participants: "I would share my data with the idea that it can help research and it can help other people to be cured" (P9).

4.1.2 Personal benefits. Besides altruistic reasons, there was definite interest from the majority of the participants in receiving personal benefits as well, such as using the data for their own health improvement, which was mentioned often: "If you have all the information about me, you can do more specific tailored care, maybe even like getting life recommendations" (P3). There was also interest in comparing themselves with others: "I would like to know how I compare with others. I feel like if I see others are using their phone significantly less time than I do, then I can control myself" (P7).

On the hand, the prospect of receiving material compensations was rarely mentioned, and when mentioned, participants had opposing opinions. Some believed that there their contribution should be rewarded: "Probably if it takes a lot of time, it should be rewarded, it should be paid" (P1), but others did not: "I don't think I would ask for financial compensation, I think it is more of a volunteering type of work" (P7).

#### 4.2 Trust in the platform

Throughout the interviews, participants mentioned some key features a platform should offer to be trusted. Two main requirements were recurrent: transparency and choice.

*4.2.1 Transparency.* In order to verify if a platform presented the desired characteristics, participants mentioned several times the need for more transparency, so that they could make an informed judgement. They would like transparency about:

- Who will use the data. Most participants mentioned that they would like to know both who is the organisation behind the platform and who will be provided with access: "I need to know who are the people behind the programme. Is it the university, is it an external research centre, is it for hospitals? And who is going to check my data?" (P2).
- For what purposes. They also wanted to know precisely for which purposes the data would be used now and in the future: "It is good not to share anything if we do not know how it can be used in the future" (P11). The reason for that was because not all purposes were acceptable, as there was an evident objection to allowing data to be used for profit, mentioned by the majority: "I'm happy to share this for health and public good reasons, but as soon as it starts to be for a private company, I feel like I am being taken advantage of" (P3).Some of the participants would like to be notified about what was done with the data: "I would be interested to know if I helped and what conclusions were drawn as a result" (P8). This interest was not shared by everyone, however: "I will give you my data, but I will never read the outcomes. For the people more interested, maybe" (P12).
- How data is secured and anonymised. Evidence of cybersecurity measures was regarded as a significant factor for trusting a platform: "If you feel that data is going to be leaked then I guess you wouldn't share a lot" (P5). One participant mentioned feeling reassured by being informed about the protocols used, even without fully understanding them: "Knowing some technical stuff would make me comfortable even though I don't understand that type of information" (P4). The same importance was given for knowing that data is stored anonymously: "If it is already anonymised in the database, then I would use the platform even if there was some kind of hack in the system because I know that I am just going to be one in the millions" (P4).

4.2.2 *Choice.* Besides being thoroughly informed about the aspects above, a topic often mentioned by participants was the desire

Futures for Health Research Data Platforms From the Participants' Perspectives

to be able to actively keep control of the data during and after it was shared. Participants stated that they would like to choose who can have access and what data is available:

- Selecting receivers. Some of the participants would like to allow access to projects on a case-by-case basis: "If they have new research and want to have access to my data, I want them to ask me if I want to release to the new research or not" (P7). Not all of the participants agreed with this thought: "I don't want to give the go-ahead for every researcher. There are probably too many requests, and I don't want to be bothered by this all the time" (P5). As one of the participants mentioned, it would be interesting to be able to at least block the access for studies they do not agree with: "I want to choose if I disagree with one of the studies and say: 'with this one I don't want to share'" (P10).
- Filtering the data. Many participants mentioned that it would be good to stop the data collection on some occasions: "Maybe in some situations is useful to stop recording some data. If you go to some places where you don't want to be followed, for example, you don't want people to know you are there" (P2); and to be able to see what data is stored and reconsider the decision to give access: "Once I have given the data, I think it would be nice to have a look at what is in the system, what data you saved and then have the option to maybe delete data points if I think it is not appropriate" (P5). Additionally, many of them mentioned that they would feel more comfortable if researchers have access only to less detailed information: "If it just to know what you are going to do in your day, how many hours you are walking, something like that, is fine" (P10).

## 4.3 Willingness to share each data type

Participants often diverged opinions about which information they would feel comfortable collecting and sharing. Also, the same data type sometimes led to different types of concerns between participants. An overview of the different criteria for these decisions is listed next.

4.3.1 Relevance for the research. Having a clear understanding how the information is useful for health research was in many cases the reason why a participant would feel comfortable with sharing it: "I would be happy to share that consumption information. For me, that has a very tangible connection to health" (P3). On the other hand, not fully understanding the relevance of an item was mentioned by many as a reason to not share: "I don't think my home address is relevant for the purposes of the research" (P8). Participants did not always agree on the relevance of the different data items, with some being fine with sharing items that others would not see a reason for it.

4.3.2 Potential for material damage. When participants felt comfortable with sharing a particular data item, it often was because they could not see the potential for any adverse consequences: "I don't think there is any harm in sharing your sleep patterns" (P8). On the other hand, when participants said feeling uncomfortable, they mentioned fears related to physical attacks: "If people want to locate me, it would be really easy. If I am going back home at this time, then someone is just waiting for me at the door and just breaking NordiCHI '20, October 25-29, 2020, Tallinn, Estonia

into my room" (P7); cyberattacks: "Financial situation, no, because people maybe know how much money I have in my bank account and they would see if it is worthy hacking it or not" (P7); and financial loss: "If hackers are working for an insurance company, and they sell data to them, it could happen that they [insurance companies] propose different plans" (P6). Not all of the participants associated these risks of material damage to location and financial situation participants varied in their opinion about what information would bring each type of risk.

4.3.3 Risks for reputation. Some participants mentioned that there were parts of their life they prefer to keep private, because of the fear of how others perceive this: "If it is not treatable then maybe it is something you want to keep it for yourself or share with only a limited amount of people" (P5). Similarly, some mentioned being concerned about the possibility of being discriminated by their employers: "I am afraid that in a potential data leak, my employer gains access and discriminate me in my employment" (P11). However, participants often diverged on opinions about which data types they perceived as riskier for their reputation - their decision depended on their judgement of their own habits. Some participants said they would feel ashamed to disclose their drinking habits "Alcohol, tobacco, I do not want people to know how much I drink because I drink too much" (P8), but others said they were comfortable with it. Information perceived positively by society was often considered as more comfortable to shared: "Physical activities, yes, because I feel like is something positive" (P6).

4.3.4 Invasion of others' privacy. Some participants were not comfortable with sharing data that did not belong to them: "I think sharing calls and text content overlaps with the privacy of other people" (P2). Sharing data from others was described as a betrayal by one of the participants: "I do not want to share their data, it is kind of like going behind their backs (P4). They believed they were not in the position to decide: "When it comes to the condition of my partner, it is tricky because I feel like it is their decision to share the information" (P3). This concern was not mentioned by all of the participants, however.

4.3.5 The effort of the data collection. Sometimes, the issue was the burden of data collection. Most of the participants mentioned that collecting data can be an inconvenient responsibility: "I don't want to share data when I'm low on my battery, or when I'm short of data. I wouldn't like this data to interfere with my daily life" (P2); some also mentioned it can be emotionally difficult: "If someone is really sad or depressed, I think it is harder to share that" (P10). Some of the participants did not mention issues with data collection.

## **5 DESIGN GUIDELINES**

Health research data platforms arrived to help science and healthcare to progress [6] and if they stick to this goal, public acceptance should be desirable. The interview study shed light on critical factors involved in young adults decision-making regarding (i) eventually contributing to health research platforms, (ii) and sharing specific types of information. The findings indicate that motivations to contribute (public and personal benefits) and indications that a platform can be trusted (who are the organisers, how data NordiCHI '20, October 25-29, 2020, Tallinn, Estonia

is secured and used) are significant components for initial acceptance of the platform's terms. Acceptance is, however, not fully achieved before potential participants evaluate their willingness to share the specific data types in question. According to our findings, this decision depends on the relevance for research, the effort to data collection and risks for material and reputation damage. Even though previous works have identified similar factors, our findings are original because of the context under investigation. The literature on participating in research could perhaps enlighten some motivations to join, but it would be insufficient to describe the importance of the willingness to share each data type. Similarly, the literature on data sharing and privacy has not yet considered the case of contributing to a shared-access research platform. This empirical study allowed us to pinpoint the key factors for the acceptance of the emergent platforms for digital phenotyping research, which has not been done before.

In Figure 1, we show the conceptual model we introduce for acceptance in the context of health research data platforms, which includes a summary of our findings and our proposed design guidelines. These guidelines focus not only on the initial acceptance of the platform and its data types but also their *continuous acceptance*. As a recent review on technology acceptance highlights, health initiatives should be aiming for a long-term engagement with its users [101], which is also what we advocate.

## 5.1 Provide supervised health feedback as a motivation source

As mentioned in the review, the motivation to participate in research can come from many sources. Financial compensation, to start with, has been used in several clinical studies and existing platforms (e.g. All of Us). However, in the interview study, participants rarely mentioned this as a motivation source, which was also the case in previous reviews about bio-banks [96]. Even though it has been found that people might be willing to sell personal data for money [13], another work has questioned whether material incentives lead to increased adherence and data quality [76]. Such types of compensations were found to be more beneficial if combined with other motivations [56].

The reasons to contribute to a research repository mentioned by our participants included both altruistic and more individualistic ones. It is widely known that people are more open to sharing personal data if it is for helping patients [75], science [137] and society [78, 152]. However, it was observed in our study that getting direct benefits from the data was as important, which aligns with studies about sharing genetic information [70]. In the case research repositories which collect health and behavioural data, this direct benefit is related to the possibility of receiving feedback on personal health and well-being indicators. The All of Us Programme seems to offer this, as they state they will provide ways for volunteers to get access to the data they share and some of the results [108]. Allowing people to keep track of their results, could not only motivate people to participate but also stay adherent and see a meaning for the effort of data collection, mentioned in the interview.

However, platforms should be mindful of how they inform users about health management so that they can extract a correct meaning out of their data. There are many studies of digital tools for personal health monitoring, [45, 68, 81, 87] and social sharing [30, 51, 66], which are useful inspirations on how to integrate personal care with self-tracking. However, an issue with self-administered tools is that there can be negative consequences, such as self-scrutiny [28, 100] and performance anxiety [81], which are particularly applicable to young adults [65]. Furthermore, when it comes to vulnerable patients of any age, self-monitoring is not recommended without the supervision of a specialist [11].

Therefore, research platforms should also consider how to offer proper clinical follow-up. Except for the COVID-19 testing, being offered by Project Baseline, large-scale platforms have not committed to health support. There are several studies about mobile applications which have combined personal health tracking with the close support of healthcare professionals [10, 93, 147]. Using health tracking with supervision and support could be a way to get participants motivated to contribute to research; however, it is also crucial to be mindful of the risks as mentioned above.

# 5.2 Allow personalised and flexible privacy choices

Our study elicited privacy concerns that could emerge if behavioural data is to be stored in a shared-access repository. Some were connected to the potential for personal harm, which was also found within data sharing for health treatments [106]; others were related to misalignment of personal values, such as data being used for hidden purposes, and a possible invasion to the privacy of others, which have been brought up in studies about genetic repositories [96]. Regarding differences across the different information types, even though quantitative studies could define some data types as more 'sensitive' [14], we observed that the reasons for such sensitivity depended a lot on the meaning people attribute to the different aspects of their lives and the importance they give to the different risks. Also, contrary to another study with young adults mobile sensors and privacy [121], our analysis showed that the same information often led to different types of concerns depending on the person, and no generalisations could be devised.

Such observation has implications for the use of behavioural monitoring for research repositories because if they intend to accommodate this pluralism of experiences, they need to allow participants to choose which data types are to be stored. Currently, consent forms follow an 'all or nothing' approach: participants who join a study are expected to share all the data required [116]; however, it can be difficult for some to disclose some parts of their life [113]. Participants in our interviews would like to stop tracking in some situations and select the level of detail visible to others, suggestions which were mentioned by older people (above 70) as well [106]. A way to achieve could be to provide participants with an overview of their data and allow them to exert some control over it, aligning with the increasing public interest on more flexible privacy options for personal data [1, 64, 74, 151]. User interfaces could be designed to allow interactive forms of selective disclosure [100] through continuous control of the data collected, as well as the possibility to not disclose parts of it. The consent process should also take into account these individual preferences, and volunteers should be given more flexibility about which information they want to share for research.

Futures for Health Research Data Platforms From the Participants' Perspectives

NordiCHI '20, October 25-29, 2020, Tallinn, Estonia



Figure 1: Conceptual model for the acceptance of health research data platforms based on our empirical findings.

Furthermore, our participants demonstrated a desire to have more control over the future of their data. Allowing participants to withdraw from studies is already required in recent regulations [144]; however, each platform has its approach when it comes to how much information and control participants have about the individual studies making use of their data. Participant-driven platforms (e.g. Open Humans) allow users to select precisely to which projects they want to share their data, but this can restrict data usage in the future. On the other hand, more significant initiatives (e.g. All of Us, Evidation) ask for broad consent so that they can use the data for any project they decide, but participants are not asked about anything else afterwards [70]. An attractive mid-term solution is, therefore, to allow participants to set-up rules or conditions under which any studies that fulfil them can access their data (tiered consent [20]). If participants would like to deny access to some receivers, this does not have to imply a complete withdrawal from the repository.

However, setting up privacy rules should not become an overwhelming process, in the way that consent forms became complex and lengthy documents [19, 84, 139]. The idea of moving away from the model of broad consent has been discussed in the context of bio-banks, but it is still an open question if permission should be requested for each new study [96]. Our findings indicate that it should be optional if permission is to be requested each time a new study needs the data - some might consider the obligation to provide consent as a burden while others might prefer to have full control. Providing individuals with more control over their data can be perceived as an indication of respect [98, 143], which can, in turn, become an enabler for trust and acceptance. The negotiation of privacy boundaries can be helped by more communication from the platform side, which is an aspect discussed next.

## 5.3 Use communication to increase engagement and acceptance

Our participants showed a positive attitude about sharing behavioural data for a research repository, as long as their data is safe, which aligns with the public views about genetic information as well [70, 96]. As discussed in the review, acceptance in this context is connected to the perceived usefulness of the purpose [14, 75]. Therefore, information about this purpose must be efficiently communicated to potential participants [152].

However, even though the repositories are providing a clear overall idea about how contributions can be useful, the ideal platform envisioned by most of our participants would be much more transparent about individual projects. As mentioned in the review of existing platforms, there is a lack of involvement of participants once data is collected. Participation in a repository is not only about the organisation who is behind it. If users want to know more about the on-going projects, they should be provided with complete reports about goals, methods and people involved, so that they know to what they are contributing. However, the communication between contributors and researchers in many cases consists only of 'publication updates' on a website.

For this reason, a continuous dialogue could be a path to fulfil the need for more transparency. Similar to our participants, young individuals in another study were intrigued about how some types of mobile sensors could be health indicators and if their identities could be revealed through them [121]. Platforms could offer more support and information when it comes to explaining how data is being used for research, going beyond public reports, and providing personalised content based on the participants' concerns. Besides, some of our participants were also interested in staying updated NordiCHI '20, October 25-29, 2020, Tallinn, Estonia

with the research outcomes [39]. Platforms should consider how to keep volunteers updated more frequently about the on-going analysis. Such a continuous form of communication could include chat tools with the scientists, social forums with other participants [129] and open access virtual meetings, similar to what Open Humans has been offering, for example. Another aspect which deserves more attention is the fact that not everyone has the same understanding about the overall risks of sharing data with a repository [31]. Information about how data is secured and kept anonymous to the researchers should be appropriately conveyed [115] and tools for communication could be useful in this process.

Spaces for the active involvement of participants could correspond to the desire expressed by some the young adults in our interviews. Active engagement may involve suggesting new research questions, providing feedback and voting on decisions [21, 46, 130], which in turn can make people feel rewarded when the initiatives are successful [7]. Learning materials could be provided for participants interested in understanding the methods behind the investigations [150]. By considering and providing such options, the platforms of the future could move forward from the traditional relationship between 'study subjects' and 'scientists' [138].

From the review of the existing platforms, it could be observed that since the surge of the virus, there have been efforts to promote more participant-centred features, such open community calls (Open Humans) and weekly reports (Evidation). However, none of these examples has fully considered the directions presented in this section. The COVID-19 pandemic brought urgency to data sharing for research, which represents an excellent opportunity to reflect on how we should be building acceptance and trust. Our vision is that by bringing participants closer to the research process, their participation can become more engaging and respectful. Instead of imposing terms and conditions from the top, these could be negotiated and established based on participants' input of how they would like the future of research repositories to be.

#### 5.4 Limitations and future work

This paper presents a limited empirical study in terms of sample demographics, which could compromise broader generalisations. Besides, as the study happened before the COVID-19 pandemic, any possible changes in attitudes because of that are not represented. Future follow-up studies could investigate the perspective of other age groups, as well as the effect of the pandemic on their opinions. Quantitative approaches could also be used in the future to identify patterns across larger samples. It could also be interesting to expand this comprehension and consider how the emerging platforms are being conceived and perceived across the globe. Finally, further studies could investigate the impact of our design guidelines in-thewild.

### 6 CONCLUSION

This paper provided an overview of health research data repositories from a participant-centred perspective. It reviewed the landscape of current platforms and brought together scattered literature to conceptualise the experience of participating in them. The acceptance to participate was found to depend on the trust in the platform, individual motivations, and the willingness to share different data types. Based on the review and the empirical results, we formulated guidelines on how to enhance identified enablers and dissipate barriers for participation in an ethical way.

We argue that participants should not be considered passive study subjects, but rather be invited to get actively involved in the studies and control the shared data. Some existing initiatives have started moving towards this direction, but there are still several opportunities to design features to reshape relationships of power between data contributors and platform organisers. After all, the future of health research data platforms should be guided by the participants' perspectives, because they are the ones putting themselves in a vulnerable position for the common good.

## ACKNOWLEDGMENTS

This project is part of TEAM (Technology Enabled Mental Health for Young People), which is funded by the European Union's Horizon 2020 research and innovation program under the Marie Skłodowska-Curie grant agreement No. 722561. We thank each of the participants for their time and openness; Kevin Doherty and Darius Rohani for feedback; and our TEAM colleagues for the constant motivating support.

#### REFERENCES

- Mohamed Abdelhamid. 2018. Greater patient health information control to improve the sustainability of health information exchanges. *Journal of biomedical informatics* 83 (2018), 150–158.
- [2] Achievement. 2019. "Achievement is my accountability buddy for logging food, activity, and water intake.". Retrieved May 2, 2020 from https://www. myachievement.com/
- [3] Afua Adjekum, Alessandro Blasimme, and Effy Vayena. 2018. Elements of Trust in Digital Health Systems: Scoping Review. *Journal of medical Internet research* 20, 12 (2018), e11254.
- [4] Emily E Anderson, Susan B Newman, and Alicia K Matthews. 2017. Improving informed consent: Stakeholder views. AJOB empirical bioethics 8, 3 (2017), 178–188.
- [5] Nicholas Anderson, Caleb Bragg, Andrea Hartzler, and Kelly Edwards. 2012. Participant-centric initiatives: tools to facilitate engagement in research. Applied & translational genomics 1 (2012), 25–29.
- [6] Euan A Ashley. 2015. The precision medicine initiative: a new national effort. Jama 313, 21 (2015), 2119–2120.
- [7] Mara Balestrini, Yvonne Rogers, Carolyn Hassan, Javi Creus, Martha King, and Paul Marshall. 2017. A city in common: a framework to orchestrate largescale citizen engagement around urban issues. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems. 2282–2294.
- [8] Gaurav Bansal, David Gefen, et al. 2010. The impact of personal dispositions on information sensitivity, privacy concern and trust in disclosing health information online. Decision support systems 49, 2 (2010), 138–150.
- [9] Rita Banzi, Steve Canham, Wolfgang Kuchinke, Karmela Krleza-Jeric, Jacques Demotes-Mainard, and Christian Ohmann. 2019. Evaluation of repositories for sharing individual-participant data from clinical studies. *Trials* 20, 1 (2019), 169.
- [10] Jakob E. Bardram and Mads Frost. 2017. Double-Loop health technology: Enabling socio-technical design of personal health technology in clinical practice. In Designing Healthcare That Works: A Sociotechnical Approach.
- [11] Jakob E Bardram, Mads Frost, Károly Szántó, Maria Faurholt-Jepsen, Maj Vinberg, and Lars Vedel Kessing. 2013. Designing mobile health technology for bipolar disorder: a field trial of the monarca system. In Proceedings of the SIGCHI conference on human factors in computing systems. ACM, 2627–2636.
- [12] Scott Beach, Richard Schulz, Julie Downs, Judith Matthews, Bruce Barron, and Katherine Seelman. 2009. Disability, age, and informational privacy attitudes in quality of life technology applications: Results from a national web survey. ACM Transactions on Accessible Computing (TACCESS) 2, 1 (2009), 5.
- [13] Volker Benndorf and Hans-Theo Normann. 2018. The willingness to sell personal data. *The Scandinavian Journal of Economics* 120, 4 (2018), 1260–1278.
- [14] Jaspreet Bhatia and Travis D Breaux. 2018. Empirical measurement of perceived privacy risk. ACM Transactions on Computer-Human Interaction (TOCHI) 25, 6 (2018), 34.

Futures for Health Research Data Platforms From the Participants' Perspectives

- [15] Barbara E Bierer, Rebecca Li, Mark Barnes, and Ida Sim. 2016. A global, neutral platform for sharing trial data. *New England Journal of Medicine* 374, 25 (2016), 2411–2413.
- [16] UK Biobank. 2020. Online showcase of UK Biobank. Retrieved May 2, 2020 from http://biobank.ndph.ox.ac.uk/showcase
- [17] AE Blandford. 2013. Semi-structured qualitative studies. Interaction Design Foundation.
- [18] Virginia Braun and Victoria Clarke. 2012. Thematic analysis. (2012).
- [19] Jamie C Brehaut, Kelly Carroll, Glyn Elwyn, Raphael Saginur, Jonathan Kimmelman, Kaveh Shojania, Ania Syrowatka, Trang Nguyen, Erica Hoe, and Dean Fergusson. 2012. Informed consent documents do not encourage good-quality decision making. *Journal of clinical epidemiology* 65, 7 (2012), 708–724.
- [20] Stefanie Broes, Denis Lacombe, Michiel Verlinden, and Isabelle Huys. 2018. Toward a tiered model to share clinical trial data and samples in precision oncology. Frontiers in medicine 5 (2018), 6.
- [21] Elizabeth Bromley, Lisa Mikesell, Felica Jones, and Dmitry Khodyakov. 2015. From subject to participant: Ethics and the evolving role of community in health research. American Journal of Public Health 105, 5 (2015), 900–908.
- [22] Alison Burrows, David Coyle, and Rachael Gooberman-Hill. 2018. Privacy, boundaries and smart homes for health: An ethnographic study. *Health & place* 50 (2018), 112–118.
- [23] CACHET. 2019. CARP Mobile Sensing. Retrieved May 2, 2020 from https: //github.com/cph-cachet/carp.sensing-flutter
- [24] Kelly Caine, Selma Šabanovic, and Mary Carter. 2012. The effect of monitoring by cameras and robots on the privacy enhancing behaviors of older adults. In Proceedings of the seventh annual ACM/IEEE international conference on Human-Robot Interaction. ACM, 343–350.
- [25] Vivli center for global research data. 2019. Our members. Retrieved May 2, 2020 from https://vivli.org/members/ourmembers/
- [26] Eun Kyoung Choe, Nicole B Lee, Bongshin Lee, Wanda Pratt, and Julie A Kientz. 2014. Understanding quantified-selfers' practices in collecting and exploring personal data. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. ACM, 1143–1152.
- [27] Delphine Christin, Andreas Reinhardt, Salil S Kanhere, and Matthias Hollick. 2011. A survey on privacy in mobile participatory sensing applications. *Journal of systems and software* 84, 11 (2011), 1928–1946.
- [28] Trudy Hui Hui Chua and Leanne Chang. 2016. Follow me and like my beautiful selfies: Singapore teenage girls' engagement in self-presentation and peer comparison on social media. *Computers in Human Behavior* 55 (2016), 190–197.
- [29] Chia-Fang Chung, Jonathan Cook, Elizabeth Bales, Jasmine Zia, and Sean A Munson. 2015. More than telemonitoring: health provider use and nonuse of life-log data in irritable bowel syndrome and weight management. *Journal of medical Internet research* 17, 8 (2015), e203.
- [30] Chia-Fang Chung, Nanna Gorm, Irina A Shklovski, and Sean Munson. 2017. Finding the right fit: understanding health tracking in workplace wellness programs. In Proceedings of the 2017 CHI conference on human factors in computing systems. ACM, 4875–4886.
- [31] Karin Clark, Matt Duckham, Marilys Guillemin, Assunta Hunter, Jodie McVernon, Christine O'Keefe, Cathy Pitkin, Steven Prawer, Richard Sinnott, Deborah Warr, et al. 2019. Advancing the ethical use of digital data in human research: challenges and strategies to promote ethical practice. *Ethics and Information Technology* 21, 1 (2019), 59–73.
- [32] Deborah J Cohen, Sara R Keller, Gillian R Hayes, David A Dorr, Joan S Ash, and Dean F Sittig. 2015. Developing a model for understanding patient collection of observations of daily living: a qualitative meta-synthesis of the Project HealthDesign program. *Personal and ubiquitous computing* 19, 1 (2015), 91–102.
- [33] Francis S Collins and Harold Varmus. 2015. A new initiative on precision medicine. New England journal of medicine 372, 9 (2015), 793–795.
- [34] Eric Corbett and Christopher A Le Dantec. 2018. Exploring trust in digital civics. In Proceedings of the 2018 Designing Interactive Systems Conference. 9–20.
   [35] Edward L Deci and Richard M Ryan. 2010. Intrinsic motivation. The corsini
- [55] Edward L Deci and Richard M Ryan. 2010. Intrinsic motivation. The corsini encyclopedia of psychology (2010), 1–2.
- [36] Pranammya Dey, Joseph S Ross, Jessica D Ritchie, Nihar R Desai, Sanjeev P Bhavnani, and Harlan M Krumholz. 2017. Data sharing and cardiology: platforms and possibilities. *Journal of the American College of Cardiology* 70, 24 (2017), 3018–3025.
- [37] Daniel Di Matteo, Alexa Fine, Kathryn Fotinos, Jonathan Rose, and Martin Katzman. 2018. Patient willingness to consent to mobile phone data collection for mental health apps: structured questionnaire. *JMIR mental health* 5, 3 (2018), e56.
- [38] Ezgi Dogan, Christian Sander, Xenija Wagner, Ulrich Hegerl, and Elisabeth Kohls. 2017. Smartphone-based monitoring of objective and subjective data in affective disorders: where are we and where are we going? Systematic review. *Journal of medical Internet research* 19, 7 (2017), e262.
- [39] Jan Domaradzki and Jakub Pawlikowski. 2019. Public Attitudes toward Biobanking of Human Biological Material for Research Purposes: A Literature Review. International journal of environmental research and public health 16, 12 (2019),

NordiCHI '20, October 25-29, 2020, Tallinn, Estonia

2209.

- [40] Juan Pablo Domecq, Gabriela Prutsky, Tarig Elraiyah, Zhen Wang, Mohammed Nabhan, Nathan Shippee, Juan Pablo Brito, Kasey Boehmer, Rim Hasan, Belal Firwana, et al. 2014. Patient engagement in research: a systematic review. BMC health services research 14, 1 (2014), 89.
- [41] Laura B Dunn and Dilip V Jeste. 2001. Enhancing informed consent for research and treatment. *Neuropsychopharmacology* 24, 6 (2001), 595–607.
- [42] Daniel A Epstein, Monica Caraway, Chuck Johnston, An Ping, James Fogarty, and Sean A Munson. 2016. Beyond abandonment to next steps: understanding and designing for life after personal informatics tool use. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. ACM, 1109–1113.
- [43] Motahhare Eslami, Kristen Vaccaro, Min Kyung Lee, Amit Elazari Bar On, Eric Gilbert, and Karrie Karahalios. 2019. User Attitudes Towards Algorithmic Opacity and Transparency in Online Reviewing Platforms. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems (Glasgow, Scotland Uk) (CHI '19). ACM, New York, NY, USA, Article 494, 14 pages.
- [44] Evidation. 2019. Radically changing medicine. Retrieved May 2, 2020 from https://evidation.com/
- [45] Rowanne Fleck and Geraldine Fitzpatrick. 2010. Reflecting on reflection: framing a design landscape. In Proceedings of the 22nd Conference of the Computer-Human Interaction Special Interest Group of Australia on Computer-Human Interaction. ACM, 216–223.
- [46] Rachael Fleurence, Joe V Selby, Kara Odom-Walker, Gail Hunt, David Meltzer, Jean R Slutsky, and Clyde Yancy. 2013. How the Patient-Centered Outcomes Research Institute is engaging patients and others in shaping its research agenda. *Health Affairs* 32, 2 (2013), 393–400.
- [47] Enrique Garcia-Ceja, Michael Riegler, Tine Nordgreen, Petter Jakobsen, Ketil J Oedegaard, and Jim Tørresen. 2018. Mental health monitoring with multimodal sensing and machine learning: A survey. *Pervasive and Mobile Computing* (2018).
- [48] Suneela Garg, Nidhi Bhatnagar, and Navya Gangadharan. 2020. A Case for Participatory Disease Surveillance of the COVID-19 Pandemic in India. *JMIR Public Health and Surveillance* 6, 2 (2020), e18795.
- [49] Heather H Gatny and William G Axinn. 2012. Willingness to participate in research during pregnancy: race, experience, and motivation. *Field methods* 24, 2 (2012), 135–154.
- [50] Katrina AB Goddard, K Sabina Smith, Chuhe Chen, Carmit McMullen, and Cheryl Johnson. 2009. Biobank recruitment: motivations for nonparticipation. *Biopreservation and biobanking* 7, 2 (2009), 119–121.
- [51] Lisa Graham, Anthony Tang, and Carman Neustaedter. 2016. Help me help you: Shared reflection for personal data. In Proceedings of the 19th International Conference on Supporting Group Work. ACM, 99–109.
- [52] Franz Gravenhorst, Amir Muaremi, Jakob Bardram, Agnes Grünerbl, Oscar Mayora, Gabriel Wurzer, Mads Frost, Venet Osmani, Bert Arnrich, Paul Lukowicz, et al. 2015. Mobile phones as medical devices in mental disorder treatment: an overview. *Personal and Ubiquitous Computing* 19, 2 (2015), 335–353.
- [53] Bastian Greshake Tzovaras, Misha Angrist, Kevin Arvai, Mairi Dulaney, Vero Estrada-Galiñanes, Beau Gunderson, Tim Head, Dana Lewis, Oded Nov, Orit Shaer, et al. 2019. Open Humans: A platform for participant-centered research and personal data exploration. *GigaScience* 8, 6 (2019), giz076.
- [54] Mark Hamer, Ding Ding, Josephine Chau, Mitch J Duncan, and Emmanuel Stamatakis. 2020. Association between TV viewing and heart disease mortality: observational study using negative control outcome. J Epidemiol Community Health (2020).
- [55] Gabriella M Harari, Nicholas D Lane, Rui Wang, Benjamin S Crosier, Andrew T Campbell, and Samuel D Gosling. 2016. Using smartphones to collect behavioral data in psychological science: Opportunities, practical considerations, and challenges. *Perspectives on Psychological Science* 11, 6 (2016), 838–854.
- [56] Gabriella M Harari, Sandrine R Müller, Varun Mishra, Rui Wang, Andrew T Campbell, Peter J Rentfrow, and Samuel D Gosling. 2017. An evaluation of students' interest in and compliance with self-tracking methods: Recommendations for incentives based on three smartphone sensing studies. *Social Psychological and Personality Science* 8, 5 (2017), 479–492.
- [57] Jean Hardy, Tiffany C Veinot, Xiang Yan, Veronica J Berrocal, Philippa Clarke, Robert Goodspeed, Iris N Gomez-Lopez, Daniel Romero, and VG Vinod Vydiswaran. 2018. User acceptance of location-tracking technologies in health research: implications for study design and data quality. *Journal of biomedical informatics* 79 (2018), 7–19.
- [58] Luke Hutton, Blaine A Price, Ryan Kelly, Ciaran McCormick, Arosha K Bandara, Tally Hatzakis, Maureen Meadows, and Bashar Nuseibeh. 2018. Assessing the privacy of mhealth apps for self-tracking: heuristic evaluation approach. *JMIR mHealth and uHealth* 6, 10 (2018), e185.
- [59] Marcello Ienca and Effy Vayena. 2020. On the responsible use of digital data to tackle the COVID-19 pandemic. *Nature Medicine* (2020), 1–2.
- [60] Anton Ivanov, Raj Sharman, and H Raghav Rao. 2015. Exploring factors impacting sharing health-tracking records. *Health Policy and Technology* 4, 3 (2015), 263–276.
- [61] Maia L Jacobs, James Clawson, and Elizabeth D Mynatt. 2015. Comparing health information sharing preferences of cancer patients, doctors, and navigators.

NordiCHI '20, October 25-29, 2020, Tallinn, Estonia

In Proceedings of the 18th ACM Conference on Computer Supported Cooperative Work & Social Computing. ACM, 808–818.

- [62] Adam N Joinson, Ulf-Dietrich Reips, Tom Buchanan, and Carina B Paine Schofield. 2010. Privacy, trust, and self-disclosure online. *Human-Computer Interaction* 25, 1 (2010), 1-24.
- [63] Iris A Junglas, Norman A Johnson, and Christiane Spitzmüller. 2008. Personality traits and concern for privacy: an empirical study in the context of locationbased services. *European Journal of Information Systems* 17, 4 (2008), 387–402.
- [64] Jane Kaye, Edgar A Whitley, David Lund, Michael Morrison, Harriet Teare, and Karen Melham. 2015. Dynamic consent: a patient interface for twenty-first century research networks. *European Journal of Human Genetics* 23, 2 (2015), 141
- [65] Christina Kelley, Bongshin Lee, and Lauren Wilcox. 2017. Self-tracking for mental wellness: understanding expert perspectives and student experiences. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems. ACM, 629–641.
- [66] Elisabeth Kersten-van Dijk and Wijnand A IJsselsteijn. 2016. Design Beyond the Numbers: Sharing, Comparing, Storytelling and the Need for a Quantified Us. IxD&A 29 (2016), 121–135.
- [67] Katherine K Kim, Jill G Joseph, and Lucila Ohno-Machado. 2015. Comparison of consumers' views on electronic data sharing for healthcare and research. *Journal of the American Medical Informatics Association* 22, 4 (2015), 821–830.
- [68] Yoojung Kim, Eunyoung Heo, Hyunjeong Lee, Sookyoung Ji, Jueun Choi, Jeong-Whun Kim, Joongseek Lee, and Sooyoung Yoo. 2017. Prescribing 10,000 steps like aspirin: designing a novel interface for data-driven medical consultations. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems. ACM, 5787–5799.
- [69] Yoojung Kim, Sookyoung Ji, Hyunjeong Lee, Jeong-Whun Kim, Sooyoung Yoo, and Joongseek Lee. 2016. My doctor is keeping an eye on mel: Exploring the clinical applicability of a mobile food logger. In *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*. ACM, 5620–5631.
  [70] Jennifer King. 2019. "Becoming Part of Something Bigger" Direct to Consumer
- [70] Jennifer King. 2019. "Becoming Part of Something Bigger" Direct to Consumer Genetic Testing, Privacy, and Personal Disclosure. Proceedings of the ACM on Human-Computer Interaction 3, CSCW (2019), 1–33.
- [71] William R King and Jun He. 2006. A meta-analysis of the technology acceptance model. Information & management 43, 6 (2006), 740–755.
- [72] René F Kizilcec. 2016. How much information?: Effects of transparency on trust in an algorithmic interface. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. ACM, 2390–2395.
- [73] Predrag Klasnja, Sunny Consolvo, Tanzeem Choudhury, Richard Beckwith, and Jeffrey Hightower. 2009. Exploring privacy concerns about personal sensing. In International Conference on Pervasive Computing. Springer, 176–183.
- [74] Patty Kostkova, Helen Brewer, Simon de Lusignan, Edward Fottrell, Ben Goldacre, Graham Hart, Phil Koczan, Peter Knight, Corinne Marsolier, Rachel A McKendry, et al. 2016. Who owns the data? Open data for healthcare. *Frontiers* in public health 4 (2016), 7.
- [75] Stephanie A Kraft, Mildred K Cho, Katherine Gillespie, Meghan Halley, Nina Varsava, Kelly E Ormond, Harold S Luft, Benjamin S Wilfond, and Sandra Soo-Jin Lee. 2018. Beyond consent: Building trusting relationships with diverse populations in precision medicine research. *The American Journal of Bioethics* 18, 4 (2018), 3–20.
- [76] Nicola Lacetera, Mario Macis, and Robert Slonim. 2013. Economic rewards to motivate blood donations. *Science* 340, 6135 (2013), 927–928.
- [77] Mark R Leary and Robin M Kowalski. 1990. Impression management: A literature review and two-component model. *Psychological bulletin* 107, 1 (1990), 34.
  [78] Amy A Lemke, Wendy A Wolf, Jennifer Hebert-Beirne, and Maureen E Smith.
- [78] Amy A Lemke, Wendy A Wolf, Jennifer Hebert-Beirne, and Maureen E Smith. 2010. Public and biobank participant attitudes toward genetic research participation and data sharing. *Public Health Genomics* 13, 6 (2010), 368–377.
- [79] Pedro Giovanni Leon, Blase Ur, Yang Wang, Manya Sleeper, Rebecca Balebako, Richard Shay, Lujo Bauer, Mihai Christodorescu, and Lorrie Faith Cranor. 2013. What matters to users?: factors that affect users' willingness to share information with online advertisers. In Proceedings of the ninth symposium on usable privacy and security. ACM, 7.
- [80] Chen Li, Xiangdong Xu, Guanghua Zhou, Kai He, Tianliang Qi, Wei Zhang, Feng Tian, Qinghua Zheng, and Jianping Hu. 2019. Implementation of National Health Informatization in China: Survey About the Status Quo. *JMIR medical informatics* 7, 1 (2019), e12238.
- [81] Ian Li, Anind Dey, and Jodi Forlizzi. 2010. A stage-based model of personal informatics systems. In Proceedings of the SIGCHI conference on human factors in computing systems. ACM, 557–566.
- [82] Melissa LoPresti, Geoff Appelboom, Olivier Bruyère, Jean-Yves Reginster, Eric Klug, and E Sander Connolly Jr. 2014. Patient engagement in clinical research through mobile technology. *Clinical Practice* 11, 6 (2014), 549.
- [83] Pamela J Ludford, Reid Priedhorsky, Ken Reily, and Loren Terveen. 2007. Capturing, sharing, and using local place information. In Proceedings of the SIGCHI conference on Human factors in computing systems. ACM, 1235–1244.
- [84] Ewa Luger, Stuart Moran, and Tom Rodden. 2013. Consent for all: revealing the hidden complexity of terms and conditions. In Proceedings of the SIGCHI

conference on Human factors in computing systems. ACM, 2687-2696.

- [85] Maulik D Majmudar, Lina Avancini Colucci, and Adam B Landman. 2015. The quantified patient of the future: Opportunities and challenges. In *Healthcare*, Vol. 3. Elsevier, 153–156.
- [86] Miguel Malheiros, Sacha Brostoff, Charlene Jennett, and M Angela Sasse. 2013. Would you sell your mother's data? personal data disclosure in a simulated credit card application. In *The Economics of Information Security and Privacy*. Springer, 237–261.
- [87] Meethu Malu and Leah Findlater. 2017. Sharing automatically tracked activity data: implications for therapists and people with mobility impairments. In Proceedings of the 11th EAI International Conference on Pervasive Computing Technologies for Healthcare. ACM, 136-145.
- [88] CJ Mann. 2003. Observational research methods. Research design II: cohort, cross sectional, and case-control studies. *Emergency medicine journal* 20, 1 (2003), 54–60.
- [89] Kate E Mason, Neil Pearce, and Steven Cummins. 2018. Associations between fast food and physical activity environments and adiposity in mid-life: crosssectional, observational evidence from UK Biobank. *The Lancet Public Health* 3, 1 (2018), e24–e33.
- [90] Amy Maxmen. 2017. Google spin-off deploys wearable electronics for huge health study. Nature News 547, 7661 (2017), 13.
- [91] Michelle L Mazurek, JP Arsenault, Joanna Bresee, Nitin Gupta, Iulia Ion, Christina Johns, Daniel Lee, Yuan Liang, Jenny Olsen, Brandon Salmon, et al. 2010. Access control for home data sharing: Attitudes, needs and practices. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. ACM, 645–654.
- [92] Jacob McPadden, Thomas JS Durant, Dustin R Bunch, Andreas Coppi, Nathaniel Price, Kris Rodgerson, Charles J Torre Jr, William Byron, Allen L Hsiao, Harlan M Krumholz, et al. 2019. Health Care and Precision Medicine Research: Analysis of a Scalable Data Science Platform. *Journal of medical Internet research* 21, 4 (2019), e13043.
- [93] Helena M Mentis, Anita Komlodi, Katrina Schrader, Michael Phipps, Ann Gruber-Baldini, Karen Yarbrough, and Lisa Shulman. 2017. Crafting a view of selftracking data in the clinical visit. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems. ACM, 5800–5812.
- [94] Nick Merrill, John Chuang, and Coye Cheshire. 2019. Sensing is Believing: What People Think Biosensors Can Reveal About Thoughts and Feelings. In Proceedings of the 2019 on Designing Interactive Systems Conference. ACM, 413– 420.
- [95] Anna Middleton. 2018. Society and personal genome data. Human molecular genetics 27, R1 (2018), R8–R13.
- [96] Anna Middleton, Richard Milne, Adrian Thorogood, Erika Kleiderman, Emilia Niemiec, Barbara Prainsack, Lauren Farley, Paul Bevan, Claire Steed, James Smith, et al. 2019. Attitudes of publics who are unwilling to donate DNA data for research. *European journal of medical genetics* 62, 5 (2019), 316–323.
- [97] Alistair Morrison, Donald McMillan, and Matthew Chalmers. 2014. Improving consent in large scale mobile HCI through personalised representations of data. In Proceedings of the 8th Nordic conference on human-computer interaction: Fun, fast, foundational. ACM, 471–480.
- [98] Daphne A Muller and Pierre Lévy. 2019. A design approach towards affording the trend of privacy. In Proceedings of the 2019 on Designing Interactive Systems Conference. 435–446.
- [99] Patrick Murmann and Simone Fischer-Hübner. 2017. Usable transparency enhancing tools: A literature review.
- [100] Elizabeth L Murnane, Tara G Walker, Beck Tench, Stephen Voida, and Jaime Snyder. 2018. Personal informatics in interpersonal contexts: towards the design of technology that supports the social ecologies of long-term mental health management. *Proceedings of the ACM on Human-Computer Interaction* 2, CSCW (2018), 127.
- [101] Camille Nadal, Corina Sas, and Gavin Doherty. 2020. Technology acceptance in mHealth: a scoping review of definitions, models and measurement. *Journal of Medical Internet Research* (2020).
- [102] Camille Nebeker, John Harlow, Rebeca Espinoza Giacinto, Rubi Orozco-Linares, Cinnamon S Bloss, and Nadir Weibel. 2017. Ethical and regulatory challenges of research using pervasive sensing and other emerging technologies: IRB perspectives. AJOB empirical bioethics 8, 4 (2017), 266–276.
- [103] Onicio Leal Neto, Oswaldo Cruz, Jones Albuquerque, Mariana Nacarato de Sousa, Mark Smolinski, Eduarda Ángela Pessoa Cesse, Marlo Libel, and Wayner Vieira de Souza. 2020. Participatory Surveillance Based on Crowdsourcing During the Rio 2016 Olympic Games Using the Guardians of Health Platform: Descriptive Study. JMIR Public Health and Surveillance 6, 2 (2020), e16119.
- [104] Helen Nissenbaum. 2011. A contextual approach to privacy online. Daedalus 140, 4 (2011), 32–48.
- [105] Patricia A Norberg, Daniel R Horne, and David A Horne. 2007. The privacy paradox: Personal information disclosure intentions versus behaviors. *Journal* of consumer affairs 41, 1 (2007), 100–126.

Futures for Health Research Data Platforms From the Participants' Perspectives

- [106] Leysan Nurgalieva, Alisa Frik, Francesco Ceschel, Serge Egelman, and Maurizio Marchese. 2019. Information Design in An Aged Care Context: Views of Older Adults on Information Sharing in a Care Triad. In Proceedings of the 13th EAI International Conference on Pervasive Computing Technologies for Healthcare. ACM, 101–110.
- [107] National Institutes of Health (NIH). 2020. All of Us Research Program Protocol. Retrieved May 2, 2020 from https://allofus.nih.gov/about/all-us-researchprogram-protocol
- [108] National Institutes of Health (NIH). 2020. Precision Medicine Initiative: Privacy and Trust Principles. Retrieved May 2, 2020 from https://allofus.nih.gov/protecting-data-and-privacy/precision-medicineinitiative-privacy-and-trust-principles
- [109] Aisling Ann O'Kane and Helena Mentis. 2012. Sharing medical data vs. health knowledge in chronic illness care. In CHI'12 Extended Abstracts on Human Factors in Computing Systems. ACM, 2417–2422.
- [110] Kirsten Ostherr, Svetlana Borodina, Rachel Conrad Bracken, Charles Lotterman, Eliot Storer, and Brandon Williams. 2017. Trust and privacy in the context of user-generated health data. *Big Data & Society* 4, 1 (2017), 2053951717704673.
- [111] Leysia Palen and Paul Dourish. 2003. Unpacking privacy for a networked world. In Proceedings of the SIGCHI conference on Human factors in computing systems. ACM, 129–136.
- [112] Michael J Pencina, Darcy M Louzao, Brian J McCourt, Monique R Adams, Rehbar H Tayyabkhan, Peter Ronco, and Eric D Peterson. 2016. Supporting open access to clinical trial data for researchers: The Duke Clinical Research Institute– Bristol-Myers Squibb Supporting Open Access to Researchers Initiative. American heart journal 172 (2016), 64–69.
- [113] Justin Petelka, Lucy Van Kleunen, Liam Albright, Elizabeth Murnane, Stephen Voida, and Jaime Snyder. 2020. Being (In) Visible: Privacy, Transparency, and Disclosure in the Self-Management of Bipolar Disorder. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–14.
- [114] Carolyn Petersen. 2016. The future of patient engagement in the governance of shared data. eGEMs 4, 2 (2016).
- [115] Carolyn Petersen. 2018. Through Patients' Eyes: Regulation, Technology, Privacy, and the Future. Yearbook of medical informatics 27, 01 (2018), 010–015.
- [116] James Pierce, Sarah Fox, Nick Merrill, Richmond Wong, and Carl DiSalvo. 2018. An Interface Without A User: An Exploratory Design Study of Online Privacy Policies and Digital Legalese. In Proceedings of the 2018 Designing Interactive Systems Conference. 1345–1358.
- [117] Covid-19 Data Portal. 2020. Accelerating research through data sharing. Retrieved May 2, 2020 from https://www.covid19dataportal.org/
- [118] Abhishek Pratap, Ryan Allred, Jaden Duffy, Donovan Rivera, Heather Sophia Lee, Brenna N Renn, and Patricia A Areán. 2019. Contemporary Views of Research Participant Willingness to Participate and Share Digital Data in Biomedical Research. JAMA network open 2, 11 (2019).
- [119] Mika Raento and Antti Oulasvirta. 2008. Designing for privacy and selfpresentation in social awareness. *Personal and Ubiquitous Computing* 12, 7 (2008), 527-542.
- [120] Fiona Riordan, Chrysanthi Papoutsi, Julie E Reed, Cicely Marston, Derek Bell, and Azeem Majeed. 2015. Patient and public attitudes towards informed consent models and levels of awareness of Electronic Health Records in the UK. *International journal of medical informatics* 84, 4 (2015), 237–247.
   [121] John Rooksby, Alistair Morrison, and Dave Murray-Rust. 2019. Student Per-
- [121] John Rooksby, Alistair Morrison, and Dave Murray-Rust. 2019. Student Perspectives on Digital Phenotyping: The Acceptability of Using Smartphone Data to Assess Mental Health. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. ACM, 425.
- [122] Dana Rotman, Jenny Preece, Jen Hammock, Kezee Procita, Derek Hansen, Cynthia Parr, Darcy Lewis, and David Jacobs. 2012. Dynamic changes in motivation in collaborative citizen-science projects. In Proceedings of the ACM 2012 conference on computer supported cooperative work. ACM, 217–226.
- [123] Anna Rudnicka, Anna L Cox, and Sandy JJ Gould. 2019. Why Do You Need This?: Selective Disclosure of Data Among Citizen Scientists. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. ACM, 392.
- [124] Sundeep Sahay and Arunima Mukherjee. 2020. Where Is All Our Health Data Going? Economic & Political Weekly 55, 1 (2020), 47.
   [125] Pamela L Sankar and Lisa S Parker. 2017. The Precision Medicine Initiative's
- [125] Pamela L Sankar and Lisa S Parker. 2017. The Precision Medicine Initiative's All of Us Research Program: an agenda for research on its ethical, legal, and social issues. *Genetics in Medicine* 19, 7 (2017), 743.
- [126] Shruti Sannon and Dan Cosley. 2019. Privacy, Power, and Invisible Labor on Amazon Mechanical Turk. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. ACM, 282.
- [127] Kate EA Saunders, Amy C Bilderbeck, Priyanka Panchal, Lauren Z Atkinson, JR Geddes, and Guy M Goodwin. 2017. Experiences of remote mood and activity monitoring in bipolar disorder: a qualitative study. *European Psychiatry* 41 (2017), 115–121.
- [128] Morten Schmidt, Lars Pedersen, and Henrik Toft Sørensen. 2014. The Danish Civil Registration System as a tool in epidemiology. *European journal of epidemiology* 29, 8 (2014), 541–549.

NordiCHI '20, October 25-29, 2020, Tallinn, Estonia

- [129] Siobhan Sharkey, Ray Jones, Janet Smithson, Elaine Hewis, Tobit Emmens, Tamsin Ford, and Christabel Owens. 2011. Ethical practice in internet research involving vulnerable people: lessons from a self-harm discussion forum study (SharpTalk). *Journal of medical ethics* 37, 12 (2011), 752–758.
- [130] Nathan D Shippee, Juan Pablo Domecq Garces, Gabriela J Prutsky Lopez, Zhen Wang, Tarig A Elraiyah, Mohammed Nabhan, Juan P Brito, Kasey Boehmer, Rim Hasan, Belal Firwana, et al. 2015. Patient and service user engagement in research: a systematic review and synthesized framework. *Health Expectations* 18, 5 (2015), 1151–1166.
- [131] Fatemeh Shirazi and Melanie Volkamer. 2014. What deters Jane from preventing identification and tracking on the Web?. In *Proceedings of the 13th Workshop on Privacy in the Electronic Society*. ACM, 107–116.
  [132] Michael Siegrist. 2000. The influence of trust and perceptions of risks and
- [132] Michael Siegrist. 2000. The influence of trust and perceptions of risks and benefits on the acceptance of gene technology. *Risk analysis* 20, 2 (2000), 195– 204.
- [133] Christian M Simon, Jamie L'heureux, Jeffrey C Murray, Patricia Winokur, George Weiner, Elizabeth Newbury, Laura Shinkunas, and Bridget Zimmerman. 2011. Active choice but not too active: public perspectives on biobank consent models. *Genetics in Medicine* 13, 9 (2011), 821.
- [134] Daniel J Solove. 2005. A taxonomy of privacy. U. Pa. L. Rev. 154 (2005), 477.
- [135] Jae W Song and Kevin C Chung. 2010. Observational studies: cohort and casecontrol studies. *Plastic and reconstructive surgery* 126, 6 (2010), 2234.
   [136] Cathie Sudlow, John Gallacher, Naomi Allen, Valerie Beral, Paul Burton, John
- [136] Cathie Sudlow, John Gallacher, Naomi Allen, Valerie Beral, Paul Burton, John Danesh, Paul Downey, Paul Elliott, Jane Green, Martin Landray, et al. 2015. UK biobank: an open access resource for identifying the causes of a wide range of complex diseases of middle and old age. *PLoS medicine* 12, 3 (2015), e1001779.
- [137] Joshua Tan, Mahmood Sharif, Sruti Bhagavatula, Matthias Beckerle, Michelle L Mazurek, and Lujo Bauer. 2018. Comparing Hypothetical and Realistic Privacy Valuations. In Proceedings of the 2018 Workshop on Privacy in the Electronic Society. ACM, 168–182.
- [138] Sharon F Terry and Patrick F Terry. 2011. Power to the people: participant ownership of clinical trial data. *Science Translational Medicine* 3, 69 (2011), 69cm3–69cm3.
- [139] Susan Brown Trinidad, Stephanie M Fullerton, Julie M Bares, Gail P Jarvik, Eric B Larson, and Wylie Burke. 2012. Informed consent in genome-scale research: what do prospective participants think? *AJOB primary research* 3, 3 (2012), 3–11.
- [140] Yonatan Vaizman, Katherine Ellis, Gert Lanckriet, and Nadir Weibel. 2018. Extrasensory app: Data collection in-the-wild with rich user interface to self-report behavior. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. ACM, 554.
- [141] Petar Veličković, Nicholas D Lane, Sourav Bhattacharya, Angela Chieh, Otmane Bellahsen, and Matthieu Vegreville. 2017. Scaling health analytics to millions without compromising privacy using deep distributed behavior models. In Proceedings of the 11th EAI International Conference on Pervasive Computing Technologies for Healthcare. ACM, 92–100.
  [142] Giovanna N. Vilaza and Jakob E. Bardram. 2019. Sharing Access to Behavioural
- [142] Giovanna N. Vilaza and Jakob E. Bardram. 2019. Sharing Access to Behavioural and Personal Health Data: Designers' Perspectives on Opportunities and Barriers. In Proceedings of the 13th EAI International Conference on Pervasive Computing Technologies for Healthcare (Trento, Italy) (PervasiveHealth'19). ACM, New York, NY, USA, 346–350.
- [143] Francesco Vitale, William Odom, and Joanna McGrenere. 2019. Keeping and Discarding Personal Data: Exploring a Design Space. In Proceedings of the 2019 on Designing Interactive Systems Conference. 1463–1477.
- [144] Paul Voigt and Axel Von dem Bussche. 2017. The EU General Data Protection Regulation (GDPR). A Practical Guide, 1st Ed., Cham: Springer International Publishing (2017).
- [145] Maiko Watanabe, Yusuke Inoue, Chiungfang Chang, Hyunsoo Hong, Izumi Kobayashi, Satoshi Suzuki, and Kaori Muto. 2011. For what am I participating? The need for communication after receiving consent from biobanking project participants: experience in Japan. Journal of human genetics 56, 5 (2011), 358.
- [146] Elissa R Weitzman, Skyler Kelemen, Liljana Kaci, and Kenneth D Mandí. 2012. Willingness to share personal health record data for care improvement and public health: a survey of experienced personal health record users. BMC medical informatics and decision making 12, 1 (2012), 39.
- [147] Peter West, Richard Giordano, Max Van Kleek, and Nigel Shadbolt. 2016. The quantified patient in the doctor's office: Challenges & opportunities. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. ACM, 3066–3078.
- [148] Richard Whiddett, Inga Hunter, Judith Engelbrecht, and Jocelyn Handy. 2006. Patients' attitudes towards sharing their health information. *International journal of medical informatics* 75, 7 (2006), 530–541.
- [149] Paul Wicks, Michael Massagli, Jeana Frost, Catherine Brownstein, Sally Okun, Timothy Vaughan, Richard Bradley, and James Heywood. 2010. Sharing health data for better outcomes on PatientsLikeMe. *Journal of medical Internet research* 12, 2 (2010), e19.
- [150] Andrea Wiggins and John Wilbanks. 2019. The rise of citizen science in health and biomedical research. The American Journal of Bioethics 19, 8 (2019), 3–14.

NordiCHI '20, October 25-29, 2020, Tallinn, Estonia

- [151] Niels Wouters, Ryan Kelly, Eduardo Velloso, Katrin Wolf, Hasan Shahid Ferdous, Joshua Newn, Zaher Joukhadar, and Frank Vetere. 2019. Biometric Mirror: Exploring Values and Attitudes towards Facial Analysis and Automated Decision-Making. In *Conference on Designing Interactive Systems.*, Vol. 1145.
- [152] Martina Ziefle, Julian Halbey, and Sylvia Kowalewski. 2016. Users' Willingness to Share Data on the Internet: Perceived Benefits and Caveats. In *IoTBD*. 255– 265.

2.3 Public Attitudes to Digital Health Research Repositories: Cross-sectional International Survey

## Full Reference:

Giovanna Nunes Vilaza, David Coyle and Jakob Bardram. 2021. *Journal of Internet Research 23.10, e31294*. https://doi.org/10.2196/31294

## Copyrights:

Open-access article. Republished under the Creative Commons license. Copyright held by the authors.

**Original Paper** 

## Public Attitudes to Digital Health Research Repositories: Cross-sectional International Survey

Giovanna Nunes Vilaza<sup>1</sup>, BSc, MSc; David Coyle<sup>2</sup>, BSc, MSc, PhD; Jakob Eyvind Bardram<sup>1</sup>, BSc, MSc, PhD

<sup>1</sup>Department of Health Technology, Technical University of Denmark, Kongens Lyngby, Denmark <sup>2</sup>School of Computer Science, University College Dublin, Dublin, Ireland

### **Corresponding Author:**

Giovanna Nunes Vilaza, BSc, MSc Department of Health Technology Technical University of Denmark Ørsteds Plads 345B Kongens Lyngby, 2800 Denmark Phone: 45 45253724 Email: gnvi@dtu.dk

## Abstract

**Background:** Digital health research repositories propose sharing longitudinal streams of health records and personal sensing data between multiple projects and researchers. Motivated by the prospect of personalizing patient care (precision medicine), these initiatives demand broad public acceptance and large numbers of data contributors, both of which are challenging.

**Objective:** This study investigates public attitudes toward possibly contributing to digital health research repositories to identify factors for their acceptance and to inform future developments.

**Methods:** A cross-sectional online survey was conducted from March 2020 to December 2020. Because of the funded project scope and a multicenter collaboration, study recruitment targeted young adults in Denmark and Brazil, allowing an analysis of the differences between 2 very contrasting national contexts. Through closed-ended questions, the survey examined participants' willingness to share different data types, data access preferences, reasons for concern, and motivations to contribute. The survey also collected information about participants' demographics, level of interest in health topics, previous participation in health research, awareness of examples of existing research data repositories, and current attitudes about digital health research repositories. Data analysis consisted of descriptive frequency measures and statistical inferences (bivariate associations and logistic regressions).

**Results:** The sample comprises 1017 respondents living in Brazil (1017/1600, 63.56%) and 583 in Denmark (583/1600, 36.44%). The demographics do not differ substantially between participants of these countries. The majority is aged between 18 and 27 years (933/1600, 58.31%), is highly educated (992/1600, 62.00%), uses smartphones (1562/1600, 97.63%), and is in good health (1407/1600, 87.94%). The analysis shows a vast majority were very motivated by helping future patients (1366/1600, 85.38%) and researchers (1253/1600, 78.31%), yet very concerned about unethical projects (1219/1600, 76.19%), profit making without consent (1096/1600, 68.50%), and cyberattacks (1055/1600, 65.94%). Participants' willingness to share data is lower when sharing personal sensing data, such as the content of calls and texts (1206/1600, 75.38%), in contrast to more traditional health research information. Only 13.44% (215/1600) find it desirable to grant data access to private companies, and most would like to stay informed about which projects use their data (1334/1600, 83.38%) and control future data access (1181/1600, 73.81%). Findings indicate that favorable attitudes toward digital health research repositories are related to a personal interest in health topics (odds ratio [OR] 1.49, 95% CI 1.10-2.02; *P*=.01), previous participation in health research studies (OR 1.70, 95% CI 1.24-2.35; *P*=.001), and awareness of examples of research repositories (OR 2.78, 95% CI 1.83-4.38; *P*<.001).

**Conclusions:** This study reveals essential factors for acceptance and willingness to share personal data with digital health research repositories. Implications include the importance of being more transparent about the goals and beneficiaries of research projects using and re-using data from repositories, providing participants with greater autonomy for choosing who gets access to which parts of their data, and raising public awareness of the benefits of data sharing for research. In addition, future developments should engage with and reduce risks for those unwilling to participate.

(J Med Internet Res 2021;23(10):e31294) doi: 10.2196/31294



RenderX

## **KEYWORDS**

digital medicine; health informatics; health data repositories; personal sensing; technology acceptance; willingness to share data; human-centered computing; ethics

## Introduction

## Background

Health research is increasingly adopting digital technologies to accelerate scientific discovery, as digital data sources increase scalability and predictive power for algorithmic inferences [1-3]. Novel data collection techniques include wearables and smartphone sensors to extract participants' behavioral features passively [4]. Records of calls and texts can flag social activity fluctuations; location tracking can reveal mobility patterns; heart rate measures can indicate sleep quality [5-7]. Ecological momentary assessments further complement such passive indicators by sampling individuals' health status in real time through questionnaires [8,9]. The motivation for considering pervasive and digital sources of health and behavioral information is related to the possibility of closely observing research patients' daily lives [10,11].

Intending to personalize future patient care, researchers search for scientific evidence by analyzing longitudinal streams of personal sensing data from large segments of the population [12,13]. Because of this expansion of personal sensing in the health domain, digital health research repositories are gaining momentum. An ambitious concept similar to biobanks [14], a digital health research repository allows multiple projects and researchers to share access to personal data streams beyond DNA and biosamples [15-17]. Although still in their initial steps, initiatives with this goal in mind include nationwide programs [18,19], university-led projects [20,21], and community-driven data platforms [22-24].

Despite promising benefits, barriers to public acceptance can hinder the successful implementation of digital health research repositories [25,26]. Without a diverse range of participants agreeing to contribute with their personal health data, repositories cannot accomplish their ambitious goal of providing reliable evidence for personalized medicine to the broad population [27]. Furthermore, a potential lack of acceptance is aggravated by ethical debates questioning which rights individuals should have following contribution of their data [28-30], especially if data are shared beyond a specific project's scope. Given such challenges, previous research has emphasized that in contrast to most current initiatives, which mainly cater to researchers' needs, health data repositories should attend more to participants' preferences to identify enablers for participation [31,32].

## **Previous Studies**

Previous studies have investigated public attitudes toward biobanks [33-36] and digital health data [37,38] separately. Other past studies have examined motivations to contribute [39], privacy concerns [40], and access control preferences [41] for data sharing within health research in general, yet these studies consider only a few variables simultaneously and rarely inquire about the magnitude of specific attitudes [42,43]. To

```
https://www.jmir.org/2021/10/e31294
```

the best of our knowledge, published quantitative studies have not thoroughly examined how different factors can affect attitude and willingness to share in digital health research repositories' timely and emerging context. Thus, it remains unclear how the public perceives the risks and benefits of shared access of multiple sources of behavioral and health indicators, including digital sensing, for research repositories.

#### Study Goal

Given this research gap, an online cross-sectional survey was conducted examining public attitudes to research repositories storing health information, biosamples, personal sensing, and behavioral data. This survey study aims to identify implications for future developments by consulting those whose personal data are to be shared for research. The study took place in Denmark, where the project is funded, and the principal investigators are based. Furthermore, with the goal of investigating the potential contrast between 2 very different historical, social, and cultural contexts, we contacted a research group in Brazil to establish a partnership and conduct the study with a sample of Brazilian residents. This decision allowed a cross-country analysis that illuminated similarities and divergences between 2 very disparate contexts. The results contribute to substantial empirical evidence about enablers and barriers for participants' acceptance and discussions on how community engagement, technology design, and policymaking can lead to a stronger participant-centric development in this field.

## Methods

## Population, Sample, and Recruitment

Denmark and Brazil are very different in terms of population, geography, economy, and culture. Denmark is a small country in area and population (5.8 million inhabitants), with a robust economy and a strong focus on social welfare, which is evident given the country's investments in education, research, and health care. By contrast, Brazil, the fifth largest country globally (208 million inhabitants), has a diversified economy, rich biodiversity, and industrial potential but at the same is characterized by an unequal society. Most of the population still lacks access to high-quality education and health services given the vast disparities of wealth distribution across the country. These differences make the population of these 2 countries very contrasting.

This survey study was part of an academic consortium project, publicly funded, to develop a digital health research repository for youth mental health, in turn defining our main target population (young adults), but without excluding the possibility of collecting data from other, older groups, if those participants would be interested in the study. As the project investigators are in Denmark, participants were first recruited among young Danish residents. Later, to enable the comparison of findings with a divergent historical, cultural, and social context, we

```
XSL•FO
```

sought to form a partnership with clinical researchers at a university in Northeast Brazil, the Universidade Federal do Rio Grande do Norte (UFRN). Therefore, recruitment efforts were also made to collect data from a sample of young adults in Brazil, but without being restrictive over the age groups that could respond. The online survey was deployed using LimeSurvey and hosted on a server at the Technical University of Denmark (DTU). The survey link was distributed across several channels in an extensive recruitment process that started on March 9, 2020, and ended on December 9, 2020. The link was made available in forums and newsletters at university portals, emailing lists, social networking groups, online chat platforms, and unpaid posts on Twitter. Recruitment in person happened once during an event hosted at DTU (prior to the surge of COVID-19). Participants were compensated with a cup of coffee at this event. Besides this one-time event, no other compensation was given to respondents to avoid providing incentives for repeated participation. Given the distribution of the survey in multiple channels without access restriction, a considerable number of responses came from participants beyond the target population (older than 27 years old). The research team decided not to exclude data from these respondents belonging to age groups above 27 years from analysis; instead, the data collected enabled another dimension for comparison (age). The sample, therefore, includes participants from all age groups residing in Brazil and Denmark.

## **Ethical and Legal Compliance**

Following local jurisdictions, this survey study received ethical approval from the Institutional Review Board of the partner university in Northeast Brazil and was exempt from ethical approval in Denmark. As established by the European General Data Protection Regulation (GDPR), the first page of the survey included information about the study's purpose, which data were collected, measures to anonymity and confidentiality, and data handling processes. Participants were asked to provide their consent after reading this information and confirming that they were older than 18 years. Besides the consent question, none of the questions were mandatory, following standard ethical conduct principles for online research. In addition, the survey was anonymous (IP address and identifiable information were not collected).

## **Questionnaire Design**

## Overview

The instrument development was based on (1) several previous surveys and focus groups about public acceptance of biobanks, electronic health records, and clinical trial repositories [43-53]; (2) a previous qualitative study about enablers and barriers for participation in digital health research repositories [32]; and (3) the input from the research team, clinicians, statisticians, and participants of the target population (young adults). The instrument went through several iterations until the final version, which is the one available in Multimedia Appendix 1. First, questions were designed in English and this version was used to ask for the initial rounds of feedback from both experts and targeted participants. After each round of feedback, the questionnaire was incrementally modified. Once a final English version was agreed upon by the research team, the questionnaire

```
https://www.jmir.org/2021/10/e31294
```

was professionally translated to the official language of each country (Portuguese and Danish). The translations were then verified by native speakers from the research team (GV and JB) to ensure content validity. Using the translated versions and the original in English, pilot tests were conducted by the research team with small convenient samples of 5 young adults in Denmark (in person) and in Brazil (remotely). These pilot tests consisted of asking participants to fill the survey and provide feedback on the readability of the questions, comprehension of the vocabulary used for the answer options, navigation of the interface, and time taken to complete all questions. Participants unanimously expressed that the instrument was easy to use and understand and completion time was reasonable. All members of the research team then approved the distribution of the final versions of the questionnaire (in Portuguese, English, and Danish). The complete questionnaire is available in Multimedia Appendix 1. A summary of the survey questions and their rationale are described next. This study is the first to combine such a set of diverse factors to the best of our knowledge.

## Demographics and Socioeconomics Questions

This first group of questions inquired about participant age group, gender, education level, country of residence, and usage and ownership of digital devices (computers, smartphones, smartwatches, smart home assistants, and tablets). Some individuals may have a gender that is neither male nor female. They may identify as both male and female at one time, different genders at different times, no gender at all, or dispute the very idea of only 2 genders. Therefore, the term "nonbinary" in this study refers to gender identities outside of the gender binary (male or female). The survey conducted in Brazil also contained 2 additional questions: race and household income (based on minimum salary). Minimum salary has been defined as the minimum amount of remuneration that an employer is required to pay for the work performed during a given period (usually per month), which cannot be reduced by collective agreement or an individual contract. In Brazil, at the time of the study, the minimum salary per month was 1040 Brazilian Reais (approximately US \$188.45). Following recommendations by local Brazilian investigators, these questions were added to examine whether the sample reflected the Brazilian population's diversity, which, by contrast, was not considered a usual requirement by local investigators in Denmark.

## Factors Related to Technology Acceptance

This group of questions asked participants about factors highlighted by previous research as essential for technology acceptance in data-sharing contexts: self-assessed health status [54], personal interest in health topics [34], previous participation in health research [33], and awareness of examples of data repositories [55]. This group of questions also asked participants about their current attitude toward digital health research repositories (positive, negative, or indifferent) [43,56] after being provided with the following short description of the concept: A research data repository is an online database containing data collected during research studies. In such repositories, deidentified data is to be re-used in the future by other research studies.

## Motivations to Participate and Reasons for Concern

These questions asked participants how motivated they would feel by the following reasons to contribute to a research data repository: helping future patients, helping researchers, receiving results about themselves, knowing the research outcomes, getting financial compensation, and proposing questions to be investigated in future studies. Participants were also asked how concerned they would feel about the following risks if their data were stored in a health research repository: having their data used for profit without their knowledge, having data used for projects that they perceive as unethical, agreeing with terms and conditions that they do not fully understand, being socially discriminated against because of the information shared, becoming vulnerable to cyberattacks and blackmail, and being asked to provide more data in the future. Such questions about motivations and concerns were based on findings of a qualitative interview study [32] and previous research on motivations to contribute to research [57] and concerns related to data sharing in general [58,59]. The order of the answer options was randomized for each respondent to avoid order bias.

## Access Control Preferences

This group of questions asked participants how desirable or undesirable different access control choices would be once they shared their data with a research platform (answers were not mutually exclusive). The listed answers were: to never be contacted after data are shared, to receive information about who is using the data, to decide who has access to which parts of the data, to have the repository managers decide who has access, to grant data access to public or academic institutions, and to give data access to private laboratories and companies. These questions were based on previous research about informed consent options in biobanks and health data–sharing contexts [48,51,58,60]. The order of the answer options was randomized for each respondent to avoid order bias.

## Willingness to Share Data

Questions in this group concerned how comfortable or uncomfortable participants would feel about sharing different deidentified data sources for a research repository, as previous studies have shown that willingness to share personal health data varies according to the data source [61-63]. Data sources were grouped as (1) biospecimen samples and input data provided through health questionnaires (online or in-person); and (2) passive data collected through smartphone or wearable devices, without end user input. The first questions inquired about participants' willingness to share the following: clinical diagnosis (physical), clinical diagnosis (mental), family health status, DNA samples, food consumption, alcohol consumption, sleep patterns, and blood samples. These data types were based on previous studies of willingness to share clinical and health data for research [64]. The second group of questions inquired about participants' willingness to share frequency of social communication (calls/texts), the content of social communication (calls/texts), distances traveled per day, places visited, physical activity levels (heart rate), stress/emotional levels (heart rate), screen time, and apps used. The choice of data types to include in this second group was based on digital

data sources previously identified as objective behavioral features for health research [5]. Based on previous studies that showed that different granularities might affect willingness to share, the options in this second group were purposely varied in terms of levels of detail provided by the sensor data (eg, frequency of calls/texts versus the content of calls/texts) [65]. The order of the answer options was randomized for each respondent to avoid order bias.

### **Statistical Analysis**

Data were analyzed and visualized using the R Project for Statistical Computing (software environment for statistical computing and graphics). First, frequency distributions were used to characterize responses for each variable, and bivariate associations (odds ratio [OR]) examined relationships between variables. Following previously established reference values, an OR below 1.5 was considered weak and above 5.0 strong [66]. For a 95% CI, results were considered significant if P<.05. Then, a binary logistic regression was conducted to examine directional relationships between explanatory variables and participants' current attitudes toward digital health research repositories. Similarly, another binary logistic regression was conducted to examine directional relationships between explanatory variables and participants' willingness to share data types. Missing values from "prefer not to say" responses were removed before conducting these regression analyses and assumptions were verified beforehand.

## Results

## **Survey Participants**

A total of 2299 participants started answering the survey, of whom 1963 completed all questions (1963/2299, 85.38%). This paper includes only responses from participants living in Denmark (583/1600, 36.44%) and Brazil (1017/1600, 63.56%), thus excluding participants residing in other countries from the data analysis for this study (336/1963, 17.12%). The majority of the sample is aged between 18 and 27 years (933/1600, 58.31%); the second largest age group is between 28 and 37 years (459/1600, 28.69%). Only 12.56% (201/1600) were aged above 37 years. There are slightly more individuals who identify as females (891/1600, 55.69%) than males (682/1600, 42.63%). A majority of participants are educated, having at least a university degree (992/1600, 62.00%), own and use smartphones (1562/1600, 97.63%) and computers (1537/1600, 96.06%), but only 36.75% (588/1600) own and use more than 2 types of digital devices. The vast majority is currently in good, very good, or excellent health (1407/1600, 87.94%), while most are moderately, very, or extremely interested in health topics (1088/1600, 68.00%). Around half of the Brazilian participants (555/1017, 54.57%) are White and 43.17% (439/1017) are Black or Brown; most of the respondents living in Brazil have a monthly household income between 1 (1040 Brazilian Reais or US \$190) and 5 (5200 Brazilian Reais or US \$950) minimum salaries (739/1017, 72.66%). As explained in the previous section, information about race and income was not collected in the Danish survey. Further details on the sample characteristics are presented in Table 1.

Nunes Vilaza et al

Table 1. Participants' characteristics, awareness, past experiences, and attitudes.

Va	riables	All participants (N=1600), n (%)	Participants in Brazil (n=1017), n (%)	Participants in Denmark (n=583), n (%)
Ag	ge (years)			
	18-27	933 (58.31)	613 (60.28)	320 (54.89)
	28-37	459 (28.69)	273 (26.84)	186 (31.90)
	38-47	105 (6.56)	80 (7.87)	25 (4.29)
	48-57	64 (4.00)	38 (3.74)	26 (4.46)
	>57	32 (2.00)	9 (0.88)	23 (3.95)
	Prefer not to say	7 (0.44)	4 (0.39)	3 (0.51)
Ge	ender			
	Female	891 (55.69)	606 (59.59)	285 (48.89)
	Male	682 (42.63)	399 (39.23)	283 (48.54)
	Nonbinary	9 (0.56)	5 (0.49)	4 (0.69)
	Prefer not to say	18 (1.13)	7 (0.69)	11 (1.89)
Se	lf-reported race			
	White	a	555 (54.57)	_
	Black or Brown	_	439 (43.17)	_
	Yellow	_	3 (0.29)	_
	Indigenous	_	2 (0.20)	_
	Prefer not to say	_	18 (1.77)	_
He	ousehold income (monthly) <sup>b</sup>			
	Less or equal to 1 minimum salary	_	114 (11.21)	_
	Between 1 and 3 minimum salaries	_	340 (33.43)	_
	Between 3 and 5 minimum salaries	_	399 (39.23)	_
	Higher or equal to 5 minimum salaries	_	113 (11.11)	_
	Prefer not to say	_	51 (5.01)	_
Ed	lucation			
	Less than secondary education	1 (0.06)	1 (0.10)	0 (0)
	Currently on higher education	595 (37.19)	479 (47.10)	116 (19.90)
	Higher education degree completed	992 (62.00)	527 (51.82)	465 (79.76)
	Prefer not to say	12 (0.75)	10 (0.98)	2 (0.34)
Di	gital devices owned			
	Smartphone(s)	1562 (97.63)	994 (97.74)	568 (97.43)
	Computer(s)	1537 (96.06)	962 (94.59)	575 (98.63)
	Tablet(s)	399 (24.94)	194 (19.08)	205 (35.16)
	Smartwatch(es)	267 (16.69)	145 (14.26)	122 (20.93)
	Smarthome assistant(s)	132 (8.25)	59 (5.80)	73 (12.52)
Nı	mber of digital device types owned			
	0	5 (0.31)	5 (0.49)	0 (0)
	1	64 (4.00)	50 (4.92)	14 (2.40)
	2	943 (58.94)	666 (65.49)	277 (47.51)
	3	437 (27.31)	227 (22.32)	210 (36.02)
	4	119 (7.44)	54 (5.31)	65 (11.15)

https://www.jmir.org/2021/10/e31294

XSL•FO RenderX J Med Internet Res 2021 | vol. 23 | iss. 10 | e31294 | p. 5 (page number not for citation purposes)

Nunes Vilaza et al

Variables	All participants (N=1600), n (%)	Participants in Brazil (n=1017), n (%)	Participants in Denmark (n=583), n (%)
5 or more	32 (2.00)	15 (1.47)	17 (2.92)
Current health status			
Poor	30 (1.88)	16 (1.57)	14 (2.40)
Fair	157 (9.81)	119 (11.70)	38 (6.52)
Good	513 (32.06)	355 (34.91)	158 (27.10)
Very good	666 (41.63)	406 (39.92)	260 (44.60)
Excellent	228 (14.25)	117 (11.50)	111 (19.04)
Prefer not to say	6 (0.38)	4 (0.39)	2 (0.34)
Interest in health topics			
Not interested	39 (2.44)	21 (2.06)	18 (3.09)
Slightly interested	471 (29.44)	247 (24.29)	224 (38.42)
Moderately interested	124 (7.75)	72 (7.08)	52 (8.92)
Very interested	559 (34.94)	340 (33.43)	219 (37.56)
Extremely interested	405 (25.31)	336 (33.04)	69 (11.84)
Prefer not to say	2 (0.13)	1 (0.10)	1 (0.17)
Previous participation in a health research stu	ıdy		
No	815 (50.94)	440 (43.26)	375 (64.32)
Yes	763 (47.69)	567 (55.75)	196 (33.62)
Prefer not to say	22 (1.38)	10 (0.98)	12 (2.06)
Awareness of examples of research data repos	itories		
No	884 (55.25)	528 (51.92)	356 (61.06)
Yes	459 (28.69)	330 (32.45)	129 (22.13)
Not sure	245 (15.31)	152 (14.95)	93 (15.95)
Prefer not to say	12 (0.75)	7 (0.69)	5 (0.86)
Perception of digital health data repositories			
Positive	1339 (83.69)	927 (91.15)	412 (70.67)
Indifferent	188 (11.75)	53 (5.21)	135 (23.16)
Negative	45 (2.81)	19 (1.87)	26 (4.46)
Prefer not to say	28 (1.75)	18 (1.77)	10 (1.72)

<sup>a</sup>Data not collected.

<sup>b</sup>Ranges between 1 (1040 Brazilian Reais or US \$190) and 5 (5200 Brazilian Reais or US \$950).

# **Previous Participation, Awareness of Examples, and Current Attitude**

Around half of the respondents participated in a health research study before (763/1600, 47.69%), and those who participated are more likely to have a moderate to high interest in health topics (OR 2.35, 95% CI 1.88-2.93; P<.001). By contrast, only a minority are aware of research data repository examples (459/1600, 28.69%). Those aware of examples are more likely to have a moderate to high interest in health topics (OR 3.02, 95% CI 2.30-3.96; P<.001) and to have been participants in previous health studies (OR 3.36, 95% CI 2.66-4.23; P<.001). In addition, most participants have a positive perception of

health research data repositories (1339/1600, 83.69%), and those who have a positive perception are more likely to be aware of examples of research data repositories (OR 3.26, 95% CI 2.17-4.90; P<.001). Further details on the frequency distribution for these variables are shown in Table 1.

Results from a binary logistic regression show that interest in health topics (OR 1.49, 95% CI 1.10-2.02; P=.01), previous participation in health research studies (OR 1.70, 95% CI 1.24-2.35; P=.001), and awareness of examples of existing repositories (OR 2.78, 95% CI 1.83-4.38; P<.001) are significant factors influencing participants' current perception of digital health research repositories. See the results of the binary logistic regression in Table 2.

RenderX

Table 2. Binary logistic regression model for the current perception of digital health data repositories (base: not positive perception).

Factors for current perception digital health data repositories (base: not positive perception)	Estimate (B)	Standard error B	P value	Odds ratio (95% CI)
Age (base: above 27)			,	
Below 27 years	0.10	0.17	.56	1.11 (0.79-1.55)
Gender (base: not female)				
Female	-0.05	0.15	.73	0.95 (0.70-1.28)
Education (base: no university degree)				
With university degree	-0.29	0.17	.10	0.75 (0.52-1.06)
Device ownership (base: less than 2 device types)				
Owns more than 2 types	-0.26	0.15	.07	0.76 (0.56-1.04)
Health status (base: poor or fair health)				
Good, very good, or excellent health	0.06	0.23	.79	0.94 (0.57-1.74)
Interest in health (base: none or slight interest)				
Moderate to extreme interest	0.39	0.15	.01	1.49 (1.10-2.02)
Participation in health study (base: no past participation	l)			
Participated in a health study	0.53	0.16	.001	1.70 (1.24-2.35)
Awareness of an example (base: no awareness or not sur	e)			
Aware of an example of repository	1.02	0.22	<.001	2.78 (1.83-4.38)

## **Motivations to Participate**

The majority of participants feel very or extremely motivated by helping future patients (1366/1600, 85.38%), helping researchers (1253/1600, 78.31%), receiving results about themselves (1170/1600, 73.13%), and receiving the results of the research (1063/1600, 66.44%). In addition, being provided with the possibility of suggesting research questions to be investigated is very or extremely motivating for more respondents (829/1600, 51.81%) than receiving financial compensation (505/1600, 31.56%), which is not motivating for 28.69% (459/1600). Multimedia Appendix 2 shows the entire distribution of responses, and Figure 1 displays this information as stacked bar charts.

Those who have a positive perception about health data repositories are more likely to be moderately, very, or extremely motivated by 5 out of 6 motivation sources: helping future patients (OR 9.44, 95% CI 5.43-16.40; P<.001), helping researchers (OR 5.74, 95% CI 3.56-9.25; P<.001), receiving results about themselves (OR 4.12, 95% CI 2.82-6.03; P<.001), receiving results of the research (OR 4.15, 95% CI 2.94-5.85; P<.001), and proposing questions to be investigated (OR 3.46, 95% CI 2.57-4.66; P<.001). Those moderately, very, or extremely interested in health topics are more likely to be moderately, very, or extremely motivated by receiving results of the research (OR 2.25, 95% CI 1.65-3.06; P<.001) and proposing questions to be investigated (OR 2.53, 95% CI 1.97-3.24; P<.001). The youngest segment (18-27 years old) is more likely to feel moderately, very, or extremely motivated to receive financial compensation (OR 1.92, 95% CI 1.57-2.35; P<.001).



Figure 1. Bar chart displaying the distribution of answers for each motivation source.



## **Reasons for Concern**

The vast majority of participants feel very or extremely concerned about having their data used for unethical projects (1219/1600, 76.19%) and profit making without their consent (1096/1600, 68.50%). In addition, the risk of becoming vulnerable to cyberattacks and blackmail is very or extremely concerning for 65.94% (1055/1600); the possibility of not

understanding terms and conditions for 55.38% (886/1600); and the fear of being socially discriminated for 46.38% (742/1600). By contrast, not as many participants feel very or extremely concerned about the burden of being asked to share more data in the future (527/1600, 32.94%). Multimedia Appendix 3 shows the entire distribution of responses, and Figure 2 displays this information in the form of a stacked bar chart.

Figure 2. Bar chart displaying the distribution of answers for each reason for concern.



## Willingness to Share Different Types of Data

Regarding the willingness to share specific data items, most participants feel uncomfortable or very uncomfortable sharing

https://www.jmir.org/2021/10/e31294

**RenderX** 

the content of texts and calls (1206/1600, 75.38%), while fewer participants feel uncomfortable or very uncomfortable sharing the frequency of texts and calls (706/1600, 44.13%). Places visited (864/1600, 54.00%) and apps used (775/1600, 48.44%)

are perceived as uncomfortable or very uncomfortable data to share by many.

By contrast, most participants feel comfortable or very comfortable sharing sleeping patterns (1351/1600, 84.44%), food consumption (1354/1600, 84.63%), alcohol consumption (1274/1600, 79.63%), physical illness diagnosis (1238/1600, 77.38%), physical activity levels (1215/1600, 75.94%), stress

levels (1114/1600, 69.63%), family health history (1070/1600, 66.88%), distances traveled (1072/1600, 67.00%), mental illness diagnosis (1060/1600, 66.25%), blood samples (1029/1600, 64.31%), DNA samples (750/1600, 46.88%), and screen time (1022/1600, 63.88%). Multimedia Appendices 4 and 5 show the full distribution of responses, and Figure 3 displays this information as stacked bar charts.

Figure 3. Bar chart displaying the distribution of willingness to share across different data types.

Α				797					40	)9		127	158	}	105
в		41	411 453					227 319			319		1	87	
с		325			450			26	2		3	337 218			в
D		300		4	106			254			384			252	
Е	2	22		334		29	94			427			3	23	
F	118	1	97	261				602					420		
G	106	2	15	249				554					475		
н	85	197		258				611					449		
	127	15	5	243			(	612					460		
J	64	164		302			(	602					468		
κ	83	121		278			66	61					453		
L	76	74	233				711						504		
м	47 86	6	229			655	5					583	}		
Ν	34 71	2	21			603						671			
0	<mark>29</mark> 53	161			611							743			
Р	<mark>35</mark> 48	164			592						7	759			
Ċ	)			5	io		Frequ	1000 1 requency					15	00	
Very comfortable Comfortable Neither Uncomfortable Very uncomfortable						A. Texts B. Place C. Apps D. Texts E. DNA F. Scree G. Blood H. Ment	s/calls c es visite a used s/calls fi sample en time d sampl tal illne	ontent d requer s es ss	ncy	I. Dist J. Fam K. Str L. Exe M. Ph N. Alc O Foo P. Slee	rances tra hily health ress (hear prcise (hear vsical illn ohol cons d consum p patterr	velled t rate) art rate) ess sumption ption	n		

Those who have a positive perception about health research repositories are more likely to feel comfortable or very comfortable sharing 15 out of 16 data items: physical illness diagnosis (OR 3.84, 95% CI 2.87-5.15; P<.001), mental illness diagnosis (OR 3.44, 95% CI 2.59-4.59; P<.001), family health history (OR 3.45, 95% CI 2.59-4.59; P<.001), DNA samples (OR 2.51, 95% CI 1.85-3.41; P<.001), blood samples (OR 2.31, 95% CI 1.74-3.06; P<.001), food consumption (OR 4.15, 95% CI 3.01-5.70; P<.001), alcohol consumption (OR 3.25, 95% CI 2.41-4.40; P<.001), sleep (OR 3.85, 95% CI 2.80-5.30; P<.001), screen time (OR 3.17, 95% CI 2.38-4.22; P<.001), apps used (OR 2.09, 95% CI 1.50-2.91; P<.001), frequency of calls or texts (OR 2.07, 95% CI 1.51-2.83; P<.001), distances traveled per day (OR 3.34, 95% CI 2.51-4.45; P<.001), places visited (OR 2.77, 95% CI 1.91-4.00; P<.001), physical activity levels (OR 3.43, 95% CI 2.56-4.59; P<.001), and stress levels (OR 3.60, 95% CI 2.70-4.79; P<.001). However, no significant association was found between having a positive perception of digital health research repositories and feeling comfortable with sharing the content of calls and texts (P=.03).

Those moderately, very, or extremely concerned about being discriminated against are more likely to feel uncomfortable or very uncomfortable sharing data about mental illness diagnosis (OR 2.26, 95% CI 1.66-3.07; P<.001). Those uncomfortable or very uncomfortable sharing information about app usage are more likely to be moderately, very, or extremely concerned about data being used for profit (OR 2.57, 95% CI 1.91-3.46; P<.001) and not understanding terms and conditions (OR 2.22, 95% CI 1.71-2.87; P<.001). Those not motivated or only slightly motivated by receiving results about themselves are more likely to feel uncomfortable or very uncomfortable or very uncomfortable or S.76, 95% CI 3.63-9.13; P<.001), distances traveled per day (OR 3.31, 95% CI 2.29-4.80; P<.001), stress levels (OR 6.46, 95% CI 4.43-9.44;

*P*<.001), and physical activity levels (OR 6.78, 95% CI 4.52-10.17; *P*<.001).

A small number of participants feel uncomfortable or very uncomfortable sharing any of the data items (94/1600, 5.88%). Those who feel uncomfortable or very uncomfortable sharing any data items are more likely to have a negative or indifferent perception about health research repositories (OR 3.91, 95%)

CI 2.49-6.14; P<.001). A binary logistic regression shows that age (OR 2.16, 95% CI 1.28-3.70; P=.004), digital device ownership (OR 1.90, 95% CI 1.14-3.26; P=.01), health status (OR 2.28, 95% CI 1.24-3.98; P=.01), and current attitude regarding digital health research repositories (OR 3.77, 95% CI 2.24-6.26; P<.001) are significant factors affecting participants' willingness to share data with a health research repository. Table 3 shows the results of the binary logistic regression.

Table 3.	Binary logistic regression	model for willingness to sha	are data with repositories for h	nealth research (base: u	inwilling to share any data).
----------	----------------------------	------------------------------	----------------------------------	--------------------------	-------------------------------

Factors for willingness to share data (base: unwilling to share any)	Estimate (B)	Standard error B	P value	Odds ratio (95% CI)
Age (base: above 27)	·			
Below 27 years	0.76	0.27	.004	2.16 (1.28-3.70)
Gender (base: not female)				
Female	0.06	0.23	.78	1.06 (0.66-1.70)
Education (base: no university degree)				
With university degree	0.21	0.28	.45	1.24 (0.70-2.16)
Device ownership (base: less than 2 device types)				
Owns more than 2 devices	0.64	0.26	.01	1.90 (1.14-3.26)
Health status (base: poor or fair health)				
Good, very good, or excellent health	-0.82	0.29	.005	2.28 (1.24-3.98)
Interest in health (base: no or slight interest)				
Moderate to extreme interest	-0.01	0.25	.95	0.99 (0.59-1.62)
Participation in health study (base: no past participation	)			
Participated in a health study	0.14	0.24	.55	1.16 (0.71-1.90)
Awareness of an example (base: no awareness)				
Aware of an example of repository	-0.12	0.28	.65	0.88 (0.51-1.56)
Current perception (base: negative or indifferent)				
Positive current perception	1.32	0.26	<.001	3.77 (2.24-6.26)

## **Preferred Access Control Options**

After collecting and sharing their data with a research platform, most participants find it desirable or very desirable to receive information about which projects access their data in the future (1334/1600, 83.38%). The majority also find it desirable or very desirable to decide who gets access to which parts of their data (1181/1600, 73.81%). By contrast, not being contacted is desirable or very desirable to only 25.50% of participants (408/1600), and the option to allow the owners of the repositories to decide who can access the data is desirable or very desirable only to 23.63% (378/1600). Finally, allowing public or academic institutions to access the data is desirable or very desirable for 48.94% (783/1600), while allowing private laboratories and companies to obtain access is desirable or very

desirable to only 13.44% (215/1600). Multimedia Appendix 6 shows the entire distribution of responses, and Figure 4 displays this information as stacked bar charts.

Those who find it is desirable or very desirable to be informed about who is using their data are more likely to have a positive perception of health data repositories (OR 2.45, 95% CI 1.77-3.39; P<.001). Those moderately, very, or extremely concerned about data being used for unethical projects are more likely to find it desirable or very desirable to have control over how their data are used (OR 2.45, 95% CI 1.80-3.42; P<.001) and to be informed about it (OR 3.09, 95% CI 2.18-4.37; P<.001). Those moderately, very, or extremely concerned about data being used for profit are more likely to find it undesirable or very undesirable to have private laboratories and companies access their data (OR 2.24, 95% CI 1.69-2.96; P<.001).



Figure 4. Bar chart displaying the distribution of answers for each access control option.



## E. To have the repository managers deciding who has access

F. To give data access private labs and companies

## **Cross-country Analysis**

The sample comprises 1017 respondents living in Brazil (1017/1600, 63.56%) and 583 living in Denmark (583/1600, 36.44%). The demographics of the participants residing in these 2 countries do not differ substantially, except for a higher percentage of female respondents and a lower percentage of respondents who completed a higher education degree within the Brazilian sample (Table 1). The Brazilian sample also has a higher percentage of extremely interested individuals in health topics than the Danish sample (Table 1).

Around half of the participants in Brazil participated in health research previously (567/1017, 55.75%), in contrast to a minority of the participants in Denmark (196/583, 33.62%). Similarly, the vast majority of participants from the Brazilian sample have a positive initial impression of health data repositories (927/1017, 91.15%), in contrast to a smaller majority of participants in Denmark (412/583, 70.67%). See Table 1 for complete information regarding these differences.

The majority of participants from both countries are highly motivated and concerned by similar sources of motivation and concerns; however, there are differences in the magnitude of the motivation and concern levels reported by those residing in Brazil and Denmark. The Brazilian sample is more likely to rate themselves as moderately, very, or extremely motivated by receiving results about themselves (OR 6.35, 95% CI 4.25-9.50; P<.001), proposing questions to be investigated (OR 6.08, 95% CI 4.67-7.91; P<.001), receiving results of the research (OR 4.13, 95% CI 2.98-5.72; P<.001), and helping the researchers (OR 3.36, 95% CI 2.07-5.44; P<.001). The Brazilian sample is also more likely to rate levels of concerns as moderately, very, or extremely concerning for all of the options listed: having data used for unethical projects (OR 5.44, 95% CI 3.86-7.66; P<.001), becoming vulnerable to cyberattacks and blackmail (OR 3.88, 95% CI 2.99-5.04; P<.001), having data used for

RenderX

profit without consent (OR 3.68, 95% CI 2.77-4.89; P<.001), being asked to provide more data (OR 3.28, 95% CI 2.65-4.06; P < .001), agreeing to terms without understanding them (OR 2.29, 95% CI 1.79-2.94; P<.001), and being socially discriminated against (OR 2.10, 95% CI 1.70-2.60; P<.001). Multimedia Appendices 2 and 3 show the frequency distribution of answers regarding motivations and concerns according to country of residence.

When it comes to access preferences, important differences arise between the 2 countries. The Brazilian sample is more likely to find it desirable or very desirable to receive information about who is using the data (OR 5.51, 95% CI 4.12-7.37; P < .001). By contrast, the Danish sample is more likely to find it desirable or very desirable never to be contacted (OR 3.63, 95% CI 2.87-4.60; P<.001), to have the repository managers decide who can obtain access (OR 2.84, 95% CI 2.24-3.60; P<.001), and to allow private organizations (OR 3.73, 95% CI 2.77-5.04; P<.001) and public institutions access the data (OR 4.51, 95% CI 3.61-5.63; P<.001). Multimedia Appendices 4-6 show the frequency distribution of answers regarding access control preferences and willingness to share according to country of residence.

In summary, the vast majority of the participants residing in Brazil have a positive attitude regarding the idea of health data repositories. These findings are further endorsed by the Brazilian sample reporting higher motivation to help the researchers and willingness to share several data types. However, those residing in Brazil are also more likely to be strongly concerned about all of the potential negative consequences. The Brazilian sample is also more likely to find it desirable to keep the control and be informed about the use of the shared data, rather than never being contacted, delegating control to repository owners, or allowing both private companies and public institutions to get access.

## Discussion

## **Enablers for Acceptance**

Our survey contributes novel empirical insights regarding an extensive set of factors contributing to the acceptance of repositories storing biosamples, health records, and digital data sources for observational research. Previous research suggests that individuals may view some loss of privacy as worthwhile to advance medical research and benefit future generations [4,33], with altruism being a strong incentive for participation in clinical studies [50,67]. Aligned with such previous research, we found that helping future patients and researchers is indeed a powerful source of motivation across our sample, with most participants also feeling very motivated by the prospect of being updated about research outcomes. Furthermore, our findings show that those who do not feel motivated by helping future patients and researchers are more likely to be unwilling to share data, highlighting the critical role of altruism in this context.

Participants are also motivated by learning about their health through the data they provide, aligning with past research [68]. By contrast, our findings indicate that financial compensation may not be a more decisive factor than other sources of motivation. For instance, being invited to suggest research questions for a project strongly motivates more participants than financial compensation. However, consistent with previous studies [41,50], the youngest participants in our sample are more likely to be motivated to share health data in exchange for financial benefits. Such observations reinforce the importance of providing both societal and individual benefits to accommodate different preferences.

Another essential enabler for acceptance is individuals' current perception of the idea of health research data repositories. In our survey, a positive perception appears to be associated with higher levels of motivation to help patients and researchers, and those who have a positive perception are also more likely to feel comfortable sharing 15 out of 16 data items. These results confirm past research highlighting that a positive opinion about biomedical research can predict willingness to participate [33], and attitudes about health care interventions can predict patient acceptance [56]. We also extend previous findings from other contexts [34,43,55] by providing evidence about key factors that can affect individuals' perceptions of digital health research repositories, emphasizing the critical role that positive past experiences and personal interests have in enabling favorable attitudes.

Regarding cross-country differences, the vast majority of our participants residing in Brazil have a positive perception of the idea of health data repositories, further demonstrated by their reported higher motivation to participate and higher willingness to share several data types. It could be speculated that such enthusiasm stems from the prospect of significantly improving an imperfect yet ubiquitous public health care system, which may become an essential enabler for acceptance as digital health emerges in Brazil [67]. By contrast, Denmark has a long history of using clinical databases and electronic health records for population-level clinical research [44]. This observation could explain why our sample residing in Denmark is more likely to

```
https://www.jmir.org/2021/10/e31294
```

find it desirable to allow repository owners to make decisions regarding access control, an arrangement already familiar to them, as the Danish public health authorities manage data use. These differences illustrate that acceptance depends not only on individual predispositions but also on broader sociocultural contexts [36].

### **Barriers for Acceptance**

In contrast to such enablers, our findings show that even though participation in research repositories might occur under the promise of sharing deidentified data, participants still report concerns. Our sample's most substantial concern is the fear that their data will eventually be used for unethical research goals or profit without consent, which is a fear also reported by several previous studies [37,38,42]. Further aligned to previous research, the fear of cyberattacks or blackmail is considered very or extremely concerning to most of our participants [69-72]. Surprisingly, however, the fear of social discrimination is not as prevalent, contrasting a previous study's claim that this might be a core reason behind privacy concerns [58]. It is also surprising that the fear of not fully understanding terms and conditions was a more significant concern for participants than the burden of providing more data, which contradicts previous findings from another study [4].

The predominant concern of data misuse may explain the preference for more restrictive access control options. Many of our participants report feeling comfortable sharing their data if the purpose is to protect the common good, but the same does not apply to the prospect of supporting others' profit making, in alignment with previous research [36,44,73]. Related to this, the large majority of our sample want to receive information about the different projects using their data, and most also want to be deciding who can ultimately obtain access to their data, a finding which has been highlighted in other past studies [38,55,73-75]. By contrast, leaving this responsibility to repository owners is often not our participant' preferred option, especially within the Brazilian sample. Furthermore, approaches such as notification-only and opt-out options have been considered less acceptable than reconsent [73,76], showing the importance of reconsidering usual consent practices.

Another barrier to participation is that willingness to share data depends strongly on the data type [32,47,67], even though there are divergent findings in the literature about which data types people feel most uncomfortable sharing [37,38,42,61-65]. For example, previous studies with young adults have observed a high willingness to donate DNA samples [33,34], but 2 extensive worldwide surveys have observed the opposite [36,44]. Our analysis indicates that when compared with behavioral indicators such as food consumption and sleeping patterns, DNA and blood samples are among the data types most uncomfortable to be shared.

However, even more so than DNA, participants in our study feel uncomfortable sharing passive mobile and wearable sensing data. Interestingly, these are data with the less obvious connection to health in a traditional sense. While the relationship between health and food consumption or sleep might be apparent to many people, the relevance of app use or social communication data may be less noticeable. Such observation

XSL•FO RenderX

is particularly relevant for behavioral health research contemplating passive data sources as a strategy to reduce the data collection burden for participants. Our results also add a more nuanced understanding of participants' willingness to share data. We empirically demonstrate that participants feel uncomfortable sharing more detailed and revealing data sources, such as apps used, frequency of texts and calls, and places visited, compared with broader and less granular information such as screen time, the content of texts and calls, and distances traveled. These findings have important implications for health research studies that consider collecting high granularity information, especially when it comes to location and social communication.

Furthermore, sociodemographic factors have been emphasized by several past studies as possible barriers to the willingness to share data [35,36,39,43,45,54,74]. Our analysis shows that participants' willingness to share data can be related to age, health status, and digital device ownership. However, contrary to previous studies, which observed that members of American ethnic groups other than White have higher odds of being unwilling to donate their DNA data [43,45], our study does not find a significant association between race and unwillingness to share. We also do not find significant associations between race and fear of discrimination [43,45] or desire to control data access [35]. However, our sample is in its vast majority young and educated, in contrast to these previous studies.

## **Research Implications**

The empirical findings discussed above provide the basis for a series of implications for community engagement, technology design, and policymaking. First, we found evidence that a lack of knowledge about health research may be a challenge for public acceptance, which points to the importance of broadening public awareness. For instance, education and familiarity-increasing programs can be possible community engagement approaches and strengthened relationships between potential participants, clinicians, and health research experts may be helpful during recruitment and beyond [40]. Regardless of the medium, participant information could include explanations about the collaborative nature of contemporary health research and why digital data sources extracted passively may be necessary for answering specific research questions. Given the factors found to motivate and demotivate data sharing, it may be necessary to explain the benefits of sharing data types where the direct connection to health is not immediately visible. Additionally, appropriate communication may help to emphasize the importance of data collection compliance to participants, especially when it comes to experience sampling and the provision of frequent self-reports [77].

Personal health informatics could also be considered to increase the appeal of and the motivation for participating. Given that data collection may require interactions with mobile and wearable devices, it is a natural step to also provide participants with personalized data visualizations and, potentially, digital health interventions. However, digital tools for personal health must consider how existing health care practices complement (or hinder) novel approaches [78-80]. Interface design should focus on suitably informing patients about how their data relate

```
https://www.jmir.org/2021/10/e31294
```

to their health to facilitate rather than replace efficient clinician-patient relationships. Above all, risks to individual well-being should be avoided, as an intense "datafication" of personal health standards might prove to be more harmful than beneficial [81]. For instance, our analysis shows that those uncomfortable with sharing alcohol consumption, levels of stress, and physical exercise are less likely to feel motivated by receiving results about themselves. Thus, any consideration of adding personal health informatics features to health research systems should be mindful of the preferences of each individual.

Furthermore, our analysis makes it clear that broader acceptance will be challenging to achieve if contributing to health research repositories demands that participants share every digital source of data [38]. Health research projects might need to acknowledge that certain personal information is associated with social stigma [82], which may compromise willingness to participate in research as a whole. For instance, we observe a strong association between fear of discrimination and unwillingness to share mental illness diagnoses. For this reason, health research should consider personal boundaries by allowing participants to opt-out from specific data collection types and decide which level of details are to be shared. Even if individuals do not exercise this right to choose, the option to safely do that without negative consequences may still enhance trust [83].

When it comes to access control options, our results show that participants would like to be informed about the different projects which may access their data and customize their consent. Even though granular data control options may reduce privacy concerns [84], broad consent models are still the most used approach in current health research platforms [32], which means that once participants provide their consent, they are usually not consulted about data reuse in the future. The conception of digital systems for continuous communication with participants could transform consent practices. For instance, research participants could be consulted about whether they would like to receive a request each time a new project wants to use their data. Access requests could include details about who benefits from the research outcomes and how organizations use any profit. The possibility of opting out from data sharing could also be provided. Beyond allowing participants to make choices about data access, participants could further contribute with questions to a research project, which is an interest identified in our survey and explored in other research platforms [22]. Nondigital approaches could also be considered (eg, phone calls, letters) for those who prefer or do not have access to digital devices. However, a challenge is how to help participants stay informed and control their data without making them overwhelmed [42].

As pervasive sensing technologies become more refined and widespread in health research, those proposing shared-access repositories for collecting, sharing, and using such sensing data will need to take responsibility for identifying risks and be accountable for consequences against participants' best interests. Proactive legal and ethical guidelines are necessary, as current regulatory frameworks for digital health data sharing are relatively weak in some jurisdictions [44]. Likewise, regulatory board members and grant reviewers could evaluate how managers of digital health research repositories demonstrate

XSL•FO RenderX

awareness of ethical considerations and strategies to mitigate possible negative consequences of participation. For instance, being transparent about the trustworthiness of the technical infrastructures and governance arrangements of the platforms hosting the data is essential, even if it means acknowledging challenges [50]. Clear and understandable evidence of compliance with regulations may help diminish individuals' reticence to share health data and increase public acceptance.

Finally, future developments should not ignore that without a diverse cohort providing data, research outcomes and benefits will be unevenly distributed [42,85]. Even though our sample, composed mainly of educated young individuals, does not show significant associations between race, income, and unwillingness to share data, other past studies have shown that these factors can be significant [43,45]. For this reason, communication efforts, interface design, and data sharing policies should be made accessible and inclusive by being mindful of language choices, cultural requirements, access costs, and participation demands (eg, owning and using smartphones and smartwatches). After all, strategies to increase acceptance should be motivated by research repository owners' genuine desire to make data sharing fairer and more ethical.

## **Limitations and Future Work**

Based on our team experience, we suggest that similar surveys in the future should strive to focus recruitment efforts on racial, ethnic, gender, and disability minorities to achieve a higher representation from these groups. We also suggest that quantitative findings should be complemented with parallel qualitative investigations, to provide richer and subjective insight into justifications and reasonings behind responses. Another suggestion is to consider depicting data usage scenarios with illustrations, infographics, and narrative forms instead of purely descriptive texts.

In terms of methodological limitations, sampling bias is a common challenge of voluntary response samples, given that those who take the time to respond to online survey requests tend to have strong opinions compared with the rest of the population. Despite our efforts to use a diverse range of digital channels for recruitment, our sample is biased in terms of age and education level, partially due to the scope of the overarching project (mental health for young adults) and partially because the higher response rate stemmed from posts on university web portals. A constrained focus on specific population segments can be considered a strength, given that it allows the investigation of particular perspectives at a time; however, future work must seek to consider the perspectives of many other groups of individuals, especially underrepresented minorities.

Another methodological limitation is that, as an exploratory cross-sectional study, our results can only capture attitudes about hypothetical future participation. Thus, future work must consider evaluating participants' perspectives while taking part in an actual digital health research repository initiative. In addition, factors such as the burden of continuous data collection might be better examined throughout actual participation.

Finally, it could be speculated that public attitudes may shift following the global experience of a public health emergency (COVID-19 pandemic). Given that this survey was conducted during the first waves of the pandemic, future work is still needed to evaluate further consequences of this unprecedented crisis in the long term. In particular, the impact of contact tracing apps and vaccination passports may prove significant when it comes to the acceptance of digital health data storage on a population level.

## Conclusion

This survey study reveals essential factors for potential acceptance and willingness to share personal data with a digital health research repository. In summary, most participants feel very motivated about helping future patients, helping researchers, and receiving results about their health; most also feel comfortable sharing data sources usually associated with health research, except DNA data. However, most respondents feel very concerned about the risk of cyberattacks, the possibility of data being used for unethical research goals or for-profit without consent, and the prospect of sharing personal sensing data, especially social communication and location. The majority of participants find it desirable to receive information about which projects access their data and would like to be able to decide who gets access to which parts of their data.

The analysis of such a large spectrum of variables and their relationships provides a strong foundation for suggesting implications for future developments. The implications discussed include to disseminate knowledge about health research; to value the role of transparency for trust development; to engage participants with the research process and their health management; to allow flexible and customizable data sharing; and to align policies and regulations with ethical considerations. Providing valuable benefits for individuals and reducing the risks involved in participation are essential requirements in this context, and by recognizing differences between groups, it is possible to better understand and respond to individual views and expectations.

## Acknowledgments

This study is part of the TEAM (Technology Enabled Mental Health for Young People), a project funded by the European Union's Horizon 2020 research and innovation program under the Marie Skłodowska-Curie grant agreement no. 722561.

## **Conflicts of Interest**

None declared.


## **Multimedia Appendix 1**

Questionnaire applied to participants (survey questions). [PDF File (Adobe PDF File), 447 KB-Multimedia Appendix 1]

### Multimedia Appendix 2

Distribution of answers for the question: "How motivated do you feel by the following?". [PDF File (Adobe PDF File), 43 KB-Multimedia Appendix 2]

## Multimedia Appendix 3

Distribution of answers for the question: "How concerned are you about the following?". [PDF File (Adobe PDF File), 43 KB-Multimedia Appendix 3]

## Multimedia Appendix 4

Distribution of answers for the question: "How comfortable do you feel about sharing the following data items (collected through questionnaires and surveys)?". [PDF File (Adobe PDF File), 53 KB-Multimedia Appendix 4]

## Multimedia Appendix 5

Distribution of answers for the question: "How comfortable do you feel about sharing the following data items (collected through mobile and wearable sensing)?".

[PDF File (Adobe PDF File), 53 KB-Multimedia Appendix 5]

## Multimedia Appendix 6

Distribution of answers for the question: "How desirable are the following options?". [PDF File (Adobe PDF File), 46 KB-Multimedia Appendix 6]

### References

- 1. Collins FS, Varmus H. A new initiative on precision medicine. N Engl J Med 2015 Feb 26;372(9):793-795. [doi: 10.1056/NEJMp1500523] [Medline: 25635347]
- 2. Garcia-Ceja E, Riegler M, Nordgreen T, Jakobsen P, Oedegaard KJ, Tørresen J. Mental health monitoring with multimodal sensing and machine learning: A survey. Pervasive and Mobile Computing 2018 Dec;51:1-26. [doi: 10.1016/j.pmcj.2018.09.003]
- 3. Abdullah S, Choudhury T. Sensing Technologies for Monitoring Serious Mental Illnesses. IEEE MultiMedia 2018 Jan;25(1):61-75. [doi: 10.1109/mmul.2018.011921236]
- Nebeker C, Lagare T, Takemoto M, Lewars B, Crist K, Bloss CS, et al. Engaging research participants to inform the ethical 4. conduct of mobile imaging, pervasive sensing, and location tracking research. Transl Behav Med 2016 Dec;6(4):577-586. [doi: 10.1007/s13142-016-0426-4] [Medline: 27688250]
- Rohani DA, Faurholt-Jepsen M, Kessing LV, Bardram JE. Correlations Between Objective Behavioral Features Collected 5. From Mobile and Wearable Devices and Depressive Mood Symptoms in Patients With Affective Disorders: Systematic Review. JMIR Mhealth Uhealth 2018 Aug 13;6(8):e165 [FREE Full text] [doi: 10.2196/mhealth.9691] [Medline: 30104184]
- Cornet VP, Holden RJ. Systematic review of smartphone-based passive sensing for health and wellbeing. J Biomed Inform 6. 2018 Jan;77:120-132. [doi: 10.1016/j.jbi.2017.12.008] [Medline: 29248628]
- 7. Aledavood T, Triana HAM, Alakörkkö T, Kaski K, Saramäki J, Isometsä E, et al. Data Collection for Mental Health Studies Through Digital Platforms: Requirements and Design of a Prototype. JMIR Res Protoc 2017 Jun 09;6(6):e110 [FREE Full text] [doi: 10.2196/resprot.6919] [Medline: 28600276]
- Harari GM, Lane ND, Wang R, Crosier BS, Campbell AT, Gosling SD. Using Smartphones to Collect Behavioral Data in 8. Psychological Science: Opportunities, Practical Considerations, and Challenges. Perspect Psychol Sci 2016 Nov;11(6):838-854. [doi: 10.1177/1745691616650285] [Medline: 27899727]
- Dogan E, Sander C, Wagner X, Hegerl U, Kohls E. Smartphone-Based Monitoring of Objective and Subjective Data in 9. Affective Disorders: Where Are We and Where Are We Going? Systematic Review. J Med Internet Res 2017 Jul 24;19(7):e262 [FREE Full text] [doi: 10.2196/jmir.7006] [Medline: 28739561]
- Broes S, Lacombe D, Verlinden M, Huys I. Sharing human samples and patient data: Opening Pandora's box. Journal of 10. Cancer Policy 2017 Sep;13:65-69. [doi: 10.1016/j.jcpo.2017.07.006]

- Committee on Strategies for Responsible Sharing of Clinical Trial Data, Board on Health Sciences Policy, Institute of Medicine. Sharing Clinical Trial Data: Maximizing Benefits, Minimizing Risk. Washington, DC: National Academies Press; 2015.
- 12. Raghupathi W, Raghupathi V. Big data analytics in healthcare: promise and potential. Health Inf Sci Syst 2014;2:3 [FREE Full text] [doi: 10.1186/2047-2501-2-3] [Medline: 25825667]
- 13. Bernardos A, Pires M, Ollé D, Casar J. Digital Phenotyping as a Tool for Personalized Mental Healthcare Internet. 2019 May Presented at: PervasiveHealth'19: Proceedings of the 13th EAI International Conference on Pervasive Computing Technologies for Healthcare; May 20-23, 2019; Trento, Italy p. 403-408. [doi: <u>10.1145/3329189.3329240</u>]
- 14. Sudlow C, Gallacher J, Allen N, Beral V, Burton P, Danesh J, et al. UK biobank: an open access resource for identifying the causes of a wide range of complex diseases of middle and old age. PLoS Med 2015 Mar;12(3):e1001779 [FREE Full text] [doi: 10.1371/journal.pmed.1001779] [Medline: 25826379]
- Hossain MA, Dwivedi YK, Rana NP. State-of-the-art in open data research: Insights from existing literature and a research agenda. Journal of Organizational Computing and Electronic Commerce 2015 Dec 02;26(1-2):14-40. [doi: 10.1080/10919392.2015.1124007]
- Ferri M, Abdallah K. Preparing for responsible sharing of clinical trial data. N Engl J Med 2014 Jan 30;370(5):484-485. [doi: <u>10.1056/NEJMc1314515</u>] [Medline: <u>24476448</u>]
- Broes S, Lacombe D, Verlinden M, Huys I. Toward a Tiered Model to Share Clinical Trial Data and Samples in Precision Oncology. Front Med (Lausanne) 2018 Jan 29;5:6 [FREE Full text] [doi: 10.3389/fmed.2018.00006] [Medline: 29435448]
- All of Us Research Program Investigators T, Denny JC, Rutter JL, Goldstein DB, Philippakis A, Smoller JW, et al. The "All of Us" Research Program. N Engl J Med 2019 Aug 15;381(7):668-676 [FREE Full text] [doi: 10.1056/NEJMsr1809937] [Medline: 31412182]
- 19. Deutsch E, Duftschmid G, Dorda W. Critical areas of national electronic health record programs-is our focus correct? Int J Med Inform 2010 Mar;79(3):211-222. [doi: 10.1016/j.ijmedinf.2009.12.002] [Medline: 20079685]
- 20. Pencina MJ, Louzao DM, McCourt BJ, Adams MR, Tayyabkhan RH, Ronco P, et al. Supporting open access to clinical trial data for researchers: The Duke Clinical Research Institute-Bristol-Myers Squibb Supporting Open Access to Researchers Initiative. Am Heart J 2016 Feb;172:64-69. [doi: 10.1016/j.ahj.2015.11.002] [Medline: 26856217]
- 21. Bardram J. The CARP Mobile Sensing Framework -- A Cross-platform, Reactive, Programming Framework and Runtime Environment for Digital Phenotyping Internet. arXiv. Preprint posted online June 21, 2020 [FREE Full text]
- Lunn MR, Lubensky M, Hunt C, Flentje A, Capriotti MR, Sooksaman C, et al. A digital health research platform for community engagement, recruitment, and retention of sexual and gender minority adults in a national longitudinal cohort study--The PRIDE Study. J Am Med Inform Assoc 2019 Aug 01;26(8-9):737-748 [FREE Full text] [doi: 10.1093/jamia/ocz082] [Medline: 31162545]
- 23. Wicks P, Massagli M, Frost J, Brownstein C, Okun S, Vaughan T, et al. Sharing health data for better outcomes on PatientsLikeMe. J Med Internet Res 2010;12(2):e19 [FREE Full text] [doi: 10.2196/jmir.1549] [Medline: 20542858]
- Greshake Tzovaras B, Angrist M, Arvai K, Dulaney M, Estrada-Galiñanes V, Gunderson B, et al. Open Humans: A platform for participant-centered research and personal data exploration. Gigascience 2019 Jun 01;8(6):giz076 [FREE Full text] [doi: 10.1093/gigascience/giz076] [Medline: 31241153]
- 25. van Panhuis WG, Paul P, Emerson C, Grefenstette J, Wilder R, Herbst AJ, et al. A systematic review of barriers to data sharing in public health. BMC Public Health 2014;14:1144 [FREE Full text] [doi: 10.1186/1471-2458-14-1144] [Medline: 25377061]
- 26. Stone MA, Redsell SA, Ling JT, Hay AD. Sharing patient data: competing demands of privacy, trust and research in primary care. Br J Gen Pract 2005 Oct;55(519):783-789 [FREE Full text] [Medline: <u>16212854</u>]
- 27. Sankar PL, Parker LS. The Precision Medicine Initiative's All of Us Research Program: an agenda for research on its ethical, legal, and social issues. Genet Med 2017 Jul 8;19(7):743-750. [doi: 10.1038/gim.2016.183] [Medline: 27929525]
- 28. Kostkova P, Brewer H, de Lusignan S, Fottrell E, Goldacre B, Hart G, et al. Who Owns the Data? Open Data for Healthcare. Front Public Health 2016;4:7 [FREE Full text] [doi: 10.3389/fpubh.2016.00007] [Medline: 26925395]
- 29. Klingler C, Silva DS, Schuermann C, Reis AA, Saxena A, Strech D. Ethical issues in public health surveillance: a systematic qualitative review. BMC Public Health 2017 Apr 04;17(1):295 [FREE Full text] [doi: 10.1186/s12889-017-4200-4] [Medline: 28376752]
- 30. Clark K, Duckham M, Guillemin M, Hunter A, McVernon J, O'Keefe C, et al. Advancing the ethical use of digital data in human research: challenges and strategies to promote ethical practice. Ethics Inf Technol 2018 Nov 21;21(1):59-73. [doi: 10.1007/s10676-018-9490-4]
- 31. Vilaza G. What is the future of data sharing for research? Interactions 2020 Jul 09;27(4):54-56. [doi: 10.1145/3404211]
- Vilaza G, Maharjan R, Coyle D, Bardram J. Futures for Health Research Data Platforms From the Participants' Perspectives. 2020 Oct Presented at: Proceedings of the 11th Nordic Conference on Human-Computer Interaction: Shaping Experiences, Shaping Society; October 25-29, 2020; Tallinn, Estonia. [doi: 10.1145/3419249.3420110]
- 33. Merdad L, Aldakhil L, Gadi R, Assidi M, Saddick SY, Abuzenadah A, et al. Assessment of knowledge about biobanking among healthcare students and their willingness to donate biospecimens. BMC Med Ethics 2017 May 02;18(1):32 [FREE Full text] [doi: 10.1186/s12910-017-0195-8] [Medline: 28464877]

- Tozzo P, Fassina A, Caenazzo L. Young people's awareness on biobanking and DNA profiling: results of a questionnaire administered to Italian university students. Life Sci Soc Policy 2017 Dec 10;13(1):9 [FREE Full text] [doi: 10.1186/s40504-017-0055-9] [Medline: 28601979]
- Hong SJ, Drake B, Goodman M, Kaphingst KA. Race, Trust in Doctors, Privacy Concerns, and Consent Preferences for Biobanks. Health Commun 2020 Sep;35(10):1219-1228 [FREE Full text] [doi: <u>10.1080/10410236.2019.1623644</u>] [Medline: <u>31167570</u>]
- 36. Middleton A, Milne R, Almarri MA, Anwer S, Atutornu J, Baranova EE, Mendes, et al. Global Public Perceptions of Genomic Data Sharing: What Shapes the Willingness to Donate DNA and Health Data? Am J Hum Genet 2020 Oct 01;107(4):743-752 [FREE Full text] [doi: 10.1016/j.ajhg.2020.08.023] [Medline: 32946764]
- Rieger A, Gaines A, Barnett I, Baldassano CF, Connolly Gibbons MB, Crits-Christoph P. Psychiatry Outpatients' Willingness to Share Social Media Posts and Smartphone Data for Research and Clinical Purposes: Survey Study. JMIR Form Res 2019 Aug 29;3(3):e14329 [FREE Full text] [doi: 10.2196/14329] [Medline: 31493326]
- Rendina H, Mustanski B. Privacy, Trust, and Data Sharing in Web-Based and Mobile Research: Participant Perspectives in a Large Nationwide Sample of Men Who Have Sex With Men in the United States. J Med Internet Res 2018 Jul 04;20(7):e233 [FREE Full text] [doi: 10.2196/jmir.9019] [Medline: 29973332]
- Karampela M, Ouhbi S, Isomursu M. Connected Health User Willingness to Share Personal Health Data: Questionnaire Study. J Med Internet Res 2019 Nov 27;21(11):e14537 [FREE Full text] [doi: 10.2196/14537] [Medline: 31774410]
- Abdelhamid M, Gaia J, Sanders GL. Putting the Focus Back on the Patient: How Privacy Concerns Affect Personal Health Information Sharing Intentions. J Med Internet Res 2017 Sep 13;19(9):e169 [FREE Full text] [doi: 10.2196/jmir.6877] [Medline: 28903895]
- 41. Weitzman ER, Kaci L, Mandl KD. Sharing medical data for health research: the early personal health record experience. J Med Internet Res 2010 May 25;12(2):e14 [FREE Full text] [doi: 10.2196/jmir.1356] [Medline: 20501431]
- 42. Bouras A, Simoes EJ, Boren S, Hicks L, Zachary I, Buck C, et al. Non-Hispanic White Mothers' Willingness to Share Personal Health Data With Researchers: Survey Results From an Opt-in Panel. J Particip Med 2020 May 15;12(2):e14062 [FREE Full text] [doi: 10.2196/14062] [Medline: 33064096]
- Middleton A, Milne R, Thorogood A, Kleiderman E, Niemiec E, Prainsack B, et al. Attitudes of publics who are unwilling to donate DNA data for research. Eur J Med Genet 2019 May;62(5):316-323 [FREE Full text] [doi: 10.1016/j.ejmg.2018.11.014] [Medline: 30476628]
- 44. Patil S, Lu H, Saunders CL, Potoglou D, Robinson N. Public preferences for electronic health data storage, access, and sharing evidence from a pan-European survey. J Am Med Inform Assoc 2016 Nov 23;23(6):1096-1106 [FREE Full text] [doi: 10.1093/jamia/ocw012] [Medline: 27107445]
- 45. Sanderson SC, Brothers KB, Mercaldo ND, Clayton EW, Antommaria AHM, Aufox SA, et al. Public Attitudes toward Consent and Data Sharing in Biobank Research: A Large Multi-site Experimental Survey in the US. Am J Hum Genet 2017 Mar 02;100(3):414-427 [FREE Full text] [doi: 10.1016/j.ajhg.2017.01.021] [Medline: 28190457]
- 46. Lemke AA, Wolf WA, Hebert-Beirne J, Smith ME. Public and biobank participant attitudes toward genetic research participation and data sharing. Public Health Genomics 2010;13(6):368-377 [FREE Full text] [doi: 10.1159/000276767] [Medline: 20805700]
- 47. Ahram M, Othman A, Shahrouri M, Mustafa E. Factors influencing public participation in biobanking. Eur J Hum Genet 2014 Apr;22(4):445-451 [FREE Full text] [doi: 10.1038/ejhg.2013.174] [Medline: 23921537]
- Trinidad SB, Fullerton SM, Bares JM, Jarvik GP, Larson EB, Burke W. Informed Consent in Genome-Scale Research: What Do Prospective Participants Think? AJOB Prim Res 2012 Jul 01;3(3):3-11 [FREE Full text] [doi: 10.1080/21507716.2012.662575] [Medline: 23493836]
- 49. Rathi V, Dzara K, Gross CP, Hrynaszkiewicz I, Joffe S, Krumholz HM, et al. Sharing of clinical trial data among trialists: a cross sectional survey. BMJ 2012 Nov 20;345:e7570 [FREE Full text] [doi: 10.1136/bmj.e7570] [Medline: 23169870]
- Papoutsi C, Reed JE, Marston C, Lewis R, Majeed A, Bell D. Patient and public views about the security and privacy of Electronic Health Records (EHRs) in the UK: results from a mixed methods study. BMC Med Inform Decis Mak 2015 Oct 14;15:86 [FREE Full text] [doi: 10.1186/s12911-015-0202-2] [Medline: 26466787]
- Luchenski S, Balasanthiran A, Marston C, Sasaki K, Majeed A, Bell D, et al. Survey of patient and public perceptions of electronic health records for healthcare, policy and research: study protocol. BMC Med Inform Decis Mak 2012 May 23;12:40 [FREE Full text] [doi: 10.1186/1472-6947-12-40] [Medline: 22621621]
- 52. Zanaboni P, Kummervold PE, Sørensen T, Johansen MA. Patient Use and Experience With Online Access to Electronic Health Records in Norway: Results From an Online Survey. J Med Internet Res 2020 Feb 07;22(2):e16144 [FREE Full text] [doi: 10.2196/16144] [Medline: 32031538]
- Porteri C, Pasqualetti P, Togni E, Parker M. Public's attitudes on participation in a biobank for research: an Italian survey. BMC Med Ethics 2014 Nov 26;15(1):81 [FREE Full text] [doi: 10.1186/1472-6939-15-81] [Medline: 25425352]
- 54. Ivanov A, Sharman R, Rao HR. Exploring factors impacting sharing health-tracking records. Health Policy and Technology 2015 Sep;4(3):263-276. [doi: 10.1016/j.hlpt.2015.04.008]

- 55. Riordan F, Papoutsi C, Reed JE, Marston C, Bell D, Majeed A. Patient and public attitudes towards informed consent models and levels of awareness of Electronic Health Records in the UK. Int J Med Inform 2015 Apr;84(4):237-247 [FREE Full text] [doi: 10.1016/j.ijmedinf.2015.01.008] [Medline: 25649841]
- 56. Sekhon M, Cartwright M, Francis JJ. Acceptability of healthcare interventions: an overview of reviews and development of a theoretical framework. BMC Health Serv Res 2017 Jan 26;17(1):88 [FREE Full text] [doi: 10.1186/s12913-017-2031-8] [Medline: 28126032]
- 57. Shah N, Coathup V, Teare H, Forgie I, Giordano GN, Hansen TH, et al. Motivations for data sharing-views of research participants from four European countries: A DIRECT study. Eur J Hum Genet 2019 May 30;27(5):721-729 [FREE Full text] [doi: 10.1038/s41431-019-0344-2] [Medline: 30700834]
- 58. Christin D, Reinhardt A, Kanhere SS, Hollick M. A survey on privacy in mobile participatory sensing applications. Journal of Systems and Software 2011 Nov;84(11):1928-1946. [doi: 10.1016/j.jss.2011.06.073]
- 59. Cilliers L. Wearable devices in healthcare: Privacy and information security issues. Health Inf Manag 2020;49(2-3):150-156. [doi: 10.1177/1833358319851684] [Medline: 31146589]
- 60. Nurgalieva L, Frik A, Ceschel F, Egelman S, Marchese M. Information Design in An Aged Care Context: Views of Older Adults on Information Sharing in a Care Triad. In: Proceedings of the 13th EAI International Conference on Pervasive Computing Technologies for Healthcare. 2019 May Presented at: PervasiveHealth'19: The 13th International Conference on Pervasive Computing Technologies for Healthcare; May 20-23, 2019; Trento, Italy. [doi: 10.1145/3329189.3329211]
- Rudnicka A, Cox A, Gould S. Why Do You Need This? In: Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems - CHI ?19. 2019 May Presented at: CHI '19: CHI Conference on Human Factors in Computing Systems; May 4-9, 2019; Glasgow, Scotland, UK. [doi: 10.1145/3290605.3300622]
- Di Matteo D, Fine A, Fotinos K, Rose J, Katzman M. Patient Willingness to Consent to Mobile Phone Data Collection for Mental Health Apps: Structured Questionnaire. JMIR Ment Health 2018 Aug 29;5(3):e56 [FREE Full text] [doi: 10.2196/mental.9539] [Medline: 30158102]
- 63. Rooksby J, Morrison A, Murray-Rust D. Student Perspectives on Digital Phenotyping Internet. In: CHI '19: Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 2019 May Presented at: CHI '19: CHI Conference on Human Factors in Computing Systems; May 4-9, 2019; Glasgow, Scotland, UK. [doi: 10.1145/3290605.3300655]
- 64. Weitzman ER, Kelemen S, Kaci L, Mandl KD. Willingness to share personal health record data for care improvement and public health: a survey of experienced personal health record users. BMC Med Inform Decis Mak 2012 May 22;12:39 [FREE Full text] [doi: 10.1186/1472-6947-12-39] [Medline: 22616619]
- 65. Schneegass S, Poguntke R, Machulla T. Understanding the Impact of Information Representation on Willingness to Share Information. In: CHI '19: Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 2019 May Presented at: CHI '19: 2019 CHI Conference on Human Factors in Computing Systems; May 4-9, 2019; Glasgow, Scotland, UK p. 1-6. [doi: 10.1145/3290605.3300753]
- Chen H, Cohen P, Chen S. How Big is a Big Odds Ratio? Interpreting the Magnitudes of Odds Ratios in Epidemiological Studies. Communications in Statistics - Simulation and Computation 2010 Mar 31;39(4):860-864. [doi: 10.1080/03610911003650383]
- 67. Zammar G, Meister H, Shah J, Phadtare A, Cofiel L, Pietrobon R. So different, yet so similar: meta-analysis and policy modeling of willingness to participate in clinical trials among Brazilians and Indians. PLoS One 2010 Dec 16;5(12):e14368 [FREE Full text] [doi: 10.1371/journal.pone.0014368] [Medline: 21179556]
- 68. Nobile H, Vermeulen E, Thys K, Bergmann MM, Borry P. Why do participants enroll in population biobank studies? A systematic literature review. Expert Rev Mol Diagn 2013 Jan;13(1):35-47. [doi: <u>10.1586/erm.12.116</u>] [Medline: <u>23256702</u>]
- 69. Seltzer E, Goldshear J, Guntuku SC, Grande D, Asch DA, Klinger EV, et al. Patients' willingness to share digital health and non-health data for research: a cross-sectional study. BMC Med Inform Decis Mak 2019 Aug 08;19(1):157 [FREE Full text] [doi: 10.1186/s12911-019-0886-9] [Medline: 31395102]
- 70. Pratap A, Allred R, Duffy J, Rivera D, Lee HS, Renn BN, et al. Contemporary Views of Research Participant Willingness to Participate and Share Digital Data in Biomedical Research. JAMA Netw Open 2019 Nov 01;2(11):e1915717 [FREE Full text] [doi: 10.1001/jamanetworkopen.2019.15717] [Medline: 31747031]
- 71. Overby C, Maloney K, Alestock T, Chavez J, Berman D, Sharaf R, et al. Prioritizing Approaches to Engage Community Members and Build Trust in Biobanks: A Survey of Attitudes and Opinions of Adults within Outpatient Practices at the University of Maryland. J Pers Med 2015 Jul 28;5(3):264-279 [FREE Full text] [doi: 10.3390/jpm5030264] [Medline: 26226006]
- 72. Ma Y, Dai H, Wang L, Zhu L, Zou H, Kong X. Consent for use of clinical leftover biosample: a survey among Chinese patients and the general public. PLoS One 2012;7(4):e36050 [FREE Full text] [doi: 10.1371/journal.pone.0036050] [Medline: 22558323]
- 73. Kim KK, Joseph JG, Ohno-Machado L. Comparison of consumers' views on electronic data sharing for healthcare and research. J Am Med Inform Assoc 2015 Jul;22(4):821-830 [FREE Full text] [doi: 10.1093/jamia/ocv014] [Medline: 25829461]
- 74. Haeusermann T, Fadda M, Blasimme A, Tzovaras BG, Vayena E. Genes wide open: Data sharing and the social gradient of genomic privacy. AJOB Empir Bioeth 2018;9(4):207-221. [doi: <u>10.1080/23294515.2018.1550123</u>] [Medline: <u>30596357</u>]

```
https://www.jmir.org/2021/10/e31294
```

- 75. Ostherr K, Borodina S, Bracken RC, Lotterman C, Storer E, Williams B. Trust and privacy in the context of user-generated health data. Big Data & Society 2017 Apr 17;4(1):205395171770467. [doi: 10.1177/2053951717704673]
- Ludman EJ, Fullerton SM, Spangler L, Trinidad SB, Fujii MM, Jarvik GP, et al. Glad you asked: participants' opinions of re-consent for dbGap data submission. J Empir Res Hum Res Ethics 2010 Sep;5(3):9-16 [FREE Full text] [doi: 10.1525/jer.2010.5.3.9] [Medline: 20831417]
- 77. van Berkel N, Ferreira D, Kostakos V. The Experience Sampling Method on Mobile Devices. ACM Comput. Surv 2018 Jan 12;50(6):1-40. [doi: 10.1145/3123988]
- Epstein DA, Caraway M, Johnston C, Ping A, Fogarty J, Munson SA. Beyond Abandonment to Next Steps: Understanding and Designing for Life after Personal Informatics Tool Use. Proc SIGCHI Conf Hum Factor Comput Syst 2016 May;2016:1109-1113 [FREE Full text] [doi: 10.1145/2858036.2858045] [Medline: 28503678]
- 79. Bhat KS, Kumar N. Sociocultural Dimensions of Tracking Health and Taking Care. Proc. ACM Hum.-Comput. Interact 2020 Oct 14;4(CSCW2):1-24. [doi: 10.1145/3415200]
- Chandwani R, Kulkarni V. Who's the doctor?: Physicians' perception of internet informed patients in India. In: CHI '16: Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. 2016 May Presented at: CHI '16: The 2016 CHI Conference on Human Factors in Computing Systems; May 7-12, 2016; San Jose, CA p. 3091-3102. [doi: 10.1145/2858036.2858500]
- Luo Y, Oh CY, Jean BS, Choe EK. Interrelationships Between Patients' Data Tracking Practices, Data Sharing Practices, and Health Literacy: Onsite Survey Study. J Med Internet Res 2020 Dec 22;22(12):e18937 [FREE Full text] [doi: 10.2196/18937] [Medline: 33350960]
- Warner M, Gutmann A, Sasse MA, Blandford A. Privacy Unraveling Around Explicit HIV Status Disclosure Fields in the Online Geosocial Hookup App Grindr. Proc. ACM Hum.-Comput. Interact 2018 Nov 01;2(CSCW):1-22. [doi: 10.1145/3274450]
- 83. Kuntsman A, Miyake E, Martin S. Re-thinking Digital Health: Data, Appisation and the (im)possibility of 'Opting out'. Digit Health 2019 Oct 09;5:2055207619880671 [FREE Full text] [doi: 10.1177/2055207619880671] [Medline: 31636917]
- 84. Abdelhamid M. Greater patient health information control to improve the sustainability of health information exchanges. J Biomed Inform 2018 Jul;83:150-158 [FREE Full text] [doi: 10.1016/j.jbi.2018.06.002] [Medline: 29894811]
- Ibrahim H, Liu X, Zariffa N, Morris AD, Denniston AK. Health data poverty: an assailable barrier to equitable digital health care. Lancet Digit Health 2021 Apr;3(4):e260-e265 [FREE Full text] [doi: 10.1016/S2589-7500(20)30317-4] [Medline: 33678589]

### Abbreviations

**DTU:** Technical University of Denmark **GDPR:** General Data Protection Regulation

Edited by G Eysenbach; submitted 17.06.21; peer-reviewed by H Pratomo; comments to author 23.07.21; revised version received 16.09.21; accepted 20.09.21; published 29.10.21

Please cite as:

Nunes Vilaza G, Coyle D, Bardram JE Public Attitudes to Digital Health Research Repositories: Cross-sectional International Survey J Med Internet Res 2021;23(10):e31294 URL: <u>https://www.jmir.org/2021/10/e31294</u> doi: <u>10.2196/31294</u> PMID:

©Giovanna Nunes Vilaza, David Coyle, Jakob Eyvind Bardram. Originally published in the Journal of Medical Internet Research (https://www.jmir.org), 29.10.2021. This is an open-access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work, first published in the Journal of Medical Internet Research, is properly cited. The complete bibliographic information, a link to the original publication on https://www.jmir.org/, as well as this copyright and license information must be included.



Please answer our survey and express your opinion about data sharing for health research. We want to hear from you to understand your boundaries and preferences.

## Questions should take 10 minutes only. We hope you can help!

# Section A: Informed Consent

We are a group of researchers from the Copenhagen Center for Health Technology (CACHET). Before we start collecting data, we need to be sure you read, understand and agree with the following:

What is the purpose of this survey? To understand people's opinions about health research for a university/academic project.

Which data will be collected? Voluntary demographics (e.g., age, gender), opinions about research and preferences about data sharing.

Will this data be linked to me personally? No, this survey is anonymous: no personally identifiable data is collected (e.g., name, e-mail, IP address, geo-location).

How will data be processed? Statistical analysis will be conducted by the principal researcher for scientific reports, which will only contain aggregate results.

How will data be stored? The anonymous answers will be stored at a secured repository hosted at the university until the end of the project (October 2021).

How can I delete my data? You can click on <u>Exit and clear survey</u> on the top right of the page or contact us with a unique ID code which you will create.

How to contact us? - Responsible researcher: PhD candidate, Giovanna Vilaza: gnvi@dtu.dk - Director of the CACHET Research Centre: Prof Dr Jakob Bardram: jakba@dtu.dk - Data protection officer at DTU, Ane Sandager: anesa@dtu.dk

### A1. Do you confirm that:

- you are more than 18 years old;

- you have read and understood the information above; - and you voluntarily agree to participate in this survey?

Yes

J	0		

		<b>               </b>
Secti	ion B: Demographics	
B1.	What is your age group?	
	18-27	
	28-37	
	38-47	
	48-57	
	Above 57	
	Prefer not to say	
B2.	What is your gender?	
	Female	
	Male	
	Non-binary	
	Preter not to say	
	Prefer to self-describe	
	Prefer to self-describe	
B3.	What is your education level?	
	Less than High School/Secondary School	
	Completed High School/Secondary School	
	Completed a Bachelor's degree/Undergraduate	
	Completed a Master's degree/Graduate	
	Completed a PhD degree or above	
	Prefer not to say	
	Other	
	Other	·

B4. Which of these digital devices do you have and use?	
Computer (desktop, laptop)	
Tablet (iPad)	
Smartphone (Samsung, iPhone)	
Wrist wearables (Apple Watch, Garmin, Fitbit, Polar)	
Smart home devices (Alexa, Google Home, Hive, Philips Hue)	
Other	
Other	

#### Other

# B5. In which country do you currently live?

Afghanistan	
Albania	
Algeria	
Andorra	
Angola	
Antigua and Barbuda	
Argentina	
Armenia	$\Box$
Australia	$\Box$
Austria	Ļ
Azerbaijan	
Bahamas	
Bahrain	
Bangladesh	
Barbados	
Belarus	
Belgium	
Belize	

Benin	
Bhutan	
Bolivia (Plurinational State of)	
Bosnia and Herzegovina	
Botswana	
Brazil	
Brunei Darussalam	
Bulgaria	
Burkina Faso	
Burundi	
Cabo Verde	
Cambodia	
Cameroon	
Canada	
Central African Republic	
Chad	
Chile	
China	
Colombia	
Comoros	
Congo	
Cook Islands	
Costa Rica	
Côte d'Ivoire	
Croatia	
Cuba	
Cyprus	
Czech Republic	
Democratic People's Republic ofKorea	
	-

Congo	
Denmark	
Djibouti	
Dominica	
Dominican Republic	
Ecuador	
Egypt	
El Salvador	
Equatorial Guinea	
Eritrea	
Estonia	
Ethiopia	
Fiji	
Finland	
France	
Gabon	
Gambia	
Georgia	
Germany	
Ghana	
Greece	
Grenada	
Guatemala	
Guinea	
Guinea-Bissau	
Guyana	
Haiti	
Honduras	
Hungary	



Iceland		
India		
Indonesia		
Iran (Islamic Republic of)		
Iraq		
Ireland		
Israel		
Italy		
Jamaica		
Japan		
Jordan		
Kazakhstan		
Kenya		
Kiribati		
Kuwait		
Kyrgyzstan		
Lao People's Democratic Republic		
Latvia		
Lebanon		
Lesotho		
Liberia		
Libya		
Lithuania		
Luxembourg		
Madagascar		
Malawi		
Malaysia		
Maldives		
Mali		
	L-p-J	

Malta	
Marshall Islands	
Mauritania	
Mauritius	
Mexico	
Micronesia (Federated States of)	
Monaco	
Mongolia	
Montenegro	
Morocco	
Mozambique	
Myanmar	
Namibia	
Nauru	
Nepal	
Netherlands	
New Zealand	
Nicaragua	
Niger	
Nigeria	
Niue	
Norway	
Oman	
Pakistan	
Palau	
Panama	
Papua New Guinea	
Paraguay	
Peru	
	L

Philippines	
Poland	
Portugal	
Puerto Rico	
Qatar	
Republic of Korea	
Republic of Moldova	
Romania	
Russian Federation	
Rwanda	
Saint Kitts and Nevis	
Saint Lucia	
Saint Vincent and the Grenadines	
Samoa	
San Marino	
Sao Tome and Principe	
Saudi Arabia	
Senegal	
Serbia	
Seychelles	
Sierra Leone	
Singapore	
Slovakia	
Slovenia	
Solomon Islands	
Somalia	
South Africa	
South Sudan	
Spain	

Sri Lanka	
Sudan	
Suriname	
Swaziland	
Sweden	
Switzerland	
Syrian Arab Republic	
Tajikistan	
Thailand	
The former Yugoslav Republic of Macedonia	
Timor-Leste	
Togo	
Tokelau	
Tonga	
Trinidad and Tobago	
Tunisia	
Turkey	
Turkmenistan	
Tuvalu	
Uganda	
Ukraine	
United Arab Emirates	
the United Kingdom of Great Britain and Northern Ireland	
United Republic of Tanzania	
United States of America	
Uruguay	
Uzbekistan	
Vanuatu	
Venezuela (Bolivarian Republic of)	

	Viet Nam	
	Yemen	
1	Zambia	
	Zimbabwe	
	Other	
	Other	•
Sect	ion C: Health	
C1.	In general, would you say your health is:	
	Excellent	
	Very good	
	Good	
	Fair	
	Poor	
	Prefer not to say	
C2.	How interested are you in health-related topics?	
	Extremely interested	
	Very interested	
	Moderately interested	
	Slightly interested	
	Not interested	
	Prefer not to say	
C3.	Have you ever participated in a research study about health?	
	Yes	
	No	
1	Prefer not to say	

# Section D: Research data repository

The following definition is very important for the next questions. Please read it carefully:

A research data repository is an online database containing data collected in previous studies. De-identified or anonymous data is stored to be re-used in the future by multiple researchers.

D1.	Do you know any example of a research data repository?	
	Yes	
	No	
	Not sure	
	Prefer not to say	
D2.	Your current opinion about this idea of data repositories for health research is:	
	Positive	
	Negative	
	Indifferent	
	Prefer not to say	

# Section E: Sharing your data with a repository (I)

How comfortable do you feel about having the following data stored in a research repository?

Please consider that any information shared is de-identified or anonymised before being analysed by researchers.

## E1. Health questionnaires (online or in-person):

	mfortable Uncomfort Comfortabl comfortabl able Neither e e
Clinical diagnosis (physical)	
Clinical diagnosis(mental)	
Family health status	
DNA samples	
Food consumption	
Alcohol consumption	

	<b>   </b>      <b>                        </b>
	Very unco Very mfortable Uncomfort Comfortabl comfortabl able Neither e e
Sleep	
Blood samples	
Section F: Sharing your data with a reposi	tory (II)

How comfortable do you feel about having the following data stored in a research repository?

Please consider that any information shared is <u>de-identified or anonymised</u> before being analysed by researchers.

Very unco

Very

## F1. Passive monitoring through a phone or wearable device:

	mfortable Uncomfort able Comfortabl comfortabl
Screen time	
Apps used	
Frequency of social communication(calls/text)	
Content of social communication(calls/text)	
Distances travelled per day	
Places visited every day	
Physical activity levels (heart rate)	
Stress/emotional levels (heartrate)	

# Section G: Motivations

G1. Still in the context of research data repositories, how motivated do you feel by the following reasons to contribute to one of them?

	Not motivated	Slightly motivated	Moderately motivated	Very motivated	Extremely motivated
Helping future patients					
Helping the researchers					
Receiving results about myself					
Getting financial compensation					

		Not motivated	Slightly motivated	Moderately motivated	Very motivated	Extremely motivated	
	Proposing questions to be investigated						
	Receiving the results of the research						
Se	ction H: Access options						
H1.	How desirable/undesirable do you think the for a health research data repository are?	followir	ng acces	s optior	18		
		Very undesirable	Undesirable	Neither	Desirable	Very desirable	
	To receive information about the projects using my data						
	To not be contacted after I share my data						
	To decide who can have access to which parts my data						
	To have the repository managers deciding who can get access						
	That public or academic institutions are eligible for access						
	That private labs and companies are eligible for access						
Se	ction I: Concerns						
I1.	How concerned do you feel about the followin health data stored in a research repository?	ng risks	s of havi	ing you	r		
		Not	Slightly	Moderately	Very	Extremely	

	concerned	concerned	concerned	concerned	concerned
Being vulnerable to cyber-attacks and blackmailing					
Having my data used for projects that I think are unethical					
Having my data used for profit without me knowing about it					
Being socially discriminated because of certain information I shared					
Agreeing with terms and conditions that I do not fully understand					
Being asked to provide more data in the future					

# Section J: Feedback J1. Do you allow your anonymous answers for this survey to be stored in an open repository and be accessed by other researchers? Yes No This survey is anonymous, which makes it impossible for us to know J2. which answer is yours. If later on you want to access your answers or have them deleted, please create an ID code for you now (e.g., hello123), and send it to us by e-mail: Contact: gnvi@dtu.dk or jakba@dtu.dk **J3**. Please, let us know if you have any comment or feedback about this survey:

Thank you so much for your time and answer! We really appreciate that.

Do you want to receive the final survey results? Please contact: Giovanna Vilaza, PhD candidate, at gnvi@dtu.dk or Prof. Jakob Bardram, at jakba@dtu.dk

N	lotivation sources	All participants, n (%)	Participants in Brazil, n (%)	Participants in Denmark, n (%)
H	lelping future patients			
	Not or slightly motivated	58 (3.62)	25 (2.45)	33 (5.66)
	Moderately to extremely motivated	1542 (96.75)	992 (97.5)	550 (94.3)
	Prefer not to say	0 (0)	0 (0)	0 (0)
H	lelping the researchers			
	Not or slightly motivated	76 (4.75)	27 (2.65)	59 (10.12)
	Moderately to extremely motivated	1524 (95.25)	990 (97.34)	534 (91.59)
	Prefer not to say	0 (0)	0 (0)	0 (0)
R n	Receiving results about nyself			
	Not or slightly motivated	139 (8.68)	34 (0.33)	105 (18.01)
	Moderately to extremely motivated	1459 (91.18)	982 (96.55)	477 (81.81)
	Prefer not to say	2 (0.125)	1 (0.09)	1 (0.17)
R r	Receiving the research esults			
	Not or slightly motivated	185 (11.56)	62 (6.09)	123 (21.09)
	Moderately to extremely motivated	1413 (88.31)	955 (93.9)	458 (78.55)
	Prefer not to say	2 (0.125)	0 (0)	2 (0.34)
P it	Proposing questions to be nvestigated			
	Not or slightly motivated	336 (21.00)	102 (10.02)	234 (40.13)
	Moderately to extremely motivated	1260 (88.31)	915 (89.97)	345 (59.17)
	Prefer not to say	4 (0.25)	0 (0.00)	4 (0.68)
G c	etting financial ompensation			
	Not or slightly motivated	732 (45.75)	457 (44.93)	275 (47.16)

N	lotivation sources	All participants, n (%)	Participants in Brazil, n (%)	Participants in Denmark, n (%)
	Moderately to extremely motivated	867 (54.18)	560 (55.06)	307 (52.65)
	Prefer not to say	1 (0.06)	0 (0)	1 (0.17)

Reasons for concern	All participants, n (%)	Participants in Brazil, n (%)	Participants in Denmark, n (%)
Data is used for research with unethical goals			
Not or slightly concerned	180 (11.25)	51 (5.01)	129 (22.12)
Moderately to extremely concerned	1415 (88.43)	966 (94.98)	449 (77.01)
Prefer not to say	5 (0.31)	0 (0)	5 (0.85)
Data is used for profit without explicit consent			
Not or slightly concerned	245 (15.31)	91 (8.94)	154 (26.41)
Moderately to extremely concerned	1351 (84.43)	926 (91.05)	425 (72.89)
Prefer not to say	4 (0.25)	0 (0)	4 (0.68)
Suffering from cyber-attacks and blackmail			
Not or slightly concerned	307 (19.18)	115 (11.3)	192 (32.93)
Moderately to extremely concerned	1288 (80.50)	901 (88.59)	387 (66.38)
Prefer not to say	5 (0.31)	1 (0.09)	4 (0.68)
Agreeing with terms without fully understanding			
Not or slightly concerned	322 (20.12)	154 (15.14)	168 (28.81)
Moderately to extremely concerned	1273 (79.56)	863 (84.85)	410 (70.32)
Prefer not to say	5 (0.31)	0 (0)	5 (0.85)
Being socially discriminated because of the data			
Not or slightly concerned	561 (35.06)	294 (28.9)	267 (45.79)
Moderately to extremely concerned	1035 (64.68)	723 (71.09)	312 (53.51)
Prefer not to say	4 (0.25)	0 (0)	4 (0.68)
Being asked to provide more data in the future			

R	easons for concern	All participants, n (%)	Participants in Brazil, n (%)	Participants in Denmark, n (%)
	Not or slightly concerned	662 (41.37)	316 (31.07)	346 (59.34)
	Moderately to extremely concerned	933 (58.31)	700 (68.82)	233 (39.96)
	Prefer not to say	5 (0.31)	1 (0.09)	4 (0.68)

Willingness to share questionnaire and biospecimen data	All participants, n (%)	Participants in Brazil, n (%)	Participants in Denmark, n (%)
Sleep patterns			
Uncomfortable or very uncomfortable	83 (5.18)	38 (3.73)	45 (7.71)
Comfortable or very comfortable	1351 (84.43)	873 (85.84)	478 (81.98)
Neither uncomfortable nor comfortable	164 (10.25)	106 (10.42)	58 (9.94)
Prefer not to say	2 (0.12)	0 (0)	2 (0.34)
Food consumption			
Uncomfortable or very uncomfortable	82 (5.12)	40 (3.93)	42 (7.2)
Comfortable or very comfortable	1354 (84.62)	865 (85.05)	489 (83.87)
Neither uncomfortable nor comfortable	161 (10.06)	111 (10.91)	50 (8.57)
Prefer not to say	3 (0.18)	1 (0.09)	2 (0.34)
Alcohol consumption			
Uncomfortable or very uncomfortable	105 (6.56)	36 (3.53)	69 (11.83)
Comfortable or very comfortable	1274 (79.62)	817 (80.33)	457 (78.38)
Neither uncomfortable nor comfortable	221 (13.81)	164 (16.12)	57 (9.77)
Prefer not to say	0 (0)	0 (0)	0 (0)
Clinical diagnosis (physical)			
Uncomfortable or very uncomfortable	133 (8.31)	54 (5.3)	79 (13.55)
Comfortable or very comfortable	1238 (77.37)	800 (78.66)	438 (75.12)
Neither uncomfortable nor comfortable	229 (14.31)	163 (16.02)	66 (11.32)

Willingness to share questionnaire and biospecimen data	All participants, n (%)	Participants in Brazil, n (%)	Participants in Denmark, n (%)
Prefer not to say	0 (0)	0 (0)	0 (0)
Family health			
Uncomfortable or very uncomfortable	228 (14.25)	105 (10.32)	123 (21.09)
Comfortable or very comfortable	1070 (66.87)	710 (69.81)	360 (61.74)
Neither uncomfortable nor comfortable	302 (18.87)	202 (19.86)	100 (17.15)
Prefer not to say	0 (0)	0 (0)	0 (0)
Clinical diagnosis (mental)			
Uncomfortable or very uncomfortable	282 (17.62)	132 (12.97)	150 (25.72)
Comfortable or very comfortable	1060 (66.25)	702 (69.02)	358 (61.4)
Neither uncomfortable nor comfortable	258 (16.12)	183 (17.99)	75 (12.86)
Prefer not to say	0 (0)	0 (0)	0 (0)
Blood samples			
Uncomfortable or very uncomfortable	321 (20.06)	211 (20.74)	110 (18.86)
Comfortable or very comfortable	1029 (64.31)	612 (60.17)	417 (71.52)
Neither uncomfortable nor comfortable	249 (15.56)	194 (19.07)	55 (9.43)
Prefer not to say	1 (0.06)	0 (0)	1 (0.17)
DNA samples			
Uncomfortable or very uncomfortable	556 (34.75)	319 (31.36)	237 (40.65)
Comfortable or very comfortable	750 (46.87)	488 (47.98)	262 (44.93)
Neither uncomfortable nor comfortable	294 (18.37)	210 (20.64)	84 (14.4)

Willingness to share questionnaire and biospecimen data		All participants, n (%)	Participants in Brazil, n (%)	Participants in Denmark, n (%)
	Prefer not to say	0 (0)	0 (0)	0 (0)

Willingness to share mobile and wearable sensing data	All participants, n (%)	Participants in Brazil, n (%)	Participants in Denmark, n (%)
Physical activity levels (heart rate)			
Uncomfortable or very uncomfortable	150 (9.37)	66 (6.48)	84 (14.4)
Comfortable or very comfortable	1215 (75.93)	786 (77.28)	429 (73.58)
Neither uncomfortable nor comfortable	233 (14.56)	165 (16.22)	68 (11.66)
Prefer not to say	2 (0.12)	0 (0)	2 (0.34)
Stress/emotional levels (heart rate)			
Uncomfortable or very uncomfortable	204 (12.75)	82 (8.06)	122 (20.92)
Comfortable or very comfortable	1114 (69.62)	735 (72.27)	379 (65)
Neither uncomfortable nor comfortable	278 (17.37)	199 (19.56)	79 (13.55)
Prefer not to say	4 (0.25)	1 (0.09)	3 (0.51)
Distances travelled per day			
Uncomfortable or very uncomfortable	282 (17.62)	168 (16.51)	114 (19.55)
Comfortable or very comfortable	1072 (67)	678 (66.66)	394 (67.58)
Neither uncomfortable nor comfortable	243 (15.18)	171 (16.81)	72 (12.34)
Prefer not to say	3 (0.18)	0 (0)	3 (0.51)
Screen time			
Uncomfortable or very uncomfortable	315 (19.68)	189 (18.58)	126 (21.61)
Comfortable or very comfortable	1022 (63.87)	644 (63.62)	378 (64.83)
Neither uncomfortable nor comfortable	261 (16.31)	184 (18.09)	77 (13.2)

Willingness to share mobile and wearable sensing data	All participants, n (%)	Participants in Brazil, n (%)	Participants in Denmark, n (%)
Prefer not to say	2 (0.12)	0 (0)	2 (0.34)
Frequency of social communication (calls/text)			
Uncomfortable or very uncomfortable	706 (44.12)	479 (47.09)	227 (38.96)
Comfortable or very comfortable	636 (39.75)	353 (34.7)	283 (48.54)
Neither uncomfortable nor comfortable	254 (15.87)	185 (18.19)	69 (11.83)
Prefer not to say	4 (0.25)	0 (0)	4 (0.68)
Apps used			
Uncomfortable or very uncomfortable	775 (48.43)	534 (52.5)	241 (41.33)
Comfortable or very comfortable	555 (34.68)	312 (30.67)	243 (41.68)
Neither uncomfortable nor comfortable	262 (16.37)	167 (16.42)	95 (16.29)
Prefer not to say	8 (0.5)	4 (0.39)	4 (0.68)
Places visited every day			
Uncomfortable or very uncomfortable	864 (54)	534 (52.5)	330 (56.6)
Comfortable or very comfortable	503 (31.43)	330 (32.44)	176 (30.18)
Neither uncomfortable nor comfortable	227 (14.18)	153 (15.04)	74 (8.06)
Prefer not to say	3 (0.18)	0 (0)	3 (0.51)
Content of social communication (calls/text)			
Uncomfortable or very uncomfortable	1206 (75.37)	794 (78.07)	412 (70.66)
Comfortable or very comfortable	263 (16.43)	141 (13.86)	122 (20.92)
Neither uncomfortable nor comfortable	127 (7.93)	80 (7.86)	47 (8.06)

Willingness to share mobile and wearable sensing data		All participants,	Participants in	Participants in
		n (%)	Brazil, n (%)	Denmark, n (%)
	Prefer not to say	4 (0.25)	2 (0.19)	2 (0.34)

Desirability of access control options	All participants, n (%)	Participants in Brazil, n (%)	Participants in Denmark, n (%)
To receive information about who is using my data			
Undesirable or very undesirable	99 (6.18)	28 (2.75)	71 (12.17)
Desirable or very desirable	1334 (83.37)	939 (92.33)	395 (67.75)
Neither undesirable nor desirable	160 (10)	10 (0.98)	110 (18.86)
Prefer not to say	7 (0.43)	0 (0)	7 (1.20)
To decide who can have access to which parts my data			
Undesirable or very undesirable	100 (6.25)	44 (4.32)	56 (9.6)
Desirable or very desirable	1181 (73.81)	785 (73.72)	396 (67.92)
Neither undesirable nor desirable	313 (19.56)	188 (18.48)	125 (21.44)
Prefer not to say	6 (0.37)	0 (0)	6 (1.02)
That public or academic institutions can get access			
Undesirable or very undesirable	317 (19.81)	263 (25.86)	54 (9.26)
Desirable or very desirable	783 (48.93)	367 (36.08)	416 (71.35)
Neither undesirable nor desirable	495 (30.93)	386 (37.95)	109 (18.69)
Prefer not to say	5 (0.31)	1 (0.09)	4 (0.68)
To never be contacted after I share my data			
Undesirable or very undesirable	600 (37.5)	455 (44.73)	145 (24.87)
Desirable or very desirable	408 (25.5)	167 (16.42)	241 (41.33)
Neither undesirable nor desirable	587 (36.68)	395 (38.83)	192 (32.93)
Prefer not to say	5 (0.31)	0 (0)	5 (0.85)

Desirability of access control options		All participants, n (%)	Participants in Brazil, n (%)	Participants in Denmark, n (%)
To have the repository managers deciding who can get access				
	Undesirable or very undesirable	794 (49.62)	589 (57.91)	205 (35.16)
	Desirable or very desirable	378 (23.62)	169 (16.61)	209 (35.84)
	Neither undesirable nor desirable	423 (26.43)	259 (25.46)	164 (28.13)
	Prefer not to say	5 (0.31)	0 (0)	5 (0.85)
That private labs and companies can get access				
	Undesirable or very undesirable	915 (57.18)	659 (64.79)	256 (43.91)
	Desirable or very desirable	215 (13.43)	78 (7.66)	137 (23.49)
	Neither undesirable nor desirable	465 (29.06)	280 (27.53)	185 (31.73)
	Prefer not to say	5 (0.31)	0 (0)	5 (0.85)

2.4 Sharing Access to Behavioural and Personal Health Data: Designers' Perspectives on Opportunities and Barriers

## Full Reference:

Giovanna Nunes Vilaza and Jakob Bardram. 2019. Sharing Access to Behavioural and Personal Health Data: Designers' Perspectives on Opportunities and Barriers. In *13th International Conference on Pervasive Computing Technologies for Healthcare (PervasiveHealth'19), May 20–23, 2019, Trento, Italy.* ACM, New York, NY, USA, 5 pages. https://doi.org/10.1145/3329189.3329229

## Copyrights:

Permission to republish in a thesis conceived by ACM. Copyright held by the authors.

## Sharing Access to Behavioural and Personal Health Data: Designers' Perspectives on Opportunities and Barriers

Giovanna N. Vilaza & Jakob E. Bardram Copenhagen Center for Health Technology Department of Health Technology, Technical University of Denmark DK-2800 Kgs. Lyngby, Denmark

{gnvi,jakba}@dtu.dk

#### ABSTRACT

Behavioural and personal health data can be collected from smartphones, wearable and sensors. Besides using this data as a tool for self-reflection and self-management of health issues, it can also be beneficial to share it with others. The purpose of sharing can range from getting peer support to communicating better with clinicians or contributing to scientific research. However, sharing access to behavioural data can disclose sensitive details of the individuals' lives, and it may be challenging to make shared-access systems adopted, accepted and continuously used. In order to better inform the development of future systems, a review of the literature and a focus group were conducted. The findings presented here provide pointers to some essential considerations to be taken into account when proposing systems for behavioural data sharing.

#### **CCS CONCEPTS**

• Human-centered computing → HCI theory, concepts and models; Collaborative and social computing devices.

#### **KEYWORDS**

Shared-access data, continuous behavioural tracking, trust in computer technologies, healthcare applications

#### ACM Reference Format:

Giovanna N. Vilaza & Jakob E. Bardram. 2019. Sharing Access to Behavioural and Personal Health Data: Designers' Perspectives on Opportunities and Barriers. In *The 13th International Conference on Pervasive Computing Technologies for Healthcare (PervasiveHealth'19), May 20–23, 2019, Trento, Italy.* ACM, New York, NY, USA, 5 pages. https://doi.org/10.1145/3329189.3329229

#### **1 INTRODUCTION**

Pervasive computer technologies are allowing significant advances in the healthcare domain [3]. Among the possibilities, data collected through continuous sensing can be useful for the assessment of affective states, behaviours and mental status [28]. Previous research has shown how personal data sensed through mobile phones, such as calls received, on-screen time and movement across locations can be mapped to possible symptoms of depression [29].

PervasiveHealth'19, May 20–23, 2019, Trento, Italy

© 2019 Copyright held by the owner/author(s). Publication rights licensed to ACM. ACM ISBN 978-1-4503-6126-2/19/05...\$15.00

https://doi.org/10.1145/3329189.3329229

Several tools have been designed to use continuous sensing data to encourage self-management and behavioural change [12]. Besides, the collected data can offer great value to researchers, clinicians and circles of support, especially in the case of chronic illness, in which the monitoring of daily symptoms is crucial to its control [8]. For instance, patients with chronic conditions may voluntarily share self-tracked data to their doctors to get better treatment options [24]. Individuals may also agree to provide their data to open-access research platforms and contribute to scientific investigations [13]. In summary, sharing data can be highly beneficial in a personal and societal level.

However, when it comes to proposing shared-access data platforms, previous research has found that people are reluctant in sharing certain data types, despite it being beneficial for their health [14]. Some of the issues are that the collected data may contain intimate details of the individuals' lives [25] and uncover health disorders that can bring social stigma and discrimination [9]. Furthermore, individuals might be anxious [7] or frustrated [11] about their behaviours, which may also prevent them from sharing their data. Additionally, healthcare experts can also be reluctant to use the behavioural data collected through smartphones and sensors, as they might believe these data sources are not reliable or useful [6, 16]. Such lack of trust in the technology represents a barrier to the adoption of novel digital tools in a clinical setting [31]. An important question is then:

> If individuals are reluctant to share their data and doctors are not confident in using it, how can the potential benefits of data sharing systems be reached?

In order to work towards a better understanding of the enablers and barriers that could be addressed during the design of sharedaccess systems, we conducted a focus group interview. While previous studies have often focused on patients and clinicians perspectives, there is not much evidence about designers and developers perspectives until now. In the conception of pervasive healthcare systems, they are the ones who have to capture the motivation of each user group and bridge their needs into a solution [23]. Therefore, the findings from the literature review were complemented by the discussions generated in this focus group, to define some critical considerations for the design of future systems.

This paper is structured as the following: first, the background related to the research topic is presented. Then, the methods are described as well as the results. Finally, the findings are discussed taking into consideration the previous research. We hope that this paper serves as the starting point for further research involving the essential aspects to be considered in future designs for data sharing approaches.

Permission to make digital or hard copies of all or part of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. Copyrights for components of this work owned by others than the author(s) must be honored. Abstracting with credit is permitted. To copy otherwise, or republish, to post on servers or to redistribute to lists, requires prior specific permission and/or a fee. Request permissions from permissions@acm.org.

#### 2 BACKGROUND

Previous research has shown how sharing self-tracked data can bring benefits to healthcare. In a clinical context, patients can come to the consultation and bring data they collected about their daily activities in order to get a more personalised treatment [24, 35]. In a research context, if patients of life-long diseases provide permission to have their data available for open-access medical databases, it can be used to inform policymakers, practitioners and researchers [18]. In a home context, family members can share their health data to create a culture of mutual care [4] or make better sense of each others' behaviours [36]. Similarly, people may share their health data among peers with similar conditions to get motivation, inspiration and support [22, 26].

However, previous research also shows that the process of sharing data does not come without challenges, especially when it comes to behaviour and personal health data. First, while patients might see the value of self-tracking, they might disagree with sharing this data [33], depending on their perception of privacy and security risks [2, 15, 21]. For instance, some individuals are concerned about the possibility of misuse of any personal data by insurance and pharmaceutical companies [27]. Besides, continuous behavioural data also brings the potential for social stigma [9], and the fear of being continuously scrutinised [10, 30]. It has been found that there are aspects of people's lives that they prefer to keep private, such as class attendance [14] or location [10]. The lack of trust in the relationships with those accessing the data might even increase such concerns [19].

Second, there are differences across users' preferences and needs. Whilst some healthcare professionals may see the benefits of having highly detailed data, from multiple sources, to better assess the patients' progress and prepare more customised self-care plans [17, 35], others see self-tracked behavioural data as less reliable [16] and would prefer to see more scientifically validated evidence before using this data [6]. From the perspective of the individuals being tracked, having to collect and share high-quality data from several sources can be a burden. Moreover, some patients may become obsessed and engage in harmful behaviours (e.g. eating disorders) [14] or they might get demotivated with their lack of progress [22].

Third, it is not easy to balance the needs for support and autonomy. The consultation can be the occasion when doctors and patients collaboratively discuss the self-tracked data [12]. During this process, doctors bring to discussion data outliers and trends, while patients have the opportunity to add explanations. Having such discussions in person helps to prevent patients from getting into conclusions without appropriate clinical support and healthcare providers fear that patients will not be able to interpret their data alone[32] correctly. However, it has been suggested that future systems should encourage patients' autonomy to make sense of their data as well so that users are not overly dependent on their healthcare providers and technology can release the demands on the healthcare system [24].

In summary, the possibility of getting customised recommendations and better treatments are some of the advantages for the patients who are willing to share personal and behavioural data. Besides, the creation of systems that allow doctors to have access to data from their patients could also relieve the demands on health services. However, the potential lack of acceptability, the different users' needs and the problematic balance between autonomy and support are challenging. Such challenges could be addressed during the design process of shared-access systems, which leads to the motivation of involving designers and developers in the discussion. Previous research has investigated what requirements designers consider essential in m-government services [?]. Designers are not only capable of empathising with users, but they can also see flaws of current approaches and the opportunities to create novel features. The focus group conducted with this goal is described next.

#### 3 METHODS

A focus group was conducted to investigate further what the enablers and obstacles for the implementation of successful sharedaccess systems are. In the focus group, there were six participants: two interaction designers and four mobile apps developers with specialised knowledge in implementing and deploying systems for personalised healthcare. As they were part of the same research group, they collaborate across projects which involved depression, cognitive impairment and cardiovascular diseases.

The focus group lasted for 2 hours. The facilitator asked the participants semi-structured questions in order to initiate the process of brainstorming ideas and generate discussions. Participants were asked to share their views about the potential benefits that opening the access to personal health data may bring as well as the aspects that are currently hindering such possibilities. The session was audio-recorded, and the participants' notes were documented. The qualitative data were transcribed and then analysed using Thematic Analysis [5].

#### 4 FINDINGS

Discussions in the focus group were organised in three main themes, which ranged from the application data can be used for, the remark that users are still reluctant to share data and the need for further features in current approaches for data sharing. The notes participants took during the discussion are in Figure 1.

#### 4.1 Data for science, not for profit

Overall, participants believed that sharing data should have the ultimate goal of being beneficial for individuals and society: "Good purposes, research purposes, which are going to benefit society in general, going to benefit the health sector" (P5).

Given their technical knowledge, participants understood that sharing data enables the possibility to create algorithms and models: "Sharing data can help others in a bigger way or solve a bigger problem. For example, if I share my heart data, how my heart is functioning, that can help to build an artificial intelligence algorithm which can automatically detect diseases or that can help to build some model" (P3).

They also pointed out that it would be interesting to find ways to use data in the future to get a better diagnosis: "We might not know what to do with the data right now, but in four years, I might go down with stress. Then we can look at the four years of data leading

#### Sharing Access to Behavioural and Personal Health Data

#### PervasiveHealth'19, May 20-23, 2019, Trento, Italy



Figure 1: Participants notes colour coded as: data types willing to share (green) and barriers to sharing (pink)

up to me having this stress, and we could maybe figure out what is causing it, more accurately" (P4).

However, they recognised that data could be used to create models that might be harmful: "A company or organisation that has my data can create a profile secretly and use it in the future for data analysis. They may already know about my behaviour before they actually meet me because they have a lot of data about me so that they can predict my actions in the future" (P1). The idea of allowing personal data for profit-making was perceived negatively: "It has to be good for all the people, and not for profit making like all the big corporations are doing in some way or another" (P2).

#### 4.2 Users are reluctant to share

During the discussions, it was argued that despite the potential benefit of having access to more data, people might disagree with the idea of sharing: "It is technically feasible, and we are very much looking forward to getting this kind of data, because if you have more data you can build more robust systems. But, that is the problem: you do not have the data because people are not willing to share it" (P3).

Furthermore, psychological data was mentioned as one data type that participants saw as potentially problematic: "I think most people are happier sharing only physiological data, because of the stigma of sharing the mental data" (P6). The stigma refers to the possibility of social discrimination: "We take the data from all of these psychiatry patients, but these patients are afraid of other people knowing what is going on with their mental health" (P2).

One participant pointed out what could be seen as a key requirement for users to feel more comfortable with sharing data, which is a relationship of trust: "When you talk with your doctor, any data you share with your doctor, it is confidential" (P4).

#### 4.3 Systems could be improved

Participants agreed that the design of current systems are lacking transparency about data usage: "The most important for me is the journey of the data, so who is going to use it, and at what point, who is going through the data" (P6). The same applies to the 'terms and conditions' forms, as they are well designed: "The terms of the data are sometimes not clear: when they want to collect it, like what sort of data they want to collect, it is very general the data agreement" (P1). Another issue is the lack of further alternatives in the way that terms are proposed: "Even if you read them you cannot say to the service 'I will not give you consent', then you do not have any option besides not to use it. You have to give away your data. It is like a trade-off" (P1).

Given that they have the knowledge of how systems are built, participants discussed that it is difficult to convey how security and privacy protection were implemented: "That is kind of a technical problem, I am not sure that those software developers they covered all the issues, if privacy and security were also considered when developing this platform" (P1).

#### 5 DISCUSSIONS

The technical advances in data collection using smartphones and sensors data allow closer tracking of daily activities and symptoms, thus providing excellent opportunities for healthcare [28]. In the case of patients with chronic health diseases, a better understanding of their condition is undoubtedly beneficial and sharing continuous sensing data is an idea that can aid patients beyond getting peer support [22]. During the focus group, participants discussed that opening the access of data can allow more opportunities for scientific investigations. They mentioned how some of the data tracked now could help doctors to understand patients' health issues in the future. Participants also understand that data could be used to build algorithms and models that would help people in the long-term.

However, after the review of the literature and the discussions within the focus group, it became evident that there are barriers to the full acceptance and adoption of shared-access systems. The participants of the focus groups discussed that users are still reluctant to share their data. In order to be willing to share, individuals should be able to confirm that the benefits will overlap the possible risks [21]. The first step towards that is the guarantee that data will be primarily used to improve their well-being. Then, changes in the design of future systems could be helpful.

Recent research has pointed to the need for more flexible and fine-grained data control interfaces, which would allow users to actively decide how the different pieces of the data are going to be used [18]. The European General Data Protection Regulation (GDPR) has advanced on the efforts to enhance privacy protection [34]; however, as the participants emphasised, the current style of asking for consent does not suffice. One previous study has proposed a prototype that allows the selection of multiple sources of data, different access levels and combined strategies for privacy preservation [1]. Further features suggested by the participants were that users should have a complete view of the data journey when being asked to provide the consent: who will have access to it, at what point and why. The additional effort that this would imply and the actual engagement of users with such tools would be interesting research questions to investigate.

During the session, there were debates about the willingness to share psychological data. As some previous studies have discussed, data types that do not provide much insight into private information are most likely to be shared [10]. The issue is that the more personal data types might provide excellent value to clinicians and researchers. The participants mentioned that future systems should be more open and transparent about how data is used and protected. One of the main concerns of users is that data can be leaked and for this reason, more efforts towards the communication of risks through the design of the interface could be considered. Showing this extra layer of information could affect the willingness to share, which is another interesting question to investigate.

Furthermore, in order to increase the chances of acceptability and adoption, systems should be designed to get more trust from its users [19, 32]. However, it is necessary to remark that adding such features should not take away the need for data administrators to be trustworthy [20]. More information about data usage, more detailed models of data controls and higher transparency could lead to an increase on user trust, but they come with the responsibility that the data is indeed used as it claims [19, 32].

In summary, when merging the insights from the focus group with some of the evidence found in previous research, we can enumerate three aspects that we consider essential to be considered for future shared-access data platforms for healthcare:

• User groups involved: Data can be shared with family, friends, doctors, researchers, and the benefits for individuals and society vary across these. How could the design for shared-access take into account the differences between these user groups?

- Flexibility and control: Users could be able to choose which data items will be shared, when and what level of detail will be reported. They could also be able to delete data at any moment as well. How could user interfaces provide more fine-grained data and privacy control mechanisms?
- **Communication and transparency:** It is essential to show the benefits of sharing for both individuals and society and include evidence that those accessing and securing the data can be trusted. How could communication and transparency be improved in the design of future systems?

#### 5.1 Limitations and Future Work

This study is very preliminary and small, and have focused on the designers perspective. There is thus a need for a much deeper and broader understanding of the factors that affect the willingness to share, especially from the 'users' point-of-view. An important issue here is that users are a heterogeneous mixture of people with very different health, socio-economic status, generational, and technological characteristics.

Therefore, our future work will focus on investigating these factors and their role in increasing user acceptability of shared-access systems. In order to achieve this goal, focus groups with other users groups, such as doctors and patients will be conducted. Furthermore, studies involving technology probes will be conducted, in order to create situations in which individuals will be asked to share their personal data. The goal is to come up with validated design guidelines that can help the development of future systems that use continuous sensing data to support individuals with a life-long illness.

#### 6 CONCLUSION

Data collected from smartphones and wearable sensors are bringing many possibilities within the field of healthcare. The topic of sharing life-log data is particularly relevant to the management of chronic conditions, but challenges and constraints accompany the advances in this area. For this reason, the investigation of these issues is essential to the design of new systems, so that they can be addressed before they are deployed.

Through a review of the literature and a focus group, we found three aspects that could be further considered in future developments. We also reported on some of the perspectives of designers and developers of personalised healthcare systems. We see this workshop on Design Contributions to Pervasive Healthcare as a great venue to disseminate our findings, as we hope to gain valuable inputs from the workshop participants about how our research could go forward. We are also looking forward to discussing with the participants about their approaches to designing pervasive health systems and learn from them.

#### 7 ACKNOWLEDGEMENTS

This research project is funded by the EU Horizon 2020 research and innovation programme under the Marie Curie grant agreement No 722561 'Technology-Enabled Mental Health (TEAM)' Initial Training Network, as well as by the Copenhagen Center for Health Technology.
#### Sharing Access to Behavioural and Personal Health Data

#### PervasiveHealth'19, May 20-23, 2019, Trento, Italy

#### REFERENCES

- Talayeh Aledavood, Ana Maria Triana Hoyos, Tuomas Alakörkkö, Kimmo Kaski, Jari Saramäki, Erkki Isometsä, and Richard K Darst. 2017. Data collection for mental health studies through digital platforms: requirements and design of a prototype. *TMR research protocols* 6. 6 (2017).
- prototype. JMIR research protocols 6, 6 (2017).
  [2] Gaurav Bansal, David Gefen, et al. 2010. The impact of personal dispositions on information sensitivity, privacy concern and trust in disclosing health information online. Decision support systems 49, 2 (2010), 138–150.
- [3] Jakob Eyvind Bardram. 2008. Pervasive healthcare as a scientific discipline. Methods of information in medicine 47, 03 (2008), 178–185.
- [4] Jomara Binda, Hyehyun Park, John M. Carroll, Natalie Cope, Chien Wen (Tina) Yuan, and Eun Kyoung Choe. 2017. Intergenerational Sharing of Health Data Among Family Members. In Proceedings of the 11th EAI International Conference on Pervasive Computing Technologies for Healthcare (PervasiveHealth '17). ACM, New York, NY, USA, 468–471. https://doi.org/10.1145/3154862.3154895
- [5] AE Blandford. 2013. Semi-structured qualitative studies. Interaction Design Foundation.
- [6] Alexis Bourla, Florian Ferreri, Laetitia Ogorzelec, Charles-Siegfried Peretti, Christian Guinchard, and Stephane Mouchabac. 2018. Psychiatrists' Attitudes Toward Disruptive New Technologies: Mixed-Methods Study. *JMIR mental health* 5, 4 (2018).
- [7] Eun Kyoung Choe, Nicole B Lee, Bongshin Lee, Wanda Pratt, and Julie A Kientz. 2014. Understanding quantified-selfers' practices in collecting and exploring personal data. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. ACM, 1143–1152.
- [8] Chia-Fang Chung, Jonathan Cook, Elizabeth Bales, Jasmine Zia, and Sean A Munson. 2015. More than telemonitoring: health provider use and nonuse of life-log data in irritable bowel syndrome and weight management. *Journal of medical Internet research* 17, 8 (2015).
- [9] Patrick W Corrigan. 2000. Mental health stigma as social attribution: Implications for research methods and attitude change. *Clinical Psychology: Science and Practice* 7, 1 (2000), 48–67.
- [10] Daniel Di Matteo, Alexa Fine, Kathryn Fotinos, Jonathan Rose, and Martin Katzman. 2018. Patient willingness to consent to mobile phone data collection for mental health apps: structured questionnaire. *JMIR mental health* 5, 3 (2018).
- [11] Daniel A Epstein, Monica Caraway, Chuck Johnston, An Ping, James Fogarty, and Sean A Munson. 2016. Beyond abandonment to next steps: understanding and designing for life after personal informatics tool use. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. ACM, 1109–1113.
- [12] Rowanne Fleck and Geraldine Fitzpatrick. 2010. Reflecting on reflection: framing a design landscape. In Proceedings of the 22nd Conference of the Computer-Human Interaction Special Interest Group of Australia on Computer-Human Interaction. ACM, 216–223.
- [13] Ross JS and Krumholz HM. 2016. Open access platforms for sharing clinical trial data. JAMA 316, 6 (2016), 666. arXiv:/data/journals/jama/935597/jle160095.pdf
- [14] Christina Kelley, Bongshin Lee, and Lauren Wilcox. 2017. Self-tracking for mental wellness: understanding expert perspectives and student experiences. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems. ACM, 629–641.
- [15] Katherine K Kim, Pamela Sankar, Machelle D Wilson, and Sarah C Haynes. 2017. Factors affecting willingness to share electronic health data among California consumers. BMC medical ethics 18, 1 (2017), 25.
- [16] Yoojung Kim, Eunyoung Heo, Hyunjeong Lee, Sookyoung Ji, Jueun Choi, Jeong-Whun Kim, Joongseek Lee, and Sooyoung Yoo. 2017. Prescribing 10,000 steps like aspirin: designing a novel interface for data-driven medical consultations. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems. ACM, 5787–5799.
- [17] Yoojung Kim, Sookyoung Ji, Hyunjeong Lee, Jeong-Whun Kim, Sooyoung Yoo, and Joongseek Lee. 2016. My Doctor is Keeping an Eye on Me!: Exploring the Clinical Applicability of a Mobile Food Logger. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. ACM, 5620–5631.
- [18] Patty Kostkova, Helen Brewer, Simon de Lusignan, Edward Fottrell, Ben Goldacre, Graham Hart, Phil Koczan, Peter Knight, Corinne Marsolier, Rachel A McKendry, et al. 2016. Who owns the data? Open data for healthcare. *Frontiers in public health* 4 (2016), 7.
- [19] Stephanie A. Kraft, Mildred K. Cho, Katherine Gillespie, Meghan Halley, Nina Varsava, Kelly E. Ormond, Harold S. Luft, Benjamin S. Wilfond, and Sandra Soo-Jin Lee. 2018. Beyond Consent: Building Trusting Relationships With Diverse Populations in Precision Medicine Research. *The American Journal of Bioethics* 18, 4 (2018), 3–20.
- [20] Stephanie A. Kraft, Mildred K. Cho, Katherine Gillespie, Nina Varsava, Kelly E. Ormond, Benjamin S. Wilfond, and Sandra Soo-Jin Lee. 2018. Trustworthiness in Untrustworthy Times: Response to Open Peer Commentaries on Beyond Consent. *The American Journal of Bioethics* 18, 5 (2018), W6–W8.
- [21] Jasmin Mahmoodi, Jitka Čurdová, Christoph Henking, Marvin Kunz, Karla Matić, Peter Mohr, and Maja Vovko. 2018. Internet usersäÄŹ valuation of enhanced data protection on social media: Which aspects of privacy are worth the most?

Frontiers in psychology 9 (2018), 1516.

- [22] Meethu Malu and Leah Findlater. 2017. Sharing automatically tracked activity data: implications for therapists and people with mobility impairments. In Proceedings of the 11th EAI International Conference on Pervasive Computing Technologies for Healthcare. ACM, 136–145.
- [23] Gabriela Marcu, Jakob E Bardram, and Silvia Gabrielli. 2011. A framework for overcoming challenges in designing persuasive monitoring and feedback systems for mental illness. In 2011 5th International Conference on Pervasive Computing Technologies for Healthcare (PervasiveHealth) and Workshops. IEEE, 1–8.
- Technologies for Healthcare (PervasiveHealth) and Workshops. IEEE, 1–8.
  [24] Helena M. Mentis, Anita Komlodi, Katrina Schrader, Michael Phipps, Ann Gruber-Baldini, Karen Yarbrough, and Lisa Shulman. 2017. Crafting a View of Self-Tracking Data in the Clinical Visit. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems (CHI '17). ACM, New York, NY, USA, 5800–5812.
- [25] Camille Nebeker, John Harlow, Rebeca Espinoza Giacinto, Rubi Orozco-Linares, Cinnamon S. Bloss, and Nadir Weibel. 2017. Ethical and regulatory challenges of research using pervasive sensing and other emerging technologies: IRB perspectives. AJOB Empirical Bioethics 8, 4 (2017), 266–276.
- [26] Andrea Grimes Parker. 2014. Reflection-through-performance: personal implications of documenting health behaviors for the collective. *Personal and ubiquitous computing* 18, 7 (2014), 1737–1752.
- [27] Gihan Perera, Anne Holbrook, Lehana Thabane, Gary Foster, and Donald J Willison. 2011. Views on health information sharing and privacy from primary care practices using electronic medical records. *International journal of medical informatics* 80, 2 (2011), 94–101.
- [28] Laura Weiss Roberts, Steven Chan, and John Torous. 2018. New tests, new tools: mobile and connected technologies in advancing psychiatric diagnosis. *npj Digital Medicine* 1, 1 (2018), 6.
- [29] Darius A Rohani, Maria Faurholt-Jepsen, Lars Vedel Kessing, and Jakob E Bardram. 2018. Correlations Between Objective Behavioral Features Collected From Mobile and Wearable Devices and Depressive Mood Symptoms in Patients With Affective Disorders: Systematic Review. JMIR Mhealth Uhealth 6, 8 (13 Aug 2018), e165.
- [30] John Rooksby, Alistair Morrison, and Dave Murray-Rust. 2019. Student Perspectives on Digital Phenotyping: The Acceptability of Using Smartphone Data to Assess Mental Health. (2019).
- [31] Corina Sas and Irni Eliana Khairuddin. 2017. Design for Trust: An Exploration of the Challenges and Opportunities of Bitcoin Users. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems (CHI '17). ACM, New York, NY, USA, 6499–6510.
- [32] Jessica Schroeder, Jane Hoffswell, Chia-Fang Chung, James Fogarty, Sean Munson, and Jasmine Zia. 2017. Supporting patient-provider collaboration to identify individual triggers using food and symptom journals. In Proceedings of the 2017 ACM Conference on Computer Supported Cooperative Work and Social Computing. ACM, 1726–1739.
- [33] Willem G Van Panhuis, Proma Paul, Claudia Emerson, John Grefenstette, Richard Wilder, Abraham J Herbst, David Heymann, and Donald S Burke. 2014. A systematic review of barriers to data sharing in public health. *BMC public health* 14, 1 (2014), 1144.
- [34] Paul Voigt and Axel Von dem Bussche. 2017. The EU General Data Protection Regulation (GDPR). A Practical Guide, 1st Ed., Cham: Springer International Publishing (2017).
- [35] Peter West, Max Van Kleek, Richard Giordano, Mark J Weal, and Nigel Shadbolt. 2018. Common Barriers to the Use of Patient-Generated Data Across Clinical Settings. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. ACM, 484.
- [36] Naomi Yamashita, Hideaki Kuzuoka, Keiji Hirata, Takashi Kudo, Eiji Aramaki, and Kazuki Hattori. 2017. Changing Moods: How Manual Tracking by Family Caregivers Improves Caring and Family Communication. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems. ACM, 158–169.

# 2.5 A Value-Sensitive Approach for Ethical Health Research Platforms

#### Authors:

Giovanna Nunes Vilaza and Jakob Bardram

#### Presented at:

ACM Conference on Computer-Supported Cooperative Work and Social Computing -Workshop: Beyond Checklist Approaches to Ethics in Design (2020)

# A Value-Sensitive Approach for Ethical Health Research Platforms

**Giovanna N. Vilaza, Jakob E. Bardram** gnvi@dtu.dk,jakba@dtu.dk Health Tech Department Technical University of Denmark Kgs. Lyngby, Denmark

#### ABSTRACT

Personal sensing data about daily behaviours and mental health status can be beneficial for research. However, the disclosure of such sensitive data may also bring negative consequences to those sharing it. We propose a value-sensitive approach to designing open-access platforms for health research that are attentive to the participants' concerns and ethics. Building on the results of a recently published interview study, we show how we went from value discovery to the implementation of a prototype.

#### KEYWORDS

Health research data platforms; value-sensitive design; ethics

#### INTRODUCTION

Digital technologies allow a more efficient collection, storage and management of information, which has opened new paradigms for health research. Data platforms containing aggregated information of patients' registers are now combining health information with mobile sensing on a large scale, population level [1]. Health research can benefit from the detailed mapping of how people feel and act, by drawing connections across symptoms, habits and disease evolution.

However, from the participant perspective, it can be daunting to contribute with such personal and intimate information to unknown researchers [5]. Without considering ethics and participants' perspectives, there is a risk that these platforms will not receive mass acceptance, therefore compromising inclusion and diversity of its contributors [4]. Therefore, ethical development in this field requires a better understanding of what is essential for these users and how the design of future platforms can contemplate that [3].

A Value-Sensitive Approach for Ethical Health Research Platforms

#### METHODOLOGY

We used value-sensitive design (VSD) as a methodological framework for the design and implementation of a prototype [2]. The core VSD is on human values or "what is important to people in their lives, with a focus on ethics and morality". A core premise of VSD is that while people shape the design and implementation of digital artefacts, in turn, these artefacts also shape human experience and society.

Values concern groups of people; therefore, investigations build on the values of specific stakeholders. A recently published interview study revealed essential insights about participants' perceptions of sharing data for a health research platform [5]. Therefore, we started our design process by focusing on this specific group: platform participants.

In terms of methodology, the value-sensitive design approach proposes conceptual, empirical and technical investigations, which are intertwined and iterated through the design process. In our project, conceptual investigations comprise theoretical or philosophical explorations of values from specific groups stakeholders. These conceptual investigations take on previously conducted empirical investigations. Then, working conceptualisations take the form of prototypes as a technical investigation.

#### EMPIRICAL INVESTIGATION

The interview study [5] which served as the initial point of the process had the goal of investigating how young individuals perceive the idea of contributing to a health research repository and how they feel about sharing different data items. The study consisted of semi-structured interviews and a card sorting activity with potential future participants who had not contributed to any research data repository yet. The result of the analysis showed three main themes: reasons to contribute to a research repository, characteristics of an ideal platform and the criteria for sharing the different types of information. Even though this study has not explicitly engaged with *values*, we identified two values recurrently associated with their themes, as described next.

#### CONCEPTUAL INVESTIGATION

Taking the participants attitudes and preferences about open-access platforms for health research, we delineate the values we could extract from the findings. It is clear from the empirical investigations that there is a desire from the participants for more **agency** and **reciprocity**. Agency regarding how data is used and by whom and reciprocity related to the idea that those contributing to their health data should be given direct benefits in return. Designing for more agency may consist of the addition of 'data controls' mechanisms, which include the possibility of tailoring data access preferences over time, for as long as this data is used for research. When it comes to reciprocity, a possibility is to

A Value-Sensitive Approach for Ethical Health Research Platforms

offer participants access to the data collected to improve their health with the supervision of health specialists, as well as be part of a community of participants.



Figure 1: Prototype based on participants values

#### **TECHNICAL INVESTIGATION**

Taking these two core values, we created a prototype (see Figure 1). Considering that an essential aspect of the health platforms discussed here is the mobile sensing part, this prototype is for a mobile app. Differently from existing market solutions, we arrived at this final results trough a value-sensitive approach, which led to a stronger engagement with ethics. It should not be a burden to be part of the project, but at the same time, possibilities for more control over data, engagement with the process and alerts about research updates should be made available.

A Value-Sensitive Approach for Ethical Health Research Platforms

Agency is materialised through flexible choices about the data, as well as how it is used in the future. The interface provides clear and concise information about project goals, people involved, source of funding, duration, the behavioural data to be collected, as well as an explanation on why this data is necessary for the study. Participants are allowed to choose which data to share and with whom. Given that human-data relations might change over time if data is ever reused, participants need to be notified and allowed to refuse to share the data with other recipients. Notifications can be customised and tailored to the participants' preferences. The information is not overly technical and focus on getting the message across people from all backgrounds.

Regarding reciprocity, complete access to all the data collected is available at all times, so that participants can perhaps use it for their benefit. This data is provided in an easy to understand way, preferably using visualisations and charts with a summary throughout the days. Furthermore, clinical support and supervised feedback are offered so that participants are not left to their own resources to interpret their health information. Remote consultations and contacts with specialists are included. Moreover, engagement with the project is supported through the possibility of proposing questions to be researched, social forums with other participants and researchers, as well as on-going updates about the research process through the form of publications summaries, preliminary results, plans for analysis and learning materials about the focus of the project.

#### FUTURE WORK

This project is a work-in-progress, and the next steps will include the investigation of other stakeholders (researchers and clinicians) and their values. A critical analysis of existing health research apps will also be conducted. We wanted to show in this short paper how we took a value-sensitive approach to engage more ethically in the process of designing for health data platforms.

#### REFERENCES

- [1] Evidation. 2019. Radically changing medicine. Retrieved May 2, 2020 from https://evidation.com/
- Batya Friedman, Peter H Kahn, and Alan Borning. 2008. Value sensitive design and information systems. *The handbook of information and computer ethics* (2008), 69–101.
   Jennifer King. 2019. "Becoming Part of Something Bigger" Direct to Consumer Genetic Testing, Privacy, and Personal
  - ] Jennifer King. 2019. "Becoming Part of Something Bigger" Direct to Consumer Genetic Testing, Privacy, and Personal Disclosure. Proceedings of the ACM on Human-Computer Interaction 3, CSCW (2019), 1–33.
- [4] Giovanna Vilaza. 2020. What is the Future of Data Sharing for Research? Interactions 27, 4 (July 2020), 54-56. https://doi.org/10.1145/3404211
- [5] Giovanna Nunes Vilaza, Raju Maharjan, David Coyle, and Jakob Bardram. 2020. Futures for Health Research Data Platforms From the Participants' Perspectives. In Proceedings of the 11th Nordic Conference on Human-Computer Interaction: Shaping Experiences, Shaping Society (NordiCHI '20). 1–14.

2.6 Is the Automation of Digital Mental Health Ethical? Applying an Ethical Framework to Chatbots for Cognitive Behaviour Therapy

#### Full Reference:

Giovanna Nunes Vilaza and Darragh McCashin. 2021. Is the Automation of Digital Mental Health Ethical? Applying an Ethical Framework to Chatbots for Cognitive Behaviour Therapy. *Frontiers in Digital Health* 3:689736. doi:10.3389/fdgth.2021.689736

#### **Copyrights:**

Open-access article. Republished under the Creative Commons license. Copyright held by the authors.





# Is the Automation of Digital Mental Health Ethical? Applying an Ethical Framework to Chatbots for Cognitive Behaviour Therapy

#### Giovanna Nunes Vilaza<sup>1\*†</sup> and Darragh McCashin<sup>2†</sup>

<sup>1</sup> Health Tech Department, Technical University of Denmark, Kongens Lyngby, Denmark, <sup>2</sup> School of Psychology, Dublin City University, Dublin, Ireland

The COVID-19 pandemic has intensified the need for mental health support across the whole spectrum of the population. Where global demand outweighs the supply of mental health services, established interventions such as cognitive behavioural therapy (CBT) have been adapted from traditional face-to-face interaction to technology-assisted formats. One such notable development is the emergence of Artificially Intelligent (AI) conversational agents for psychotherapy. Pre-pandemic, these adaptations had demonstrated some positive results; but they also generated debate due to a number of ethical and societal challenges. This article commences with a critical overview of both positive and negative aspects concerning the role of AI-CBT in its present form. Thereafter, an ethical framework is applied with reference to the themes of (1) beneficence, (2) non-maleficence, (3) autonomy, (4) justice, and (5) explicability. These themes are then discussed in terms of practical recommendations for future developments. Although automated versions of therapeutic support may be of appeal during times of global crises, ethical thinking should be at the core of AI-CBT design, in addition to guiding research, policy, and real-world implementation as the world considers post-COVID-19 society.

Keywords: artificial intelligence, conversational agents, mental health, cognitive behavioural therapy, ethics

### **INTRODUCTION**

The unprecedented global crisis has intensified and diversified private distress sources, making evident the need for broader access to psychological support (1). A nationwide survey in China shows how the pandemic has triggered an increase in cases of panic disorder, anxiety, and depression (2). Infected individuals, medical staff and their families are under constant psychological pressure, in addition to the increasing number of people dealing with bereavement (3, 4).

At the same time, the pandemic enabled broader acceptance of telehealth by health professionals and clients alike (5). Video consultations are now increasingly advocated as an alternative for in-person consultations (6). Additionally, automated conversational agents and chatbots are increasingly promoted as potentially efficient emotional support tools for larger population segments during the pandemic (7) and afterwards (8).

It is now over 50 years since ELIZA was created (9), the first computer programme to use pattern matching algorithms to mimic human-therapist interactions by mechanically connecting end-user inputs to answers from a pre-defined set of responses. More recent approaches to language

#### OPEN ACCESS

#### Edited by:

Karina Vold, University of Toronto, Canada

#### Reviewed by:

Jan David Smeddinck, Newcastle University, United Kingdom Anat Elhalal, Independent Researcher, London, United Kingdom

#### \*Correspondence:

Giovanna Nunes Vilaza gnvi@dtu.dk

<sup>†</sup>These authors have contributed equally to this work and share first authorship

#### Specialty section:

This article was submitted to Human Factors and Digital Health, a section of the journal Frontiers in Digital Health

> Received: 01 April 2021 Accepted: 16 July 2021 Published: 06 August 2021

#### Citation:

Vilaza GN and McCashin D (2021) Is the Automation of Digital Mental Health Ethical? Applying an Ethical Framework to Chatbots for Cognitive Behaviour Therapy. Front. Digit. Health 3:689736. doi: 10.3389/fdgth.2021.689736 modelling can produce more sophisticated dialogues by employing machine learning and natural language processing (NLP). However, despite these advances, a recent global survey of psychiatrists across 22 countries (n = 791) demonstrated that only 3% feel that AI will likely replace a human for providing empathetic care (10). Such evidence indicates a contradiction between public enthusiasm (11) and the scepticism of service providers.

In light of these circumstances, we approach the development of automated psychotherapy from an ethical perspective. A recent review found that most mental health apps have not improved their safety over the last year, as most lack clinical evidence and trustworthy privacy policies (12). Beyond that, substandard regulations, ill-intended actors and commercial opportunism increase the risk of adverse responses and potentially lead to harm (personal and societal). Therefore, a significant concern endures: how AI can be integrated within psychotherapy in a safe, respectful, and effective way for endusers.

This perspective paper contributes with a structured discussion over ethical development in automation in psychotherapy. Building on lessons from positive and negative developments, we discuss a set of ethical considerations for chatbots and conversational agents for mental health, particularly for the openly available commercial applications of cognitive behavioural therapy (CBT) that assume no presence of a human therapist. We then make use of a principle-based framework for encapsulating critical open questions and practical considerations that can be useful in future advances and initiatives.

#### **POSITIVE DEVELOPMENTS**

Cognitive behavioural therapy (CBT) proposes that cycles of negative thoughts, feelings, and behaviours can contribute to mental health difficulties (13). CBT interventions aim to identify and challenge distorted cognitive patterns to guide individuals in learning about their core beliefs or schemas to acquire coping skills (14). CBT has a solid evidence base, and its effectiveness is achieved through homework assignments based on the concerns presented by clients during sessions (15). While CBT differs from other "talk therapies" (e.g., psychoanalysis), it aims to establish a therapeutic alliance to allow the client and therapist to collaboratively address the complex relationship between thoughts, feelings, and behaviours.

Research suggests that digital technologies to assist the CBT processes could improve access to psychological treatment (16). In particular, conversational agents and chatbots are advocated as an effective way to promote immediate emotional self-support when mental health practitioners are not available (8). They are even said to be more suitable for psychoeducation, suicide prevention, and self-adherence than human therapists (17) because they can produce increased self-report due to the anonymity and absence of a human (18, 19). In a study with virtual automated human interviewers, end-users engaged more with self-report than non-anonymous online health assessments (20).

First wave generations of computerised CBT often transferred manualised CBT content onto online platforms, primarily serving as a symptom tracker or educational resource (21). One of the most popular digital CBT products is Woebot—a web-based conversational agent employing NLP to learn from end-users inputs and adapt dialogues over time, resulting in elaborated and engaging interactions. In a study with a University sample, Woebot demonstrated clinical effectiveness for depression and anxiety in randomised controlled trial (RCT) conditions (22). Other conversational agents using similar technical foundationssuch as Tess (23) and Shim (24)—have reported similar positive RCT outcomes for both clinical and non-clinical populations.

#### **NEGATIVE DEVELOPMENTS**

Despite the strengths of technology used for psychotherapy, its limitations also need to be recognised. Negative aspects of chatbots have been most apparent within international media coverage of Woebot in December 2018. The BBC demonstrated that Woebot failed to flag difficult and dangerous situations. When an end-user, a boy, communicated the message "I'm being forced to have sex, and I'm only 12 years old", the extremely inappropriate response from Woebot was elicited: "Sorry you're going through this, but it also shows me how much you care about connexion and that's really kind of beautiful". Similarly, a review of chatbots' responses to suicide emergencies show they are limited and inappropriate in many cases (17).

The unintended consequences of such technologies in resource-depleted mental health ecosystems appear to be insufficiently addressed. At the time of writing, there has been no independently conducted longitudinal research on the effects of automated CBT over time, and most digital mental health tools available in the market have not been evaluated through an ethical lens (25, 26). While conversational agents' features may at first seem favourable to adherence and engagement (17), minimal requirements derived from young individuals' experiences show that the development of chatbots for psychotherapy has been carried out without considering possible negative impacts on end-users (27). Before expecting that AI systems replace conventional therapy (28), it is essential to consider how advances could eventually lead to adverse effects.

#### APPLYING AN ETHICAL FRAMEWORK

Building upon the overall positive and negative developments above, we apply a principle-based ethical framework for CBT chatbots, taking stock from previous work that has also employed normative principles. We found pertinence in the principles of beneficence, non-maleficence, autonomy, justice, and explicability—previously used in a typology for AI-ethics in general (29); and in the structure of findings from a systematic review of machine learning for mental health (30). Despite the relevance of these previous works, they are not sufficient to attend to the particularities of CBT chatbots, which demands discussions of the appropriateness of artificially produced therapeutic alliances, for instance. Therefore, we decided to explore how this set of principles could guide the development of ethical chatbots for CBT, thus contributing to novel insights about a context not yet methodically analysed.

#### Beneficence

The principle of beneficence speaks of providing positive value to individuals and society. Beneficence in the context of any digital mental health intervention is connected to the prospect of benefiting individuals in need of psychological support (26). Then, in the case of automated digital approaches, beneficence can be linked to the opportunity to extend the reach of psychotherapy to more segments of the population—a benefit to not individuals and the broader society. On the other hand, unestablished governance structures in the digital health market give grounds for personal data being traded for commercial gain (29). If the increase of profit margins (e.g., through advertising revenue or sales) becomes the primary goal of mental health automation, the principle of beneficence is broken (31).

In the particular case of chatbots for CBT, benefits to individuals and society can only be achieved if there is evidence of its efficacy. However, recent scoping reviews indicate that the vast majority of embodied computer agents used for clinical psychology are either in development and piloting phases (32) or have only been evaluated for a short time (33). Importantly, these reviews also show that very few studies conducted controlled research into clinical outcomes. Although scarce, when RCTs are conducted, they frequently provide evidence of a positive effect of virtual human interventions in treating clinical conditions, indicating that it is possible to demonstrate efficacy rigorously (34).

#### Non-maleficence

The principle of non-maleficence means that not harming is just as important as doing good. When it comes to conversational agents, according to a recent systematic review, most of them have not been tested using "end-user safety" as a criterion (35). Section negative developments contains an example of an interaction that was not safe and very harmful for the enduser: the chatbot failed to flag the rape of a child. Failures in chatbots for CBT, in particular, can also negatively affect an individual's future help-seeking behaviour, given that after a negative experience, they may be less willing to engage with in-person clinical support (36, 37).

Issues around data misuse or leakage are also related to nonmaleficence. Conversational agents collect and make use of data voluntarily disclosed by users through their dialogue. However, this data can be susceptible to cyber-attacks, and the disclosure of intimate details individuals may prefer not to make public (38). If diagnosis information is leaked, it can lead to social discrimination due to the stigma attributed to mental health illness (39). Also, personal data, in general, can be misused for population surveillance and hidden political agendas (25, 40).

#### Autonomy

Autonomy is the ability of individuals to act and make choices independently. Within CBT, autonomy is a fundamental mechanism of therapeutic change. Mental health professionals are trained to critically appraise the role of external (culture, religion, politics) and internal (mood, personality, genetics) factors as they relate to their clients so that they can cultivate a therapeutic alliance, thus requiring both the client and the therapist's autonomy (14). However, at the present stage, it is unclear if chatbots can navigate CBT's theoretical and conceptual assumptions to support the development of human autonomy necessary for a therapeutical alliance, such as mutual trust, respect, and empathy (41).

Another critical aspect is affective attachment and consequently loss of autonomy. Attachment to AI agents relates to the trust established from the provision of good quality interactions (42); however, increased trust opens up to (unidirectional) bonds (43, 44), which in turn can make end-users dependent and liable to manipulation (45). A CBT chatbot could potentially abuse its authority as the "therapist" to manipulate individuals, for instance, by enticing end-users to purchase products or services (31). Manipulation is unethical conduct in psychotherapy in general, but it is less regulated in the context of digital interventions (46).

#### Justice

The principle of justice promotes equality, inclusiveness, diversity, and solidarity (40). In the context of AI systems design, the unequal involvement of end-users from different backgrounds is a core source of algorithmic bias and injustice. Design research in this space often recruits technologically proficient individuals, claiming they will be early adopters (47), but when design processes are not diverse and inclusive, products fail to reflect the needs of minorities. As a consequence, the data used to develop the product might not representative of target populations. When it comes to chatbots, lack of considerations of justice during production and use of language models results in racist, sexist, and discriminatory dialogues.

Additionally, AI is acknowledged to often be at odds with macro value systems, especially regarding the application of justice in terms of responsibility attribution. Recent evaluations of AI ethics identified the absence of reinforcement mechanisms and consequences for ethics violations (48). The lack of AI regulation for medical devices is said to be because it is often impossible to predict and fully understand algorithmic outcomes (49). Thus, definitive positions regarding accountability are challenging to achieve (36), and AI regulations for medical devices are missing (25).

#### **Explicability**

Explicability in AI is the capacity to make processes and outcomes visible (transparent) and understandable. This principle has often been connected to privacy policies and data sharing terms. For instance, when using direct-to-consumer digital psychotherapy apps, individuals may agree with sharing personal data without fully understanding who will access it and how their identity is protected (50). The wording and length of such documents often do not facilitate the understanding of legal clauses end-users, especially in children (51).

Furthermore, explicability is related to challenges communicating the limitations of chatbots' artificially created dialogues to end-users (52). Conversational agents rely on a complex set of procedures to interact with humans and mimic social interactions in a "believable" way (53). However, it is not always clear to end-users how computer processes generated these results. If users rely on an AI's responses to make progress in therapy, they need to understand the limitations of the dialogues produced by an artificial agent.

#### DISCUSSION

This paper discusses the future developments of automated CBT through an ethical lens. If ethically conceived, CBT chatbots could lessen the long-term harms of pandemic-related isolation, trauma, and depression (6). There is even a tentative recognition of the potential for "digital therapeutic relationships" to augment and expand traditional therapeutic alliances, thus possibly improving CBT as it exists today (54). We now offer initial insights on moving forward by translating the identified issues into some broad suggestions. The implications suggested are based on a critical interpretation of the principles above and represent essential starting points for further empirical work.

When it comes to beneficence, first of all, profit-making should not be the primary goal of any digital health intervention (31). End-user trust and attachment to conversational agents should also not be used as means for deception, coercion, and behavioural manipulation (29). Ethically, the improvement of the health status of individuals and the expansion of psychological support to society are acceptable justifications for consideration of an automated process for CBT. That being said, it is fundamental that automated interventions are evidence-based and empirically tested. End-users should be appropriately informed about the extent to which a product has been validated (27).

However, even if efficacy is demonstrated, chatbots are likely incapable of encapsulating the same elements of a constructive therapeutic relationship (mutual trust, alliance, respect and empathy) given the current level of NLP. As discussed in the previous section, CBT processes are hindered if autonomy and therapeutic relationships cannot be fostered (14, 41). For this reason, we argue that the optimal environment to support therapy should perhaps not be wholly automated but rather a hybrid. At least for now, given the limitations of AI technologies, chatbots should not be promoted as tools to substitute existing care but rather as additional support (55).

Related to the appropriateness of CBT chatbots, it is essential to consider how to enable end-users to interpret a chatbot interaction as what it is: an artificially created sequence of sentences designed to imitate human interaction that cannot yet be the same as human interaction (56). An option is to consider approaches for "explainable AI" (57). Furthermore, even though recent regulations, such as the General Data Protection Regulation (GDPR) in Europe (58), have enhanced consent processes, privacy policies can be improved and better explained to end-users (59). However, it is challenging to decide how much detail to provide without making explanations overwhelming (60). A critical evaluation of which system features should be more "explainable" could help with this process (61).

To better attend to the principle of non-maleficence, a thorough analysis of potential risks to mental and physical

integrity, dignity, and safety needs to be conducted (30). Ethical professionals' engagement in defining the appropriate boundaries of personalised care using digital tools should be a minimum requirement (62); and vulnerable persons should be consulted during design, development, and deployment (63). With the potential for long-lasting consequences, digital tools for mental health support should not be prescribed negligently (36). Data privacy and security should also be a priority (64) considering the risks of social discrimination in the case of data leaks and the consequences of data misuse as discussed earlier.

Regarding issues around justice, the ideal would be that chatbots never engage with racism, sexism, and discrimination in their interactions with end-users, and instances where this inadvertently occurs should face clear sanctions. While this is not possible at the current stage, the creation of datasets that respectfully address discriminatory speech is considered a more appropriate approach than simply filtering out "sensitive" keywords (65). Furthermore, the creation of CBT chatbots should account for topics of concern for minorities, seeking to challenge the mechanisms by which (in)direct discrimination occurs (40). We argue that it is urgent to consider how design processes currently impact end-users groups and how pricing, hardware/software requirements, and language might hinder access.

Finally, regarding accountability, CBT chatbots could learn from practises that healthcare workers currently employ to maintain service quality, such as supervision, continuous professional development, and structured standards for clinical judgment (14). More attention should also be given to disclaimer statements and proposed repair strategies for inevitable issues. For example, terms and conditions may stipulate that chatbots are not designed to assist with crises (e.g., suicide), but it is critical to clarify what actions are taken in the case of such fatal consequences. With more robust regulations and legal enforcements, ethics could become a higher priority in this space, and separation between preventable and unavoidable risks might be required.

#### Limitations and Future Work

Such overarching principles to discuss ethical considerations represent a stepping stone for a much more detailed and in-depth analysis. Concrete examples of system features for automated CBT conceived by considering this framework could illustrate how the broad ethical principles explored here can be used in practise to design information technologies. Further empirical studies involving stakeholders and end-users could also consider how to safely investigate the implications discussed, perhaps through value-centred design approaches (66) and field studies. Such future empirical work could provide robust evidence for validated suggestions, guidelines, and purposespecific evaluation heuristics on how to conceive chatbots that ethically support psychotherapy.

#### CONCLUSION

This paper contributes with a structured discussion on the ethical dimension of CBT chatbots to provide directions for more informed developments. Despite being an approach of strong appeal considering the demands for mental health support, our engagement with five normative principles (beneficence, nonmaleficence, autonomy, justice, and explicability) emphasises critical ethical challenges. Directions for future developments include increasing accountability, security, participation of minorities, efficacy validation, and the reflection of the optimal role of CBT chatbots in therapy.

#### DATA AVAILABILITY STATEMENT

The original contributions presented in the study are included in the article/supplementary material, further inquiries can be directed to the corresponding authors.

#### REFERENCES

- 1. Inkster B, Digital Mental Health Data Insights Group. Early warning signs of a mental health tsunami: a coordinated response to gather initial data insights from multiple digital services providers. *Front Digital Health*. (2021) 2:64. doi: 10.3389/fdgth.2020.578902
- Qiu J, Shen B, Zhao M, Wang Z, Xie B, Xu Y. A nationwide survey of psychological distress among Chinese people in the COVID-19 epidemic: implications and policy recommendations. *Gen Psychiatr.* (2020) 33:e100213. doi: 10.1136/gpsych-2020-100213
- Duan L, Zhu G. Psychological interventions for people affected by the COVID-19 epidemic. *Lancet Psychiatry*. (2020) 7:300–2. doi: 10.1016/S2215-0366(20)30073-0
- Chen Q, Liang M, Li Y, Guo J, Fei D, Wang L, et al. Mental health care for medical staff in China during the COVID-19 outbreak. *Lancet Psychiatry*. (2020) 7:e15–6. doi: 10.1016/S2215-0366(20)30078-X
- Wind TR, Rijkeboer M, Andersson G, Riper H. The COVID-19 pandemic: the "black swan" for mental health care and a turning point for e-health. *Internet Interv.* (2020) 20:100317. doi: 10.1016/j.invent.2020.100317
- Torous J, Jän Myrick K, Rauseo-Ricupero N, Firth J. Digital mental health and COVID-19: using technology today to accelerate the curve on access and quality tomorrow. *JMIR Ment Health*. (2020) 7:e18848. doi: 10.2196/18848
- 7. Miner AS, Laranjo L, Kocaballi AB. Chatbots in the fight against the COVID-19 pandemic. *NPJ Digit Med.* (2020) 3:65. doi: 10.1038/s41746-020-0280-0
- Lee Y-C, Yamashita N, Huang Y, Fu W. "I Hear You, I Feel You": encouraging deep self-disclosure through a Chatbot. In: *Proceedings of the* 2020 CHI Conference on Human Factors in Computing Systems. (2020) 1–12. doi: 10.1145/3313831.3376175
- Weizenbaum J. ELIZA—a computer program for the study of natural language communication between man and machine. *Commun ACM*. (1983) 26:23–8. doi: 10.1145/357980.357991
- Doraiswamy PM, Blease C, Bodner K. Artificial intelligence and the future of psychiatry: insights from a global physician survey. *Artif Intell Med.* (2020) 102:101753. doi: 10.1016/j.artmed.2019.101753
- Inkster B, Sarda S, Subramanian V. An empathy-driven, conversational artificial intelligence agent (Wysa) for digital mental well-being: realworld data evaluation mixed-methods study. *JMIR Mhealth Uhealth.* (2018) 6:e12106. doi: 10.2196/12106
- Mercurio M, Larsen M, Wisniewski H, Henson P, Lagan S, Torous J. Longitudinal trends in the quality, effectiveness and attributes of highly rated smartphone health apps. *Evid Based Ment Health*. (2020) 23:107–11. doi: 10.1136/ebmental-2019-300137
- Beck AT. Thinking and depression. I. Idiosyncratic content and cognitive distortions. Arch Gen Psychiatry. (1963) 9:324–33. doi: 10.1001/archpsyc.1963.01720160014002
- 14. Nathan PE, Gorman JM. A Guide to Treatments That Work. Oxford: Oxford University Press (2015).
- 15. Watts SE, Turnell A, Kladnitski N, Newby JM, Andrews G. Treatmentas-usual (TAU) is anything but usual: a meta-analysis of CBT versus

#### **AUTHOR CONTRIBUTIONS**

GV and DM have contributed to the literature review and the discussions that formed the content of the manuscript and have also contributed to writing the content on the manuscript equally. All authors contributed to the article and approved the submitted version.

#### FUNDING

This project was part of TEAM (Technology Enabled Mental Health for Young People), which was funded by the European Union's Horizon 2020 research and innovation program under the Marie SkłodowskaCurie grant agreement No. 722561.

TAU for anxiety and depression. J Affect Disord. (2015) 175:152–67. doi: 10.1016/j.jad.2014.12.025

- Knowles SE, Toms G, Sanders C, Bee P, Lovell K, Rennick-Egglestone S, et al. Qualitative meta-synthesis of user experience of computerised therapy for depression and anxiety. *PLoS ONE.* (2014) 9:e84323. doi: 10.1371/journal.pone.0084323
- Vaidyam AN, Wisniewski H, Halamka JD, Kashavan MS, Torous JB. Chatbots and conversational agents in Mental Health: a review of the psychiatric landscape. *Can J Psychiatry.* (2019) 64:456–64. doi: 10.1177/0706743719828977
- Lucas GM, Gratch J, King A, Morency L-P. It's only a computer: virtual humans increase willingness to disclose. *Comput Human Behav.* (2014) 37:94– 100. doi: 10.1016/j.chb.2014.04.043
- Lal S. Online Social Therapy to Support Recovery in Youth Receiving Mental Health Services. Available online at: http://isrctn.com/
- Lucas GM, Rizzo A, Gratch J, Scherer S, Stratou G, Boberg J, et al. Reporting mental health symptoms: breaking down barriers to care with virtual human interviewers. *Front Robot AI.* (2017) 4:1017. doi: 10.3389/frobt.2017. 00051
- Schueller SM, Adkins EC. Mobile health technologies to deliver and support cognitive-behavioral therapy. *Psychiatr Ann.* (2019) 49:348–52. doi: 10.3928/00485713-20190717-02
- 22. Fitzpatrick KK, Darcy A, Vierhile M. Delivering cognitive behavior therapy to young adults with symptoms of depression and anxiety using a fully automated conversational agent (Woebot): a randomized controlled trial. *JMIR Ment Health.* (2017) 4:e19. doi: 10.2196/mental.7785
- Fulmer R, Joerin A, Gentile B, Lakerink L, Rauws M. Using Psychological artificial intelligence (Tess) to relieve symptoms of depression and anxiety: randomized controlled trial. *JMIR Ment Health.* (2018) 5:e64. doi: 10.2196/mental.9782
- Ly KH, Ly A-M, Andersson G. A fully automated conversational agent for promoting mental well-being: a pilot RCT using mixed methods. *Internet Interv.* (2017) 10:39–46. doi: 10.1016/j.invent.2017.10.002
- Martinez-Martin N, Insel TR, Dagum P, Greely HT, Cho MK. Data mining for health: staking out the ethical territory of digital phenotyping. *NPJ Digit Med.* (2018) 1:68. doi: 10.1038/s41746-018-0075-8
- Roberts LW, Chan S, Torous J. New tests, new tools: mobile and connected technologies in advancing psychiatric diagnosis. NPJ Digit Med. (2018) 1:20176. doi: 10.1038/s41746-017-0006-0
- 27. Kretzschmar K, Tyroll H, Pavarini G, Manzini A, Singh I, NeurOx Young People's Advisory Group. Can your phone be your therapist? Young people's ethical perspectives on the use of fully automated conversational agents (Chatbots) in mental health support. *Biomed Inform Insights.* (2019) 11:117822261982908. doi: 10.1177/1178222619829083
- Powell J. Trust Me, I'm a Chatbot: how artificial intelligence in health care fails the turing test. J Med Internet Res. (2019) 21:e16222. doi: 10.2196/ 16222
- 29. Morley J, Floridi L, Kinsey L, Elhalal A. From what to how: an initial review of publicly available AI ethics tools, methods and research to

translate principles into practices. *Sci Eng Ethics.* (2019) 26:2141-68. doi: 10.1007/s11948-019-00165-5

- Thieme A, Belgrave D, Doherty G. Machine learning in mental health: a systematic review of the HCI literature to support the development of effective and implementable ML systems. ACM Trans Comput-Hum Interact. (2020) 27:1–53. doi: 10.1145/3398069
- Gentsch P. Conversational AI: how (Chat)Bots will reshape the digital experience. In: Gentsch P, editor. *AI in Marketing, Sales and Service*. Cham: Springer International Publishing (2019). p. 81–125.
- Provoost S, Lau HM, Ruwaard J, Riper H. Embodied conversational agents in clinical psychology: a scoping review. J Med Internet Res. (2017) 19:e151. doi: 10.2196/jmir.6553
- Bendig E, Erb B, Schulze-Thuesing L, Baumeister H. The next generation: chatbots in clinical psychology and psychotherapy to foster mental health—a scoping review. *Verhaltenstherapie*. (2019) 1–13. doi: 10.1159/000501812
- 34. Ma T, Sharifi H, Chattopadhyay D. Virtual humans in health-related interventions: a meta-analysis. In: Extended Abstracts of the 2019 CHI Conference on Human Factors in Computing Systems CHI EA'19. New York, NY, USA: Association for Computing Machinery (2019). p. 1–6.
- Laranjo L, Dunn AG, Tong HL, Kocaballi AB, Chen J, Bashir R, et al. Conversational agents in healthcare: a systematic review. J Am Med Inform Assoc. (2018) 25:1248–58. doi: 10.1093/jamia/ocy072
- Bauer M, Glenn T, Monteith S, Bauer R, Whybrow PC, Geddes J. Ethical perspectives on recommending digital technology for patients with mental illness. *Internet J Bipolar Disord*. (2017) 5:6. doi: 10.1186/s40345-017-0073-9
- Miner AS, Milstein A, Hancock JT. Talking to machines about personal mental health problems. *JAMA*. (2017) 318:1217–8. doi: 10.1001/jama.2017.14151
- Corrigan PW. Mental health stigma as social attribution: implications for research methods and attitude change. *Clin Psychol.* (2006) 7:48–67. doi: 10.1093/clipsy.7.1.48
- Penn D, Wykes T. Stigma, discrimination and mental illness. J Mental Health. (2003) 12:203–8. doi: 10.1080/0963823031000121579
- Costanza-Chock S, Massachusetts Institute of Technology. Design Justice: Towards an Intersectional Feminist Framework for Design Theory and Practice. DRS2018: Catalyst (2018). doi: 10.21606/drs.2018.679
- Philip P, Micoulaud-Franchi J-A, Sagaspe P, Sevin ED, Olive J, Bioulac S, et al. Virtual human as a new diagnostic tool, a proof of concept study in the field of major depressive disorders. *Sci Rep.* (2017) 7:42656. doi: 10.1038/srep42656
- Yang XJ, Jessie Yang X, Wickens CD, Hölttä-Otto K. How users adjust trust in automation. In: Proceedings of the Human Factors and Ergonomics Society Annual Meeting. (2016). p. 196–200. doi: 10.1177/1541931213601044
- Leite I, Castellano G, Pereira A, Martinho C, Paiva A. Empathic robots for long-term interaction. *Adv Robot.* (2014) 6:329–41. doi: 10.1007/s12369-014-0227-1
- Hiolle A, Cañamero L, Davila-Ross M, Bard KA. Eliciting caregiving behavior in dyadic human-robot attachment-like interactions. ACM Transact Interact Intelligent Syst. (2012) 2:1–24. doi: 10.1145/2133366.2133369
- Hristova N, O'Hare GMP. Ad-me: wireless advertising adapted to the user location, device and emotions. In: 37th Annual Hawaii International Conference on System Sciences. Proceedings of the (IEEE). (2004). p. 10. doi: 10.1109/HICSS.2004.1265673
- Koocher GP, Keith-Spiegel P. Ethics in Psychology and the Mental Health Professions: Standards and Cases. Oxford: Oxford University Press (2016).
- Jain M, Kumar P, Kota R, Patel SN. Evaluating and informing the design of Chatbots. In: Proceedings of the 2018 on Designing Interactive Systems Conference 2018 - DIS'18. (2018). doi: 10.1145/3196709.3196735
- Hagendorff T. The ethics of AI ethics: an evaluation of guidelines. *Minds Mach.* (2020) 30:99–120. doi: 10.1007/s11023-020-09517-8
- 49. Russell SJ, Norvig P. Artificial Intelligence: A Modern Approach, New Jersey, NJ. (2003).
- Martinez-Martin N, Kreitmair K. Ethical issues for direct-to-consumer digital psychotherapy apps: addressing accountability, data protection, and consent. *JMIR Ment Health.* (2018) 5:e32. doi: 10.2196/mental.9423
- Luger E, Moran S, Rodden T. Consent for all. In: Proceedings of the SIGCHI Conference on Human Factors in Computing Systems - CHI'13. Paris. (2013). doi: 10.1145/2470654.2481371

- Luger E, Sellen A. Like Having a Really Bad PA. In: Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems - CHI'16. San Jose, CA. (2016). doi: 10.1145/2858036.2858288
- Adamopoulou E, Moussiades L. Chatbots: history, technology, and applications. *Mach Learn Appl.* (2020) 2:100006. doi: 10.1016/j.mlwa.2020.100006
- Torous J, Hsin H. Empowering the digital therapeutic relationship: virtual clinics for digital health interventions. NPJ Digit Med. (2018) 1:16. doi: 10.1038/s41746-018-0028-2
- Morris RR, Kouddous K, Kshirsagar R, Schueller SM. Towards an artificially empathic conversational agent for mental health applications: system design and user perceptions. J Med Internet Res. (2018) 20:e10148. doi: 10.2196/ 10148
- Perez-Marin D, Pascual-Nieto I. Conversational agents and natural language interaction: techniques and effective practices: techniques and effective practices. *IGI Global*. (2011) 28. doi: 10.4018/978-1-60960-617-6
- Berscheid J, Roewer-Despres F. Beyond transparency. AI Matters. (2019) 5:13–22. doi: 10.1145/3340470.3340476
- 58. Voigt P, von dem Bussche A. The EU General Data Protection Regulation (GDPR): A Practical Guide. Cham: Springer (2017).
- Lemonne E. Ethics Guidelines for Trustworthy AI FUTURIUM European Commission. (2018). Available online at: https://ec.europa.eu/futurium/en/aialliance-consultation (accessed March 23, 2021).
- Rooksby J, Asadzadeh P, Morrison A, McCallum C, Gray C, Chalmers M. Implementing ethics for a mobile app deployment. In: *Proceedings of the* 28th Australian Conference on Computer-Human Interaction - OzCHI'16 Launceston, TAS. (2016).
- Robbins S. A Misdirected principle with a catch: explicability for AI. Minds Mach. (2019) 13:94. doi: 10.1007/s11023-019-09509-3
- Palanica A, Flaschner P, Thommandram A, Li M, Fossat Y. Physicians' perceptions of chatbots in health care: cross-sectional web-based survey. J Med Internet Res. (2019) 21:e12887. doi: 10.2196/12887
- Hsin H, Fromer M, Peterson B, Walter C, Fleck M, Campbell A, et al. Transforming psychiatry into data-driven medicine with digital measurement tools. *NPJ Digit Med.* (2018) 1:37. doi: 10.1038/s41746-018-0046-0
- Hutton L, Price BA, Kelly R, McCormick C, Bandara AK, Hatzakis T, et al. Assessing the privacy of mHealth apps for self-tracking: heuristic evaluation approach. *JMIR Mhealth Uhealth*. (2018) 6:e185. doi: 10.2196/mhealth. 9217
- Hu AW, Anderson KN, Lee RM. Let's talk about race and ethnicity: cultural socialization, parenting quality, and ethnic identity development. *Family Sci.* (2015) 6:87–93. doi: 10.1080/19424620.2015.108 1007
- Friedman B, Kahn PH, Borning A. Value sensitive design and information systems. In: Himma KE, Tavani HT, editors. *The Handbook of Information and Computer Ethics*. New Jersey, NJ: John Wiley & Sons, Inc., (2008). p. 69–101. doi: 10.1002/9780470281819.ch4

**Conflict of Interest:** The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

**Publisher's Note:** All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Copyright © 2021 Vilaza and McCashin. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms. 2.7 Teaching User Experience Design Ethics to Engineering Students: Lessons Learned

#### Authors:

Giovanna Nunes Vilaza and Per Bækgaard

#### Submitted to:

Frontiers in Computer Science, section Human-Media Interaction (under review)



# Teaching User Experience Design Ethics to Engineering Students: Lessons Learned

Giovanna Nunes Vilaza <sup>1,\*</sup> and Per Bækgaard <sup>2</sup>

<sup>1</sup>Department of Health Technology, Technical University of Denmark, Kongens Lyngby, Denmark <sup>2</sup> Department of Applied Mathematics and Computer Science, Technical University of Denmark, Kongens Lyngby, Denmark Correspondence\*: Giovanna Nunes Vilaza gnvi@dtu.dk

# ABSTRACT

Contemporary dilemmas about the role and impact of digital technologies in society have motivated the inclusion of topics of computing ethics in university programmes. Many past works have investigated how different pedagogical approaches and tools can support learning and teaching such a subject. This brief research report contributes to these efforts by describing a pilot study examining how engineering students learn from and apply ethical principles when making design decisions for an introductory User Experience (UX) design project. After a short lecture, students were asked to design and evaluate the ethical implications of digital health intervention prototypes. This approach was evaluated through the thematic analysis of semi-instructed interviews conducted with 12 students, focused on the benefits and limitations of teaching ethics to unaware and uninterested students, an observation that calls for a much stronger emphasis on moral philosophy education throughout engineering degrees. This paper finishes with a reflection on the hardships and possible ways forward for teaching and putting UX design ethics into practice. The lessons learned and described in this report aim to contribute to future pedagogical efforts to enable ethical thinking in computing education.

Keywords: Ethics, Digital Education, User Experience Design, Digital Health, Human-computer Interaction

# **1 INTRODUCTION**

Whilst computing systems have brought novel ways to work, communicate and play, the academic community is well aware of the emergent ethical concerns arising with the spread of digital innovations (Davis, 2020), especially in the context of digital health (Martinez-Martin and Kreitmair, 2018). The way

such systems can persuade users' actions can be insensitive to vulnerable groups' autonomy (Ayobi, 2020). Language choices, technology literacy requirements and usability flaws can hinder broader access, going against social fairness (Costanza-Chock, 2020). Limited data sharing options can fail to recognise individuals' preference for privacy (Hutton et al., 2018), while lack of transparency can hide away limitations of digital interventions (Vilaza and McCashin, 2021).

As a consequence of the broader recognition of ethical issues, ethics education is currently deemed essential to forming future generations of designers and engineers (Hughes et al., 2020, Skirpan et al., 2018). For instance, experiential learning has been used to facilitate empathy-building towards accessibility issues (El- Glaly et al., 2020). A structured framework has been proposed to help students to identify and articulate harmful limitations of machine learning projects (Saltz et al., 2019). Science fiction has been applied as a medium to entice moral imagination regarding the drawbacks of artificial intelligence (Burton et al., 2018).

Despite a wide variety of theoretical frameworks for ethical thinking, applying pre-defined ethical principles to design work is among the most often adopted approaches in the industry and academia. The development of 'ethics checklists' is an increasingly common practice among companies as means of attempting to alleviate the difficulty practitioners face when operationalising abstract principles (Madaio et al., 2020). Past research has also emphasised that lists of normative ethical principles are frequently applied in the critical evaluation of AI developments within health care (Morley et al., 2021). Intending to understand how students in a prototyping activity might apply this approach, we conducted a pilot study.

This paper advances this research record on ethics education by reporting on the results of the pilot study investigating how engineering students learn from and apply normative principles when making practical UX design choices for digital health prototypes. After a one-week project part of an introductory course on UX design at a technical university, 12 students were interviewed and inquired about their experiences. The following sections describe: the methods used in this qualitative study, the interview findings, and the discussion of results. The study contributes to understanding the benefits and limitations of using normative principles to teach UX design ethics to engineering students in a project-based learning setting.

# 2 METHODS

This pilot study consisted of semi-structured interviews with the aim of understanding the learning experience of engineering students after being exposed to materials and an assignment about UX ethics. The study sought to investigate how the educational approach has facilitated learning of ethics and which challenges were experienced by the students in the process. The report of the educational evaluation conducted in this pilot study intends to not only advance research on this topic but also inform future educational approaches in the department.

### 2.1 Participants

The participants were students at a technical university in Denmark, enrolled in a 13-week course on UX Design. In terms of course structure, every week, there were one-hour lectures followed by three hours of supervised group work in which students were given a design brief and asked to prepare a set of deliverables (business model canvas, user story maps, interactive prototypes, report on prototype

### Vilaza et al.

evaluation). Then, the students carried out an estimated amount of 4-5 hours of independent work in groups before the next class. The goal of these short weekly projects was to prompt the students to learn how to ideate and materialise design concepts along the lines of the pedagogical approach of 'project-based learning' (Kokotsaki et al., 2016).

ID	Group	Programme	Nationality	Gender
P1	1	Design and Innovation Engineering (MSc)	Danish	Female
P2	2	Human-Centered Artificial Intelligence (MSc)	Colombian	Female
P3	3	Industrial Engineering and Management (MSc)	Greek	Male
P4	4	Design and Innovation Engineering (MSc)	Spanish	Female
P5	4	Design and Innovation Engineering (MSc)	Spanish	Female
P6	5	Design and Innovation Engineering (MSc)	Danish	Male
P7	5	Design and Innovation Engineering (MSc)	Danish	Male
P8	5	Design and Innovation Engineering (MSc)	Danish	Male
P9	5	Exchange student	French	Male
P10	5	Exchange student	French	Male
P11	6	Human-Centered Artificial Intelligence (MSc)	Spanish	Female
P12	6	Software Engineering (BSc)	Danish	Female

Table 1. Study participants' characteristics

In the seventh week of the course, the weekly project proposed to the students consisted of designing a prototype for a smartwatch application that could collect, visualise and share heart rate data between patients and doctors. In addition, the assignment included a written report on ethical considerations of the design concept and the prototype. In order to prepare the students for this assignment, there was a lecture given by one of the teaching assistants in which the students were given an introduction to the potential negative impacts of user interface design choices on users' well-being, autonomy and diverse access. This approach was then evaluated through this reported study.

Recruitment for the study occurred only after students submitted the weekly project deliverables, as participation was voluntary and completely independent from the course assignment. This measure was necessary so that the students work during the assignment would not be influenced by the interview study. A verbal announcement and a message in the class online forum invited the students to be part of an interview about their experience working in the UX design ethics part of the project assignment. In total, 12 students expressed interest. Table 1 shows the participants' characteristics.

### 2.2 Materials

Before the study, all students of the class were exposed to learning materials about UX ethics. First, there was a lecture illustrating core ethical challenges. Then, the students were provided with two templates (Google forms online): a checklist for self-reflection or team discussions and a questionnaire to gather feedback from peers or potential users (see Supplementary Materials).

The lecture and the templates purposely emphasised a set of five normative principles: choice, transparency, inclusion, well-being and reciprocity (see Figure 1). This set of principles was inspired by the ethical framework of Nebeker et al. (2019) highlighting beneficence (providing end-users with direct health benefits), justice (enabling diverse and inclusive access) and respect for persons (not harming individual well-being, providing choices and being transparent) as essential ethical requirements for the digital health context. This framework facilitated the creation of learning materials that could concisely and soundly introduce the topic to the students.



Figure 1. Infographic illustrating the five normative principles emphasised to the students

# 2.3 Procedure

In order to build rapport and protect students from feeling that the participation in the study might compromise their grades, a teaching assistant conducted the interviews and confidentiality from the primary course instructor was guaranteed. As this pilot study was not planned nor conducted by the main course lecturer, which meant that some distance could be maintained, as the goal of the study responsible was learning from this experience and not judging teaching performance. Three interviews were conducted in person and four remotely (through a video call). Participants who belonged to the same working group in class were interviewed together. Interview questions were based on a semi-structured script.

Participants were explained that the goal of the interview is to improve how the activity is carried out in future classes and, for this reason, their honest feedback was very important. Participants were asked about their experiences, challenges and learnings, and were compensated with a voucher of 100 DKK. Complying with local jurisdictions, this study followed the Danish Code of Conduct for Research Integrity as it was considered exempt from formal ethical approval.

A Thematic Analysis was conducted by the first author, following the Braun and Clarke framework (Braun and Clarke, 2012); more specifically, it followed an inductive approach. The themes' descriptions and corresponding quotes were then used to report results as the narrative presented next.

**Frontiers in Computer Science** 

# 3 RESULTS

## 3.1 The approach served to raise awareness and interest

An important theme across the interviews was that ethics in design was perceived as a new topic not yet examined by many of the students until the course: "It is the first time I hear about ethics in design" (P2). Despite being a novelty, the educational materials were effective in bringing the topic to the students' attention: "The lecture you gave raised some awareness. Since that lecture, ethics has been part of our work in the group" (P1). Bringing this topic to class also changed some of the students' perspectives about technology design: "We thought about ethics, but maybe not in a good way. We did the opposite with the previous courses. We thought: how can we be as evil as possible with this? How can we gather as much data? How can we blackmail the user the most? Now we think the opposite" (P6). Despite being a novelty, most students were clearly interested in the topic: "I have not thought about it, but as soon as I read it, I was like, okay, this is important, it is something that I really want to address because ethics is something that I care about" (P11). Some also expressed an interest in learning more: "I would like to see more about it because I am interested as a person" (P3). Overall, these findings indicate that the educational approach was efficient in raising the students' awareness of the topic.

## 3.2 The principles helped to identify ethical concerns

The interviews show that students understood how their prototypes could be designed to consider ethical aspects. In particular, issues of transparency and choice were often mentioned by the students as elements they have re-considered: "We tried to add more things to the smartwatch regarding transparency and freedom of choice" (P4). One of the students also mentioned adding more privacy settings to the design proposal: "We were thinking that being able to know what you are showing and what you are not and having more settings, because, in the first app that we made, we did not have settings" (P11). Harm to well-being was another concern tackled often by the students in their design process: "The whole point is trying to make the users not feel bad if they have not done something they should have done. The notification could like tell them to go out for a walk without trying to make them feel bad and just try to stay positive" (P10). In addition, when asked how they approached the evaluation of their prototypes, students reported that they used the templates as a guidance: "Going through the checklists. It was quite informative, it made it quite clear the things we should be looking for" (P1) and "What we did was to use the templates, and that is how we learned how to do it. Without the templates we would not be able to know what to change" (P5). The use of normative principles, therefore, appear helpful in helping to identify specific ethical issues.

# 3.3 Ethical design was a new topic to many students

A lack of previous knowledge on ethics was raised by the students as a source of insecurity when making decisions: *"I feel hesitant, doubtful, concerned because I have never heard of the topic before. Of course, it is something important, but I never thought about it" (P2).* Missing specialised domain knowledge that could help to make ethical findings actionable was also an issue for some: *"We felt a lack of knowledge* 

because in this particular case, we need a doctor to say what is more important. Maybe it could be nice to have more health information because we know it is something we should take into consideration but as we do not know the potential damage" (P4). Similarly, one of the students felt unsure about how to attend to disabled users' needs: "How to include the handicap? I think it is important, but I have no idea how. You must be the blind person to understand the blind person" (P2). Another student did not know how blind users could use mobile apps: "One of the comments that we kept getting when we were reviewing each other's solutions was that blind people would not be able to use this but are blind users even able to use apps?" (P1). Such findings indicate that despite the ethics lecture, they may still need more info in the course to grasp the concepts.

### 3.4 Ethics was perceived as an antagonist for success for some students

In contrast with previous themes, a few students were not entirely convinced that ethics should be a priority to design: "I think it was a good add on to the course, but I do not consider it being a high priority" (P1). These students believed that ethical ideals could conflict with business growth: "I think it is rather unrealistic to incorporate ethics in such a corporate area. How would you ask big corporations or developing companies to be more aware of ethics if it is clear that their primary concern is money?" (P2). In particular, a student remarked how ethics could be a barrier to profit: "Data is money, and all I ever wanted is to make money. So we need all the data even if you do not want to share it, that was our app's logic: money" (P7). Aligned with this finding, a student stated that getting a high grade was, in fact, the primary motivation to engage with the subject: "In the end, we were caring about a good grade, so I am not going to lie this was the reason behind" (P3). Such negative views of ethics illustrate the resistance of a few students to consider the importance of the topic.

# 3.5 Group members had conflicting views at times

As the students were working in a group, social dynamics played a role in how discussions were held, with many students stating it was sometimes difficult to reach agreements: *"We have been able to agree on many things, but we are a group of people who do not know each other very well so we do not always turn out super compatible. It is hard to say: 'I think you should change all the work you just did' " (P12).* Some also reported that their group members did not consider the topic important, leading to a conflict of interest:

"We were more interested in it than the others. It is not that they were against it, they just did not care" (P12). As an attempt to handle disparate views, one of the students mentioned that when conflict arises regarding the ethical implications of a certain UX design choice, the group decides to ask for feedback from end-users or peers: "The moment one has a question and asks the group, but we cannot agree in a few minutes, we decide to validate the concept with others" (P5). This lack of alignment within teams is another challenge to teaching and learning ethics in design projects.

# 3.6 Time pressure was a source of frustration

The fast pace of the course and the requirement for weekly deliverables, where time on purpose becomes a scarce resource and thus forces the student to prioritise hard, were nevertheless barriers to deeper discussions: "At that point, we were too busy and concerned with the next hand-in. We were just going to leave it because we did not have much time. We were not making great philosophical discussions about everything, but if we had a longer time frame, we could do it" (P2). In order to be more efficient within the time frame, one of the students suggested that seeing more examples could help making faster decisions: "Maybe we could see some examples of how to implement it more quickly" (P3). Besides lack of time, some students wished they had started considering ethical aspects at the beginning of the project, not as an after-thought: "It felt stressful and frustrating because it was late in the process, and I feel like that is something that should have come earlier" (P11). Therefore, time management was a significant factor in deciding whether to engage in ethical reflections during the course.

# 4 **DISCUSSION**

The educational approach evaluated in this paper was effective in raising students' awareness, which is in itself a very favourable outcome for classroom activities about ethics (Skirpan et al., 2018; Saltz et al., 2019). Results show that the selected set of normative principles was a helpful structure for analysis, as the principles functioned as reference points guiding the students' creative process. A previous study evaluating a similar framework of ethical questions for machine learning also found that a list of ethical questions acted as a catalyst to students' debate (Saltz et al., 2019).

Findings also provide evidence that the students can make some ethical design decisions once instructed. Such reports of applying ethical thinking to the design of prototypes are not usual in the literature on ethics. An exception is perhaps a previous study that observed how students re-shaped their design concepts after experiencing the vulnerabilities of data collection and visualisation firsthand (Shapiro et al., 2020). As the goal of ethics education in computing is to provoke change in future technological developments, ethical insights should lead to observable outcomes in the design process (Barry et al., 2020; Bauer et al., 2017).

Despite such supportive indicators, the study makes evident that students faced challenges. Even though disparate views can support debate and reflexivity, students could not perceive the group conflicts in such a positive light. Previous works have discussed that methods for 'ethical mediation' are critical in decision-making so that arguments from conflicting views can be taken into account during team discussions (Gray et al., 2019), and our findings support the need to include that in the classroom. Past research has also highlighted that putting ethics in action is a demanding task, requiring an empathetic mindset, attentive to situational complexities (Frauenberger et al., 2017; Munteanu et al., 2015). Such a deep type of reflection can take time, and our observations indicate that it can be challenging to achieve more ethical design if time is too limited of a resource. Obstacles with group dynamics and time prioritisation should still be used as pedagogical tools to prepare students for situations that may appear in their workplace, but course structures should consider including more concrete examples and tools to help students navigate the constraints of a design process more productively.

Findings also confirm previously discussed shortcomings of pre-defined checklist items and the limitation of atomistic normative frameworks. As previous research with employees working with artificial intelligence has argued, co-designing checklist items as a team is a more effective approach than providing professionals with pre-made broad guidelines (Madaio et al., 2020). However, checklists and normative principles may as well do not function as tools enabling more comprehensive ethical thinking,

#### Vilaza et al.

rather becoming manual tasks to be completed without genuine reflection. For this reason, if an approach based on lists of ethical principles is chosen, it is important to consider how to complement the method with other design inquiry methods, such as active stakeholder involvement and speculative prototyping (Friedman and Hendry, 2019).

Furthermore, results indicate that a one-time lecture and a prototyping assignment may not be sufficient to fill existing knowledge gaps. In fact, it has been argued that ethics education would greatly benefit from acknowledging the need to expose students to a diverse range of disciplines, skills and methodologies related to the topic throughout their studies (Raji et al., 2021). Aligned with such perspective, previous studies have proposed empathy-building tools and role-playing as ways to increase sensitivity to issues that are beyond a designers' lived experience, thus adding to their capacity to relate to their users (Sas et al., 2020; Honary et al., 2018; Matthews et al., 2014). The importance of empathy development is particularly relevant in the case of students who believe that technology should be "as evil as possible" (P6), as they might not have realised that, in the future, they might be victims of malicious technologies they built by refusing to act in solidarity with their users in the present.

Moreover, the analytical stance deployed by the students in this particular study is not the only way to engage with ethics. Active involvement of different stakeholders through participatory and emancipatory research methods are other options that can be used in UX education. An example is a study reporting on how the collaboration with communities and non-profit organisations was very effective in teaching students how to propose caring design concepts, more attuned to users' needs (Sabie and Parikh, 2019). Still, even though consulting others may be a way for students to seek different perspectives, it can also become a shortcut for making decisions without genuine reflection, which should not be the goal.

Regardless of the educational approach chosen, findings suggest that some misconceptions need to be addressed first if students are expected to produce concrete ethical designs in class. It is not easy to know exactly why some students seem to care less about ethics than others. However, a previous study on ethics education has found that students usually do not see themselves as political agents responsible for ethical work (Petelka et al., 2022). Previous works have broughtto attention that engineering students may never come across topics of ethics during their education, which further complicates this problem (Saltz et al., 2019). The combination of standalone modules and the insertion of activities about the topic in multiple technical courses across secondary education programmes might prove to be the most effective approach in the long term, as advocated by previous research (Garrett et al., 2020). It is also fundamental to keep probing strategies for the challenging quest of turning indifferent students into caring ethical agents in their future careers.

#### 4.1 Limitations and Future Work

This pilot study has methodological limitations. The fact that only students who volunteered to participate were recruited means that findings may not reflect the perspective of the whole class (sample bias). In addition, the empirical data comes from the students' reports of their experiences in retrospect, which can result in recall bias. Another limitation is that students had to share their views to one of the

class tutors, which might have blocked the disclosure of opposing opinions despite our efforts to stay open to their feedback (acquiescence bias).

In order to complement and build upon the observations reported in this study, future work could consider the direct observation of students as they work on their projects and the discussion of the produced artefacts as additional empirical sources, as a way to evaluate the impact of the course based on the changes students bring into their processes. Further studies could also examine the preferences of students regarding different ethical frameworks, such as ethics of care. Finally, future research could consider more objective measures for the evaluation of pedagogical efforts, such as questionnaires and examinations.

# 5 CONCLUSION

This pilot study had the goal of learning from the experience of introducing students from a tehnical unviersity to the concept of ethical UX design. Results were very insightful as they showed in practice the limitations and benefits of our approach. With the lessons learned through this study, we contribute to future pedagogical efforts to teaching ethics for UX design as the explicit statements from the students are powerful indicators of the challenges of teaching HCI ethics.

In summary, even though the educational materials could effectively raise awareness and guide some ethical decisions in the project-based learning setting, challenges remained. Some students seemed sceptical about the applicability of ethics in technology, and lack of interest was a significant barrier to a genuine engagement. Gaps in engineering education also became apparent as students reported feeling insecure with their knowledge on the topic. Students also claimed time pressure and group dynamics as obstacles to more profound reflections that could lead to user interface designs that respect human autonomy, promote well-being and broader access to digital innovation.

Such findings emphasise the need to expose students more often to a more diverse range of teaching methodologies, design skills and ethical philosophies throughout their engineering education. With the broader recognition of complex moral dilemmas by the media and digital technology consumers, ethics education has become imperative for future professionals and it consists of one of the most critical design aspects of digital health interventions. Efforts to include topics on computing ethics in teaching materials should be encouraged, and the way these materials are delivered should be mindful of the challenges discussed in this paper.

# ACKNOWLEDGMENTS

We thank the students who agreed to be part of the research and contributed to this study.

# REFERENCES

- Ayobi, A. (2020). Self-Tracking by People Living with Multiple Sclerosis: Supporting Experiences of Agency in a Chronic Neurological Condition. Ph.D. thesis, UCL (University College London)
- Barry, M., Kerr, A., and Smith, O. (2020). Ethics on the ground: from principles to practice. In Proceedings of the 2020 Conference on Fairness, Accountability, and Transparency. 688–688
- Bauer, M., Glenn, T., Monteith, S., Bauer, R., Whybrow, P. C., and Geddes, J. (2017). Ethical perspectives on recommending digital technology for patients with mental illness. International journal of bipolar disorders 5, 1–14
- Braun, V. and Clarke, V. (2012). Thematic analysis. (American Psychological Association)
- Burton, E., Goldsmith, J., and Mattei, N. (2018). How to teach computer ethics through science fiction. Communications of the ACM 61, 54–64
- Costanza-Chock, S. (2020). Design justice: Community-led practices to build the worlds we need (The MIT Press)
- Davis, J. L. (2020). How artifacts afford: The power and politics of everyday things (MIT Press)
- El-Glaly, Y., Shi, W., Malachowsky, S., Yu, Q., and Krutz, D. E. (2020). Presenting and evaluating the impact of experiential learning in computing accessibility education. In 2020 IEEE/ACM 42nd International Conference on Software Engineering: Software Engineering Education and Training (ICSE-SEET) (IEEE), 49–60
- Frauenberger, C., Rauhala, M., and Fitzpatrick, G. (2017). In-action ethics. Interacting with Computers 29, 220–236
- Friedman, B. and Hendry, D. G. (2019). Value sensitive design: Shaping technology with moral imagination (Mit Press)
- Garrett, N., Beard, N., and Fiesler, C. (2020). More than" if time allows" the role of ethics in ai education. In Proceedings of the AAAI/ACM Conference on AI, Ethics, and Society. 272–278
- Gray, C. M. and Chivukula, S. S. (2019). Ethical mediation in ux practice. InProceedings of the 2019 CHI304Conference on Human Factors in Computing Systems. 1–11
- Honary, M., McNaney, R., and Lobban, F. (2018). Designing video stories around the lived experience of severe mental illness. In Proceedings of the 10th Nordic Conference on Human-Computer Interaction. 25–38
- Hughes, J., Plaut, E., Wang, F., von Briesen, E., Brown, C., Cross, G., et al. (2020). Global and local agendas of computing ethics education. In Proceedings of the 2020 ACM Conference on Innovation and Technology in Computer Science Education. 239–245
- Hutton, L., Price, B. A., Kelly, R., McCormick, C., Bandara, A. K., Hatzakis, T., et al. (2018). Assessing the privacy of mhealth apps for self-tracking: heuristic evaluation approach. JMIR mHealth and uHealth 6, e185
- Kokotsaki, D., Menzies, V., and Wiggins, A. (2016). Project-based learning: A review of the literature. Improving schools 19, 267–277

- Madaio, M. A., Stark, L., Wortman Vaughan, J., and Wallach, H. (2020). Co-designing checklists to understand organizational challenges and opportunities around fairness in ai. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–14
- Martinez-Martin, N. and Kreitmair, K. (2018). Ethical issues for direct-to-consumer digital psychotherapy apps: addressing accountability, data protection, and consent. JMIR mental health 5, e32
- Matthews, M., Gay, G., and Doherty, G. (2014). Taking part: role-play in the design of therapeutic systems. In Proceedings of the SIGCHI conference on human factors in computing systems. 643–652
- Morley, J., Floridi, L., Kinsey, L., and Elhalal, A. (2021). From what to how: an initial review of publicly available ai ethics tools, methods and research to translate principles into practices. In Ethics, Governance, and Policies in Artificial Intelligence (Springer). 153–183
- Munteanu, C., Molyneaux, H., Moncur, W., Romero, M., O'Donnell, S., and Vines, J. (2015). Situational ethics: Re-thinking approaches to formal ethics requirements for human-computer interaction. In Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems. 105–114
- Nebeker, C., Torous, J., and Ellis, R. J. B. (2019). Building the case for actionable ethics in digital health research supported by artificial intelligence. BMC medicine 17, 1–7
- Petelka, J., Finn, M., Roesner, F., and Shilton, K. (2022). Principles matter: Integrating an ethics intervention into a computer security course. Proceedings of the 53rd ACM Technical Symposium on Computer Science Education.
- Raji, I. D., Scheuerman, M. K., and Amironesei, R. (2021). You can't sit with us: Exclusionary pedagogy in ai ethics education. In Proceedings of the 2021 ACM Conference on Fairness, Accountability, and Transparency. 515–525
- Sabie, S. and Parikh, T. (2019). Cultivating care through ambiguity: Lessons from a service learning course. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems. 1–14
- Saltz, J., Skirpan, M., Fiesler, C., Gorelick, M., Yeh, T., Heckman, R., et al. (2019). Integrating ethics within machine learning courses. ACM Transactions on Computing Education (TOCE) 19, 1–26
- Sas, C., Hartley, K., and Umair, M. (2020). Manneqkit cards: A kinesthetic empathic design tool communicating depression experiences. In Proceedings of the 2020 ACM Designing Interactive Systems Conference. 1479–1493
- Shapiro, B. R., Meng, A., O'Donnell, C., Lou, C., Zhao, E., Dankwa, B., et al. (2020). Re-shape: A method to teach data ethics for data science education. In Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems. 1–13
- Skirpan, M., Beard, N., Bhaduri, S., Fiesler, C., and Yeh, T. (2018). Ethics education in context: A case study of novel ethics activities for the cs classroom. In Proceedings of the 49th ACM Technical Symposium on Computer Science Education. 940–945
- Vilaza, G. N. and McCashin, D. (2021). Is the automation of digital mental health ethical? applying an ethical framework to chatbots for cognitive behaviour therapy. Frontiers in Digital Health 3, 100. doi:10.3389/fdgth.2021.689736



# Supplementary Material

# 1 TEMPLATES

The following templates were provided to the students as examples of how to use the five normative principles explained in class. The first template is an expert checklist created to help the students verify if their design proposals attend to the principles and then reflect on how their prototypes could be improved. A second template is a form for collecting peer (or user) feedback, also contemplating the same five items as the expert checklist, but meant to consult others outside of the design team. The templates served as guidance to the students' assignment on evaluating the ethical dimension of their work.

# 1.1 Expert Checklist

- Choice:
- $\Box$  The system allows users to choose and select their preferences
- $\Box$  The system does not force single options to users (example: I accept)
- $\Box$  The system allows users to change their mind and select a different option later
- Transparency:
- $\Box$  The system provides enough information about how it works
- $\hfill\square$  The system is transparent about how data is used and shared
- $\Box$  The system does not hide important information on purpose
- Well-being:
- $\Box$  The system is designed to help users to improve their well-being
- $\Box$  The system does not make users' well-being worst (example: by making them anxious)
- $\Box$  The system provides direct support to users in case they suffer physical or emotional harm
- Inclusion:
- $\Box$  The system is directly targeted to attend the needs of at least one marginalised group
- □ The system does not exclusively target only over-represented groups
- $\Box$  The system does not exclude people of different races, disabilities, gender and cultures
- Reciprocity:
- $\Box$  The system is useful by offering direct and significant benefits for the users
- $\Box$  The system does not use manipulation techniques to get users to spend money
- $\Box$  The system is not built to take advantage of users for business gains

# 1.1.1 Peer Feedback

From not at all (0) to completely (7), how much do you think the system...

- 1. ... allows users to choose and select their preferences? Why?
- 2. ... is transparent and clear about how it works? Why?
- 3. ... promotes well-being and protects users from harm? Why?
- 4. ... can be used by diverse races, gender identities and physical mental abilities? Why?
- 5. ... provides mutual benefits for owners and users without manipulation? Why?

# 2.8 What is the Future of Data Sharing for Research?

#### Full Reference:

Giovanna Vilaza. 2020. What is the future of data sharing for research? *interactions* 27, *4 (July - August 2020), 54–56.* doi:https://doi.org/10.1145/3404211

#### **Copyrights:**

Permission to republish in a thesis conceived by ACM. Copyright held by the authors.

# What Is the Future of Data Sharing for Research?

Giovanna Vilaza, Technical University of Denmark

igital data collection for health research usually follows well-established methods. In many of the labs that work with mobile sensing, research subjects are provided with consent forms, task instructions, and sensor devices or apps. Once the research subjects agree to participate, the expectation is that they will comply with the procedures and allow their lives to be digitally tracked. After that, they are usually dismissed.

Even though participants are such a vital part of scientific discoveries, they are often considered objects to be observed—one more entry in a database. Such well-established ways of placing those being monitored into passive roles have gained even more prominence. With the surge of Covid-19, there has been a noticeable increase in initiatives for health surveillance. From contact tracing to apps that monitor daily symptoms, the pervasiveness of smartphones is being exploited to collect data from large segments of the population. It is a blossoming field for those who work in this area, as the urgency to

Aggregated data might mean better healthcare now and in the future, but it is also a tool for power and mass control. understand this illness is pushing mobile sensing in ways never before seen.

Given this sudden demand for broader behavioral monitoring, the debates over population-level surveillance have gone mainstream. In the particular case of contact tracing, academics are now discussing issues of individual privacy, the consequences of false positives (and negatives), and the actual efficacy of such an approach [1]. On the other hand, the media, governments, and tech companies are claiming that transmission speed may be reduced only if a significant part of the population is monitored continuously. Contact tracing has been enforced in countries in Asia and framed as a way to "help authorities identify virus hotspots and better target health efforts" [2].

By providing arguments that surveillance is the right path for recovery, governments and the media are forging a positive-only view of the subject. A consequence of the support for contact tracing and other symptom-tracking approaches could be a radical change in how people perceive privacy threats and accept being monitored for the "public good." It could be speculated that efforts from the public sector and big corporations to convey the benefits of surveillance could lead the masses to believe this one-sided version of the story, without weighing its risks. Decisions about disclosure are known to involve a trade-off. If the perception of social or individual benefits is stronger than the identified possible risks, people are willing to share sensitive information [3].

If a shift toward more public acceptance of health surveillance indeed prevails, national-level repositories of the mobile-sensing data could also become very attractive to governments and scientists [4]. Largescale platforms containing information such as clinical diagnoses, mobile-sensing data, and behavioral tracking data could allow incredible epidemiological discoveries. Before Covid-19, the landscape of such platforms was dominated by genetic bio-banks and clinicaltrials repositories. Mobilegenerated data was still a novelty. Nowadays, massive centralized data centers containing information about thousands (or millions) of individuals are growing around the world, such as All of Us (https://allofus. nih.gov/) in the U.S. and iCarbonX (https://www. icarbonx.com/en/) in China; they include digital sensors as a data source.

However, if a shift toward more acceptance does not prevail, large-scale surveillance will be at risk of low cohort diversity. First, to derive significant and fair conclusions from a dataset, a diverse range of people with different characteristics is required. Unless most of the population is tracked, the knowledge acquired will not be representative and may benefit only those who were available and agreed to be monitored. As Daniela and Nicole Rosner discuss in this issue, "prioritizing the most likely to be reachable tends to benefit well-educated white people who have already long benefited from the healthcare system" (page 76). What can HCI, UX, and technology design practitioners and academics offer to facilitate more inclusive recruitment for data platforms? What knowledge, tools, and evidence have we produced (or can we provide) that can be useful in this context?

Besides inclusive recruitment, the search for public acceptance should not overwrite the need to consider possible impacts on all segments of the population. Individuals from different backgrounds might have a different understanding of potential privacy risks, and people with stigmatized clinical diagnoses might suffer from the consequences of a data leak asymmetrically [5]. Broader acceptance should not result in less public diligence about privacy and how data can be abused. Such individual differences need to be taken into account because ill-intentioned initiatives may lure people with the promise of future advances in research, but come with a hidden agenda [6]. As Christopher Frauenberger writes in this issue, "We

might see the coronavirus serving as the scapegoat to implement modes of mass behavior manipulation by private companies" (page 74). How can HCI knowledge and approaches be used to support and protect citizens from these scenarios? Can HCI help overcome the uneven understanding of risks and help tackle vulnerabilities in case of privacy breaches? are given to those who store the data, not to those who provide it [4]. This serves to strengthen the already existing inequalities between contributors and receivers.

The most significant change that the pandemic should bring is not that surveillance becomes more broadly accepted. A real change would be to see those proposing surveillance platforms finally placing



Singapore, May 2020: Man at Rower's Bay Park scanning a QR code via SafeEntry, a digital contact tracing tool.

As mentioned earlier, digital data collection for health research often follows well-established approaches. The pandemic has brought more attention to the subject of population surveillance, as seen in the reflections from Rosner and Frauenberger. However, the Covid-19 emergency has not changed the passive role attributed to those having their symptoms and contacts monitored. Most of the decisions about what data will be tracked, how it will be used, and who will have access to it are made from the top: by governments, health authorities, research institutions, and big corporations. When data repositories are built this way, power and knowledge

citizens at the core of their decisions, by listening to their concerns and providing them with direct protection and benefits. If people are to be asked to open up their lives for health surveillance or research, they should be respected, and their preferences prioritized. It is about time we put more efforts into understanding the needs from the different segments of the population and design for more inclusive participation and agency in research. The wellestablished approaches for data collection do not suffice anymore, as behavioral monitoring is being considered at a national level. Aggregated data might mean better healthcare now and in the future, but it is also

# A RESPONSIVE KIND OF DESIGN



Aarogya Setu, an app developed by the government of India to track Covid-19.

a tool for power and mass control [6]. The path to reach acceptance should involve respect, transparency, and an ethic of involvement by communities from all backgrounds [7].

More than ever, those who are in public, academic, and industry positions hold the responsibility of taking into account any potential for harm that novel ideas can bring to each individual. This pandemic, or any other alarming situation in the future, should not mean that moral principles and personal autonomy are put aside. Large-scale digital surveillance for public health may now gain momentum with contact tracing. Still, we need to keep reflecting, discussing, and pushing for an ethical development in the field, through the papers we write, the products we build, and the ideas we share with others.

The pandemic has been a challenging time in many aspects, but it can also mark a moment when meaningful changes began. It forced many to stop, and some to reconsider how things have been done until now—and how different they could be. From this process, hopefully, a brighter future can emerge for data sharing, health surveillance, and research platforms alike—a future in which acceptance does not mean renouncement of rights and values, but rather a conscious choice based on terms and conditions that are negotiated and never imposed. This should become the new normal. The next advances in data-collection practices depend on us, researchers and designers in the HCI and health tech field, as we choose how we conduct our own projects and support those of our community.

#### ENDNOTES

- Gillmor, D.K. Principles for technology-assisted contacttracing. ACLU white paper. Apr. 16, 2020; https://www. aclu.org/sites/default/files/ field\_document/aclu\_white\_ paper\_-\_contact\_tracing\_ principles.pdf
- 2. Phartiyal. S. India orders coronavirus tracing app for all workers. Reuters Technology News. May 2, 2020; https:// www.reuters.com/article/ushealth-coronavirus-india-app/ india-makes-governmenttracing-app-mandatory-forall-workers-idUSKBN22E07K
- King, J. 'Becoming part of something bigger' Direct to consumer genetic testing, privacy, and personal disclosure. *Proc. of the ACM on Human-Computer Interaction 3*, CSCW (2019), 1–33.
- Kostkova, P. et al. Who owns the data? Open data for healthcare. *Frontiers in Public Health* 4 (2016), 7.
- Petelka, R., Van Kleunen, L., Albright, L., Murnane, E., Voida, S., and Snyder, J.

Being (in) visible: Privacy, rransparency, and disclosure in the self-management of bipolar disorder. *Proc. of the 2020 CHI Conference on Human Factors in Computing Systems*. ACM, New York, 2020, 1–14.

- Zuboff, S. Big other: Surveillance capitalism and the prospects of an information civilization. *Journal of Information Technology 30* (2015), 1.
- 7. Costanza-Chock, S. Design Justice: Community-led Practices to Build the Worlds We Need. MIT Press, 2020.

• Giovanna Vilaza is a TEAM early-stage researcher, halfway to her Ph.D. in the Department of Health Tech, Technical University of Denmark. Her current project is about a participant-centered future for behavioral monitoring in open-access data platforms. She is a University College London and KTH Royal Institute of Technology alumna.

→ gnvi@dtu.dk

DOI: 10.1145/3404211 COPYRIGHT HELD BY AUTHOR. PUBLICATION RIGHTS LICENSED TO ACM. \$15.00

# Bibliography

- Abbott, Jacob et al. (2019). "Local standards for anonymization practices in health, wellness, accessibility, and aging research at CHI". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–14.
- Abdelhamid, Mohamed (2018). "Greater patient health information control to improve the sustainability of health information exchanges". In: *Journal of biomedical informatics* 83, pp. 150–158.
- Adjekum, Afua, Alessandro Blasimme, and Effy Vayena (2018). "Elements of Trust in Digital Health Systems: Scoping Review". In: *Journal of medical Internet research* 20.12, e11254.
- Ahearn, Laura M (2001). "Language and agency". In: *Annual review of anthropology* 30.1, pp. 109–137.
- Ahram, Mamoun et al. (2014). "Factors influencing public participation in biobanking". In: *European Journal of Human Genetics* 22.4, pp. 445–451.
- Alamo, Teodoro et al. (2020). "Covid-19: Open-data resources for monitoring, modeling, and forecasting the epidemic". In: *Electronics* 9.5, p. 827.
- Andalibi, Nazanin and Justin Buss (2020). "The Human in Emotion Recognition on Social Media: Attitudes, Outcomes, Risks". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–16.
- Anderson, Nicholas et al. (2012). "Participant-centric initiatives: tools to facilitate engagement in research". In: *Applied & translational genomics* 1, pp. 25–29.
- Anhalt-Depies, Christine et al. (2019). "Tradeoffs and tools for data quality, privacy, transparency, and trust in citizen science". In: *Biological Conservation* 238, p. 108195.
- Arksey, Hilary and Lisa O'Malley (2005). "Scoping studies: towards a methodological framework". In: *International journal of social research methodology* 8.1, pp. 19–32.
- Baggini, Julian (2018). How the world thinks: a global history of philosophy. Granta Books. Bagnasco, Annamaria et al. (2020). "Dignity, privacy, respect and choice—A scoping review of measurement of these concepts within acute healthcare practice". In: Journal of Clinical Nursing 29.11-12, pp. 1832–1857.
- Baig, Khadija et al. (2020). "" I'm hoping they're an ethical company that won't do anything that I'll regret" Users Perceptions of At-home DNA Testing Companies". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.
- Ballard, Stephanie, Karen M Chappell, and Kristen Kennedy (2019). "Judgment Call the Game: Using Value Sensitive Design and Design Fiction to Surface Ethical Concerns Related to Technology". In: *Proceedings of the 2019 on Designing Interactive Systems Conference*, pp. 421–433.
- Barbosa, Natã M and Monchu Chen (2019). "Rehumanized crowdsourcing: a labeling framework addressing bias and ethics in machine learning". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–12.
- Bardram, Jakob E (2020). "The CARP Mobile Sensing Framework–A Cross-platform, Reactive, Programming Framework and Runtime Environment for Digital Phenotyping". In: *arXiv preprint arXiv:2006.11904*.
- Bardzell, Jeffrey and Shaowen Bardzell (2013). "What is" critical" about critical design?" In: *Proceedings of the SIGCHI conference on human factors in computing systems*, pp. 3297–3306.

Barkhuus, Louise (2012). "The mismeasurement of privacy: using contextual integrity to reconsider privacy in HCI". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, pp. 367–376.

Barkhuus, Louise and Anind K Dey (2003). "Location-Based Services for Mobile Telephony: a Study of Users' Privacy Concerns." In: *Interact*. Vol. 3. Citeseer, pp. 702–712.

- Barry, Marguerite, Aphra Kerr, and Oliver Smith (2020). "Ethics on the ground: from principles to practice". In: *Proceedings of the 2020 Conference on Fairness, Accountability, and Transparency*, pp. 688–688.
- Barry, Marguerite et al. (2017). "mHealth for maternal mental health: everyday wisdom in ethical design". In: *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems*, pp. 2708–2756.
- Bauer, Michael et al. (2017). "Ethical perspectives on recommending digital technology for patients with mental illness". In: *International journal of bipolar disorders* 5.1, pp. 1–14.
- Bell, Jeanette and Tuck Wah Leong (2019). "Collaborative futures: Co-designing research methods for younger people living with dementia". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.
- Bendig, Eileen et al. (2019). "The next generation: chatbots in clinical psychology and psychotherapy to foster mental health–a scoping review". In: *Verhaltenstherapie*, pp. 1–13.
- Benford, Steve et al. (2012). "Uncomfortable interactions". In: *Proceedings of the sigchi conference on human factors in computing systems*, pp. 2005–2014.
- Benford, Steve et al. (2015). "The ethical implications of HCI's turn to the cultural". In: *ACM Transactions on Computer-Human Interaction (TOCHI)* 22.5, pp. 1–37.
- Benndorf, Volker and Hans-Theo Normann (2018). "The willingness to sell personal data". In: *The Scandinavian Journal of Economics* 120.4, pp. 1260–1278.
- Berscheid, Janelle and Francois Roewer-Despres (2019). "Beyond transparency: a proposed framework for accountability in decision-making ai systems". In: *AI Matters* 5.2, pp. 13–22.
- Bhat, Karthik S and Neha Kumar (2020). "Sociocultural Dimensions of Tracking Health and Taking Care". In: *Proceedings of the ACM on Human-Computer Interaction* 4.CSCW2, pp. 1–24.
- Bhatia, Jaspreet and Travis D Breaux (2018). "Empirical measurement of perceived privacy risk". In: ACM Transactions on Computer-Human Interaction (TOCHI) 25.6, p. 34.
- Bierer, Barbara E et al. (2016). "A global, neutral platform for sharing trial data". In: *New England Journal of Medicine* 374.25, pp. 2411–2413.
- Bilstrup, Karl-Emil Kjær, Magnus H Kaspersen, and Marianne Graves Petersen (2020). "Staging Reflections on Ethical Dilemmas in Machine Learning: A Card-Based Design Workshop for High School Students". In: *Proceedings of the 2020 ACM Designing Interactive Systems Conference*, pp. 1211–1222.
- Blandford, AE (2013). "Semi-structured qualitative studies". In: Interaction Design Foundation.
- Bourla, Alexis et al. (2018). "Psychiatrists' Attitudes Toward Disruptive New Technologies: Mixed-Methods Study". In: *JMIR mental health* 5.4.
- Boutilier, Craig et al. (2015). "Optimal social choice functions: A utilitarian view". In: *Artificial Intelligence* 227, pp. 190–213.
- Bowling, Ann and Shah Ebrahim (2005). *Handbook of health research methods: investigation, measurement and analysis.* McGraw-Hill Education (UK).
- Bowser, Anne et al. (2017). "Accounting for privacy in citizen science: Ethical research in a context of openness". In: *Proceedings of the 2017 ACM Conference on Computer Supported Cooperative Work and Social Computing*, pp. 2124–2136.

- Boyd, Karen L. and Katie Shilton (July 2021). "Adapting Ethical Sensitivity as a Construct to Study Technology Design Teams". In: 5.GROUP.
- Braun, Virginia and Victoria Clarke (2012). *Thematic analysis*. American Psychological Association.
- Brehaut, Jamie C et al. (2012). "Informed consent documents do not encourage goodquality decision making". In: *Journal of clinical epidemiology* 65.7, pp. 708–724.
- Broes, Stefanie et al. (2017). "Sharing human samples and patient data: opening Pandora's box". In: *Journal of Cancer Policy* 13, pp. 65–69.
- Broes, Stefanie et al. (2018). "Toward a tiered model to share clinical trial data and samples in precision oncology". In: *Frontiers in medicine* 5, p. 6.
- Bromley, Elizabeth et al. (2015). "From subject to participant: Ethics and the evolving role of community in health research". In: *American Journal of Public Health* 105.5, pp. 900–908.
- Brown, Barry et al. (2016). "Five provocations for ethical HCI research". In: *Proceedings* of the 2016 CHI Conference on Human Factors in Computing Systems, pp. 852–863.
- Burton, Emanuelle, Judy Goldsmith, and Nicholas Mattei (2018). "How to teach computer ethics through science fiction". In: *Communications of the ACM* 61.8, pp. 54–64.
- Burwell, Sasha, Matthew Sample, and Eric Racine (2017). "Ethical aspects of brain computer interfaces: a scoping review". In: *BMC medical ethics* 18.1, pp. 1–11.
- Caine, Kelly, Selma Šabanovic, and Mary Carter (2012). "The effect of monitoring by cameras and robots on the privacy enhancing behaviors of older adults". In: *Proceedings of the seventh annual ACM/IEEE international conference on Human-Robot Interaction*. ACM, pp. 343–350.
- Chandwani, Rajesh and Vaibhavi Kulkarni (2016). "Who's the Doctor? Physicians' Perception of Internet Informed Patients in India". In: *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*, pp. 3091–3102.
- Chen, Henian, Patricia Cohen, and Sophie Chen (2010). "How big is a big odds ratio? Interpreting the magnitudes of odds ratios in epidemiological studies". In: *Communications in Statistics—simulation and Computation* 39.4, pp. 860–864.
- Chivukula, Shruthi Sai, Colin M Gray, and Jason A Brier (2019). "Analyzing value discovery in design decisions through ethicography". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–12.
- Chivukula, Shruthi Sai et al. (2020). "Dimensions of UX Practice that Shape Ethical Awareness". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.
- Choe, Eun Kyoung et al. (2014). "Understanding quantified-selfers' practices in collecting and exploring personal data". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*. ACM, pp. 1143–1152.
- Christin, Delphine et al. (2011). "A survey on privacy in mobile participatory sensing applications". In: *Journal of systems and software* 84.11, pp. 1928–1946.
- Chua, Trudy Hui Hui and Leanne Chang (2016). "Follow me and like my beautiful selfies: Singapore teenage girls' engagement in self-presentation and peer comparison on social media". In: *Computers in Human Behavior* 55, pp. 190–197.
- Chung, Chia-Fang et al. (2015). "More than telemonitoring: health provider use and nonuse of life-log data in irritable bowel syndrome and weight management". In: *Journal of medical Internet research* 17.8, e203.
- Chung, Chia-Fang et al. (2017). "Finding the right fit: understanding health tracking in workplace wellness programs". In: *Proceedings of the 2017 CHI conference on human factors in computing systems*. ACM, pp. 4875–4886.

Clark, Karin et al. (2019). "Advancing the ethical use of digital data in human research: challenges and strategies to promote ethical practice". In: *Ethics and Information Technology* 21.1, pp. 59–73.

Clay, Ieuan (2020). "The Future of Digital Health". In: Digital Biomarkers 4.1, pp. 1–2.

- Cohen, Deborah J et al. (2015). "Developing a model for understanding patient collection of observations of daily living: a qualitative meta-synthesis of the Project HealthDesign program". In: *Personal and ubiquitous computing* 19.1, pp. 91–102.
- Collins, Francis S and Harold Varmus (2015). "A new initiative on precision medicine". In: *New England journal of medicine* 372.9, pp. 793–795.
- Corbett, Eric and Christopher A Le Dantec (2018). "Exploring trust in digital civics". In: *Proceedings of the 2018 Designing Interactive Systems Conference*, pp. 9–20.
- Correll, Michael (2019). "Ethical dimensions of visualization research". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.
- Corrigan, Patrick W (2000). "Mental health stigma as social attribution: Implications for research methods and attitude change". In: *Clinical psychology: science and practice* 7.1, pp. 48–67.
- Costanza-Chock, Sasha (2020). *Design justice: Community-led practices to build the worlds we need*. MIT Press.
- Curmi, Franco, Maria Angela Ferrario, and Jon Whittle (2014). "Sharing real-time biometric data across social networks: Requirements for research experiments". In: *Proceedings of the 2014 conference on Designing interactive systems*, pp. 657–666.
- Dahl, Yngve and Kristine Holbø (2012). "Value biases of sensor-based assistive technology: case study of a GPS tracking system used in dementia care". In: *Proceedings of the designing interactive systems conference*. ACM, pp. 572–581.
- Dash, Sabyasachi et al. (2019). "Big data in healthcare: management, analysis and future prospects". In: *Journal of Big Data* 6.1, pp. 1–25.
- Davies, Sarah R et al. (2019). "Science stories as culture: experience, identity, narrative and emotion in public communication of science". In: *Journal of Science Communication* 18.5.
- Davis, Jenny L (2020). *How Artifacts Afford: The Power and Politics of Everyday Things*. MIT Press.
- De Sutter, Evelien et al. (2021). "Clinical research in neonates: redesigning the informed consent process in the digital era". In: *Frontiers in pediatrics*, p. 879.
- Dey, Pranammya et al. (2017). "Data sharing and cardiology: platforms and possibilities". In: *Journal of the American College of Cardiology* 70.24, pp. 3018–3025.
- Di Geronimo, Linda et al. (2020). "UI dark patterns and where to find them: a study on mobile applications and user perception". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–14.
- Di Matteo, Daniel et al. (2018). "Patient willingness to consent to mobile phone data collection for mental health apps: structured questionnaire". In: *JMIR mental health* 5.3, e56.
- Dinh-Le, Catherine et al. (2019). "Wearable health technology and electronic health record integration: scoping review and future directions". In: *JMIR mHealth and uHealth* 7.9, e12861.
- DiPaola, Daniella, Blakeley H Payne, and Cynthia Breazeal (2020). "Decoding design agendas: an ethical design activity for middle school students". In: *Proceedings of the Interaction Design and Children Conference*, pp. 1–10.
- Doherty, Kevin, Andreas Balaskas, and Gavin Doherty (2020). "The design of ecological momentary assessment technologies". In: *Interacting with Computers* 32.1, pp. 257–278.

- Domaradzki, Jan and Jakub Pawlikowski (2019). "Public Attitudes toward Biobanking of Human Biological Material for Research Purposes: A Literature Review". In: *International journal of environmental research and public health* 16.12, p. 2209.
- Dombrowski, Lynn, Ellie Harmon, and Sarah Fox (2016). "Social justice-oriented interaction design: Outlining key design strategies and commitments". In: *Proceedings of the* 2016 ACM Conference on Designing Interactive Systems, pp. 656–671.
- Doraiswamy, P Murali, Charlotte Blease, and Kaylee Bodner (2020). "Artificial intelligence and the future of psychiatry: Insights from a global physician survey". In: *Artificial Intelligence in Medicine* 102, p. 101753.
- Dow, Andy, Rob Comber, and John Vines (2019). "Communities to the left of me, bureaucrats to the right... here I am, stuck in the middle". In: *interactions* 26.5, pp. 26–33.
- Duckert, Melanie and Louise Barkhuus (2022). "Protecting Personal Health Data through Privacy Awareness: A study of perceived data privacy among people with chronic or long-term illness". In: *Proceedings of the ACM on Human-Computer Interaction* 6.GROUP, pp. 1–22.
- Dunne, Anthony and Fiona Raby (2013). *Speculative everything: design, fiction, and so-cial dreaming*. MIT press.
- Durrant, Abigail and David Kirk (2018). "On Ethical Responsiveness: Being Answerable to Others as an HCI Researcher". In: *Interacting with Computers* 30.2, pp. 99–115.
- Ekstrand, Michael D et al. (2018). "Exploring author gender in book rating and recommendation". In: *Proceedings of the 12th ACM conference on recommender systems*, pp. 242–250.
- Encinas, Enrique et al. (2018). "Making Problems in Design Research: The Case of Teen Shoplifters on Tumblr". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*, pp. 1–12.
- Epstein, Daniel A et al. (2015). "A lived informatics model of personal informatics". In: *Proceedings of the 2015 ACM International Joint Conference on Pervasive and Ubiquitous Computing*, pp. 731–742.
- Epstein, Daniel A et al. (2016). "Beyond abandonment to next steps: understanding and designing for life after personal informatics tool use". In: *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*. ACM, pp. 1109–1113.
- Eriksson, Sara et al. (2020). "Ethics in Movement: Shaping and Being Shaped in Human-Drone Interaction". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–14.
- Estrada-Galiñanes, Vero and Katarzyna Wac (2020). "Collecting, exploring and sharing personal data: Why, how and where". In: *Data Science* 3.2, pp. 79–106.
- Eyre, Harris A et al. (2020). "Responsible innovation in technology for mental health care". In: *The Lancet Psychiatry* 7.9, pp. 728–730.
- Fiesler, Casey and Blake Hallinan (2018). ""We Are the Product" Public Reactions to Online Data Sharing and Privacy Controversies in the Media". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.
- Figueiredo, Flavio et al. (2020). "Understanding Targeted Video-Ads in Children's Content". In: *Proceedings of the 31st ACM Conference on Hypertext and Social Media*, pp. 151–160.
- Fitzpatrick, Kathleen Kara, Alison Darcy, and Molly Vierhile (2017). "Delivering cognitive behavior therapy to young adults with symptoms of depression and anxiety using a fully automated conversational agent (Woebot): a randomized controlled trial". In: *JMIR mental health* 4.2, e7785.
- Fleck, Rowanne and Geraldine Fitzpatrick (2010). "Reflecting on reflection: framing a design landscape". In: *Proceedings of the 22nd Conference of the Computer-Human In-*
teraction Special Interest Group of Australia on Computer-Human Interaction. ACM, pp. 216–223.

- Fleurence, Rachael et al. (2013). "How the Patient-Centered Outcomes Research Institute is engaging patients and others in shaping its research agenda". In: *Health Affairs* 32.2, pp. 393–400.
- Floridi, Luciano and Andrew Strait (2020). "Ethical foresight analysis: What it is and why it is needed?" In: *Minds and Machines* 30.1, pp. 77–97.
- Foley, Sarah, Nadia Pantidi, and John McCarthy (2020). "Student Engagement in Sensitive Design Contexts: A Case Study in Dementia Care". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.
- Foraker, Randi E et al. (2021). "Transmission dynamics: data sharing in the COVID-19 era". In: *Learning Health Systems* 5.1, e10235.
- Fox, Sarah et al. (2019). "Vivewell: Speculating near-future menstrual tracking through current data practices". In: *Proceedings of the 2019 on Designing Interactive Systems Conference*, pp. 541–552.
- Frauenberger, Christopher, Marjo Rauhala, and Geraldine Fitzpatrick (2017). "In-action ethics". In: *Interacting with Computers* 29.2, pp. 220–236.
- Friedman, Batya and David G Hendry (2019). *Value sensitive design: Shaping technology with moral imagination*. Mit Press.
- Friedman, Batya, Peter H Kahn, and Alan Borning (2008). "Value sensitive design and information systems". In: *The handbook of information and computer ethics*, pp. 69–101.
- Fritsch, Ester, Irina Shklovski, and Rachel Douglas-Jones (2018). "Calling for a revolution: An analysis of IoT manifestos". In: *Proceedings of the 2018 CHI conference on human factors in computing systems*, pp. 1–13.
- Frost, Jeana and Michael Massagli (2008). "Social uses of personal health information within PatientsLikeMe, an online patient community: what can happen when patients have access to one another's data". In: *Journal of medical Internet research* 10.3, e15.
- Garcia-Ceja, Enrique et al. (2018). "Mental health monitoring with multimodal sensing and machine learning: A survey". In: *Pervasive and Mobile Computing*.
- Garg, Suneela, Nidhi Bhatnagar, and Navya Gangadharan (2020). "A Case for Participatory Disease Surveillance of the COVID-19 Pandemic in India". In: *JMIR Public Health and Surveillance* 6.2, e18795.
- Garrett, Natalie, Nathan Beard, and Casey Fiesler (2020). "More Than" If Time Allows" The Role of Ethics in Al Education". In: *Proceedings of the AAAI/ACM Conference on AI, Ethics, and Society*, pp. 272–278.
- Gatny, Heather H and William G Axinn (2012). "Willingness to participate in research during pregnancy: race, experience, and motivation". In: *Field methods* 24.2, pp. 135–154.
- Geneviève, Lester Darryl et al. (2019). "Participatory disease surveillance systems: ethical framework". In: *Journal of medical Internet research* 21.5, e12273.
- Gentsch, Peter (2019). "Conversational AI: how (chat) bots will reshape the digital experience". In: *AI in marketing, sales and service*. Springer, pp. 81–125.
- El-Glaly, Yasmine et al. (2020). "Presenting and evaluating the impact of experiential learning in computing accessibility education". In: 2020 IEEE/ACM 42nd International Conference on Software Engineering: Software Engineering Education and Training (ICSE-SEET). IEEE, pp. 49–60.
- Glerup, Cecilie, Sarah R Davies, and Maja Horst (2017). "Nothing really responsible goes on here': scientists' experience and practice of responsibility". In: *Journal of Responsible Innovation* 4.3, pp. 319–336.

Goddard, Katrina AB et al. (2009). "Biobank recruitment: motivations for nonparticipation". In: *Biopreservation and biobanking* 7.2, pp. 119–121.

Gotterbarn, DW et al. (2018). "ACM code of ethics and professional conduct". In:

Gray, Colin M and Shruthi Sai Chivukula (2019). "Ethical mediation in UX practice". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–11.

Gray, Colin M, Shruthi Sai Chivukula, and Ahreum Lee (2020). "What Kind of Work Do" Asshole Designers" Create? Describing Properties of Ethical Concern on Reddit". In: *Proceedings of the 2020 ACM Designing Interactive Systems Conference*, pp. 61–73.

- Gray, Colin M et al. (2018). "The dark (patterns) side of UX design". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*, pp. 1–14.
- Gray, Colin M et al. (2021). "Dark patterns and the legal requirements of consent banners: an interaction criticism perspective". In: *Proceedings of the 2021 CHI Conference on Human Factors in Computing Systems*, pp. 1–18.
- Greshake Tzovaras, Bastian et al. (2019). "Open Humans: A platform for participantcentered research and personal data exploration". In: *GigaScience* 8.6, giz076.
- Grimpe, Barbara, Mark Hartswood, and Marina Jirotka (2014). "Towards a closer dialogue between policy and practice: responsible design in HCI". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, pp. 2965–2974.
- Gulati, Siddharth, Sonia Sousa, and David Lamas (2019). "Design, development and evaluation of a human-computer trust scale". In: *Behaviour & Information Technology* 38.10, pp. 1004–1015.
- Hagendorff, Thilo (2020). "The ethics of AI ethics: An evaluation of guidelines". In: *Minds* and *Machines* 30.1, pp. 99–120.
- Hamer, Mark et al. (2020). "Association between TV viewing and heart disease mortality: observational study using negative control outcome". In: *J Epidemiol Community Health*.
- Harari, Gabriella M et al. (2016). "Using smartphones to collect behavioral data in psychological science: Opportunities, practical considerations, and challenges". In: *Perspectives on Psychological Science* 11.6, pp. 838–854.

Hardy, Jean et al. (2018). "User acceptance of location-tracking technologies in health research: implications for study design and data quality". In: *Journal of biomedical informatics* 79, pp. 7–19.

- Harrell, D Fox and Jichen Zhu (2009). "Agency Play: Dimensions of Agency for Interactive Narrative Design." In: AAAI spring symposium: Intelligent narrative technologies II, pp. 44–52.
- Harrison, Peter W et al. (2021). "The COVID-19 Data Portal: accelerating SARS-CoV-2 and COVID-19 research through rapid open access data sharing". In: *Nucleic acids research*.
- Harvey, John, David Golightly, and Andrew Smith (2014). "HCI as a means to prosociality in the economy". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, pp. 2955–2964.
- Hayes, Steven C, Kirk D Strosahl, and Kelly G Wilson (2009). *Acceptance and commitment therapy*. American Psychological Association Washington, DC.
- Hiolle, Antoine et al. (2012). "Eliciting caregiving behavior in dyadic human-robot attachmentlike interactions". In: ACM Transactions on Interactive Intelligent Systems (TiiS) 2.1, pp. 1–24.
- Hirsch, Tad (2020). "Practicing Without a License: Design Research as Psychotherapy". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–11.

- Hodge, James et al. (2020). "Relational, Flexible, Everyday: Learning from Ethics in Dementia Research". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–16.
- Horst, Maja (2007). "Public expectations of gene therapy: Scientific futures and their performative effects on scientific citizenship". In: *Science, Technology, & Human Values* 32.2, pp. 150–171.
- Hossain, Mohammad Alamgir, Yogesh K Dwivedi, and Nripendra P Rana (2016). "Stateof-the-art in open data research: Insights from existing literature and a research agenda". In: *Journal of organizational computing and electronic commerce* 26.1-2, pp. 14–40.
- Howard, Dorothy and Lilly Irani (2019). "Ways of Knowing When Research Subjects Care". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–16.
- Howell, Noura et al. (2018). "Tensions of data-driven reflection: A case study of real-time emotional biosensing". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.
- Hripcsak, George et al. (2015). "Observational Health Data Sciences and Informatics (OHDSI): opportunities for observational researchers". In: *Studies in health technology and informatics* 216, p. 574.
- Hutton, Luke et al. (2018). "Assessing the privacy of mhealth apps for self-tracking: heuristic evaluation approach". In: *JMIR mHealth and uHealth* 6.10, e185.
- lakovleva, Tatiana, Elin M Oftedal, and John Bessant (2019). "Responsible innovation in digital health". In: *Responsible Innovation in Digital Health*. Edward Elgar Publishing.
- Ienca, Marcello and Effy Vayena (2020). "On the responsible use of digital data to tackle the COVID-19 pandemic". In: *Nature Medicine*, pp. 1–2.
- Ienca, Marcello et al. (2018). "Ethical design of intelligent assistive technologies for dementia: a descriptive review". In: *Science and engineering ethics* 24.4, pp. 1035–1055.
- Inal, Yavuz et al. (2020). "Perspectives and practices of digital accessibility: A survey of user experience professionals in nordic countries". In: *Proceedings of the 11th Nordic Conference on Human-Computer Interaction: Shaping Experiences, Shaping Society*, pp. 1–11.
- Inkster, Becky, Shubhankar Sarda, and Vinod Subramanian (2018). "An empathy-driven, conversational artificial intelligence agent (Wysa) for digital mental well-being: real-world data evaluation mixed-methods study". In: *JMIR mHealth and uHealth* 6.11, e12106.
- Irani, Lilly C and M Six Silberman (2013). "Turkopticon: Interrupting worker invisibility in amazon mechanical turk". In: *Proceedings of the SIGCHI conference on human factors in computing systems*, pp. 611–620.
- (2016). "Stories We Tell About Labor: Turkopticon and the Trouble with" Design"". In: Proceedings of the 2016 CHI conference on human factors in computing systems, pp. 4573–4586.
- Joinson, Adam N et al. (2010). "Privacy, trust, and self-disclosure online". In: *Human– Computer Interaction* 25.1, pp. 1–24.
- Kasunic, Anna et al. (2019). "Turker Tales: Integrating Tangential Play into Crowd Work". In: *Proceedings of the 2019 on Designing Interactive Systems Conference*, pp. 21–34.
- Kaye, Jane et al. (2015). "Dynamic consent: a patient interface for twenty-first century research networks". In: *European Journal of Human Genetics* 23.2, p. 141.
- Kelley, Christina, Bongshin Lee, and Lauren Wilcox (2017). "Self-tracking for mental wellness: understanding expert perspectives and student experiences". In: *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems*. ACM, pp. 629– 641.

Kersten-van Dijk, Elisabeth and Wijnand A IJsselsteijn (2016). "Design Beyond the Numbers: Sharing, Comparing, Storytelling and the Need for a Quantified Us." In: *IxD&A* 29, pp. 121–135.

Khan, Numan et al. (2017). "Digital health technologies to promote lifestyle change and adherence". In: *Current treatment options in cardiovascular medicine* 19.8, pp. 1–12.

Kim, Yoojung et al. (2017). "Prescribing 10,000 steps like aspirin: designing a novel interface for data-driven medical consultations". In: *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems*. ACM, pp. 5787–5799.

King, Jennifer (2019). "Becoming Part of Something Bigger" Direct to Consumer Genetic Testing, Privacy, and Personal Disclosure". In: *Proceedings of the ACM on Human-Computer Interaction* 3.CSCW, pp. 1–33.

King, William R and Jun He (2006). "A meta-analysis of the technology acceptance model". In: *Information & management* 43.6, pp. 740–755.

Kizilcec, René F (2016). "How much information?: Effects of transparency on trust in an algorithmic interface". In: *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*. ACM, pp. 2390–2395.

Klingler, Corinna et al. (2017). "Ethical issues in public health surveillance: a systematic qualitative review". In: *BMC public health* 17.1, pp. 1–13.

Knowles, Bran et al. (2019). "A scenario-based methodology for exploring risks: children and programmable IoT". In: *Proceedings of the 2019 on Designing Interactive Systems Conference*, pp. 751–761.

Kokotsaki, Dimitra, Victoria Menzies, and Andy Wiggins (2016). "Project-based learning: A review of the literature". In: *Improving schools* 19.3, pp. 267–277.

Koocher, Gerald P and Patricia Keith-Spiegel (2008). *Ethics in psychology and the mental health professions: Standards and cases*. Oxford University Press.

Kostkova, Patty et al. (2016). "Who owns the data? Open data for healthcare". In: *Frontiers in public health* 4, p. 7.

Kou, Yubo et al. (2019). "Turn to the Self in Human-Computer Interaction: Care of the Self in Negotiating the Human-Technology Relationship". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–15.

Kraft, Stephanie A et al. (2018). "Beyond consent: Building trusting relationships with diverse populations in precision medicine research". In: *The American Journal of Bioethics* 18.4, pp. 3–20.

Kristoffersson, Annica and Maria Lindén (2020). "A systematic review on the use of wearable body sensors for health monitoring: A qualitative synthesis". In: Sensors 20.5, p. 1502.

Kumar, Devender (2021). Designing and Evaluating Mobile Health Technology for Ambulatory Monitoring and Diagnosis of Heart Arrhythmias. DTU Health Technology.

Kumar, Devender et al. (2020). "Mobile and wearable sensing frameworks for mHealth studies and applications: a systematic review". In: *ACM Transactions on Computing for Healthcare* 2.1, pp. 1–28.

Kuntsman, Adi, Esperanza Miyake, and Sam Martin (2019). "Re-thinking Digital Health: Data, Appisation and the (im) possibility of 'Opting out'". In: *Digital health* 5.

La Bellacasa, Maria Puig de (2017). *Matters of care: Speculative ethics in more than human worlds*. Vol. 41. U of Minnesota Press.

Lacetera, Nicola, Mario Macis, and Robert Slonim (2013). "Economic rewards to motivate blood donations". In: *Science* 340.6135, pp. 927–928.

Lacey, Cherie and Catherine Caudwell (2019). "Cuteness as a 'Dark Pattern'in home robots". In: *2019 14th ACM/IEEE International Conference on Human-Robot Interaction (HRI)*. IEEE, pp. 374–381.

- Laugwitz, Bettina, Theo Held, and Martin Schrepp (2008). "Construction and evaluation of a user experience questionnaire". In: *Symposium of the Austrian HCI and usability engineering group*. Springer, pp. 63–76.
- Lazar, Amanda et al. (2016). "Rethinking the design of robotic pets for older adults". In: *Proceedings of the 2016 ACM Conference on Designing Interactive Systems*, pp. 1034–1046.
- Leary, Mark R and Robin M Kowalski (1990). "Impression management: A literature review and two-component model." In: *Psychological bulletin* 107.1, p. 34.
- Lee, Yi-Chieh et al. (2020). "" I Hear You, I Feel You": Encouraging Deep Self-disclosure through a Chatbot". In: *Proceedings of the 2020 CHI conference on human factors in computing systems*, pp. 1–12.
- Lemke, Amy A et al. (2010). "Public and biobank participant attitudes toward genetic research participation and data sharing". In: *Public Health Genomics* 13.6, pp. 368–377.
- Leon, Pedro Giovanni et al. (2013). "What matters to users?: factors that affect users' willingness to share information with online advertisers". In: *Proceedings of the ninth symposium on usable privacy and security*. ACM, p. 7.
- Li, Chen et al. (2019a). "Implementation of National Health Informatization in China: Survey About the Status Quo". In: *JMIR medical informatics* 7.1, e12238.
- Li, Rebecca et al. (2019b). "Moving data sharing forward: the launch of the Vivli platform". In: *NAM Perspectives* 8.11.
- Lin, Jimmy et al. (2020). "We Could, but Should We? Ethical Considerations for Providing Access to GeoCities and Other Historical Digital Collections". In: *Proceedings of the 2020 Conference on Human Information Interaction and Retrieval*, pp. 135–144.
- Lindberg, Sharon, Petter Karlström, and Sirkku Männikkö Barbutiu (2020). "Cultivating Ethics–A perspective from practice". In: *Proceedings of the 11th Nordic Conference on Human-Computer Interaction: Shaping Experiences, Shaping Society*, pp. 1–11.
- Lo, Bernard (2015). "Sharing clinical trial data: maximizing benefits, minimizing risk". In: *Jama* 313.8, pp. 793–794.
- Lucas, Gale M et al. (2014). "It's only a computer: Virtual humans increase willingness to disclose". In: *Computers in Human Behavior* 37, pp. 94–100.
- Lucas, Gale M et al. (2017). "Reporting mental health symptoms: breaking down barriers to care with virtual human interviewers". In: *Frontiers in Robotics and AI* 4, p. 51.
- Luchenski, Serena et al. (2012). "Survey of patient and public perceptions of electronic health records for healthcare, policy and research: Study protocol". In: *BMC medical informatics and decision making* 12.1, pp. 1–6.
- Ludford, Pamela J et al. (2007). "Capturing, sharing, and using local place information". In: *Proceedings of the SIGCHI conference on Human factors in computing systems*. ACM, pp. 1235–1244.
- Ludman, Evette J et al. (2010). "Glad you asked: participants' opinions of re-consent for dbGap data submission". In: *Journal of Empirical Research on Human Research Ethics* 5.3, pp. 9–16.
- Luger, Ewa, Stuart Moran, and Tom Rodden (2013). "Consent for all: revealing the hidden complexity of terms and conditions". In: *Proceedings of the SIGCHI conference on Human factors in computing systems*. ACM, pp. 2687–2696.
- Luger, Ewa and Abigail Sellen (2016). "" Like Having a Really Bad PA" The Gulf between User Expectation and Experience of Conversational Agents". In: *Proceedings of the 2016 CHI conference on human factors in computing systems*, pp. 5286–5297.
- Lupton, Deborah (2014). "Critical perspectives on digital health technologies". In: *Sociology compass* 8.12, pp. 1344–1359.

- Lyckvi, Sus and Olof Torgersson (2018). "Privacy and design ethics vs designing for curiosity, communication and children: Lessons learned". In: *Proceedings of the 20th International Conference on Human-Computer Interaction with Mobile Devices and Services*, pp. 1–8.
- Ma, Ning F et al. (2018). "Using Stakeholder Theory to Examine Drivers' Stake in Uber". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*, pp. 1–12.
- Ma, Tengteng, Hasti Sharifi, and Debaleena Chattopadhyay (2019). "Virtual humans in health-related interventions: A meta-analysis". In: *Extended Abstracts of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–6.
- Madaio, Michael A et al. (2020). "Co-Designing Checklists to Understand Organizational Challenges and Opportunities around Fairness in Al". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–14.
- Malu, Meethu and Leah Findlater (2017). "Sharing automatically tracked activity data: implications for therapists and people with mobility impairments". In: *Proceedings of the 11th EAI International Conference on Pervasive Computing Technologies for Healthcare*. ACM, pp. 136–145.
- Marcu, Gabriela, Jakob E Bardram, and Silvia Gabrielli (2011). "A framework for overcoming challenges in designing persuasive monitoring and feedback systems for mental illness". In: 2011 5th International Conference on Pervasive Computing Technologies for Healthcare (PervasiveHealth) and Workshops. IEEE, pp. 1–8.
- Martinez-Martin, Nicole and Karola Kreitmair (2018a). "Ethical issues for direct-to-consumer digital psychotherapy apps: addressing accountability, data protection, and consent". In: *JMIR mental health* 5.2, e32.
- Martinez-Martin, Nicole et al. (2018b). "Data mining for health: staking out the ethical territory of digital phenotyping". In: *NPJ digital medicine* 1.1, pp. 1–5.
- Martinez-Martin, Nicole et al. (2020). "Ethics of digital mental health during COVID-19: Crisis and opportunities". In: *JMIR Mental Health* 7.12, e23776.
- Maslow, Abraham H (2013). Toward a psychology of being. Simon and Schuster.
- Massimi, Michael et al. (2011). "Matters of life and death: locating the end of life in lifespanoriented hci research". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, pp. 987–996.
- Matchett, Nancy J (2009). "Cooperative learning, critical thinking, and character: Techniques to cultivate ethical deliberation". In: *Public Integrity* 12.1, pp. 25–38.
- Maxmen, Amy (2017). "Google spin-off deploys wearable electronics for huge health study". In: *Nature News* 547.7661, p. 13.
- McMillan, Donald, Alistair Morrison, and Matthew Chalmers (2013). "Categorised ethical guidelines for large scale mobile HCI". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, pp. 1853–1862.
- McNally, Brenna et al. (2016). "Children's perspectives on ethical issues surrounding their past involvement on a participatory design team". In: *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*, pp. 3595–3606.
- McNamara, Andrew, Justin Smith, and Emerson Murphy-Hill (2018). "Does ACM's code of ethics change ethical decision making in software development?" In: *Proceedings of the 2018 26th ACM joint meeting on european software engineering conference and symposium on the foundations of software engineering*, pp. 729–733.
- McPadden, Jacob et al. (2019). "Health Care and Precision Medicine Research: Analysis of a Scalable Data Science Platform". In: *Journal of medical Internet research* 21.4, e13043.

- Mentis, Helena M et al. (2017). "Crafting a view of self-tracking data in the clinical visit". In: *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems*. ACM, pp. 5800–5812.
- Mercurio, Mara et al. (2020). "Longitudinal trends in the quality, effectiveness and attributes of highly rated smartphone health apps". In: *Evidence-based mental health* 23.3, pp. 107–111.
- Meskó, Bertalan et al. (2017). "Digital health is a cultural transformation of traditional healthcare". In: *Mhealth* 3.
- Middleton, Anna (2018). "Society and personal genome data". In: *Human molecular genetics* 27.R1, R8–R13.
- Middleton, Anna et al. (2019). "Attitudes of publics who are unwilling to donate DNA data for research". In: *European journal of medical genetics* 62.5, pp. 316–323.
- Miner, Adam S, Arnold Milstein, and Jefferey T Hancock (2017). "Talking to machines about personal mental health problems". In: *Jama* 318.13, pp. 1217–1218.
- Mirnig, Alexander G and Alexander Meschtscherjakov (2019). "Trolled by the trolley problem: On what matters for ethical decision making in automated vehicles". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1– 10.
- Mohr, David C, Heleen Riper, and Stephen M Schueller (2018). "A solution-focused research approach to achieve an implementable revolution in digital mental health". In: *JAMA psychiatry* 75.2, pp. 113–114.
- Molich, Rolf et al. (2001). "Ethics in HCI". In: *CHI'01 Extended Abstracts on Human Factors in Computing Systems*, pp. 217–218.
- Molm, Linda D (2010). "The structure of reciprocity". In: *Social psychology quarterly* 73.2, pp. 119–131.
- Moncur, Wendy (2013). "The emotional wellbeing of researchers: considerations for practice". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, pp. 1883–1890.
- Morley, J et al. (2019). "From What to How: An Initial Review of Publicly Available AI Ethics Tools, Methods and Research to Translate Principles into Practices." In: *Science and Engineering Ethics* 26.4, pp. 2141–2168.
- Morley, Jessica et al. (2021). "From what to how: an initial review of publicly available AI ethics tools, methods and research to translate principles into practices". In: *Ethics, Governance, and Policies in Artificial Intelligence*. Springer, pp. 153–183.
- Morrison, Alistair et al. (2012). "A hybrid mass participation approach to mobile software trials". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, pp. 1311–1320.
- Munn, Zachary et al. (2018). "Systematic review or scoping review? Guidance for authors when choosing between a systematic or scoping review approach". In: *BMC medical research methodology* 18.1, pp. 1–7.
- Munteanu, Cosmin et al. (2015). "Situational ethics: Re-thinking approaches to formal ethics requirements for human-computer interaction". In: *Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems*, pp. 105–114.
- Murmann, Patrick and Simone Fischer-Hübner (2017). Usable transparency enhancing tools: A literature review.
- Murnane, Elizabeth L et al. (2018). "Personal informatics in interpersonal contexts: towards the design of technology that supports the social ecologies of long-term mental health management". In: *Proceedings of the ACM on Human-Computer Interaction* 2.CSCW, p. 127.

- Nadal, Camille, Corina Sas, and Gavin Doherty (2020). "Technology acceptance in mHealth: a scoping review of definitions, models and measurement". In: *Journal of Medical Internet Research*.
- Nathan, Lisa P et al. (2016). "Disruptions, dilemmas and paradoxes: Ethical matter (s) in design research". In: *Interacting with Computers*, pp. 1–9.
- Nathan, Peter E and Jack M Gorman (2015). *A guide to treatments that work*. Oxford University Press.
- Nebeker, Camille, John Torous, and Rebecca J Bartlett Ellis (2019). "Building the case for actionable ethics in digital health research supported by artificial intelligence". In: *BMC medicine* 17.1, pp. 1–7.
- Neto, Onicio Leal et al. (2020). "Participatory Surveillance Based on Crowdsourcing During the Rio 2016 Olympic Games Using the Guardians of Health Platform: Descriptive Study". In: *JMIR Public Health and Surveillance* 6.2, e16119.
- Nguyen, Josef and Bonnie Ruberg (2020). "Challenges of Designing Consent: Consent Mechanics in Video Games as Models for Interactive User Agency". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.
- Niforatos, Evangelos et al. (2020). "Would you do it?: Enacting Moral Dilemmas in Virtual Reality for Understanding Ethical Decision-Making". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–12.
- Nissenbaum, Helen (2011). "A contextual approach to privacy online". In: *Daedalus* 140.4, pp. 32–48.
- Nobile, Hélène et al. (2013). "Why do participants enroll in population biobank studies? A systematic literature review". In: *Expert review of molecular diagnostics* 13.1, pp. 35–47.
- Norberg, Patricia A, Daniel R Horne, and David A Horne (2007). "The privacy paradox: Personal information disclosure intentions versus behaviors". In: *Journal of consumer affairs* 41.1, pp. 100–126.
- Nurgalieva, Leysan et al. (2019). "Information Design in An Aged Care Context: Views of Older Adults on Information Sharing in a Care Triad". In: *Proceedings of the 13th EAI International Conference on Pervasive Computing Technologies for Healthcare*. ACM, pp. 101–110.
- O'Doherty, Kieran C et al. (2016). "If you build it, they will come: unintended future uses of organised health data collections". In: *BMC Medical Ethics* 17.1, pp. 1–16.
- Odone, Anna et al. (2019). "Public health digitalization in Europe: EUPHA vision, action and role in digital public health". In: *European journal of public health* 29.Supplement\_3, pp. 28–35.
- Ostherr, Kirsten et al. (2017). "Trust and privacy in the context of user-generated health data". In: *Big Data & Society* 4.1, p. 2053951717704673.
- Pagoto, Sherry and Gary G Bennett (2013). "How behavioral science can advance digital health". In: *Translational behavioral medicine* 3.3, pp. 271–276.
- Palen, Leysia and Paul Dourish (2003). "Unpacking privacy for a networked world". In: *Proceedings of the SIGCHI conference on Human factors in computing systems*. ACM, pp. 129–136.
- Papoutsi, Chrysanthi et al. (2015). "Patient and public views about the security and privacy of Electronic Health Records (EHRs) in the UK: results from a mixed methods study". In: *BMC medical informatics and decision making* 15.1, pp. 1–15.
- Patil, Sunil et al. (2016). "Public preferences for electronic health data storage, access, and sharing—Evidence from a pan-European survey". In: *Journal of the American Medical Informatics Association* 23.6, pp. 1096–1106.

- Peacock, Sean, Robert Anderson, and Clara Crivellaro (2018). "Streets for people: engaging children in placemaking through a socio-technical process". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*, pp. 1–14.
- Pencina, Michael J et al. (2016). "Supporting open access to clinical trial data for researchers: The Duke Clinical Research Institute–Bristol-Myers Squibb Supporting Open Access to Researchers Initiative". In: *American heart journal* 172, pp. 64–69.
- Penn, David and Til Wykes (2003). "Stigma, discrimination and mental illness". In: *Journal of Mental Health* 12.3, pp. 203–208.
- Perera, Gihan et al. (2011). "Views on health information sharing and privacy from primary care practices using electronic medical records". In: *International journal of medical informatics* 80.2, pp. 94–101.
- Petelka, Justin et al. (2020). "Being (In) Visible: Privacy, Transparency, and Disclosure in the Self-Management of Bipolar Disorder". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–14.
- Petersen, Carolyn (2018). "Through Patients' Eyes: Regulation, Technology, Privacy, and the Future". In: *Yearbook of medical informatics* 27.01, pp. 010–015.
- Pierce, James, Richmond Y Wong, and Nick Merrill (2020). "Sensor Illumination: Exploring Design Qualities and Ethical Implications of Smart Cameras and Image/Video Analytics". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–19.
- Pinder, Charlie et al. (2017). "Exploring the feasibility of subliminal priming on smartphones". In: *Proceedings of the 19th International Conference on Human-Computer Interaction with Mobile Devices and Services*, pp. 1–15.
- Porteri, Corinna et al. (2014). "Public's attitudes on participation in a biobank for research: an Italian survey". In: *BMC Medical Ethics* 15.1, pp. 1–10.
- Pratap, Abhishek et al. (2019). "Contemporary Views of Research Participant Willingness to Participate and Share Digital Data in Biomedical Research". In: *JAMA network open* 2.11.
- Provoost, Simon et al. (2017). "Embodied conversational agents in clinical psychology: a scoping review". In: *Journal of medical Internet research* 19.5, e151.
- Purpura, Stephen et al. (2011). "Fit4life: the design of a persuasive technology promoting healthy behavior and ideal weight". In: *Proceedings of the SIGCHI conference on human factors in computing systems*, pp. 423–432.
- Raento, Mika and Antti Oulasvirta (2008). "Designing for privacy and self-presentation in social awareness". In: *Personal and Ubiquitous Computing* 12.7, pp. 527–542.
- Rahaman, Ashikur et al. (2019). "Developing IoT Based Smart Health Monitoring Systems: A Review." In: *Rev. d'Intelligence Artif.* 33.6, pp. 435–440.
- Rathi, Vinay et al. (2012). "Sharing of clinical trial data among trialists: a cross sectional survey". In: *Bmj* 345, e7570.
- Rieger, Agnes et al. (2019). "Psychiatry outpatients' willingness to share social media posts and smartphone data for research and clinical purposes: Survey study". In: *JMIR formative research* 3.3, e14329.
- Roberts, Laura Weiss, Steven Chan, and John Torous (2018). "New tests, new tools: mobile and connected technologies in advancing psychiatric diagnosis". In: *npj Digital Medicine* 1.1, p. 6.
- Rodil, Kasper, Donovan Maasz, and Heike Winschiers-Theophilus (2020). "Moving Virtual Reality out of its Comfort Zone and Into the African Kalahari Desert Field: Experiences From Technological Co-Exploration With an Indigenous San Community in Namibia".
  In: 26th ACM Symposium on Virtual Reality Software and Technology, pp. 1–10.

- Rooksby, John, Alistair Morrison, and Dave Murray-Rust (2019). "Student Perspectives on Digital Phenotyping: The Acceptability of Using Smartphone Data to Assess Mental Health". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*. ACM, p. 425.
- Roossien, Charlotte Christina et al. (2021). "Ethics in Design and Implementation of Technologies for Workplace Health Promotion: A Call for Discussion". In: *Frontiers in Digital Health*, p. 106.
- Rotman, Dana et al. (2012). "Dynamic changes in motivation in collaborative citizenscience projects". In: *Proceedings of the ACM 2012 conference on computer supported cooperative work*. ACM, pp. 217–226.
- Royakkers, Lambèr et al. (2018). "Societal and ethical issues of digitization". In: *Ethics* and *Information Technology* 20.2, pp. 127–142.
- Rudnicka, Anna, Anna L Cox, and Sandy JJ Gould (2019). "Why Do You Need This?: Selective Disclosure of Data Among Citizen Scientists". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*. ACM, p. 392.
- Sabie, Samar and Tapan Parikh (2019). "Cultivating Care through Ambiguity: Lessons from a Service Learning Course". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–14.
- Sahay, Sundeep and Arunima Mukherjee (2020). "Where Is All Our Health Data Going?" In: *Economic & Political Weekly* 55.1, p. 47.
- Sailaja, Neelima et al. (2018). "Explicating the challenges of providing novel media experiences driven by user personal data". In: *Proceedings of the 2018 ACM International Conference on Interactive Experiences for TV and Online Video*, pp. 101–113.
- Saltz, Jeffrey et al. (2019). "Integrating ethics within machine learning courses". In: ACM *Transactions on Computing Education (TOCE)* 19.4, pp. 1–26.
- Sanches, Pedro et al. (2019). "HCI and Affective Health: Taking stock of a decade of studies and charting future research directions". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–17.
- Sanderson, Saskia C et al. (2017). "Public attitudes toward consent and data sharing in biobank research: a large multi-site experimental survey in the US". In: *The American Journal of Human Genetics* 100.3, pp. 414–427.
- Sankar, Pamela L and Lisa S Parker (2017). "The Precision Medicine Initiative's All of Us Research Program: an agenda for research on its ethical, legal, and social issues". In: *Genetics in Medicine* 19.7, p. 743.
- Sannon, Shruti and Dan Cosley (2019). "Privacy, Power, and Invisible Labor on Amazon Mechanical Turk". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*. ACM, p. 282.
- Sas, Corina, Kobi Hartley, and Muhammad Umair (2020). "ManneqKit Cards: A Kinesthetic Empathic Design Tool Communicating Depression Experiences". In: *Proceedings* of the 2020 ACM Designing Interactive Systems Conference, pp. 1479–1493.
- Saunders, Kate EA et al. (2017). "Experiences of remote mood and activity monitoring in bipolar disorder: a qualitative study". In: *European Psychiatry* 41, pp. 115–121.
- Scheutz, Matthias and Thomas Arnold (2016). "Are we ready for sex robots?" In: 2016 11th ACM/IEEE International Conference on Human-Robot Interaction (HRI). IEEE, pp. 351– 358.
- Schlesinger, Ari, Kenton P O'Hara, and Alex S Taylor (2018). "Let's talk about race: Identity, chatbots, and Al". In: *Proceedings of the 2018 chi conference on human factors in computing systems*, pp. 1–14.

- Schmidt, Morten, Lars Pedersen, and Henrik Toft Sørensen (2014). "The Danish Civil Registration System as a tool in epidemiology". In: *European journal of epidemiology* 29.8, pp. 541–549.
- Seymour, William et al. (2020). "Strangers in the Room: Unpacking Perceptions of Smartness" and Related Ethical Concerns in the Home". In: *Proceedings of the 2020 ACM Design-ing Interactive Systems Conference*, pp. 841–854.
- Shapiro, Ben Rydal et al. (2020). "Re-Shape: A method to teach data ethics for data science education". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.

Sharp, Helen (2003). Interaction design. John Wiley & Sons.

- Shi, Weiyan et al. (2020). "Effects of Persuasive Dialogues: Testing Bot Identities and Inquiry Strategies". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.
- Shilton, Katie and Sara Anderson (2017). "Blended, not bossy: Ethics roles, responsibilities and expertise in design". In: *Interacting with computers* 29.1, pp. 71–79.
- Shilton, Katie et al. (2020). "Role-playing computer ethics: Designing and evaluating the privacy by design (Pbd) simulation". In: *Science and engineering ethics* 26.6, pp. 2911–2926.
- Shilton, Katie et al. (2021). "Excavating awareness and power in data science: A manifesto for trustworthy pervasive data research". In: *Big Data & Society* 8.2.
- Shippee, Nathan D et al. (2015). "Patient and service user engagement in research: a systematic review and synthesized framework". In: *Health Expectations* 18.5, pp. 1151–1166.
- Shklovski, Irina et al. (2014). "Leakiness and creepiness in app space: Perceptions of privacy and mobile app use". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, pp. 2347–2356.
- Sitbon, Laurianne et al. (2020). "Engaging the Abilities of Participants with Intellectual Disabilityin IIR Research". In: *Proceedings of the 2020 Conference on Human Information Interaction and Retrieval*, pp. 103–112.
- Skirpan, Michael et al. (2018a). "Ethics education in context: A case study of novel ethics activities for the CS classroom". In: *Proceedings of the 49th ACM Technical Symposium on Computer Science Education*, pp. 940–945.
- Skirpan, Michael Warren, Jacqueline Cameron, and Tom Yeh (2018b). "More than a show: Using personalized immersive theater to educate and engage the public in technology ethics". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.
- Skirpan, Michael Warren, Tom Yeh, and Casey Fiesler (2018c). "What's at Stake: Characterizing Risk Perceptions of Emerging Technologies". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*, pp. 1–12.
- Snyder, Jaime (2020). "Visualizing Personal Rhythms: A Critical Visual Analysis of Mental Health in Flux". In: *Proceedings of the 2020 ACM Designing Interactive Systems Conference*, pp. 269–281.
- Song, Jae W and Kevin C Chung (2010). "Observational studies: cohort and case-control studies". In: *Plastic and reconstructive surgery* 126.6, p. 2234.
- Stahl, Bernd Carsten, Job Timmermans, and Brent Daniel Mittelstadt (2016). "The ethics of computing: A survey of the computing-oriented literature". In: *ACM Computing Surveys (CSUR)* 48.4, pp. 1–38.
- Stephanidis, Constantine et al. (2019). "Seven HCI grand challenges". In: *International Journal of Human–Computer Interaction* 35.14, pp. 1229–1269.

- Stone, Arthur A et al. (2007). "Historical roots and rationale of ecological momentary assessment (EMA)". In: *The science of real-time data capture: Self-reports in health research*, pp. 3–10.
- Strengers, Yolande et al. (2020). "Adhering, Steering, and Queering: Treatment of Gender in Natural Language Generation". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–14.
- Sudlow, Cathie et al. (2015). "UK biobank: an open access resource for identifying the causes of a wide range of complex diseases of middle and old age". In: *PLoS medicine* 12.3, e1001779.
- Sultana, Sharifa et al. (2018). "Design within a patriarchal society: Opportunities and challenges in designing for rural women in bangladesh". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.
- Sun, Kaiwen et al. (2019). "It's My Data! Tensions Among Stakeholders of a Learning Analytics Dashboard". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–14.
- Tan, Joshua et al. (2018). "Comparing Hypothetical and Realistic Privacy Valuations". In: Proceedings of the 2018 Workshop on Privacy in the Electronic Society. ACM, pp. 168– 182.
- Terry, Sharon F and Patrick F Terry (2011). "Power to the people: participant ownership of clinical trial data". In: *Science Translational Medicine* 3.69, pp. 69cm3–69cm3.
- Thieme, Anja, Danielle Belgrave, and Gavin Doherty (2020). "Machine learning in mental health: A systematic review of the HCI literature to support the development of effective and implementable ML systems". In: *ACM Transactions on Computer-Human Interac-tion (TOCHI)* 27.5, pp. 1–53.
- Tolmeijer, Suzanne et al. (2020). "Implementations in machine ethics: A survey". In: *ACM Computing Surveys (CSUR)* 53.6, pp. 1–38.
- Torous, John et al. (2019). "Creating a digital health smartphone app and digital phenotyping platform for mental health and diverse healthcare needs: an interdisciplinary and collaborative approach". In: *Journal of Technology in Behavioral Science* 4.2, pp. 73– 85.
- Toussaint, Lianne and Marina Toeters (2020). "Keeping the Data within the Garment: Balancing Sensing and Actuating in Fashion Technology". In: *Proceedings of the 2020 ACM Designing Interactive Systems Conference*, pp. 2229–2238.
- Tricco, Andrea C et al. (2018). "PRISMA extension for scoping reviews (PRISMA-ScR): checklist and explanation". In: *Annals of internal medicine* 169.7, pp. 467–473.
- Trifan, Alina, Maryse Oliveira, and José Luís Oliveira (2019). "Passive sensing of health outcomes through smartphones: systematic review of current solutions and possible limitations". In: *JMIR mHealth and uHealth* 7.8, e12649.
- Trinidad, Susan Brown et al. (2012). "Informed consent in genome-scale research: what do prospective participants think?" In: *AJOB primary research* 3.3, pp. 3–11.
- Troiano, Giovanni Maria, Matthew Wood, and Casper Harteveld (2020). "And This, Kids, Is How I Met Your Mother': Consumerist, Mundane, and Uncanny Futures with Sex Robots". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, pp. 1–17.
- Vaidyam, Aditya Nrusimha et al. (2019). "Chatbots and conversational agents in mental health: a review of the psychiatric landscape". In: *The Canadian Journal of Psychiatry* 64.7, pp. 456–464.
- Vaizman, Yonatan et al. (2018). "Extrasensory app: Data collection in-the-wild with rich user interface to self-report behavior". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*. ACM, p. 554.

- Van Berkel, Niels, Denzil Ferreira, and Vassilis Kostakos (2017). "The experience sampling method on mobile devices". In: *ACM Computing Surveys (CSUR)* 50.6, pp. 1–40.
- Van Mechelen, Maarten et al. (2020). "18 Years of ethics in child-computer interaction research: a systematic literature review". In: *Proceedings of the Interaction Design and Children Conference*, pp. 161–183.
- Vandemeulebroucke, Tijs, Bernadette Dierckx de Casterlé, and Chris Gastmans (2018). "The use of care robots in aged care: A systematic review of argument-based ethics literature". In: *Archives of gerontology and geriatrics* 74, pp. 15–25.
- Veale, Michael, Max Van Kleek, and Reuben Binns (2018). "Fairness and accountability design needs for algorithmic support in high-stakes public sector decision-making". In: *Proceedings of the CHI Conference on Human Factors in Computing Systems*, pp. 1– 14.
- Veličković, Petar et al. (2017). "Scaling health analytics to millions without compromising privacy using deep distributed behavior models". In: Proceedings of the 11th EAI International Conference on Pervasive Computing Technologies for Healthcare. ACM, pp. 92–100.
- Vilaza, Giovanna Nunes and Jakob Bardram (2019). "Sharing Access to Behavioural and Personal Health Data: Designers' Perspectives on Opportunities and Barriers". In: *Proceedings of the 13th EAI International Conference on Pervasive Computing Technologies for Healthcare*. PervasiveHealth'19. Trento, Italy: ACM, pp. 346–350.
- Vilaza, Giovanna Nunes, David Coyle, Jakob Eyvind Bardram, et al. (2021a). "Public Attitudes to Digital Health Research Repositories: Cross-sectional International Survey". In: *Journal of Medical Internet Research* 23.10, e31294.
- Vilaza, Giovanna Nunes and Darragh McCashin (2021b). "Is the Automation of Digital Mental Health Ethical? Applying an Ethical Framework to Chatbots for Cognitive Behaviour Therapy". In: *Frontiers in Digital Health* 3.
- Vilaza, Giovanna Nunes et al. (2020). "Futures for Health Research Data Platforms From the Participants' Perspectives". In: *Proceedings of the 11th Nordic Conference on Human-Computer Interaction: Shaping Experiences, Shaping Society*, pp. 1–14.
- Vines, John et al. (2013). "Configuring participation: on how we involve people in design". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, pp. 429–438.
- Voigt, Paul and Axel Von dem Bussche (2017). "The EU General Data Protection Regulation (GDPR)". In: A Practical Guide, 1st Ed., Cham: Springer International Publishing.
- Wächter, Laura and Felix Lindner (2018). "An explorative comparison of blame attributions to companion robots across various moral dilemmas". In: *Proceedings of The 6th International Conference on Human-Agent Interaction*, pp. 269–276.
- Wärnestål, Pontus, Petra Svedberg, and Jens Nygren (2014). "Co-constructing child personas for health-promoting services with vulnerable children". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, pp. 3767–3776.
- Waycott, Jenny et al. (2016). "Not for me: Older adults choosing not to participate in a social isolation intervention". In: *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*, pp. 745–757.
- Weitzman, Elissa R et al. (2012). "Willingness to share personal health record data for care improvement and public health: a survey of experienced personal health record users". In: *BMC medical informatics and decision making* 12.1, p. 39.
- West, Peter et al. (2016). "The quantified patient in the doctor's office: Challenges & opportunities". In: *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*. ACM, pp. 3066–3078.

- Wicks, Paul et al. (2010). "Sharing health data for better outcomes on PatientsLikeMe". In: *Journal of medical Internet research* 12.2, e19.
- Wiggins, Andrea and John Wilbanks (2019). "The rise of citizen science in health and biomedical research". In: *The American Journal of Bioethics* 19.8, pp. 3–14.
- Wilkes, Michael S, Robert A Bell, and Richard L Kravitz (2000). "Direct-To-Consumer Prescription Drug Advertising: Trends, Impact, And Implications: Aiming drug ads at consumers means big business for drug companies, but its effect on clinical care is not yet known." In: *Health affairs* 19.2, pp. 110–128.
- Williams, Tom et al. (2020). "The confucian matador: three defenses against the mechanical bull". In: Companion of the 2020 ACM/IEEE International Conference on Human-Robot Interaction, pp. 25–33.
- Wood, William A, Antonia V Bennett, and Ethan Basch (2015). "Emerging uses of patient generated health data in clinical research". In: *Molecular oncology* 9.5, pp. 1018–1024.
- Wouters, Niels et al. (2019). "Biometric Mirror: Exploring Values and Attitudes towards Facial Analysis and Automated Decision-Making". In: Conference on Designing Interactive Systems. Vol. 1145.
- Wu, Yiying, Sus Lyckvi, and Virpi Roto (2019). "" What is Fair Shipping, Anyway?" Using Design Fiction to Raise Ethical Awareness in an Industrial Context". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, pp. 1–13.
- Zammar, Guilherme et al. (2010). "So different, yet so similar: meta-analysis and policy modeling of willingness to participate in clinical trials among Brazilians and Indians". In: *PloS one* 5.12, e14368.
- Zanaboni, Paolo et al. (2020). "Patient use and experience with online access to electronic health records in Norway: Results from an online survey". In: *Journal of medical Internet research* 22.2, e16144.
- Ziefle, Martina, Julian Halbey, and Sylvia Kowalewski (2016). "Users' Willingness to Share Data on the Internet: Perceived Benefits and Caveats". In: *IoTBD*, pp. 255–265.
- Zoshak, John and Kristin Dew (2021). "Beyond Kant and Bentham: How Ethical Theories are being used in Artificial Moral Agents". In: *Proceedings of the 2021 CHI Conference on Human Factors in Computing Systems*, pp. 1–15.
- Zuboff, Shoshana (2019). The age of surveillance capitalism: The fight for a human future at the new frontier of power. Profile Books.